FEASIBILITY STUDY REPORT FOR HERTEL LANDFILL SITE PLATTEKILL, NEW YORK VOLUME 1



EPA WORK ASSIGNMENT NO. 003-2LH7

JULY, 1991



REGION II

ALTERNATIVE REMEDIAL CONTRACTING STRATEGY (ARCS) FOR HAZARDOUS WASTE REMEDIAL SERVICES

EPA Contract No. 68-S9-2001

TAMS CONSULTANTS, Inc.

and

TRC ENVIRONMENTAL CONSULTANTS, Inc.

EXECUTIVE SUMMARY

A Feasibility Study (FS) for the Hertel Landfill site, located in Plattekill, New York (National Priority List No. 810, August 1990) is being conducted for the U.S. Environmental Protection Agency - Region II by TAMS Consultants, Inc. and TRC Environmental Consultants, Inc. The purpose of this FS is to evaluate a range of remedial alternatives which will provide a basis for the selection of an alternative that is protective of human health and the environment. The FS is based on an earlier Remedial Investigation (RI) which investigated the nature and extent of contamination at the site and characterized environmental impact and potential health risks posed by the site.

The Hertel Landfill site operated as an active landfill between the mid 1960's and the late 1970's, receiving municipal wastes and other wastes, including paint wastes, oily wastes, printing wastes fibrous materials, drums, engine blocks and farm equipment, as evidenced by their presence at or near the surface of the site.

The extent of the landfill area and contamination associated with the presence of the landfill was identified and described in the Remedial Investigation Report (TAMS/TRC, July 1991). The investigation indicated that the landfill covers approximately 13 acres of the 80-acre site, consists primarily of household refuse with some metal debris, and varies in thickness to over 16.5 feet. Other distinct areas of waste disposal include surficial paint wastes, and oily wastes.

The samples collected and analyzed from the environmental media at the site provided an overview of contaminant types and distribution. Volatile organic compounds, base/neutral/acid extractable compounds, and metals

ES-1

(inorganics) were the primary contaminants detected at the site and were

distributed as follows:

- The volatile organic compounds (VOCs) were distributed in samples of soil, ground water, sediment, seep water and surface water adjacent to the fill. The most commonly occurring VOCs identified at the site were aromatic hydrocarbons which were present in samples of ground water, subsurface soil, seep water and seep sediment, but not present in surface water samples. Other VOCs, including chlorinated hydrocarbons, were detected in samples of subsurface or surface soils, ground water, seep water, seep sediment, surface water and sediment samples. In most cases, the VOCs were not observed in sediment or surface water samples from downgradient of the site. VOCs exceeded Federal or New York standards for ground water and surface water. No VOCs exceeded New York State Sediment Criteria Guidance.
- The base/neutral/acid extractable (BNA) compounds were identified in all the media sampled on site. Polynuclear aromatic hydrocarbons (PAHs), a subset of the BNA compounds, were present in samples from surface and subsurface soils, on-site ground water, seep water and seep sediment. PAHs were not detected in on-site surface water or sediment. Phenols and phenolic compounds were detected in on-site subsurface soil, ground water and surface water samples. Phenols were not detected in off-site surface water and were detected in only one downgradient sediment sample. Phthalate esters were present all sampled media, including background samples of soil, sediment and ground water. Other BNA compounds are present in samples from on-site soil, ground water, seep water, seep sediment and wetland sediment, but were not detected in downgradient surface water or sediment samples. BNAs exceeded Federal or New York standards for ground water and surface water. For one seep sediment sample, three PAHs exceeded the New York State Sediment Criteria Guidance.
- Metals and other inorganics are widespread in nature and their presence must be compared with natural background in order to determine if landfill impacts are present. Aluminum, arsenic, barium, calcium, copper, chloride, cadmium, chromium, iron, potassium, magnesium, manganese, lead, mercury, sodium, zinc and cyanide all appeared at elevated concentrations in one or more samples from the various media sampled. Calcium, chloride, potassium, magnesium, manganese and sodium, all considered leachate indicator parameters, were noted at above-background concentrations in off-site surface water samples, although the concentrations decreased with distance from the landfill. Inorganics were also detected at elevated levels in downgradient sediment samples. Dissolved arsenic, iron, magnesium, manganese and sodium (as detected in filtered ground water samples) exceeded New York ground water standards. Ten metals exceeded the New York Sediment Criteria Guidance.

A health and environmental risk assessment was conducted to quantitatively and qualitatively assess the potential impacts of the landfill on human and ecological health. For the human health component of the risk assessment, both current and future land use scenarios were considered. The primary cancer and non-cancer risks were associated with the future use of the site as a residential area. The routes of exposure of most concern included dermal contact with soil, ingestion of ground water and ingestion of soil. The chemicals of primary concern include PAHs, arsenic, and chromium in soil and ground water, and manganese in ground water.

The feasibility study process uses the information on the nature and extent of contamination and associated health and environmental risks developed during the RI to develop and evaluate potential remedial alternatives and their overall protection of human health and the environment. The initial step in the Feasibility Study was the identification and screening of soil/waste and ground water remedial technologies on the basis of technical implementability. Because the majority of the surface water/sediment contaminants were attributable to leachate seeps, which would be addressed in the remediation of the soil/waste and ground water, remediation of the surface water/sediment as a separate matrix was not considered. Technology process options were then evaluated on the basis of effectiveness, implementability and cost, and representative process options to be used in the development of remedial alternatives were selected. Nine remedial alternatives, combining remedial technologies applicable to the soil/waste and ground water, were developed based on guidelines for alternative development specified in the National Contingency Plan. An initial screening of the nine alternatives on the basis of effectiveness, implementability and cost resulted in the elimination of four alternatives (referred to as Alternatives 3, 5, 6, and 7) from further consideration.

ES~3

Five alternatives were retained for detailed analysis and were evaluated on the basis of the following seven criteria:

Short-term effectiveness; Long-term effectiveness and permanence; Implementability; Reduction of toxicity, mobility or volume through treatment; Compliance with ARARs; Overall protection of human health and the environment; and Cost

The five alternatives retained for detailed analyses are summarized in Table ES-1. The results of the detailed analyses, by alternative, are presented in Tables ES-2 through ES-8.

TABLE ES-1 HERTEL LANDFILL FEASIBILITY STUDY

8

F

.

REMEDIAL ALTERNATIVES ALTERNATIVE DESCRIPTIONS

Alternative 1:	No action	No remedial activities conducted: continued ground water monitoring.
Alternative 2:	Site use restrictions, capping	Site use restrictions and ground water use restrictions: capping of the landfill area (12.2 acres) to limit exposure to contaminants and infiltration of precipitation.
Alternative 2A:	Site use restrictions, capping and slurry wall construction	Site use restrictions and ground water use restrictions: capping of the landfill area (12.2 acres) to limit exposure to contaminants and infiltration of precipitation. slurry wall (1800 feet in length. 40 feet deep) to minimize ground water flow through landfilled waste materials.
Alternative 4:	Site use restrictions. capping: Ground water extraction with on-site physical treatment	Site use restrictions and ground water use restrictions; capping of the landfill area (12.2 acres) to limit exposure to contaminants and infiltration of precipitation. ground water extraction (approximately 14,000 gallons per day) with treatment using chemical precipitation for inorganic removal and carbon adsorption for organic removal.
Alternative 4A:	Site use restrictions. capping: Ground water extraction with on-site innovative treatment using UV oxidation and membrane microfiltration	Site use restrictions and ground water use restrictions; capping of the landfill area (12.2 acres) to limit exposure to contaminants and infiltration of precipitation. ground water extraction (approximately 14.000 gallons per day) with treatment using membrane microfiltration for inorganic removal and UV oxidation for organic treatment.

1

ł

.

F

Ĩ

Ŧ

•

		HERTEL LANDFILL FEASIBILITY STUDY COMPARISON AMONG ALTERNATIVES SHORT-TERM EFFECTIVENESS
Alternative 1:	No action	No remedial activities conducted: therefore no short-term risks other than existing baseline risks result. Remedial response objectives not achieved
Alternative 2:	Site use restrictions. capping	Non-cancer risks to on-site workers exceed acceptable levels: personal protective equipment required Effective in limiting soil exposure risks within a short time frame: no protection against contaminated ground water migration
Alternative 2A:	Site use restrictions, capping and slurry wall construction	Non-cancer risks to on-site workers exceed acceptable levels: personal protective equipment required Effective in limiting soil exposure risks and continued contamination of ground water to a limited time frame
Alternative 4:	Site use restrictions. capping: Ground water extraction with on-site physical treatment	Non-cancer risks to on-site workers exceed acceptable levels: personal protective equipment required Effective in limiting soil exposure risks and migration of contaminated ground water to a limited time frame
Alternative 4A:	Site use restrictions. capping: Ground water extraction with on-site innovative treatment using UV oxidation and membrane microfiltration	Mon-cancer risks to on-site workers exceed acceptable levels: personal protective equipment required Effective in limiting soil exposure risks and migration of contaminated ground water to a limited time frame

¢ TABLE ES-

TABLE ES-3 HERTEL LANDFILL FEASIBILITY STUDY COMPARISON AMONG ALTERNATIVES LONG-TERM EFFECTIVENESS AND PERMANENCE

æ

F

E

Alternative 1:	No action	Baseline risks remain unchanged
Alternative 2:	Site use restrictions. capping	Surface and subsurface soil/waste contaminants are untreated but contained; Long-term monitoring of containment area and ground water required; Potential long-term risks associated with contaminated ground water migration not addressed
Alternative 2A:	Site use restrictions, capping and slurry wall construction	Surface and subsurface soil/waste contaminants are untreated but contained: Long-term contact between ground water and waste materials minimized or eliminated, thereby limiting potential for contaminated ground water migration: Long-term monitoring of containment area and ground water required
Alternative 4:	Site use restrictions. capping: Ground water extraction with on-site physical treatment	Surface and subsurface soil/waste contaminants are untreated but contained; Contaminated ground water migration minimized through pumping and on-site treatment; Long-term monitoring of containment areas and ground water required; may not be effective in maintaining cleanup levels in ground water once pump and treat operations are discontinued
Alternative 4A:	Site use restrictions, capping: Ground water extraction with on-site innovative treatment using UV oxidation and membrane microfiltration	Surface and subsurface soil/waste contaminants are untreated but contained; Contaminated ground water migration minimized through pumping and on-site treatment; Long-term monitoring of containment areas and ground water required; may not be effective in maintaining cleanup levels in ground water once pump and treat operations are discontinued

2

F

TABLE ES-4 HERTEL LANDFILL FEASIBILITY STUDY COMPARISON AMONG ALTERNATIVES IMPLEMENTABILITY

ALTER- NATIVE	DESCRIPTION OF ALTERNATIVE	TECHNICAL FEASIBILITY	ADMINISTRATIVE FEASIBILITY	AVAILABILITY OF SERVICES AND MATERIALS
1:	No action	Installation of additional monitoring wells is feasible	No inter-agency coordination required	Suppliers of services and equipment readily available
2:	Site use restrictions, capping	Construction easily implemented: Cap could inhibit future remedial activites in capped area or could enhance activities such as ground water extraction	Requires compliance with wetlands protection regulations	Suppliers of services and equipment readily available
2A:	Site use restrictions, capping and slurry wall construction	Construction fairly easily implemented. although presence of boulders and slopes could complicate slurry wall construction; Cap could inhibit future remedial activites in capped area	Requires compliance with wetlands protection regulations	Suppliers of services and equipment readily available
4:	Site use restrictions, capping: Ground water extraction with on-site physical treatment	Construction easily implemented; Ground water treatment technologies easily implemented; Cap could inhibit future remedial activites in capped area	Requires compliance with wetlands protection regulations and authorization to discharge treated ground water to surface water	Suppliers of services and equipment readily available
4A:	Site use restrictions. capping: Ground water extraction with on-site innovative treatment using UV oxidation and membrane microfiltration	Construction easily implemented: Technical problems possible with ground water treatment system: Cap could inhibit future remedial activities in capped area	Requires compliance with wetlands protection regulations and authorization to discharge treated ground water to surface water	Suppliers of services and equipment for ground water treatment system limited

.

E

1

I

L

1

L

.

L

1

TABLE ES-5 HERTEL LANDFILL FEASIBILITY STUDY COMPARISON AMONG ALTERNATIVES REDUCTION OF TOXICITY (T), MOBILITY (M) OR VOLUME (V) THROUGH TREATMENT

Ē

T

Alternative 1:	No action	No reductions in T. M or V: Site conditions remain unchanged
Alternative 2:	Site use restrictions, capping	No reduction in T or V; M of ground water contamination is decreased through containment and prevention of leachate production; ground water table remains within waste materials
Alternative 2A:	Site use restrictions, capping and slurry wall construction	No reduction in T or V; M of ground water contamination is decreased through prevention of leachate production and minimization of contact between waste materials and ground water table
Alternative 4:	Site use restrictions, capping: Ground water extraction with on-site physical treatment	T of ground water reduced through treatment: M of ground water reduced through pumping: No reduction of T or V of soil/waste: M of soil/waste is decreased through containment: Permanence of ground water T reductions not well-defined since ground water table resumes contact with wastes upon discontinuation of pump and treat system
Alternative 4A:	Site use restrictions, capping: Ground water extraction with on-site innovative treatment	T of ground water reduced through treatment: M of ground water reduced through pumping: No reduction of T or V of soil/waste: M of soil/waste is decreased through containment: Permanence of ground water T reductions not well-defined since ground water table resumes contact with wastes upon discontinuation of pump and treat system

T

TABLE ES-6 HERTEL LANDFILL FEASIBILITY STUDY COMPARISON AMONG ALTERNATIVES COMPLIANCE WITH ARARS

NUMBER AND DESCRIPTION OF ALTERNATIVE		CHEMICAL-SPECIFIC	LOCATION-SPECIFIC	ACTION-SPECIFIC
1:	No action	ARARs not attained for sediment, ground water or surface water: No applicable soil ARARs	Not applicable	Not applicable
2:	Site use restrictions, capping	ARARs not attained for sediment or ground water: No applicable soil ARARs: Cap designed to prevent leachate seeps, thereby reducing surface water contaminant levels	Construction activities to comply with wetlands requirements	Cap design and surface water drainage design to comply with landfill closure requirements
2A:	Site use restrictions, capping and slurry wall construction	ARARs not attained for sediment or ground water although minimization of contact between the ground water table and soil/waste may result in reduced ground water contaminant levels; No applicable soil ARARs; Cap designed to prevent leachate seeps. thereby reducing surface water contaminant levels	Construction activities to comply with wetlands requirements	Cap design and surface water drainage design to comply with landfill closure requirements
4:	Site use restrictions. capping: Ground water extraction with on-site physical treatment	ARARs not attained for sediment: No applicable soil ARARs: Ground water treatment to attain ARARs: Maintenance of ARARs in ground water following discontinuation of pump and treat operations is not guaranteed: Cap designed to prevent leachate seeps, thereby reducing surface water contaminant levels	Construction activities to comply with wetlands requirements	Cap design, surface water drainage design and ground water discharge standards to comply with landfill closure requirements and NPDES requirements
4A:	Site use restrictions, capping: Ground water extraction with on-site innovative treatment using UV oxidation and membrane microfiltration	ARARs not attained for sediment: No applicable soil ARARs; Ground water treatment expected to attain ARARs although innovative technologies not well proven; Maintenance of ARARs in ground water following discontinuation of pump and treat operations is not guaranteed; Cap designed to prevent leachate seeps, thereby reducing water contaminant levels	Construction activities to comply with wetlands requirements	Cap design, surface water drainage design and ground water discharge standards to comply with landfill closure requirements and NPDES requirements

: •

L

L

.

Ł

1

•

l

E

ł

ł

TABLE ES-7 HERTEL LANDFILL FEASIBILITY STUDY COMPARISON AMONG ALTERNATIVES OVERALL PROTECTION OF HUMAN HEALTH AND THE ENVIRONMENT

Alternative 1:	No action	Baseline risks remain unchanged: Least protective alternative
Alternative 2:	Site use restrictions, capping	Provides soil/waste contaminant containment but risks associated with ground water contaminant migration not addressed: Effective in the short-term: Does not attain chemical-specific ARARs
Alternative 2A:	Site use restrictions, capping and slurry wall construction	Provides soil/waste contaminant containment: Long-term risks associated with ground water contaminant migration minimized through elimination of contact between the soil/waste and ground water table: Effective in the short-term; Does not attain chemical-specific ARARs
Alternative 4:	Site use restrictions. capping: Ground water extraction with on-site physical treatment	Provides soil/waste contaminant containment and treatment of contaminated ground water: Good short-term effectiveness: Long-term maintenance of ground water ARARs following discontinuation of pump and treat system may not be possible
Alternative 4A:	Site use restrictions, capping: Ground water extraction with on-site innovative treatment using UV oxidation and membrane microfiltration	Provides soil/waste contaminant containment and treatment of contaminated ground water: Short-term effectiveness dependent on effectiveness of innovative ground water treatment methods: Long-term maintenance of ground water ARARs following discontinuation of pump and treat system may not be possible

4

		COMPARISON AMON CC	IG ALTERNATIVES ST		•	
	Alternative	Duration	TOTAL CAPITAL COST	ANNUAL O&M COST	* PRESENT WORTH - O&M COST	** TOTAL PRESENT WORTH
1	No action	30 yrs - G.W. mon. (1)	\$ 28,000 \$	132,000	\$ 2,033,000	\$ 2,509,000
7	Site use restrictions, capping	30 yrs - G.W. mon. (1) 30 yrs - cap maint.	3,482,000	163,000	2,503,000	7,182,000
2A	Site use restrictions, capping and slurry wall construction	30 yrs - G.W. mon. (1) 30 yrs - cap maint. 30 yrs - slurry wall maint.	8,406,000	171,000	2,626,000	13,238,000
4	Site use restrictions, capping; Ground water extraction with on-site physical treatment	17 yrs - G.W. mon. (1) 30 yrs - cap maint. 12 yrs - G.W. extr. & trtmt	3,989,000	316,000	3,322,000	8,774,000
4 A	Site use restrictions, capping; Ground water extraction with on-site innovative treatment using UV oxidation and membra microfiltration	17 yrs - G.W. mon. (1) 30 yrs - cap maint. 12 yrs - G.W. extr. & trtmt. .ne	3,955,000	267,000	2,884,000	8,207,000
	r Based on 5% discount r r Includes 20% contingen Includes 4 rounds of s	ate cy on all components amples per year				

TABLE ES-8 HERTEL LANDFILL FEASIBILITY STUDY

				PAGE
		E2	ECUTIVE SUMMARY	ES-1
1.0		IN	TRODUCTION	1-1
	1.1		Site Description	1-2
	1.2		Site History	1-3
	1.3		Nature and Extent of Contamination	1-5
		1.3.1	Volatile Organic Compounds (VOCs)	1-5
		1.3.2	Base/Neutral/Acid Extractable Compounds (BNAs).	1-6
		1.3.3	Metals and Inorganics	1-6
	1.4		Contaminant Fate and Transport	1-7
	1.5		Baseline Risk Assessment	1-9
2.0		FU	JRTHER IDENTIFICATION OF POTENTIAL APPLICABLE	
		C	OR RELEVANT AND APPROPRIATE REQUIREMENTS	2-1
		2.1.1	Potential Federal Chemical-Specific ARARs/ TBCs	2-2
		2.1.2	Potential New York Chemical-Specific ARARs/	
	~ ~		Detertial Legation Specific ADADa (TPCs	2-2
	2.2	2 2 1	Potential Location-Specific ARARS/IBCS	2-2
		2. 2.1	TBCs	2-3
		2.2.2	Potential State Location-Specific ARARs/ TBCs	2-4
	2.3		Potential Action-Specific ARARs/TBCs	2-4
		2.3.1	Potential Federal Action-Specific ARARs/	2-5
		2.3.2	Potential State Action-Specific ARARs/TBCs	2-6
9.0		II	DENTIFICATION AND SCREENING OF TECHNOLOGIES	3-1
	3.1		Remedial Action Objectives and Cleanup Criteria .	3-2
	3.2		General Response Actions	3–6
	3.3		Identification and Screening of Technologies	
			and Process Options	3–10
		3.3.1	Technology Screening	3–11
		3.3.2	Process Option Screening	3-11
1. 0		DI	EVELOPMENT AND SCREENING OF ALTERNATIVES	4-1
	4.1		Development of Alternatives	4-1
	4.2		Screening of Alternatives	4-2
		4.2.1		4-2
		4.2.2	Common Major Elements	4-3
		4.2.3	Alternative 1 - No Action	4-0
		4.2.3.1		4-0
		4.2.3.2	Evaluation	4-0
		4.4.4	Arternative 2 - Site use Restrictions and	1_4
		4 2 4 1	Description	4-6
		4 2 4 2	Evaluation	4-7
				- /

-iii-

(Continued)

PAGE

	4.2.5	Alternate 2A - Site Use Restrictions,	
		Capping, Slurry Wall	4-8
	4.2.5.1	Description	4-8
	4.2.5.2	Evaluation	4-8
	4.2.6	Alternative 3 - Excavation of Contaminated	
		Soil and Waste Materials with Off-Site	
		Disposal and Ground Water Collection	
		with Off-Site Treatment	4-9
	4.2.6.1	Description	4-9
	4.2.6.2	Evaluation	4-11
	4.2.7	Alternative 4 - Site Use Restrictions,	
		Capping, Ground Water Extraction with	
		On-Site Physical Treatment	4-12
	4.2.7.1	Description	4-12
	4.2.7.2	Evaluation	4-13
	4.2.8	Alternative 4A - Site Use Restrictions,	
		Capping, Ground Water Extraction with	
		On-Site Innovative Treatment	4-14
	4.2.8.1	Description	4-14
	4.2.8.2	Evaluation	4-15
	4.2.9	Alternative 5 - Excavation and On-Site	
		Incineration of Soils and Wastes, with	
		Ground Water Extraction and On-Site	
		Physical Treatment	4-16
	4.2.9.1	Description	4-16
	4.2.9.2	Evaluation	4-18
	4.2.10	Alternative 6 - Excavation and Off-Site	
		Treatment of Soils and Wastes, Ground Water	
		Extraction and On-Site Biological Treatment .	4-19
	4.2.10.1	Description	4-19
	4.2.10.2	Evaluation	4-20
	4.2.11	Alternative 7 - In Situ Vitrification with	
		Ground Water Extraction and On-Site	
		Chemical Treatment	4-21
	4.2.11.1	Description	4-21
	4.2.11.2	Evaluation	4-22
4.3	Se	election of Alternatives for Detailed Analysis .	4-23
	4.3.1	Effectiveness	4-23
	4.3.2	Implementability	4-24
	4.3.3	Cost	4-25
	4.3.4	Selection of Alternatives for Further	
		Consideration	4-25

(Continued)

5.0		DI	ETAILED ANALYSIS OF ALTERNATIVES	5–1
	5.1		Introduction	5-1
	5.2		Common Elements and Considerations	5-1
	5.3		Individual Analysis of Alternatives	5-11
		5.3.1	Alternative 1 - No Action with Continued	
			Ground Water Monitoring	5-11
		5.3.1.1	Description	5-11
		5.3.1.2	Criteria Assessment	5-12
		5.3.2	Alternative 2 - Site Use Restrictions and	
		0.012		5-15
		5.3.2.1	Description	5-15
		5.3.2.2	Criteria Assessment	5-16
		5 3 3	Alternative 2λ - Site Use Restrictions.	
		5.5.5	Capping Slurry Wall	5-21
		5331	Description	5-21
		5 3 3 2	Criteria Accessment	5-23
		534	Alternative A - Site Use Pestrictions	5 25
		J.J. I	Capping and Ground Water Extraction with	
			On-Site Dhysical Treatment	5-26
		5 3 4 1	Description	5-26
		5 2 4 2		5-30
		5 2 5	$\begin{array}{c} \text{Criteria Assessment} \\ \text{Alternative } A \\ \text{A} \\ \text{Site Use Pestrictions} \end{array}$	5-50
		5.5.5	Compine and Cround Water Extraction with	
			Capping and Ground water Extraction with	5-34
		5 2 5 1		5_34
		5.3.5.1		5_27
		5.3.5.4	Criteria Assessment	5-37
	5.4		Comparative Analysis of Alternatives	5-41
		5.4.1	Short-Term Effectiveness	5-42
		5.4.2	Long-Term Effectiveness	5-44
		5.4.3	Implementability	5-46
		5.4.4	Reduction of Toxicity, Mobility and Volume Through Treatment	5-47
		5.4.5	Compliance with ARARS	5-47
		546	Overall Protection of Human Health and the	5 17
		5.1.0	Environment	5-49
		5 4 7	Cost	5-50
		5 4 7 1	Present Worth Comparative Analysis	5-50
		5 4 7 2	Sensitivity Analysis	5-51
		5.1.7.4		
6.0			REFERENCES	6-1

-v-

(Continued)

LIST OF TABLES

TABLE

- 2-1 Preliminary Identification of Federal Contaminant-Specific ARARs and TBCs
- 2-2 Preliminary Identification of State Contaminant-Specific ARARs and TBCs
- 2-3 Preliminary Identification of Federal Location-Specific ARARs and TBCs
- 2-4 Preliminary Identification of State Location-Specific ARARs and TBCs
- 2-5 Preliminary Identification of Federal Action-Specific ARARs and TBCs
- 2-6 Preliminary Identification of State Action-Specific ARARs and TBCs
- 3-1 Summary of Ground Water Contaminants Detected Above ARARs and TBCs and Associated Risk-Based Cleanup Levels
- 3-2 Summary of Soil/Sediment Contaminants Detected Above ARARs and TBCs and Associated Risk-Based Cleanup Levels
- 3-3 Summary of Surface Water Contaminants Detected Above ARARs and TBCs and Associated Risk-Based Cleanup Levels
- 3-4 General Response Actions By Environmental Media
- 3-5 Soil/Waste Matrix Technology Screening
- 3-6 Ground Water Technology Screening
- 3-7 Soil/Waste Matrix Process Option Screening
- 3-8 Ground Water Process Option Screening
- 3-9 Representative Process Option Summary
- 4-1 Preliminary Remedial Alternative Summary
- 4-2 Range of Remedial Alternatives
- 4-3 Preliminary Remedial Cost Estimates

(Continued)

LIST OF TABLES

(Continued)

TABLE	
5-1	Remedial Alternatives Undergoing Detailed Analysis
5–2	Summary of Cancer Risk Estimates - Road Improvements and Clearing Vegetation
5–3	Summary of Chronic Hazard Index Estimates - Road Improvements and Clearing Vegetation
5-4	Summary of Cancer Risk Estimates - Cap Construction
5-5	Summary of Chronic Hazard Index Estimates - Cap Construction
5-6	Summary of Cancer Risk Estimates - Children
5-7	Summary of Chronic Hazard Index Estimates - Children
5-8	Summary of Cancer Risk Estimates - Well Construction
5-9	Summary of Chronic Hazard Index Estimates - Well Construction
5-10	Comparison Among Alternatives - Short-Term Effectiveness
5-11	Comparison Among Alternatives - Long-Term Effectiveness and Permanence
5-12	Comparison Among Alternatives - Implementability
5-13	Comparison Among Alternatives - Reduction of Toxicity, Mobility or Volume Through Treatment
5-14	Comparison Among Alternatives - Compliance with ARARs
5-15	Comparison Among Alternatives - Overall Protection of Human Health and the Environment
5-16	Comparison Among Alternatives - Cost
5-17	Sensitivity Analysis

(Continued)

LIST OF FIGURES

FIGURE

- 1-1 Feasibility Study Approach
- 1-2 Site Location Map
- 1-3 Hertel Landfill Site Map (Disposal Areas)
- 1-4 Landfill Area Topography
- 3-1 Monitoring Well Locations Within Landfill Area
- 5-1 Proposed Monitoring Well Locations
- 5-2 Proposed Extraction Well Locations
- 5-3 Alternative 2
- 5-4 Final Cap Detail
- 5-5 Alternative 2A
- 5-6 Hydrogeologic Effect of an Upgradient Slurry Wall
- 5-7 Alternatives 4 and 4A
- 5-8 Chemical Precipitation Schematic
- 5-9 Carbon Treatment System Schematic
- 5-10 DuPont/Oberlin Microfiltration System
- 5-11 UV Oxidation Unit

(Continued)

APPENDICES

Α

- DETAILED COST ESTIMATES
- B EXPOSURE EQUATIONS, FUGITIVE DUST MODELING AND DOSE CALCULATIONS
- C GROUND WATER MODELING INFORMATION

THIS PAGE LEFT INTENTIONALLY BLANK

1.0 INTRODUCTION

TAMS Consultants, Inc. (TAMS) and TRC Environmental Consultants, Inc. (TRC) are conducting a Remedial Investigation/Feasibility Study (RI/FS) of the Hertel Landfill Site, located in Plattekill, New York. The investigation of the site, which was ranked 810 on the National Priority List (NPL) as of August 1990, is being conducted in accordance with the requirements of the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA), as amended by the Superfund Amendments and Reauthorization Act (SARA). The study is being performed for the U.S. Environmental Protection Agency (USEPA) - Region II under the Alternate Remedial Contracting Strategies (ARCS) program (Contract No. 68-S9-2001, WA No. 003-2LH7).

The purpose of the RI conducted at the Hertel Landfill site was to investigate physical characteristics of the site, identify potential sources of contamination, determine the nature and extent of contamination, and characterize risk and environmental impact. The findings of that investigation were described and evaluated in the Remedial Investigation (RI) Report for the Hertel Landfill Site, Plattekill, New York, dated July 1991. Detailed background information was also provided in the RI Report. Other pertinent reports prepared by TAMS and TRC and others on the Hertel Landfill site prior to conducting the FS include:

- Final RI/FS Work Plan for the Hertel Landfill Site, September, 1989
- Final RI/FS Field Operations Plan (FOP) for the Hertel Landfill Site, October 1989
- Site Analysis, Hertel Landfill, 1990

The purpose of the Feasibility Study, presented herein, is to identify and evaluate alternatives for mitigating contamination and controlling its effects on public health or the environment. By evaluating remedial solutions

selected from the range of technologies available for site cleanup, a response can be formulated which is technically feasible, protects public health and the environment, is cost-effective, and is consistent with applicable or relevant and appropriate environmental standards. The Feasibility Study process was formulated by the U.S. EPA to properly implement CERCLA. The National Oil and Hazardous Substances Pollution Contingency Plan (NCP, 40 CFR Part 300) establishes the framework for performing the Feasibility Study. Further definition of the FS process is provided in <u>Guidance for Conducting</u> <u>Remedial Investigations and Feasibility Studies under CERCLA</u> (USEPA, Interim Final, October 1988).

Figure 1-1 provides a summary of the approach being used in this investigation to formulate an appropriate remedial response for the Hertel Landfill site. The FS report incorporates this approach into the following report format:

Section	Description of Contents
1	Introduction/Background Information
2	Assessment of Applicable or Relevant and Appropriate Requirements (ARARS)
3	Identification and Screening of Technologies
4	Development and Screening of Alternatives
5	Detailed Analysis of Alternatives
6	References

1.1 Site Description

The Hertel Landfill site is located in the town of Plattekill, Ulster County, New York, just south of US Route 44/NY Route 55, as shown in Figure 1-2. The site covers approximately 80 acres, of which the landfill and associated surficial waste materials occupy approximately 13.2 acres. A site map showing disposal areas is provided in Figure 1-3.

The topography of the site is generally flat with a variable gentle slope to the east. A topographic map of the landfill area is presented in Figure 1-4. The landfill area is covered with rocky soil and wastes with patches of grass and small shrubs. Abundant vegetation covers the remainder of the site.

Wetlands border the site to the north, south and east, and cover approximately 13 percent of the total area of the site. A small unnamed stream (NYSDEC stream classification number H-128-6-2-1-2) crosses the southern and eastern area of the site and flows in a northeasterly direction, bordering the east side of the fill area.

There are no permanent structures located on the site. A locked gate exists across the unpaved main access road near Routes 44/55; however, there is no perimeter fence.

The area surrounding the site (including the site) is zoned residential, although a small industry is located to the west, adjacent to the site. Private residences are located to the north and east along Routes 44/55 and along connector roads which intersect Routes 44/55. These residences receive their potable water from private wells.

1.2 Site History

The Hertel Landfill was established in 1963 as a municipal waste landfill. It operated until March 1977, when the revocation of the landfill permit by the Ulster County Department of Health and the passage of a Town of Plattekill ordinance prohibiting the disposal of out-of-town garbage resulted in the permanent closing of the site. No landfilling operations or other activities are currently performed under the present proprietor, Environmental Landfills, Inc. (ELI), of New Windsor, New York.

Initial environmental investigations began at the site in 1981, when five ground water monitoring wells were installed by Wehran Engineering, Inc. under the direction of the State of New York. Sampling and analysis of ground water from two downgradient pairs of deep/shallow wells identified the presence of organic and inorganic contaminants.

Previous investigations also identified a number of suspected waste disposal areas at the site including the following:

Disposal Area #1 - engine block and oil waste materials
Disposal Area #2 - trailer wreckage and scattered drums
Disposal Area #3 - oil stain area and sanitary waste
Disposal Area #4 - farm equipment debris
Disposal Area #5 - printing waste
Disposal Area #6 - fibrous material piles
Disposal Area #7 - paint waste and municipal landfill
Disposal Area #8 - possible rubber waste

These areas are indicated in Figure 1-3.

RI investigations were conducted by TAMS/TRC at the site between September 1989 and October 1990. The scope of these investigations included geophysical surveys, soil gas screening, test pit excavation, soil borings, and monitoring well installation. Environmental samples collected and submitted for chemical analyses included surface water, sediment, ground water, surface soil and subsurface soil samples. Also included in these samples were leachate seep water and sediment samples. Environmental sampling was conducted off-site as well as on-site. Hydraulic testing, ecological sampling, sampling of private wells in the vicinity of the site and geological characterization activities were also conducted. A complete description of sampling activities and analytical results is presented in the Remedial Investigation (RI) Report for the Hertel Landfill Site (TAMS/TRC, July 1991).

1.3 Nature and Extent of Contamination

The samples collected and analyzed during the RI investigations provided an overview of contaminant types and distribution. No significant "hot spot" areas were identified in association with the suspected disposal areas, with the possible exception of the detection of elevated soil gas readings in the paint waste area (Disposal Area #7) and a floating hydrocarbon product in a monitoring well located within Disposal Area #3, the oil stain and sanitary waste area. The existence of suspected Disposal Area #8 was not verified by the field investigations.

Volatile organic compounds, base/neutral/acid extractable compounds and metals (inorganics) were the primary contaminants detected at the site and were distributed as discussed in the following sections.

1.3.1 Volatile Organic Compounds (VOCs)

The volatile organic compounds (VOCs) were distributed in samples of soil, ground water, sediment, seep water and surface water collected adjacent to the fill. The most commonly occurring VOCs identified at the site were aromatic hydrocarbons (such as toluene, ethylbenzene and xylenes) which were present in ground water at concentrations ranging to 64 ppb, in subsurface soil at concentrations ranging to 310 ppb, in leachate sediment at concentrations ranging to 970 ppb and in leachate seep water at concentrations ranging to 8 ppb. Aromatic VOCs were not detected in non-leachate surface water samples, however. Other VOCs, such as carbon disulfide and chloroform, were also detected in various media at the site. In most cases, VOCs were not observed in sediment or surface water samples collected downgradient of the site.

TAMS/TRC

1.3.2 Base/Neutral/Acid Extractable Compounds (BNAs)

The base/neutral/acid extractable (BNA) compounds were identified in all the media sampled on site. Polynuclear aromatic hydrocarbons (PAHs), including naphthalene, phenanthrene and pyrene, were present in surface soils at concentrations ranging to 3,100 ppb, in subsurface soils at concentrations ranging to 1,100 ppb, in leachate sediment samples at concentrations ranging to 1,500 ppb and in leachate seep samples at concentrations ranging to 4 ppb. Naphthalene was also detected in ground water samples at concentrations ranging to 39 ppb. PAHs were not detected in on-site surface water or sediment.

Phenols and phenolic compounds, such as 4-methylphenol, were detected in on-site subsurface soil at concentrations ranging to 340 ppb, and in ground water and surface water samples at concentrations ranging to 110 ppb. Phenols were not detected in off-site surface water and were detected in only one downgradient sediment sample. Phthalate esters, including bis(2-ethylhexyl)phthalate, were present in all sampled media, including background samples of soil, sediment and ground water. Other BNA compounds are present in samples from on-site soil, ground water, seep water, seep sediment and wetland sediment, but were not detected in downgradient surface water or sediment samples.

1.3.3 Metals and Inorganics

Metals and other inorganics are widespread in nature and their presence must be compared with natural background levels in order to determine if landfill impacts are present. Aluminum, arsenic, barium, calcium, copper, chloride, cadmium, chromium, iron, potassium, magnesium, manganese, lead, mercury, sodium, zinc and cyanide all appeared at elevated concentrations in

one or more samples from the various media sampled. The maximum concentrations of barium (4,490 ppm), cadmium (113 ppm), copper (319 ppm), lead (1,170 ppm), mercury (1.6 ppm) and zinc (615 ppm) detected in surface or subsurface soils exceeded the maximum on-site background levels detected for these metals. Calcium, chloride, potassium, magnesium, manganese and sodium, all considered leachate indicator parameters, were noted at above-background concentrations in off-site surface water samples, although the concentrations generally decreased with distance from the landfill. Inorganics were also detected at elevated levels in downgradient sediment samples.

1.4 Contaminant Fate and Transport

Potential routes of migration, contaminant persistence and observed contaminant migration were considered in evaluating the fate and transport of the site contaminants identified during the RI investigations.

Potential routes of contaminant migration at the Hertel Landfill site consist of the following:

- Migration of contaminated surface soils off-site due to overland runoff, airborne dust, and physical tracking off-site.
- Migration of surface soil contaminants via leaching through subsurface soils and subsequent ground water migration, or through volatilization into the ambient air.
- Migration of waste and subsurface soil contaminants via leaching into the ground water and subsequent ground water migration.
- Migration of contaminated ground water through overburden and bedrock materials or through discharge of shallow ground water to wetland areas and surface waters.
- Migration of contaminated leachate seep and stream/wetland sediments through leaching, erosion, surface water transport and redeposition mechanisms.

The presence of the various contaminant types was evaluated with respect to the existing migration pathways to provide an understanding of contaminant potential migration. which persistence and VOCs, are subject to volatilization and biodegradation by naturally occurring bacteria, were rather widespread in their detection at the site. The most commonly occurring VOCs were the aromatic hydrocarbons. Volatilization, biodegradation and adsorption are the major removal mechanisms for these compounds. The individual aromatic hydrocarbon-benzene compounds vary in their degree of mobility. Other VOCs were present at the site, but do not show a readily discernable pattern of Potential migration of these contaminants appears to be mainly occurrence. via ground water migration, with shallow contamination discharging to seep locations and surface water/wetland areas. Sediments at seep locations also exhibit VOCs.

BNAs were identified in all media sampled at the site. PAHs, detected in surface and subsurface soils, and leachate seep sediment but absent from surface water samples and most ground water samples, are persistent and readily adsorbed to organic carbon in soils. Phenols and phenolic compounds are generally more soluble in water than other BNA compounds and do not tend to adsorb to soils. Phenolic compounds were detected in several ground water samples but were typically absent from soil, sediment and surface water samples. Phthalate compounds were detected in all media sampled at the site, although they were predominantly detected in sediment and soil samples. They generally exhibit low solubility and would not be particularly amenable to water transport. Other BNA compounds detected appeared to represent isolated occurrences and were mainly detected in soils, sediment, ground water and leachate seep water samples.

Pesticides were detected in sediment and soil samples, and PCBs were detected in subsurface soil samples from test pits. Both pesticides and PCBs have an affinity for organics in soils which tends to render them immobile and persistent.

Inorganics have an affinity for soils which reduce their mobility in soils, although pH extremes may render them mobile. The concentrations of metals at the site generally exceed site background levels in all media sampled. However, elevated levels were generally not detected in filtered ground water samples.

1.5 Baseline Risk Assessment

A baseline health and environmental risk assessment was conducted to quantitatively and qualitatively assess the potential impacts of the landfill on human health and ecological health. For the human health component of the risk assessment, both current and future land use scenarios were considered. These scenarios included a current/recreational use scenario, a future/ construction use scenario, and a future/residential use scenario. The routes of exposure of most concern included dermal contact with soil, ingestion of ground water and ingestion of soil. The chemicals of primary concern include PAHs, arsenic and chromium in soil and ground water, and manganese in ground water.

The primary cancer and non-cancer risks were associated with the future use of the site as a residential area, due mainly to the evaluation of ground water ingestion under this scenario. Arsenic and manganese in the ground water primarily contributed to the overall risk associated with ground water ingestion, while the presence of arsenic, chromium and carcinogenic PAH compounds contributed to the risks posed by dermal contact and ingestion with

TAMS/TRC

on-site soils. The evaluation of the current/recreational use scenario identified cancer and non-cancer risks exceeding acceptable risk ranges, mainly due to dermal contact with arsenic and PAH compounds in the soil, for children and adult receptors.

There were no federal threatened or endangered species located on-site. The benthic macroinvertebrate study conducted on-site was inconclusive; the potential exists for site contaminants to produce adverse effects to aquatic organisms. Additionally, there is some indication that the potential exists for elevated inorganics (selenium, cadmium, and mercury) in soil to produce adverse environmental effects.

TAMS/TRC

2.0 FURTHER IDENTIFICATION OF POTENTIAL APPLICABLE OR RELEVANT AND APPROPRIATE REQUIREMENTS

Potential Applicable or Relevant and Appropriate Requirements (ARARs) were initially identified in the Final RI/FS Work Plan (TAMS, September 1989). Contaminant-specific ARARs were then used to evaluate site-specific contaminant levels in the RI Report (TAMS/TRC, July 1991). This section provides an overview of potentially applicable chemical-specific, location-specific and action-specific ARARs and To-Be-Considered guidance (TBCs), both on the federal and state levels, which will be used in this report to evaluate remedial alternatives. A comprehensive and conservative approach has been used in developing these lists of ARARs/TBCs, which reflect the types, quantities and extent of contaminants detected at the Hertel Landfill site, locational considerations, and the types of remedial actions likely to be required to mitigate the public health and environmental threats posed by a potential release of contaminants.

Applicable requirements are those clean-up standards, standards of control, or other substantive environmental protection requirements, criteria or limitations promulgated under federal or state law which specifically address a hazardous substance, pollutant, contaminant, remedial action, location, or other circumstance at a CERCLA site. Relevant and appropriate requirements are those federal and state requirements that, while not "applicable" to a hazardous substance, pollutant, contaminant, remedial action, location or other circumstance at a CERCLA site, address problems or situations sufficiently similar to those encountered at the CERCLA site that their use is well suited to the particular site. To-be-considered (TEC) material are non-promulgated advisories or guidance issued by federal or state agencies that, although not legally binding, can be used in determining the level of clean-up for protection of health or the environment (USEPA, 1988a).

TAMS/TRC

2.1.1 Potential Federal Chemical-Specific ARARs/TBCs

Potential federal chemical-specific ARARs and TBC criteria are presented in Table 2-1. Because ground water contamination has been detected upgradient of private potable well locations, drinking water criteria such as Maximum Contaminant Levels (MCLs) and Maximum Contaminant Level Goals (MCLGs) have been included in the ARARs matrix. For surface waters, Ambient Water Quality Criteria (AWQC), promulgated under the Clean Water Act, represent potential ARARs. The Clean Air Act establishes maximum concentrations for particulates and fugitive dust emissions and could be applicable to remedial alternatives which impact ambient air.

2.1.2 Potential New York Chemical-Specific ARARs/TBCs

Potential New York State chemical-specific ARARs and TBC criteria are presented in Table 2-2. Potential chemical-specific ARARs for ground water remediation include New York MCLs (10 NYCRR 5) and New York Ground Water Quality Standards (6 NYCRR 703). Surface water criteria include New York Surface Water Quality Standards (6 NYCRR 701). Air criteria include New York Ambient Air Quality Standards (6 NYCRR 256 and 257).

2.2 Potential Location-Specific ARARs/TBCs

An area's location is a fundamental determinant of its impact on human health and the environment. Location-specific ARARs are restrictions placed on the concentrations of hazardous substances or the conduct of activities solely because they are in a specific location (USEPA, 1988a). Some examples of these unique locations include floodplains, wetlands, coastal areas, historic places and sensitive ecosystems or habitats.

The Hertel Landfill site includes wetlands within the boundaries of the site. No floodplain has been delineated in association with the unnamed stream which flows along the eastern portion of the site (personal communication, Town of Plattekill). The federal- or state-protected status of flora and fauna at the site was evaluated during the RI; certain state-protected species (13 plants/animal) were identified.

2.2.1 Potential Federal Location-Specific ARARs/TBCs

Based on a review of locational features, it appears that federally promulgated location-specific ARARs and TBC criteria designed to protect landmarks, historical and archeological sites, floodplains and coastal areas are not applicable to the Hertel Landfill site. However, federal requirements for the protection of wetlands, wildlife species, farmlands and cultural sites may be applicable to the evaluation of remedial alternatives. The potential federal location-specific ARARs for the site are presented in Table 2-3.

Wetlands regulations, including Executive Order 11990 and Wetlands Construction and Management Procedures (40 CFR 6, Appendix A) and Section 404 of the Clean Water Act, may apply to any remedial action which impacts the stream or wetlands at the Hertel Landfill site. The National Wetland Inventory map for the Clintondale topographic map has not yet been prepared. During the RI, however, on-site wetlands were identified and mapped on the basis of U.S. Army Corps of Engineers guidelines and methodology. The Fish and Wildlife Coordination Act (16 USC 661 et seq.) would require appropriate actions to protect wildlife if modifications to the stream are proposed. The Farmland Protection Policy Act would require evaluation of potential impacts on significant/important farmlands. A cultural resources survey would be

TAMS/TRC

required to determine if an area of cultural value would be impacted by the remedial actions.

2.2.2 Potential State Location-Specific ARARs/TBCs

The State of New York has promulgated regulations for the protection of wetlands and wildlife. The potential state location-specific ARARs and TBCs are presented in Table 2-4. The New York Freshwater Wetlands Law (ECL Article 24.71 in Title 23) and New York State Freshwater Wetlands Permit Requirements and Classification (6 NYCRR 663 and 664) may apply to alternatives which could impact wetlands or surface water at the site. The Endangered and Threatened Species of Fish and Wildlife Requirements (6 NYCRR 182) may apply to activities which impact areas inhabited by endangered species. Α red-shouldered hawk, listed as a threatened species under the New York State Environmental Conservation Law, was identified during the environmental survey of the site, as well as thirteen plant species listed in the DEC 193.3 Protected Native Plants list, pursuant to Section 9-1503 of the Environmental Conservation Law.

2.3 Potential Action-Specific ARARs/TBCs

Based on the identification of contaminants in various on-site media, remediation activities may be required and numerous state and federal requirements would apply to the implementation of these activities. Potential action-specific ARARs/TBCs cannot be well-defined until remedial alternatives are developed and response actions defined. A discussion of potential actionspecific ARARs/TBCs pertaining to such general response actions as no action, institutional controls, diversion, containment, ground water collection, treatment, decontamination and disposal is provided in the following sections.
2.3.1 Potential Federal Action-Specific ARARs/TBCs

Numerous federally promulgated action-specific ARARs and TBC criteria could potentially affect the implementation of remedial measures. The primary regulatory requirements potentially applicable to the Hertel Landfill site appear in Table 2-5.

The primary federal administrative requirements which will guide remediation are those established under CERCLA and SARA. The revised NCP (40 CFR Part 300) incorporates SARA Title III requirements that alternatives satisfy ARARs, and utilize technologies that will provide a permanent reduction in the toxicity, volume and mobility of contamination, to the extent practicable.

Additional potential federal requirements include those pertaining to worker health and safety, as established under the Occupational Safety and Health Act (OSHA). ARARs associated with treatment, storage and disposal actions include RCRA requirements governing administrative (e.g., permitting) and substantive (design) issues. RCRA requirements include Subtitle C (Hazardous Waste Management) requirements and Subtitle D (Solid Waste Management requirements). The applicability of Subtitle C requirements is dependent on the following criteria:

- If a listed or characteristic waste (as defined under RCRA) was treated, stored or disposed of after the RCRA effective date (11/19/80), Subtitle C requirements apply.
- If the remedial activity applied to the site constitutes treatment, storage or disposal, as defined by RCRA, Subtitle C requirements apply.

The federal Clean Air Act (CAA) and Clean Water Act (CWA) are also potentially applicable to the evaluation of remedial activities which result in discharges to water bodies or ambient air.

2.3.2 Potential State Action-Specific ARARs/TBCs

New York State has promulgated regulations similar to those of the federal government. Potential state action-specific ARARs are presented in Table 2-6. New York Hazardous Waste (6 NYCRR 370-372) and Solid Waste Management Facility (6 NYCRR 360) regulations establish performance standards and design requirements for storage, containment, treatment, and disposal options. The New York Pollutant Discharge Elimination System (6 NYCRR 750-757) and Air Pollution Control Regulations (6 NYCRR 200-221) are potentially applicable to remedial activities which would result in discharges to water bodies or ambient air.

TAMS/TRC

3.0 IDENTIFICATION AND SCREENING OF TECHNOLOGIES

The overall technology screening and identification process was outlined previously in Figure 1-1. A discussion of the specific activities involved is presented below.

In order to identify and screen potential remedial technologies, an initial identification of remedial action objectives and cleanup criteria is required. Regulatory criteria and risk-based levels are considered in evaluating cleanup criteria.

Once remedial action objectives are developed, general response actions are identified which satisfy the objectives. An initial evaluation is made of the areas and volumes of media to which the general response actions will be applied.

The general response actions are then used to develop a list of potential remedial technologies for each environmental matrix to be remediated. An initial screening of the technologies is conducted based on the technical implementability of the various technologies. Specific site characteristics or waste characteristics typically limit the applicability of certain technologies and these characteristics are considered in determining which technologies are not appropriate for further consideration.

For those technologies which pass the initial screening, the associated technology process options are evaluated in greater detail to allow the selection of one process option to represent each technology type. The representative process option provides a basis for developing performance specifications which are used in evaluating that technology type; however, the specific process actually used to implement the remedial action may not be selected until the remedial design phase. To select a representative process, each process option is evaluated on the basis of effectiveness,

TAMS/TRC

implementability, and cost, with the greatest focus on effectiveness factors. Innovative technologies are either carried through the screening as a selected process option (if there is a reasonable belief that they offer potential for better treatment performance or implementability, few or lesser adverse impacts than other available approaches, or lower costs than demonstrated technologies) or are "represented" by another process option of the same technology type.

3.1 Remedial Action Objectives and Cleanup Criteria

Remedial response objectives are developed in order to set goals for protecting human health and the environment early in the alternative development process. The goals should be as specific as possible but should not unduly limit the range of alternatives that can be developed. For the Hertel Landfill FS, the results of the RI have been used to define specific contaminants of interest and allowable exposures based on the baseline risk assessment and ARARs/TBCs. Remedial response objectives have been developed for the various media at the site, specifically the ground water, soils and surface water/sediment, as presented below.

Ground Water

For ground water, the overall contaminants of interest include VOCs (benzene, chlorobenzene, ethylbenzene, toluene, xylenes and others), BNAs (naphthalene, phthalate esters and phenols) and metals (arsenic, iron and manganese). These contaminants are associated with three general areas of ground water contamination identified at the Hertel Landfill site. New York MCLs were exceeded for several organic compounds in the vicinity of monitoring wells MW-6S and MW-7S, in the central portion of the landfill. The New York

MCL for toluene was exceeded in well MW-13S (31 ppb), located to the north of the general landfill area within Disposal Area #1, the suspected engine block/oil waste disposal area. A third area of ground water contamination is indicated by the presence of contaminants in leachate seeps along the eastern toe of the landfill and in one surface water sample collected from the northern end of the landfill, also impacted by a leachate seep.

The major area of ground water contamination appears to be centered around the locations of monitoring wells MW-6S and MW-7S, with a floating hydrocarbon layer detected in well MW-6S. See Figure 3-1 for monitoring well locations. Contaminants detected in ground water samples from these wells include benzene, toluene, chlorobenzene, ethylbenzene and various phenolic compounds. Leachate indicator parameters were also detected at the highest concentrations in samples collected from wells MW-6S and MW-7S. Leachate indicator parameters were detected at lesser concentrations in monitoring wells located at the toe of the landfill and at even lower concentrations in wells located adjacent to the landfill. Samples collected from both the deep overburden wells and from bedrock well MW-6D also exhibited leachate indicator parameters. Bedrock monitoring well MW-6D was the only bedrock well to exhibit Target Compound List organic contaminants when sampled. Two BNA compounds, phenol and di-n-octylphthalate, were detected in MW-6D during the June 1990 sampling round at levels exceeding New York MCLs. During the August 1990 sampling round, however, these compounds were not detected.

A second potential area of ground water contamination was identified at the location of monitoring well MW-13S. During the two rounds of ground water sampling, toluene was detected at levels of 33 and 31 ppb respectively, levels which exceed the New York MCL for this compound.

3–3

The third area of ground water contamination, as indicated by the water quality of leachate seep samples, exhibited low levels of several volatile organics and semivolatile organics, and elevated levels of inorganics which were typically not detected at elevated levels in other surface water/sediment samples (see surface water/sediment discussion below). The ground water table is found within the waste material along the eastern toe of the landfill, where most of the seeps are located. In the areas where elevated inorganics are detected in the leachate seeps, ground water samples from shallow monitoring wells did not exhibit similar inorganics levels. Therefore, while the analytical results for seep samples are considered to be indicative of the ground water quality at these locations, inorganic contaminants may be concentrated near the water table as a result of leachate production.

Potential exposure pathways to ground water contaminants include human ingestion and potential exposure due to discharge of shallow ground water to surface water bodies. A summary of specific ground water contaminants which exceed current ARARs/TBCs is presented in Table 3-1. Also included in the table are leachate seep contaminant levels which exceed current ARARs/TBCs, as well as risk-based cleanup levels for those contaminants for which calculated risks exceed acceptable risk levels. Based on this evaluation, the remedial action objective for ground water is as follows:

Prevent exposure to VOCs, BNAs, and inorganics at levels exceeding acceptable risk-based cleanup levels or ARARs/TBCs, as indicated in Table 3-1, due to ground water ingestion.

Soils

For soils, the contaminants of interest include arsenic, chromium and other metals. These contaminants were detected over the entire landfill area, with no predictive pattern of distribution observed. It should be noted that

the maximum level of arsenic detected in surface soils (SS-22: 109 ppm) was detected at an upgradient background sample location. This level of detection provided the basis for the identification of risk in association with the presence of arsenic in surface soils at the Hertel Landfill site. Its presence at such an elevated level in a background sample, however, is not easily explained and confirmation sampling would be recommended prior to further evaluation of potential remedial options in this area.

The potential exposure pathways to surface and subsurface soil under current and future scenarios include inhalation, dermal contact and ingestion. A summary of specific soil contaminants which exceed current ARARs/TBCs is presented in Table 3-2. Because no federal or New York chemical-specific ARARs/TBCs have been developed for soil contaminants, New Jersey Department of Environmental Protection (NJDEP) Soil Action Levels are provided as a basis of comparison. For those contaminants for which calculated risks exceed acceptable risk levels, risk-based cleanup levels are presented. Based on this information, the remedial action objective for soils is as follows:

Prevent exposures to inorganics at levels exceeding acceptable risk-based cleanup levels or ARARs/TBCs, as indicated in Table 3-2, and prevent migration of contaminants that could result in ground water contamination in excess of acceptable risk-based levels or ARARs.

Surface Water and Sediments

For surface water and sediments, the contaminants of interest include inorganic analytes. Elevated levels of PAHs were detected in leachate seep sediment samples and in an off-site stream/wetland sediment sample (SED-20), located near the point where the stream crosses under US 44/55. The off-site PAH contamination is not thought to be attributable to on-site contaminants,

3–5

due to the lack of detected contamination at sediment sample locations between the site and the SED-20 location. At the SED-20 location, the identified contaminants may be attributable to the presence of the adjacent roadway.

The potential exposure pathways to surface water/sediment contaminants include dermal contact and ingestion. Allowable contaminant levels based on ARARs/TBCs for surface water are summarized in Table 3-3. Because exposures to surface water (dermal contact or ingestion) did not contribute significantly to total pathway risk calculations, risk-based cleanup levels were not calculated. A similar comparison of risk-based levels and ARARs for sediment contaminants was previously presented in Table 3-2. In both tables, leachate seep/sediment sample results are tabulated separately from surface water/stream sediment sample results. Based on this information, the remedial action objective for surface water and sediments is as follows:

Prevent exposure to inorganics in sediments and surface water at levels exceeding acceptable risk-based cleanup levels or ARARs/TBCs, as indicated in Tables 3-2 and 3-3 respectively, and prevent releases of contaminants from sediments into surface water that could result in excessive contaminant levels in surface water.

3.2 General Response Actions

General response actions are those remedial actions which will satisfy the remedial objectives. General response actions for the contaminated media at the Hertel Landfill site were formulated based on the results of the environmental investigation and risk assessment.

Ground water, soils, surface water and sediment were evaluated in determining appropriate general response actions. For each of these media, an initial determination has been made of the areas or volumes to which the general response actions may be applied, as described below. In determining these volumes/areas of media, consideration has been given to site conditions,

the nature and extent of contamination, acceptable exposure levels and potential exposure routes.

Ground Water

In order to provide a preliminary estimate of the volume of ground water potentially requiring treatment, an examination of the extent of ground water contamination at levels exceeding MCLs was made. Initially the area surrounding monitoring wells MW-6S and MW-7S is considered (see Figure 3-1). MCLs are exceeded for several volatile compounds in wells MW-6S and MW-7S (6/90 & 8/90). Monitoring well MW-5S exhibited ethylbenzene at a level (6 ppb) slightly exceeding the NY MCL during the August 1990 round of ground water sampling. All other shallow wells surrounding MW-6S and MW-7S exhibited no volatile organic contaminants at levels exceeding MCLs, except at MW-13S, as described previously. Ground water elevations in well MW-6S ranged from 620.09 to 622.34 ft above the National Geodetic Vertical Datum (NGVD) and in well MW-7S they ranged from 628.28 to 628.57 ft NGVD. The depth to bedrock from the ground water table at these two well locations ranged from approximately 20 to 27 feet. Depth to bedrock increases to the east and decreases to the west. Assuming an average total saturated depth to bedrock of 25 feet, an average depth from the water table to the base of the fill material of 5 feet, an areal extent of contamination of 5 acres, a waste material porosity of 52%, and an approximate underlying formation porosity of 30%, the total volume of contaminated ground water in the general area of wells MW-6S and MW-7S is estimated to be approximately 14 million gallons.

To estimate the volume of contaminated ground water associated with the detection of toluene in monitoring well MW-13S, it is assumed that contamination extends in an approximate 150-foot radius around the well (half

TAMS/TRC

3–7

the distance to the nearest monitoring well), with a total contaminated depth from the water table to the bedrock of 30 feet. Assuming a formation porosity of 30%, the volume of contaminated water in the area of well MW-13S is estimated to be approximately 5 million gallons. Due to the lack of monitoring wells located to the north, east and west of well MW-13S, additional investigations would be required to verify the accuracy of this estimate.

The volume of near-surface ground water contaminated with inorganics, as exhibited by the leachate seep samples, can be estimated by assuming a contaminated depth of 2 to 4 feet (based on test pit observations), an areal extent of 2 acres (based on the presence of leachate seeps along the northern and eastern toes of the landfill area) and porosities ranging from 30% to 52%. The estimated contaminated water volume ranges from approximately 0.4 to 1 million gallons. For calculation purposes, a volume of 0.8 million gallons will be used.

Based on this analysis, the total volume of contaminated ground water is estimated to be approximately 20 million gallons.

Soils/Wastes

Surface soil and subsurface soil samples collected from the Hertel Landfill site have exhibited various contaminants in various locations across the landfill area. Based on soil boring logs, cover material does not appear to be continuous across the site at a significant depth. At MW-5S and MW-6S, surface soil depths were recorded at eight inches and nine inches respectively. At MW-7S, a mixture of soil and debris was recorded to a depth of four feet. Test pit subsurface soil samples were collected from soil materials within the landfill materials or from soil materials at the base of

the waste materials. Due to the mixed nature of soil and waste, the soil/waste matrix has been considered as a single separate environmental media at the site. The total volume of the soil/waste matrix is considered to consist of the landfill area of 12.2 acres, extending to an average depth of 15 feet (assuming an average waste depth of 12 feet based on test pits and soil borings, and 3 additional feet of contaminated soil beneath the base of the waste). In addition, a 1 acre area of surface debris located to the north of the landfill area is considered to be contaminated to a depth of 2 feet from the surface. For remedial purposes, the total volume of the soil/waste matrix based on these assumptions is estimated to be approximately 300,000 cubic yards.

As discussed in Section 3.1, a significant level of arsenic was detected at one background surface soil sample location. However, no resampling of this area was conducted to confirm this analysis and, if accurate, no surrounding information is available to evaluate the vertical or horizontal extent of the contamination in this area. Therefore, no separate volume has been calculated for this potential area of contamination. Additional confirmation sampling would be required to evaluate the potential remediation of this area. If the area of contamination proved to be relatively small in area and volume, its remediation could be addressed within the remediation of the soil/waste matrix.

Surface Water/Sediment

Surface water and sediment samples collected from the adjacent stream and on-site wetlands for the most part exhibit no significant levels of organic contaminants. Elevated levels of typical leachate indicator parameters and inorganics detected in the samples indicate the impacts of the presence of the

landfill on the surface water/sediment. Leachate seep water and sediment samples exhibited low levels of organics and elevated inorganics. The leachate seeps are generally located within the landfill area and, therefore, the impacted sediment samples are located within the soil/waste matrix and have been addressed in the soil/waste matrix discussion presented above. The ground water table is found within the waste material along the eastern toe of the landfill where most of the seeps are located. Therefore, the seeps are considered to be indicative of the near-surface ground water quality at these locations and were addressed in the ground water discussion presented above. Based on the general lack of contamination attributable to the Hertel Landfill site within surface water/sediment samples other than the leachate seep samples, separate volumes associated with surface water and sediment sample contamination have not been calculated and general response actions have not been developed for these media.

A listing of medium-specific general response actions developed for the media discussed above is provided in Table 3-4.

3.3 Identification and Screening of Technologies and Process Options

The general response actions are developed further through the identification and screening of remedial technologies which could potentially meet the remedial action objectives and cleanup criteria. Following a of the remedial technologies on screening the basis of technical implementability, the process options associated with each technology are screened based on effectiveness, implementability and cost. Representative process options are chosen for inclusion in the comprehensive remedial alternatives developed for the site.

3.3.1 Technology Screening

The technology screening was performed to evaluate technologies for the remediation of the soil/waste matrix and ground water. Tables 3-5 and 3-6 present the screening results for these two media of concern. Each table includes brief descriptions of the individual technologies or process options, and comments on their applicability to the site. The technologies or technology options which do not pass the screening process on the basis of technical implementability are shaded in the figures and will not be retained for further consideration.

3.3.2 Process Option Screening

Upon identification of those technologies which are technically implementable based on the site and waste characteristics, the process options are further evaluated to allow the selection of a representative process option for each technology type. The process options are evaluated on the basis of effectiveness, implementability, and cost. Process option evaluations are presented in Tables 3-7 and 3-8 for each of the soil/waste and ground water matrices. The selected process options are indicated with an asterisk in Tables 3-7 and 3-8, and are summarized in Table 3-9.

THIS PAGE LEFT INTENTIONALLY BLANK

j,

ĩ

ց,

4.0 DEVELOPMENT AND SCREENING OF ALTERNATIVES

The technologies and process options developed in Section 3 are combined to form a range of remedial alternatives which address site cleanup to varying degrees and meet the criteria set forth in the NCP for the types of remedial alternatives which must be considered. These criteria include the following:

- For alternatives which provide control of the source of contamination, the range of alternatives should include the following:
 - A range of alternatives in which treatment that reduces the toxicity, mobility, or volume of the hazardous substances is a principal element. This range should include an alternative that removes or destroys hazardous substances to the maximum extent feasible, eliminating or minimizing the need for long-term management.
 - One or more alternatives that involve little or no treatment, but provide protection of human health and the environment primarily by preventing or controlling exposure to hazardous substances through engineering controls and/or institutional control.
- For ground water response actions, a limited number of remedial alternatives should be developed that attain site-specific remediation levels within different restoration time periods utilizing one or more different technologies.
- The development of one or more innovative treatment technologies for further consideration.
- The no action alternative.

For the alternatives which are developed, general descriptions of the alternatives are provided. The alternatives then undergo an initial screening process. These activities are described further in Section 4.2.1.

4.1 Development of Alternatives

Remedial response objectives, as presented in Section 3.1, are used as a guide in the development of remedial alternatives. It is at this point in the Feasibility Study that medium-specific actions are combined to form sitewide remedial alternatives. Table 4-1 summarizes the alternatives which have been developed for the Hertel Landfill site. Table 4-2 presents the technologies which passed the screening process and depicts the way the technologies have been combined into remedial alternatives which meet the NCP criteria for the types of alternatives to be considered.

4.2 Screening of Alternatives

For each of the alternatives presented in Table 4-2, a description of the alternative and an evaluation of the alternative based on effectiveness, implementability and cost criteria are presented. Following this individual screening and analysis, a comparative analysis of the alternatives is performed based on the three screening criteria, and the alternatives to be retained for detailed analysis are selected.

4.2.1 Introduction

The alternative descriptions presented below include such information as the location of excavation or containment areas and the volumes of soil, waste and ground water to be collected, excavated or treated. The thought process used in the development of alternatives is also described. Process options which passed the process screening but are represented by another process option in the alternative are noted.

A preliminary screening of alternatives is performed subsequent to alternative description. The objective of alternative screening is to narrow the list of potential alternatives that will be evaluated in detail, by evaluating and comparing them on the basis of effectiveness, implementability and cost. This screening aids in streamlining the feasibility study process while ensuring that the most promising alternatives are being considered. The

TAMS/TRC

range of treatment alternatives initially developed is retained, where practicable, through the screening process, with comparisons typically focusing on similar alternatives, the most promising of which is carried forward for further analysis.

The effectiveness screening evaluates the effectiveness of each alternative in protecting human health and the environment through reduction of toxicity, mobility or volume. Both long- and short-term effectiveness is considered. Evaluation on the basis of implementability takes into consideration the technical and administrative feasibility of constructing, operating and maintaining a remedial action alternative. The final evaluation criterion, cost, involves the estimation of both capital and operation and maintenance (O&M) costs associated with each alternative. Because of the level of refinement of alternative development, cost estimates may not be as accurate as those developed during the detailed analysis of alternatives. However, estimates are comparative in terms of relative accuracy to allow cost decisions to be made at this point.

4.2.2 Common Major Elements

Several of the alternatives described in the following sections have common major elements which do not change in scope from one alternative to another. These elements are initially described below and are then referenced in the individual alternative descriptions which follow. The major elements common to several alternatives include:

- Ground water monitoring
- Ground water extraction

FS REPORT

Ground Water Monitoring

Ground water monitoring would involve the continued monitoring of ground water quality at the site and downgradient of the site. The purpose of this monitoring program would be to identify changes in on-site ground water quality or off-site migration of contamination, and to predict potential impacts to surrounding private potable wells. Both the deep overburden and bedrock ground water quality would be monitored on a quarterly basis. Wells to be sampled would include select existing wells and additional wells to be installed. The installation of additional monitoring wells would be required to further refine ground water flow directions and provide additional downgradient water quality information. For the purposes of the initial screening, it is assumed that twelve wells will be monitored (five bedrock and seven overburden wells), of which four bedrock wells and two overburden wells will be new wells.

Also included in the ground water monitoring program is the sampling and analysis of downgradient private potable wells. It is assumed that fifteen wells will be sampled on an annual basis to confirm that private well water quality is not impacted by contaminant migration.

Ground Water Extraction

Ground water extraction would involve the extraction of contaminated ground water using extraction wells. French drains, another ground water extraction process option, could also provide a means of extraction for contaminated shallow ground water, if necessary. The purpose of a ground water extraction system would be to depress the water table to a level below the bottom of the fill materials, in order to minimize contact between the fill material and the ground water, and thereby minimize leachate production.

TAMS/TRC

The extracted ground water would then, in most cases, be treated on-site and discharged to surface water. Due to the relatively impermeable subsurface conditions identified during remedial investigation well development and sampling, a preliminary evaluation indicates that approximately 22 extraction wells will be required to capture the contamination. The estimated total pumping rate for these wells is approximately 14,000 gallons per day. Based on an assumed ground water treatment period equivalent to the time required to remove three pore volumes of water from the contaminated ground water area (see Section 3.2), and the estimated total contaminated ground water volume of 20 million gallons, it is estimated that the ground water extraction and treatment system would require operation over a 12-year period.

4.2.3 Alternative 1 - No Action

4.2.3.1 Description

The no action alternative would involve no remedial response activities at the Hertel Landfill site, although it would include continued ground water monitoring. No removal or treatment of currently impacted ground water or soil/waste materials would be conducted. Consideration of the no action alternative is required by the NCP.

4.2.3.2 Evaluation

Effectiveness

The no action alternative would provide no reduction in toxicity, mobility or volume of contamination. It would also provide no direct protection of human health or the environment, although continued ground water monitoring would allow for the identification of off-site contaminant migration and potential subsequent response actions.

TAMS/TRC

Implementability

The no action alternative with continued ground water monitoring would require the installation of additional monitoring wells and periodic sampling activities. Overall, it would be relatively easy to implement.

Cost

The only cost associated with the no action alternative would be that associated with monitoring well installation and continued ground water sampling. The cost of the no action alternative is initially estimated at a present worth value of \$2,500,000, assuming a 30-year monitoring period.

4.2.4 Alternative 2 - Site Use Restrictions and Capping

4.2.4.1 Description

This alternative was developed as an alternative which would meet the NCP's requirement for consideration of an alternative which utilizes containment with little or no treatment. The alternative consists of the institution of site use restrictions, including fencing and deed restrictions limiting future site development and on-site potable well installation. It also includes capping of the soil/waste matrix (landfill area) with a multi-layer cap to minimize infiltration, leachate production and any infiltration-induced migration of leachate.

A multi-layer cap, designed to meet the requirements of current New York solid waste landfill closure regulations, would be constructed over an area encompassing the areal extent of the existing landfill, as previously indicated in Figure 3-1. The total cap area would be approximately 12.2 acres. The cap would not encompass the one acre area identified as exhibiting surface debris only; the surface debris would be moved to within the area to be capped prior to cap construction.

TAMS/TRC

Process options which passed the screening and which are represented by the chosen process options are presented below:

Medium	Chosen Option	Other Options Which Passed Screening
Soil/Waste	Deed/access restrictions	None
	Multi-layer cap	Clay cap; synthetic cap; asphalt cap; concrete cap
Ground Water	Deed restrictions	Alternate water supply
	Multi-layer cap	Clay cap; synthetic cap; asphalt cap; concrete cap

4.2.4.2 Evaluation

Effectiveness

Alternative 2 would provide no reduction in toxicity or volume of contamination but it would limit the mobility of contaminant migration to some extent by eliminating infiltration as a leachate source. Exposure to potential risks posed by the site would be limited by ground water use restrictions and site use restrictions. Short-term effectiveness would be impacted by the required disruption of surface vegetation and soils during cap construction. Verification of long-term effectiveness would require continued ground water monitoring.

Implementability

Alternative 2 would be relatively easy to implement. It would require removal of all surface vegetation in the general landfill area, grading, and construction of a multi-layer cap. Overall implementability would be good.

Cost

The main cost factors associated with Alternative 2 would be the cap construction cost and the long-term ground water monitoring cost. The cost of the capping alternative is initially estimated at a present worth value of \$7,000,000.

4.2.5 Alternate 2A - Site Use Restrictions, Capping, Slurry Wall

4.2.5.1 Description

This alternative was developed as a slight variation of Alternative 2, and therefore is referred to as Alternative 2A. Like Alternative 2, it is a containment alternative involving no treatment. The only difference between Alternative 2 and Alternative 2A is the inclusion of a slurry wall as an additional containment measure. The slurry wall would be constructed upgradient of the landfill area to act as a barrier to ground water flow through the in-place waste materials. In conjunction with the leachate minimization provided by the multi-layer cap, the slurry wall would minimize contact between waste materials and the ground water, and thereby minimize the potential for contaminant migration off-site.

The only additional process option incorporated into Alternative 2A is the slurry wall. Because no other process options passed the screening, the slurry wall does not represent any other process options.

4.2.5.2 Evaluation

Effectiveness

The effectiveness of Alternative 2A would be similar to the effectiveness of Alternative 2, with the added benefit of minimizing ground water flow through the waste materials. The presence of an upgradient slurry wall would

lower the water table within the downgradient portions of the landfill. Therefore, the alternative would minimize leachate generation due to infiltration through the presence of a cap, and by limiting ground water flow through the waste materials.

Implementability

Alternative 2A would be more difficult to implement than Alternative 2. The construction of a slurry wall would be complicated by the presence of subsurface boulders and by the sloping surface of the upgradient side of the landfill. Flat areas are preferred for slurry preparation and backfill mixing. The presence of boulders near the bottom of a slurry wall may lead to variations in trench depth and reduced slurry wall efficiency which, in turn, could lead to piping failure.

Cost

The cost of Alternative 2A would be identical to the cost of Alternative 2, with the added cost of slurry wall construction. The cost of the capping/slurry wall alternative is initially estimated at a present worth value of \$15,000,000.

4.2.6 Alternative 3 - Excavation of Contaminated Soil and Waste Materials with Off-Site Disposal and Ground Water Collection with Off-Site Treatment

4.2.6.1 Description

This alternative was developed to meet the NCP's requirement for consideration of an alternative which minimizes the need for long-term management. The alternative consists of excavation of the contaminated soil/waste matrix within the general landfill area and transportation of the materials off-site for disposal at a licensed disposal facility. Contaminated ground water is also extracted and transported off-site for treatment at a licensed wastewater treatment facility. The long-term management associated with an on-site ground water treatment system is eliminated in this alternative, as compared to other alternatives which include ground water treatment.

Excavation and off-site transportation of the soil/waste matrix would require the handling of an estimated 300,000 cubic yards of material. Following excavation, the site would require grading and seeding to allow for revegetation of the excavated area. An off-site disposal site would have to be identified which would have sufficient capacity to receive the soil/waste materials.

Contaminated ground water would be extracted via ground water extraction wells as previously described in Section 4.2.2. As ground water would be extracted, it would temporarily be stored on-site for subsequent off-site transport to a commercial treatment facility. The identification of a facility licensed to accept the contaminated ground water for treatment would be required. Aqueous waste treatment facilities capable of treating organic and inorganic wastes which are located in the State of New York include CECOS International, Chemical Waste Management of New York and Frontier Chemical Waste Process, Inc. (Environmental Information, Ltd., 1990). All of these facilities are located in the Niagara Falls/Model City area of New York. For the purposes of this evaluation it is assumed that extracted ground water will be sent to CECOS International, where treatment processes include chemical precipitation, oxidation, devolatilization, carbon adsorption and advanced biological treatment.

Process options which passed the screening and which are represented by the chosen process options are presented below:

Medium	Chosen Option	Other Options Which Passed Screening
Soil/Waste	Off-site Landfill	On-site Landfill
Ground Water	Extraction wells	Well points, french drains
	Commercial treatment facility	None

4.2.6.2 Evaluation

Effectiveness

Alternative 3 would be effective in terms of reducing the volume of contaminated soil/waste matrix materials on-site but would provide no reduction in the toxicity of these materials. Mobility would be reduced by the containment features of the off-site disposal facility. Contaminated ground water would also be removed from the site, and a reduction in toxicity would be achieved at a commercial wastewater treatment facility. Long-term effectiveness would be good, based on the complete removal of contaminated soil/waste materials from the site and the treatment of contaminated ground water. In the short-term, additional exposures to waste materials would be realized during the excavation and off-site transport process.

Implementability

The administrative feasibility of identifying a permitted landfill with sufficient capacity to readily accept the estimated 300,000 cubic yards of soil/waste materials is very low. Additionally, the technical feasibility of transporting an estimated 14,000 gallons of contaminated ground water off-site daily over a period of twelve years is also very low.

Cost

Transportation and disposal/treatment costs for both the soil/waste matrix and the contaminated ground water are extremely high. The cost for the off-site landfill/off-site ground water treatment alternative is initially estimated at a present worth value of \$190,000,000.

4.2.7 <u>Alternative 4 - Site Use Restrictions, Capping, Ground Water</u> Extraction with On-Site Physical Treatment

4.2.7.1 Description

This alternative was developed as one of several alternatives which provide a range in terms of the extent of reduction in toxicity, mobility or volume of hazardous substances offered by the alternative. It consists of the institution of site use restrictions, including fencing and deed restrictions, and capping of the soil/waste matrix (landfill area) with a multi-layer cap to minimize infiltration, leachate production and any infiltration-induced migration of leachate. In these respects, Alternative 4 is identical to Alternative 2. However, Alternative 4 combines these containment features with a ground water extraction and on-site physical treatment system.

As with Alterative 2, a multi-layer cap, designed to meet the requirements of current New York solid waste landfill closure regulations, would be constructed over the 12.2 acre area extent of the existing landfill.

The ground water extraction system would consist of extraction wells, as described in Section 4.2.2. Upon extraction, however, the ground water would be treated on-site using physical treatment systems. Carbon adsorption is the chosen treatment process for organic contaminants, while chemical precipitation is the chosen treatment process for the inorganics. Following on-site treatment, the effluent would be discharged to surface water.

Process options which passed the screening and which are represented by the chosen process options are presented below:

		Other Options
Medium	Chosen Option	Which Passed_Screening
Soil/Waste	Deed/access restrictions	None
	Multi-layer cap	Clay cap; synthetic cap; asphalt cap; concrete cap
Ground Water	Deed restrictions	Alternate water supply
	Multi-layer cap	Clay cap; synthetic cap; asphalt cap; concrete cap
	Extraction wells	Well points; french drains
	Physical treatment:	
	Organic - carbon adsorption	Air stripping; resin adsorption
	Inorganics - precipitation	Ion exchange; membrane micro- filtration
	On-site discharge - Surface water	Injection wells; water infil- tration galleries

4.2.7.2 Evaluation

Effectiveness

The effectiveness of Alternative 4 would be similar to that of Alternative 2, the capping and site use restriction alternative, with the added benefit of ground water extraction and treatment. By extracting ground water, an additional element of hydraulic control would be provided. Ground water would be treated on-site using conventional treatment methodologies, and discharged to the surface water on-site. The long-term effectiveness of the ground water extraction and treatment system, which would be operated over an estimated twelve year period, would be expected to be good.

Implementability

Alternative 4 would be fairly easy to implement, although the administrative feasibility of discharging to surface water may pose problems. The capping and ground water treatment technologies are commonly used and should pose no significant barriers to implementation.

Cost

The cost for Alternative 4 is similar to that of Alternative 2, with the added cost of ground water extraction and treatment. The cost of the capping/site use restriction/physical ground water treatment alternative is initially estimated at a present worth value of \$8,200,000.

4.2.8 <u>Alternative 4A - Site Use Restrictions, Capping, Ground Water</u> Extraction with On-Site Innovative Treatment

4.2.8.1 Description

This alternative was developed as a slight variation of Alternative 4, and therefore is referred to as Alternative 4A. Like Alternative 4, it is a containment alternative which includes ground water extraction and treatment. The only difference between Alternative 4 and Alternative 4A is the type of ground water treatment considered. While Alternative 4 includes traditional physical treatment methods, Alternative 4A involves more innovative water treatment methods. Organic contaminants would be treated using UV oxidation, while inorganic contaminants would be treated using a membrane microfiltration process.

UV oxidation consists of a reactor module in which influent is exposed to UV radiation, ozone and hydrogen peroxide to oxidize organic compounds. The process is successful in treating organic constituents, especially halogenated organics. The inorganic treatment system consists of membrane

microfiltration. The microfiltration system uses an automatic pressure filter and Tyvek filter material to filter solid particles from liquid wastes.

The membrane microfiltration inorganic treatment method is a comparable process option to the chemical precipitation treatment method included in Alternative 4. Therefore, the only additional process option incorporated into Alternative 4A is the organic treatment method, UV oxidation. Because no other process options passed the screening for chemical treatment methods, the UV oxidation treatment process does not represent any other process options.

4.2.8.2 Evaluation

Effectiveness

The effectiveness of Alternative 4A in comparison to Alternative 4 is difficult to gauge due to the innovative nature of the ground water treatment systems employed. UV oxidation has been proven in initial testing in the treatment of organics. Membrane microfiltration has not been tested to the same degree as UV oxidation. However, based on the residual contamination exhibited in filtered ground water samples as compared to unfiltered samples, it is expected that the membrane microfiltration unit would be effective in the removal of inorganics from the ground water, although all ARARs may not be achieved. Treatability studies would be required to further define the effectiveness of these treatment methods.

Implementability

Due to the innovative nature of the ground water treatment systems, implementation would be somewhat more difficult than implementation of Alternative 4, which uses conventional treatment methods. Treatability studies would be required prior to final design of the treatment units. As

4–15

with Alternative 4, the administrative feasibility of discharging treated ground water to the surface water may pose a problem.

Cost

The cost for Alternative 4A is similar to that of Alternative 4, with the only difference being the cost of the ground water treatment system used. The innovative nature of the treatment system complicates the cost estimation process, since treatment costs are not yet well-established. The cost of the capping/site use restriction/innovative ground water treatment alternative is initially estimated at a present worth value of \$7,900,000.

4.2.9 <u>Alternative 5 - Excavation and On-Site Incineration of Soils and</u> Wastes, with Ground Water Extraction and On-Site Physical Treatment

4.2.9.1 Description

This alternative was developed as another of several alternatives which provide a range in terms of the extent of reduction in toxicity, mobility or volume of hazardous substances offered by the alternative. It includes excavation of the contaminated soil/waste matrix within the general landfill area, and subsequent on-site incineration of these materials. A rotary kiln incinerator is the chosen incineration process option. Contaminated ground water is also extracted and treated on-site using carbon adsorption for organics removal and precipitation for inorganics removal.

Excavation and on-site incineration of the soil/waste matrix would require the handling of an estimated 300,000 cubic yards of material. Assuming 240,000 cubic yards of the total consist of waste materials at a typical in-place density for municipal landfills of 800 lb/cu.yd. (Tchobanoglous, et.al., 1977), and the remaining 60,000 cubic yards consist of soils at an average density of 100 lb/cu.ft., the total tonnage of materials requiring

incineration would be 177,000 tons. Based on this estimated tonnage, a large incinerator (greater than 40,000,000 Btu/hr) would be most cost effective to use (Cudahy, 1989). There are several vendors which supply large rotary kiln incinerators.

Following excavation and incineration, the incinerator residue would require testing and handling in accordance with federal and New York State regulations. For the purposes of the evaluation of this alternative, it is assumed that the ash would be disposed of at a hazardous waste landfill. The excavation area would subsequently require grading and seeding to allow for revegetation.

Ground water extraction and on-site treatment would be as described for Alternative 4.

Process options which passed the screening and which are represented by the chosen process options are presented below:

Medium	Chosen Option	Other Options Which Passed Screening
Soil/Waste	Rotary kiln incinerator	Fluidized bed; infrared incinerator
Ground Water	Extraction wells	Well points, french drains
	Physical treatment:	
	Organic - carbon adsorption	Air stripping; resin adsorption
	Inorganics - precipitation	Ion exchange; membrane micro- filtration
	On-site discharge - Surface water	Infiltration galleries, injection wells

4.2.9.2 Evaluation

Effectiveness

Alternative 5 would be effective in that the source of ground water and surface water contamination would be removed and treated. Soil and waste materials would be excavated and treated on-site in an incinerator. Organic contaminants would be destroyed in the incineration process. Inorganics would remain in the ash residue, which would be transported off-site for disposal at a licensed land disposal facility. Short-term effectiveness would be limited by the site disruption which would occur during excavation and by the air associated with incineration activities. The emissions long-term effectiveness of the ground water extraction and treatment system, which would operate over an estimated twelve year period, would be expected to be good.

Implementability

The technical implementability of this alternative would depend on the commercial availability of the required incineration system, while the administrative implementability would be dependent on the ability of the system to meet the substantive requirements applicable to incinerators. Mobile incinerators are available from vendors but the public opposition to such a remedial response could be expected to be significant.

Cost

The cost for Alternative 5 would be high, due to the high cost of mobilizing and operating an on-site incinerator and due to the costs associated with off-site disposal of the ash residues. The cost of the on-site incineration/physical ground water treatment alternative is initially estimated at a present worth value of \$120,000,000.

4.2.10 <u>Alternative 6 - Excavation and Off-Site Treatment of Soils and</u> Wastes, Ground Water Extraction and On-Site Biological Treatment

4.2.10.1 Description

This alternative was developed as another of several alternatives which provide a range in terms of the extent of reduction in toxicity, mobility or volume of hazardous substances offered by the alternative. It includes excavation of the contaminated soil/waste matrix within the general landfill area, and subsequent off-site treatment of these materials. The specific type of treatment offered would be dependent upon the vendor selected to provide the treatment; however, the vendor would be required to meet certain performance standards and comply with regulations such as land ban regulations in meeting these standards. For the purposes of evaluating this alternative, an off-site rotary kiln incinerator is used as a representative potential treatment option. Contaminated ground water is also extracted and treated on-site using a biological treatment system for treatment of organic contaminants and precipitation for treatment of inorganics.

Excavation and off-site treatment of the soil/waste matrix would require the handling of an estimated 300,000 cubic yards of material. Following excavation and off-site transport, the excavation area would require grading and seeding to allow for revegetation. Based on the assumed incineration treatment method, the soil/waste materials would be transported to a licensed incineration facility. The nearest licensed hazardous waste incinerators include Rollins Environmental Services Inc., located in Bridgeport, New Jersey and Ross Incineration Services Inc., located in Grafton, Ohio. The incinerator residue would require testing and handling in accordance with federal and state regulations and would be handled by the operator of the incinerator.

4–19

Ground water extraction would be as described in Section 4.2.2. Treatment would be accomplished using a biological treatment system, a bioreactor, for treatment of organics. For treatment of inorganics, chemical precipitation is the chosen treatment method.

Process options which passed the screening and which are represented by the chosen process options are presented below:

Medium	Chosen Option	Other Options Which_Passed_Screening
Soil/Waste	Off-site treatment	Various
Ground Water	Extraction wells	Well points, french drains
	Organics - Bioreactor	None
	Inorganics - Precipitation	Ion exchange; membrane micro- filtration
	On-site discharge - Surface water	Infiltration galleries; injection wells

4.2.10.2 Evaluation

Effectiveness

The effectiveness of this alternative would be dependent upon the off-site treatment method employed. For the purposes of this evaluation, it has been assumed that off-site incineration will be the proposed treatment method. The effectiveness of this alternative would be comparable to that of Alternative 5, since the same soil/waste treatment methods are used.

Implementability

Alternative 6 would not have the difficulties associated with on-site incineration, as described for Alternative 5, but would have difficulties

associated with its implementation due to the required off-site transport of excavated materials and availability of off-site incineration facilities.

Cost

The cost for Alternative 6 would be high, due to the high cost of transportation and off-site incineration of the soil/waste materials. The cost of the off-site soil/waste treatment/physical ground water treatment alternative is initially estimated at a present worth value of \$510,000,000.

4.2.11 Alternative 7 - In Situ Vitrification with Ground Water Extraction and On-Site Innovative Treatment

4.2.11.1 Description

This alternative was developed as an alternative which employs innovative treatment technologies. It consists of in situ vitrification of the soil/waste matrix and ground water extraction with on-site treatment. The ground water treatment methods employed include UV oxidation for treatment of organics and membrane microfiltration for treatment of inorganics.

In situ vitrification is a process where an electrical network is used to melt the soil/waste materials at temperatures of 1600° to 2000° C. Organic pollutants are pyrolized while inorganics are immobilized within the vitrified mass, which forms a glasslike monolith upon cooling. The ground water extraction system was previously described in Section 4.2.2. The proposed ground water treatment system was previously described in Section 4.2.8 for Alternative 4A.

Process options which passed the screening and which are represented by the chosen process options are presented below:

Medium	Chosen Option	Which Passed Screening
Soil/Waste	In situ vitrification	None
Ground Water	Extraction wells	Well points, french drains
	Organics - UV oxidation	None
	Inorganics - Membrane microfiltration	Ion exchange; precipitation
	On-site discharge - Surface water	Infiltration galleries; injection wells

4.2.11.2 Evaluation

Effectiveness

The innovative nature of Alternative 7 complicates the evaluation of the effectiveness of the alternative. The effectiveness of in situ vitrification as applied to a municipal waste landfill has not been previously proven. As described for Alternative 4A, while UV oxidation has been tested successfully in treating organic-contaminated ground water, membrane microfiltration is not as fully developed. The long-term effectiveness of in situ vitrification, if effective in its application, is expected to be good, due to the glasslike monolith that is formed by the process.

Implementability

The implementability of this alternative is limited by the limited availability of vendors which provide the treatment systems. The administrative feasibility could also be limited by the unproven nature of these treatment methodologies. Treatability studies would be required to confirm the effectiveness of these methodologies prior to final design and implementation.
Cost

The cost for Alternative 7 would be expected to be high, mainly due to the mobilization and operation expenses of the in situ vitrification system. The cost of the on-site soil/waste treatment/physical ground water treatment alternative is initially estimated at a present worth value of \$110,000,000.

4.3 Selection of Alternatives for Detailed Analysis

A comparative analysis of the individual alternative screenings based on the three evaluation criteria is conducted to allow the elimination of certain alternatives from the detailed analysis process.

4.3.1 Effectiveness

With respect to long-term effectiveness, those alternatives which involve reductions in the toxicity, mobility or volume of contamination and contaminant sources will provide the greatest protection. With respect to short-term effectiveness, those alternatives which are protective during the construction and implementation period are most effective.

For the remedial alternatives developed, those alternatives which provide the greatest long-term effectiveness, due to removal/treatment of contaminated soils and waste materials, typically provide the least amount of short-term protectiveness, due to the required disruption of the waste materials. Alternatives 5, 6, and 7 provide the greatest long-term effectiveness with Alternatives 5 and 6 having similar short-term potential adverse impacts associated with their implementation. Alternative 3 provides long-term effectiveness with respect to the Hertel Landfill site, but since it involves no reduction in soil/waste toxicity, it may not be protective of the environment at the off-site disposal area. Alternatives 2 and 4 are similar

in the degree of short-term effectiveness provided, with less short-term impacts than those expected for soil/waste excavation alternatives. Alternatives 4 and 4A provide a reduction in ground water contamination but do not address remediation of the soil/waste matrix. Alternatives 2 and 2A do not provide an active ground water remediation system, but their containment features could result in a long-term reduction in mobility of contamination. Alternative 1, the no action alternative, provides no treatment or containment of contamination. However, through continued ground water monitoring, potential migration of contamination to off-site private wells could be identified prior to impacting human health.

4.3.2 Implementability

Implementability is a measure of the technical and administrative feasibility of constructing, operating, and maintaining a remedial action alternative. Alternative 1 is the most implementable alternative from a construction standpoint. Alternatives 2 and 4 are also fairly readily implemented. The administrative feasibility of Alternatives 3 and 6 is limited by the lack of available off-site disposal/treatment sites with sufficient capacity to accept the entire volume of materials to be excavated from the Hertel Landfill site. Alternative 5 is technically feasible; the administrative feasibility is not as great as for the containment alternatives, but it is feasible. The technical feasibility of Alternative 7 is questionable due to the unproven nature of in situ vitrification with respect to municipal waste materials. The implementability constraints could provide the basis for elimination of Alternatives 3, 6 and 7 from further consideration.

4.3.3 Cost

Preliminary remedial costs for the individual alternatives are summarized in Table 4-3. Alternative 1 is the lowest cost alternative. Alternatives 2, 4 and 4A are comparative in terms of cost. Alternative 2A is slightly higher in cost. Alternatives 5 and 7 are an order of magnitude higher in cost than Alternatives 1, 2 and 4 but they provide treatment of the contamination source that the other alternatives do not. Alternatives 3 and 6 are significantly greater in cost than the other alternatives evaluated, which could provide the basis for elimination of these alternative from further consideration.

4.3.4 Selection of Alternatives for Further Consideration

As described in the previous sections, Alternative 3 may be considered for elimination from further analysis on the basis of implementability and cost. Because Alternative 3 provides a level of effectiveness similar to Alternative 4 in terms of soil/waste containment and ground water treatment, it will not be retained for further consideration.

Alternatives 5, 6, and 7 (excavation & onsite incineration, excavation and off-site treatment, in-situ vitrification, respectively) will be deleted from further analysis in consideration of the following factors:

- a) In accordance with the NCP, containment technologies are generally deemed appropriate for waste that poses a relatively low long-term threat or where treatment of the waste is impracticable (e.g., mixed waste of widely varying composition, as is generally found at municipal landfill sites).
- b) Similarly, areas of highly toxic and/or mobile material that constitute a principal threat and would justify treatment, have not been identified on this site.
- c) The costs of construction and the long-term costs to operate and maintain these alternatives are grossly excessive when compared to their overall effectiveness.

THIS PAGE LEFT INTENTIONALLY BLANK

ť

۲Ż

5.0 DETAILED ANALYSIS OF ALTERNATIVES

Based on the initial screening of alternatives presented in Section 4, three alternatives are retained for further consideration. These alternatives are summarized in Table 5-1.

5.1 Introduction

In this section, the assembled remedial alternatives are described and evaluated in detail. The descriptions are presented by first discussing work components common to several alternatives, then describing each alternative completely. Detailed analyses are then presented. Per the requirements of the NCP, the detailed analyses are divided into seven evaluation categories, including the following:

• Long-Term Effectiveness and Permanence

- Reduction of Toxicity, Mobility and Volume through Treatment
- Short-Term Effectiveness
- Implementability
- Cost
- Compliance with ARARs
- Overall Protection of Human Health and the Environment

Detailed cost estimates for the remedial alternatives are presented in Appendix A.

The two remaining evaluation criteria required by the NCP, Support Agency Acceptance and Community Acceptance, are applied after completion of the FS.

Following individual analyses of the alternatives, a comparative analysis against the seven evaluation criteria is performed.

5.2 Common Elements and Considerations

Prior to the presentation of the specific alternative descriptions, a discussion of certain work elements common to a number of the alternatives, as

well as a discussion of certain considerations which may affect alternative implementation or assessment, is presented.

Mobilization

Prior to any major site work, equipment must be mobilized and the locations of on-site operations established. It is anticipated that, at a minimum, an office trailer will be necessary for control of site operations. Access to the site will be closely monitored; warning signs will be posted and temporary fencing will separate the area where greater health and safety precautions are necessary (the "hot" zone). The office trailer will be situated near the existing site access road. Heavy equipment will be stored in a restricted area.

Equipment Decontamination

Equipment decontamination facilities will be required during remedial construction activities. Provision of such facilities will require construction of a decon pit and collection of decon wastes. After completion of remedial activities, the residual wastewaters will require analysis and disposal at a permitted facilty.

Clearing

For alternatives which require construction activities in the landfill area itself, clearing of existing vegetation will be required. Additionally, some degree of clearing will be required in areas where additional monitoring wells or ground water extraction wells are proposed.

Run-On/Run-Off Controls

For a number of the alternatives considered, run-on/run-off controls will be required to prevent run-on from entering areas where remedial activities are being conducted and to likewise control run-off from remediated areas. Run-on/run-off controls will consist of ditching around the remedial areas and, for alternatives which potentially could result in significant erosion during construction activities, sedimentation control may be required to protect the on-site wetland areas. Sedimentation control could consist of sedimentation basins, interceptor dikes/berms, diversion channels or slope control. For the purposes of cost analyses, construction of a sedimentation basin has been assumed. In accordance with New York landfill closure regulations, drainage control structures associated with capping alternatives would be designed to protect the final cover from, at a minimum, the peak discharge of a 24-hour, 25-year frequency storm.

Access Road Reconstruction

Due to the flooding which currently occurs on the unpaved access road to the site, reconstruction and drainage improvements would be required prior to the implementation of any remedial alternatives requiring regular access to the site. Additional attention to road improvements would be required for those alternatives which require site access by heavy equipment.

Short-Term Risk Estimation for Common Work Elements

Short-term risks to remedial workers associated with incidental ingestion of soil, dermal contact with soil and inhalation of fugitive dusts were quantified in relation to road improvements, and clearing of vegetation from the landfill area. Fugitive dusts were assumed to originate from wind

TAMS/TRC

erosion, vehicular activity over unpaved roads and, in some instances, loading and dumping activities (EPA, 1985). These risks were estimated for workers without the use of personal protective equipment. Use of personal protective equipment would minimize short-term risks.

For potential carcinogens, risks are estimated as probabilities. The compound-specific potency factors for carcinogens are generally estimated through the use of mathematical extrapolation models (e.g., the linearized multistage model). These models estimate the largest possible linear slope, within a 95% confidence interval, at low extrapolated doses. Thus, the potency factor is characterized as a 95% upperbound estimate, such that the true risk is not likely to exceed the upperbound estimate and may be lower.

The evaluation of risk from noncarcinogenic health hazards is based on the use of RfDs (EPA, 1991; EPA, 1989). RfDs are estimates of daily exposure to the population (including sensitive subpopulations) that are likely to be without appreciable risk of deleterious effects for the defined exposure period. The RfD is calculated by dividing the NOAEL or LOAEL derived from animal or human studies by an uncertainty factor, which is multiplied by a modifying factor. RfDs incorporate uncertainty factors which serve as a conservative downward adjustment of the numerical value and reflect scientific judgment regarding the data used to estimate the RfD. For example, a factor of 10 is used to account for variations in human sensitivity (i.e., to protect sensitive subpopulations) when the data stems from human studies involving average, healthy subjects. An additional factor of 10 may also be used for each of the following:

- extrapolation from chronic animal studies to humans,
- extrapolation from a lowest observable adverse effect level (LOAEL) to a no observed adverse effect level (NOAEL), and

FS REPORT

extrapolation from subchronic to chronic studies.

Finally, based on the level of certainty of the study and database, an additional modifying factor (between zero and ten) may be used.

The results of the quantitative risk analysis are presented in two basic forms. In the case of human health effects associated with exposure to potential carcinogens, risk estimates are expressed as the lifetime probability of additional cancer risk associated with the given exposure. In numerical terms, these are presented in scientific notation in this report. Thus, a lifetime risk of 1E-04 means a lifetime incremental risk of one in ten thousand; a lifetime risk of 1E-06 means an incremental lifetime risk of one

In the cases of exposure to non-carcinogens, the Hazard Index Ratio is used. The fundamental principles used to construct the RfD utilized in calculating the Hazard Index Ratio are predicated on long term or chronic (usually measured in years) exposures and health effects.

Cancer and non-cancer health risks are discussed for scenarios associated with each remedial alternative. In each case, daily doses of the compounds of concern have been calculated for each exposure pathway modeled, and these doses were used to calculate cancer risk levels and hazard index ratios. Dose equations, assumptions and estimates are presented in Appendix B. Cancer risk levels are the lifetime probability of excess cancer due to the exposure pathways emanating from use of the site. Cancer risk levels are derived by multiplying exposure dose by the appropriate cancer slope factor for each compound and exposure route. Non-cancer health risk is quantitated by the hazard index ratio which is the ratio of the exposure dose to the RfD (both in mg/kg/day). The calculated level of cancer risk can be compared to the acceptable total site risk range (1E-04 to 1E-06) for evaluating the need for

TAMS/TRC

remediation, as stated in the "National Oil and Hazardous Substances Pollution Contingency Plan, Final Rule" (EPA, 40 CFR Part 300, March 8), and in the Superfund Human Health Evaluation Manual (1E-04 to 1E-07) (EPA, 1989a). Regarding non-carcinogenic health hazards the Superfund Human Health Evaluation Manual (EPA, 1989) states that:

"When the total hazard index for an exposed individual or group of individuals exceeds unity, there may be concern for potential non-cancer health effects."

Thus, the cancer risk and hazard index ratios that constitute a concern are >1E-04 and >1E+00, respectively.

Appendix B contains cancer risk and hazard index ratios for all scenarios, pathways and contaminants of concern. Tables 5-2 and 5-3 summarize cancer risk levels and hazard index ratios estimated for common remediation activities including road improvements and clearing of vegetation from the site. These activities will be incorporated into all proposed alternatives, and thus risks associated with these activities are included in the total risk associated with each remedial alternative.

Exposure of remedial workers to contaminants during road construction/ improvement and vegetation clearing is associated with a total cancer risk of 2E-06, which is within the target (acceptable) risk range (Table 5-2).

These site preparation activities are also associated with a total hazard index ratio of 10 (Table 5-3), which is an order of magnitude above the target HI value. Inhalation of airborne chemicals absorbed to fugitive dust is responsible for most of this risk.

Exposure equations, fugitive dust modeling and dose calculations used in evaluating these risks are presented in Appendix B.

Ground Water Monitoring

Quarterly ground water monitoring will be conducted for 5 years (i.e., 20 rounds of samples) subsequent to completion of ground water remediation for remedial alternatives which include ground water treatment (Alternatives 4 and 4A). Long-term (30 year) ground water monitoring is also included in the no action alternative (Alternative 1) and in capping alternatives which include no ground water treatment (Alternatives 2 and 2A). The monitoring system will consist of existing monitoring wells and six new wells to be installed. The proposed locations of additional ground water monitoring wells are provided in Figure 5-1. Of the six proposed wells, two are deep overburden wells and four are bedrock wells. The proposed wells have been located to provide a more comprehensive definition of water quality within the deep overburden and shallow bedrock. These wells, along with existing wells MW-1D, MW-2D, MW-10S, MW-11S, MW-W1D and MW-W2D, will comprise the ground water monitoring network. Also included in the ground water monitoring plan are annual sampling and analysis of private potable wells near the site. For costing purposes, sampling of fifteen private wells has been assumed.

Ground Water Extraction/Discharge

A ground water extraction/discharge system is proposed for several of the alternatives under consideration. The extraction of ground water will allow for the collection of contaminated ground water, and potentially floating product, for subsequent treatment and discharge. Discharge will be to surface water.

In arriving at the proposed extraction system configuration, the primary objectives were: 1) to capture and extract ground water from the entire areal extent of the landfill, especially from areas where contaminant levels

TAMS/TRC

exceeding MCLs have been detected, 2) to depress the water table below the fill (by at least 6 feet in the central portion) so that the fill becomes effectively isolated from the saturated zone, and 3) to eliminate the breakout of leachate at the toe (eastern side) of the landfill. The extraction of ground water from the bedrock aquifer was not a design objective, since ARARs were not exceeded in bedrock well samples. Ground water extraction could result in upward migration of ground water from the bedrock towards the extraction wells.

The proposed ground water extraction well locations are provided in Figure 5-2. Further ground water studies would be conducted during the design stage, should an alternative involving ground water extraction be chosen for remediation of the site. Final design details (number of extraction wells, well locations, etc.) would be determined at that time.

A preliminary configuration of the proposed wellfield, as presented in Figure 5-2, was determined using an analytical computer model (THWELLS, Version 2.0) based on the Theis equation for uniform, non-steady flow in a homogeneous, isotropic confined aquifer of uniform thickness and infinite in areal extent. The assumed hydraulic parameters were as follows:

- Each well will penetrate 15 feet into the saturated zone, to allow for the extraction of shallow, contaminated ground water;
- The average hydraulic conductivity equals 1.2 feet per day, based on the hydraulic conductivities calculated for shallow wells located adjacent to and within the landfill area; and
- Specific yield equals 0.2 (dimensionless).

In determining the well spacing, it was considered desirable, if possible, to use an individual well pumping rate of less than 500 gallons/day, to allow for poor well performance in zones of low hydraulic conductivity.

Using the model, it was determined that, within the landfill, a wellfield consisting of 21 extraction wells, spaced 100 feet apart and each pumping 325 to 350 gallons/day, would provide adequate capture of the contaminated ground water, isolate the fill materials from the water table, and mitigate the leachate breakout. In the area of contamination around well MW-13S, using an average hydraulic conductivity of 12.5 ft/day, it was determined that one well, located midway between the adjacent wetlands and extracting approximately 700 gallons/day, would provide capture for the contaminated ground water in that area. A more detailed discussion of the THWELLS model, including the model's input and output, is provided in Appendix C.

The estimated total withdrawal rate for the proposed extraction system, based on the computer model, is approximately 8,000 gallons/day (5.6 gallons per minute or gpm). A total extraction rate of 10 gpm (14,400 gallons/day) will be used as a basis for the design of the ground water treatment system. The amount of time required to remove a volume of water equal to the contaminant plume volume based on the 10 gpm extraction rate is approximately 3.8 years. A ground water remediation period of twelve years will allow for the removal of three pore volumes.

Ejector-type pumps have been included in the cost estimates for the ground water extraction system. The pumps fill by gravity and discharge using compressed air. With a minimal number of moving parts, ejector pumps are suitable for pumping at very low flow rates and can adapt to variable flow rates as well. Ejector pumps operate pneumatically; therefore, use of electricity in the immediate landfill area may be minimized. Ejector pumps are also designed specifically for the extraction of floating product phases.

Compliance with Wetlands Requirements

The New York Department of Environmental Conservation Freshwater Wetlands Permit Requirements Regulations (6 NYCRR Part 663) identify the guidelines and procedural requirements applicable to various activities conducted in or adjacent to freshwater wetlands. The regulations identify activities which require permits prior to implementation as well as the relative compatibility of the activities with wetlands and their adjacent areas. While CERCLA Section 121(e) states that no federal, state or local permits are required for any removal or remediation activities conducted entirely on-site, remedies must meet the substantive requirements of the law. A wetland evaluation will be performed prior to initiation of Remedial Design so that a complete assessment of potential wetland impacts can be made.

Activities which are part of the following remedial alternatives and which may be impacted by the Wetlands Permit Requirements include the following:

- Draining and altering water levels;
- Constructing, expanding or substantially modifying drainage ditches;
- Clear-cutting trees and vegetation other than trees;
- Grading;
- Constructing roads;
- Drilling a well;
- Constructing a building; and
- Installing utility service.

Requirements vary depending on whether the activity is conducted within the wetland area or within an adjacent area, defined as within 100 feet, measured horizontally, of the boundary of the wetland. Activities which require a permit and are identified in the regulations as being compatible or usually incompatible must meet three test conditions for a permit to be issued. These include: (i) the activity would be compatible with preservation, protection and conservation of the wetland and its benefits; <u>and</u> (ii) would result in no more than insubstantial degradation to, or loss of, any part of the wetland,

TAMS/TRC

and (iii) would be compatible with the public health and welfare. Activities which require a permit and are identified in the regulations as being incompatible must meet certain standards based on the class of wetland affected before a permit will be issued. As stated earlier, remedial activities at the Hertel Landfill site must meet the substantive requirements of the wetlands regulations.

5.3 Individual Analysis of Alternatives

In Section 4.2, preliminary descriptions of alternatives, documentation of the logic behind alternative selection, and a discussion of acceptable process options were presented. Prior to conducting a detailed analysis of remaining alternatives, the various elements of the alternatives, implementation of the elements, the way the elements would be combined to form a complete alternative, the time frame required to achieve cleanup and the common elements included in each alternative are described. Following the alternative description, the alternatives are evaluated against each of the nine evaluation criteria described in Section 5.1.

5.3.1 Alternative 1 - No Action with Continued Ground Water Monitoring

5.3.1.1 Description

The no action alternative involves no remedial actions to reduce toxicity, mobility or volume of contamination at the Hertel Landfill site. The site would remain in its present condition, with no additional barriers to access. The alternative does include the installation of additional monitoring wells and continued ground water monitoring, to identify off-site migration of ground water contamination, should it occur. A description of the ground water monitoring network was provided in Section 5.2.

TAMS/TRC

5.3.1.2 Criteria Assessment

Short-Term Effectiveness - The short-term impacts of the no action alternative would be unchanged from the current risks posed by the Hertel Landfill site. No elevated short-term risks would result from the implementation of this alternative; however, the potential for human and environmental exposure would not be reduced either. A discussion of the risks posed by current environmental conditions at the site is presented in Section 1.5. Based on the current/recreational use scenario, cancer and non-cancer risks exceeding acceptable risk ranges were identified based on dermal contact with arsenic (based on the arsenic concentration detected in a background sample) and PAH compounds in the soil, for children and adult receptors. Remedial action objectives would not be achieved.

Long-Term Effectiveness and Permanence - The no action alternative would provide no reduction in risk and does not utilize controls to treat or manage contamination or contaminant sources. A discussion of the risks posed by current environmental conditions at the site is presented in Section 1.5. In addition to the risks posed by the site under the present use scenario, risks would be exceeded for future site development scenarios. Long-term (30-year) ground water monitoring is included in the no action alternative to provide an indication of potential ground water contaminant migration off-site towards private wells. While long-term monitoring is likely to identify any off-site migration, it provides no remedial action should migration be identified. The selection of the no action alternative would require review should continued monitoring identify migration towards off-site potable wells or should development of the site be proposed. The NCP [300.430(f)(4)(ii)] requires that, if a remedial alternative is selected which results in hazardous

substances remaining at the site above levels that allow for unlimited use and unrestricted exposure, the lead agency shall review such action no less than once every five years after initiation of the alternative (i.e., six reviews in a 30 year period; therefore, at a minimum, a five-year review would be required if the no action alternative were implemented.)

<u>Implementability</u> - The no action alternative is the most easily implemented alternative. The technical feasibility of this alternative is limited only by the difficulties in installing deep monitoring wells which were encountered during the remedial investigations. To address these difficulties, tubex drilling, which performed satisfactorily in the RI, is the proposed method of well installation. Tubex drilling services and well construction/monitoring materials are readily available. Once the wells are installed, continued ground water monitoring is easily implemented and future remedial action, if required, would not be hampered by other on-site remedial activities. No administrative feasibility problems would be expected if this alternative were selected.

Reduction of Toxicity, Mobility or Volume Through Treatment – The no action alternative does not include any treatment methods other than naturally occurring degradation processes such as in situ biological degradation and volatilization of organic contaminants. Considering the length of time in which the landfill, the source of contamination, has been in-place at the site, the short-term risks associated with providing no treatment of the source are not expected to pose imminent hazards to human health or the environment above those identified in the baseline risk assessment. However,

long-term risks may be associated with the lack of containment/treatment provided by the no action alternative.

<u>Compliance with ARARs</u> - The no action alternative does not attain all chemical-specific ARARs for contaminants detected in the ground water, surface water or sediment (no federal or state chemical-specific ARARs for soils are available). It should be noted, however, that ARARs may be waived under certain circumstances called out in the NCP, including situations where compliance with the requirement will result in greater risk to human health or the environment than other alternatives, where compliance is technically unpracticable from an engineering perspective or where the alternative will attain a standard of performance that is equivalent to that required under the otherwise applicable standard, through use of another method or approach.

Overall Protection of Human Health and the Environment - The no action alternative may not provide long-term protection of human health and the environment, because it does not address potential risks through the elimination, reduction, or control through treatment, engineering or institutional controls of the source of identified contamination. Risks associated with dermal contact with soils based on the current recreational Similarly, potential future risks use of the site are not addressed. associated with ground water ingestion are not addressed. However, through continued ground water monitoring, the alternative provides the opportunity to review the no action decision, should monitoring ever identify ground water contaminant migration, and allows for a future review of the additional potential risks to human health and the environment associated with the identified migration. A review of the remedial decision is required at a

minimum of every five years under the NCP's requirements for sites where contaminants remain following remediation. Should the site ever be developed for an alternative use, protection of human health and the environment may not be provided.

<u>Cost</u> - The only cost associated with the no action alternative would be the cost of monitoring well installation and continued monitoring of ground water quality. Due to the lack of treatment afforded by the no action alternative, a 30-year monitoring period has been assumed. The cost estimate includes \$52,000 in direct capital costs, \$6,000 in indirect capital costs and \$132,000 in annual operation and maintenance costs (\$2,033,000 present value). The total present worth value of this alternative, including contingency, is estimated at \$2,509,000. A detailed cost estimate is presented in Appendix A.

5.3.2 Alternative 2 - Site Use Restrictions and Capping

5.3.2.1 Description

Alternative 2 consists of the institution of site use restrictions and capping of the landfill area. It includes the following common work elements described in Section 5.2: Mobilization, Equipment Decontamination, Clearing, Run-on/Run-off Controls, Access Road Reconstruction, Ground Water Monitoring, and Compliance with Wetlands Requirements.

Site use restrictions would include deed restrictions limiting future site development and use, restrictions on the installation of potable wells on-site, and access restrictions in the form of secure fencing around the landfill portion of the site and posting of warning signs.

Capping of the site would initially require clearing of the landfill area and grading and filling of the area to provide an evenly graded surface. It

is estimated that 80,000 cubic yards of fill will need to be brought to the site in order to accomplish this. By providing a consistently sloped base for cap construction, the final cap would subsequently promote runoff. The existing topography is approximately at an average 8% slope across the landfill area, with individual areas ranging as steep as a 33% slope. The landfill area would be surface-graded to an average 5% to 8% slope during cap construction activities. Figure 5-3 indicates the landfill area to be capped under this alternative.

The cap itself would be designed in accordance with the closure requirements set forth in 6 NYCRR Part 360 for Solid Waste Management Facilities. The final cap would consist of a gas venting layer (bounded on its upper and lower surfaces by a filter layer), a low permeability barrier soil cover or geomembrane cover, a barrier protection layer, and topsoil. Following construction of the cap, the surface would be stabilized by planting a vegetative cover. Figure 5-4 provides a cross-section of the final cap design assumed in preparing a cost estimate for this alternative. Evaluation of landfill gas properties would be required during the remedial design stage to determine if a passive gas venting system, as indicated in Figure 5-4, would be adequate or if treatment of the off-gases would be required.

5.3.2.2 Criteria Assessment

<u>Short-Term Effectiveness</u> - Implementation of this alternative would result in disruptions of surface and near-surface soil and waste materials. Tables 5-4 and 5-5 summarize cancer risk levels and hazard index ratios estimated for remediation associated with Alternative 2. The tables present these risks on a chemical-by-chemical basis so that the major factors which drive the risk can be readily ascertained.

TAMS/TRC

Exposure of remedial workers to contaminants while conducting activities associated with remediating the site is associated with a total cancer risk of 1E-05 (including site preparation activities) which is within the acceptable (target) cancer risk range (Tables 5-2 and 5-4).

Site remediation under Alternative 2 is associated with a total hazard index ratio of 20 (Tables 5-3 and 5-5), which is above the target HI value. Inhalation of airborne chemicals adsorbed to fugitive dusts is responsible for most of this risk.

Personal protective equipment and dust control measures could be implemented to minimize the potential for airborne contaminants to impact remedial workers and the surrounding community, respectively. Likewise, surface run-off controls could minimize the impacts of surface water run-off to wetlands and the stream during implementation. The time required to complete construction of the final cap is estimated at four months.

Implementation of site use restrictions would have no short-term adverse impacts to the adjacent community or the environment. Fence construction activities would take place outside of the contaminated limits and would not be expected to pose a risk to construction workers. Construction of the fence could be conducted within the period that cap construction was conducted.

Long-Term Effectiveness – Alternative 2 would not treat the sources of contamination at the site, although it would provide containment through the elimination of exposure to surface soil contaminants and minimization of the infiltration of precipitation through the waste materials. Site use restrictions would limit the accessibility of the site and also limit exposure to identified on-site risks. Long-term residual risks associated with Alternative 2 were addressed in relation to future recreational use of the

site following completion of proposed remedial measures. Potential receptor populations and exposure routes include children playing in or near surface water resulting in incidental ingestion and dermal contact with sediments.

Exposure of children to contaminants while playing on-site is associated with a total cancer risk of 2E-05 (Table 5-6), which is within the target risk range. Similarly, the HI value associated with future recreational use of the site is 0.1, which is also less than the target HI value of 1 (Table 5-7). An additional potential future use exposure includes ingestion of ground water from potable private wells downgradient from the site. Results obtained from the field investigations indicate no current impacts on private wells. However, because this alternative does not address ground water flow through waste material, there is a potential for future impacts.

Future activities which could compromise the integrity of the cap would not be allowed through the implementation of deed restrictions. The NCP requires that the lead agency shall review such an action no less than once every five years after initiation of the alternative (see Section 5.3.1.2). Therefore, a five-year review would be required if Alternative 2 were implemented in order to confirm the continued protectiveness of the alternative.

Long-term management, in the form of maintenance of the cap, drainage control system and ground water monitoring system, would be required. The technologies involved in capping a site are not extremely complicated and the maintenance of such a system following installation should not be difficult. Similarly, replacement, should the cap fail or be damaged, would not be expected to pose risks greater than those of initial installation and should not be difficult to repair.

Implementability - The technical feasibility of cap and fence construction Current site conditions provide minimal obstacles to is good. cap construction, with the exception of the presence of large boulders on-site. Capping technologies are well-developed and fairly reliable, as long as the cap is maintained and site uses are restricted to limit potential damage to the cap system. Due to the relative simplicity of the cap's design, reliability is expected to be good. Certain future remedial actions, such as excavation and disposal/treatment, could not be implemented without compromising the integrity of the cap. However, the presence of the cap does not preclude the use of other remedial technologies and could enhance the implementation of certain technologies, such as ground water extraction. Equipment and supplies necessary to construct a cap and fence are available, although some lead time would be required to identify supply sources, due to the significant volumes of capping materials required.

Administrative feasibility of cap and fence construction is good, although the administrative feasibility of instituting site use restrictions requires the proper authority and coordination within state and local agencies. Site use restrictions have been included in recommended remedial action decisions at other CERCLA sites in New York.

Reduction of Toxicity, Mobility and Volume Through Treatment - Alternative 2 provides no treatment nor associated reduction in contaminant toxicity. The provision of an impermeable cap will limit the infiltration of precipitation and the subsequent volume of leachate production, however, and will therefore limit the mobility of contaminant migration to a certain degree. However, the presence of the water table within the fill materials is not mitigated by this

alternative, thereby providing a continued potential source of ground water contamination.

Compliance with ARARs Alternative 2 does not attain a11 chemical-specific ARARs for contaminants detected in the ground water, surface water, or sediment (no federal or state chemical-specific ARARs for soils are available). As described in Section 5.3.1.2, ARARs may be waived under certain circumstances. By designing the final cap system to meet New York Solid Waste Management Facility closure regulations, the alternative will comply with action-specific ARARs. Access road reconstruction, final cap construction or other remedial activities, as previously identified in Section 5.2, conducted within 100 feet of the boundary of a wetland will also have to be conducted in compliance with New York Freshwater Wetlands Regulations.

Overall Protection of Human Health and the Environment - This alternative provides short-term and long-term protection of human health and the environment through the utilization of engineering and institutional controls to limit leachate production and site accessibility. However, it does not address potential risks associated with the possible off-site migration of existing ground water contamination through either ground water treatment or volume reduction. Through continued ground water monitoring, it does provide the opportunity to review the decision, should monitoring ever identify off-site ground water contaminant migration, and allows for a future review of the additional potential risks to human health and the environment associated with the identified migration. Such a review is required at a minimum of every five years under the NCP's requirements for sites where contaminants remain following remediation.

FS REPORT

<u>Cost</u> - The cost associated with Alternative 2 includes site clearing, grading, cap construction and maintenance, and long-term monitoring costs. These costs include \$3,054,000 in direct capital costs, \$427,000 in indirect capital costs and \$163,000 in annual operation and maintenance costs (\$2,503,000 net present value). The present worth value of this alternative, including contingency, is estimated at \$7,182,000. A detailed cost estimate is presented in Appendix A.

5.3.3 Alternative 2A - Site Use Restrictions, Capping, Slurry Wall

5.3.3.1 Description

For Alternative 2A, site use restrictions and capping would be as described for Alternative 2 in Section 5.3.2. Therefore, the description presented herein addresses the only component not evaluated previously, the slurry wall. A slurry wall would be constructed upgradient of the landfill area to act as a barrier to ground water flow through the in-place waste materials. Figure 5-5 presents the proposed site layout for this alternative. Implementation of the following common elements, as described in Section 5.2, would also be required: Mobilization, Equipment Decontamination, Clearing, Run-on/Run-off Controls, Access Road Reconstruction, Ground Water Monitoring, and Compliance with Wetlands Requirements.

The impact of an upgradient slurry wall on the hydrogeology of a disposal site is presented in Figure 5-6. To determine the downgradient impacts of a slurry wall on the water table within the waste materials at the Hertel Landfill site, a numerical computer simulation was run using MODFLOW, the USGS three-dimensional finite difference flow model (McDonald and Harbaugh, 1988). The model was run as a one-layer simulation, and the aquifer parameters used were the same as those used in the analytical extraction wellfield model (Section 5.2). First, constant-head boundaries were placed upgradient and

downgradient from the landfill area to establish flow through the modeled site, and the model was roughly calibrated to the ground water contours for May 22, 1990 (Figure 3-6, RI Report, TAMS/TRC, 1991). Then, the nodes encompassing the proposed wall location were designated zero flux boundaries, and the model was rerun. The resulting nodal hydraulic heads were compared to the pre-wall head values to evaluate the wall's effectiveness in depressing the water table below the fill. The model indicated that the wall's effect would be to lower the water table at least 6 feet in the central portion of the landfill (beneath the base of the fill material), and that the wall would eliminate the leachate breakout at the toe of the landfill. It should be noted that, in simulating one layer only, the model did not account for potential upward ground water flow from the bedrock into the overlying materials.

The slurry wall design is dependent on site characteristics. Where low permeability is required, a soil/bentonite wall is preferred over a cement/ bentonite wall; however, soil/bentonite walls are typically limited to areas where the maximum slope along the trench line is on the order of 2% or less. Due to the existing slopes at the Hertel Landfill site, it is assumed that a cement/bentonite wall will be required.

The width of the wall is designed based on the hydraulic head across the trench. Generally, for a soil bentonite wall, the trench should "have a width of 0.5 to 0.75 feet per 10 feet of hydrostatic head on the wall" (Case, 1982). Because a cement/bentonite wall has greater strength, a wall of lesser thickness will stand up to the same hydrostatic pressure. For the purposes of this evaluation, it is assumed that the slurry wall will be approximately 1800 feet long, located as indicated in Figure 5-5. The wall will be keyed into the underlying bedrock, with an average depth of 40 feet, and the width of the wall will be 3 feet, based on an assumed maximum hydrostatic offset of 40 feet.

To monitor the effectiveness of the slurry wall system upon completion, the installation of observation wells will be required. Proposed observation well locations are provided in Figure 5-5. For the purposes of this evaluation, the installation of 8 observation wells has been assumed. These wells, along with existing monitoring wells in the fill area, will be monitored to confirm the effectiveness of the slurry wall in maintaining the ground water table at a level below the base of the fill material. If hydrostatic pressures upgradient of the slurry wall become excessive, drainage systems may be required to redirect the ground water around the slurry wall. In addition, in the potential event that the wall does not provide the desired water table depression within the landfill, it may be necessary to install extraction wells, either within or outside the walled area, to aid in this process.

5.3.3.2 Criteria Assessment

The following criteria assessments address, for the most part, only the evaluation of the slurry wall component of Alternative 2A; the remaining components were addressed during the assessment of Alternative 2, as presented in Section 5.3.2.

<u>Short-Term Effectiveness</u> - The construction of a slurry wall is not expected to result in significant impacts to the adjacent community or to remedial workers, since the slurry wall is located outside of the boundaries of the landfill area. Short-term risks associated with this alternative are similar to Alternative 2. Potential environmental impacts could result from the on-site construction activities. Run-off control measures would be required to protect wetlands and surface waters during construction.

TAMS/TRC

According to Miller (1979), soil/bentonite slurry walls are normally installed at rates of 25 to 100 linear feet per day. For the purposes of this evaluation, it is assumed that the wall will be constructed at the rate of 50 linear feet per day. Therefore, the time required to construct the slurry wall would be approximately 3 months, allowing for mobilization and demobilization periods. Assuming materials excavated during slurry wall construction could be used in grading the area of the cap installation and cap construction following completion of the slurry wall construction, the total construction time frame would be approximately 7 months.

Long-Term Effectiveness - As with Alternative 2, the construction of a slurry wall would not treat the sources of contamination at the site, although it would provide containment through the elimination of exposure to surface soil contaminants, and minimization of leachate production through the elimination of infiltration and ground water flow through in-place waste materials.

Long-term residual risks associated with this alternative are similar to Alternative 2. However, impacts on potable wells will be restricted to those stemming from ground water remaining under the cap following installation of the slurry wall. No new interaction of ground water with waste material would be expected.

The slurry wall is relatively maintenance-free but monitoring of ground water levels both upgradient and downgradient of the wall would be required to provide an indication of the wall's integrity. In uncontaminated environments, as is the case with an upgradient wall location, slurry walls have shown to be effective in the long-term. Per the requirements of the NCP (see Section 5.3.2), a five-year review would be required if Alternative 2A were

implemented, in order to confirm the continued protectiveness of the alternative.

Implementability - The technical and administrative feasibility of slurry wall construction is good. Slurry wall technologies are well-developed but construction at the site may be complicated by the presence of subsurface boulders and by the sloping surface of the upgradient side of the landfill. The presence of boulders near the bottom of a slurry wall may lead to variations in trench depth and reduced slurry wall efficiency. The proposed depth of the wall may also result in special construction equipment requirements. The presence of a slurry wall would have little effect on future remedial activities.

Reduction of Toxicity, Mobility and Volume Through Treatment - The construction of a slurry wall will provide no treatment nor associated reduction in contaminant toxicity. However, the slurry wall will lower the ground water table below the waste material, thus eliminating ground water/waste contact, and thereby preventing the flow of upgradient, uncontaminated ground water through the fill material. While existing ground water contaminants will not be treated, the potential for future generation of ground water contamination will be reduced.

<u>Compliance with ARARs</u> - The construction of a slurry wall will minimize migration of ground water through the landfill but will not attain all chemical-specific ARARs for contaminants currently detected in the ground water. As described in Section 5.3.1.2, ARARs may be waived under certain circumstances. Site disruption associated with slurry wall construction and

other activities previously identified in Section 5.2 will have to be conducted in compliance with location-specific wetlands requirements. The alternative will comply with action-specific ARARs.

Overall Protection of Human Health and the Environment - The addition of a slurry wall to the original components of Alternative 2 provides an additional degree of long-term protection of human health and the environment through the utilization of engineering controls (a slurry wall) to limit leachate production and ground water contamination. The slurry wall helps to prevent future contamination of ground water, but does not address the potential risks associated with existing ground water contamination through either ground water treatment or volume reduction. As with Alternative 2, a five-year review of the decision would be required under the NCP, should this alternative be chosen.

<u>Cost</u> - The cost associated with Alternative 2A is the same as Alternative 2, with the additional costs associated with slurry wall construction. The total costs for Alternative 2A include \$7,246,000 in direct capital costs, \$1,159,000 in indirect capital costs and \$171,000 in annual operation and maintenance costs (\$2,626,000 net present value). The present worth value of this alternative, including contingency, is estimated at \$13,238,000. A detailed cost estimate is presented in Appendix A.

5.3.4 <u>Alternative 4 - Site Use Restrictions, Capping and Ground Water</u> Extraction with On-Site Physical Treatment

5.3.4.1 Description

For Alternative 4, site use restrictions and capping would be as described for Alternative 2 in Section 5.3.2. Therefore, the description presented

5-26

TAMS/TRC

herein addresses the proposed on-site physical ground water treatment system. Figure 5-7 presents the proposed site layout for this alternative. Implementation of the following common elements, as described in Section 5.2, would also be required: Mobilization, Equipment Decontamination, Clearing, Run-on/Run-off Controls, Access Road Reconstruction, Ground Water Monitoring, Ground Water Extraction/Discharge, and Compliance with Wetlands Requirements.

Ground water would be extracted via ground water extraction wells, as described in Section 5.2, at an estimated combined rate of approximately 10 gallons per minute (14,400 gallons per day). The extracted ground water would be pumped to an on-site treatment system consisting of a chemical precipitation system for inorganic removal and carbon adsorption for organics removal. The treatment systems would be operated to reduce contaminant levels in extracted ground water in accordance with federal and state discharge requirements prior to discharge to surface water.

The chemical precipitation treatment scheme will consist of a filtration unit to remove gross solids prior to treatment, a flow equalization tank, and the chemical reaction/precipitation system. The precipitation system will include the following:

- Reaction tank including mixers and pH control instrumentation;
- Chemical feed system, including a storage tank, mixers, level instrumentation and metering equipment;
- Clarifier;
- pH adjustment tank;
- Filter; and
- Solidification/stabilization system.

A schematic of the system is provided in Figure 5-8.

The proposed precipitation process involves the initial removal of entrained solids through filtration, thereby resulting in reduced reagent costs and smaller equipment sizing for the remainder of the treatment train. A flow equalization tank is also provided prior to treatment. From the equalization tank, the ground water enters the reaction tank, where a reagent is added to adjust the pH of the wastestream to the level required to precipitate the optimal quantity of inorganic contaminants. Single-stage batch precipitation is expected to be suitable for treatment of the influent ground water stream.

The selection of an applicable precipitation reagent is dependent upon the pH, pollutant loading, and waste/reagent compatibility. For the purposes of this assessment, hydroxide precipitation is assumed. Hydroxide precipitation is a proven method for removing most of the metals identified at elevated concentrations in the ground water at the Hertel Landfill site, including arsenic, chromium (trivalent), copper, iron, lead, manganese, and zinc. Complexing/chelating agents, such as soda ash (Na₂Co₃), can improve the efficiency of the liquid/solid separation and can be effective in the removal of other inorganics such as magnesium. Determination of the final treatment scheme and appropriate reagents will require additional wastestream characterization and treatability study testing.

A clarifier follows the reaction tank. In the clarifier, flow is decreased to a point where solids with a specific gravity greater than that of the liquid settle to the bottom. The supernatant/liquid is drawn off and discharged to a pH adjustment tank for neutralization. The solids are discharged to a holding tank for subsequent dewatering. Dewatering is accomplished using mechanical dewatering equipment such as a belt filter. Once dewatered the sludge may require stabilization/solidification prior to off-site landfill disposal. The use of binding media such as cement-based compounds, lime-based pozzolanic materials or organic polymers can result in a filter cake which passes TCLP limits, potentially allowing for its disposal as

TAMS/TRC

a non-hazardous waste. Based on published sludge generation rates (Palmer, et.al., 1988) and the maximum concentration of iron in filtered ground water samples, a maximum sludge generation rate of 8,200 lb per year is estimated (i.e., 1 lb/hr).

The carbon adsorption unit will consist of two active adsorption columns operated in series. Additional columns will be temporarily stored on-site to allow replacement of spent columns as needed. A schematic of the treatment unit is presented in Figure 5-9. The first column will provide the major portion of the treatment with the second column acting as a polishing column. Based on a design flow rate of 10 gallons per minute (gpm), an assumed carbon usage of 0.5 lb carbon per 1,000 gallons treated (Stenzel, M.H., and Gupta, U.S., 1985), and a typical application rate of 2 to 10 gpm per square foot, it is estimated that a system consisting of two drum-sized columns will be appropriate. Each column will contain approximately 6.5 cubic feet of Water will be delivered directly from the chemical precipitation carbon. system to the top of the first carbon column, and effluent will be collected in a 1,000-gallon equalization tank prior to discharge to the surface water. Based on the assumed carbon usage rate, one of the drums will require replacement every 25 days, or approximately once a month. The system will be operated in a cyclic fashion, with the first adsorber drum replaced, the second adsorber drum replacing the first adsorber drum, and a new adsorber drum replacing the second adsorber drum, as indicated below:

Typical Cycle Schedule

Day	<u>Column A</u>	<u>Column B</u>	<u>Column C</u>
0-25	lst Adsorber	2nd Adsorber	(Replaced)
26-50	(Replaced)	lst Adsorber	2nd Adsorber
51-75	2nd Adsorber	(Replaced)	lst Adsorber

The use of drum-sized adsorbers allows for the replacement of spent drums as needed. The maintenance of a three month supply of replacement drums allows for spent drum pickup and replacement at 90-day intervals, in accordance with RCRA requirements for maximum temporary hazardous waste storage periods.

The major operation and maintenance activities will be the disposal of the filter cake and sludge, and the replacement of spent carbon. Stabilized sludge will be sent off-site for disposal as a non-hazardous waste, provided TCLP requirements are met. The spent carbon will be removed through the pickup and removal of the spent drummed columns and replacement with new columns. Service contracts with carbon suppliers can be arranged to perform this service, which is expected to be required once every three months.

5.3.4.2 Criteria Assessment

The following criteria assessment addresses only the evaluation of the ground water extraction and treatment components of Alternative 4; the remaining components (capping and site use restrictions) were addressed during the assessment of Alternative 2, as presented in Section 5.3.2.2.

<u>Short-Term Effectiveness</u> - In the short-term, installation of ground water extraction wells could result in potentially added risks to on-site construction workers. Short-term risks to remediation workers involved in road improvements, clearing of vegetation and installation of a cap are described under Alternative 2 (Tables 5-2 through 5-7). Additional short-term risks associated with installation of twenty-two extraction wells were estimated for remedial workers in the absence of any personal protective equipment.

Tables 5-8 and 5-9 summarize cancer risk levels and hazard index ratios estimated for installation of extraction wells under Alternative 4. Exposure of remedial workers to contaminants is associated with a total cancer risk of 5E-05 (including site preparation, capping and extraction well installation), which is within the acceptable risk range.

Site remediation under this alternative is associated with a total HI of 30, which is above the target HI value of 1. The primary contribution to this risk is inhalation of contaminants absorbed to fugitive dusts (Tables 5-3, 5-5 and 5-9). The use of personal protective equipment is likely to minimize these risks. No significant added risks to the adjacent community or the environment are anticipated as a result of extraction well or treatment system installation. Protection against contaminated ground water migration would be provided upon initiation of extraction and control of ground water gradients.

For Alternative 4, the time required to meet remedial response objectives will depend on the time required to meet ground water quality standards through the operation of a pump and treat system. For the purposes of this evaluation, it has been assumed that the treatment of three pore volumes of ground water will provide adequate treatment. At the assumed extraction rate of 14,400 gallons per day, ground water treatment would continue for a period of 12 years.

Long-Term Effectiveness and Permanence - Chemical precipitation and carbon adsorption are effective methods for the removal of metals and organics in contaminated ground water. Precipitation is a proven method for removing most of the metals identified at the Hertel Landfill site. Through stabilization, the filter cake treatment residual may pass TCLP limits thus allowing for its disposal as a non-hazardous waste. Spent carbon will be transported off-site

for disposal, and, therefore, may present minimal off-site residual contamination.

Ground water monitoring will be continued for five years after the completion of ground water treatment at the site. The potential for additional ground water contamination to occur following discontinuation of the pumping system operation, and subsequent waste/ground water contact exists and would require monitoring. Because waste materials would remain in-place under this alternative, a five-year review of the action would be required (see Section 5.3.1.2).

Residual risks associated with long-term future use of the site are not expected to be significant based on restricted future use of the site. That is, implementation of on-site ground water treatment restricts future use of the site for other purposes.

Implementability - The construction of a ground water extraction system would be fairly easily implemented. The construction and implementation of a chemical precipitation system and carbon adsorption system would also be relatively easy. Both chemical precipitation and carbon adsorption are well-proven technologies with minimal technical problems expected. These two treatment systems are not expected to pose difficulties in implementing future remedial actions or monitoring difficulties. Ongoing monitoring of the systems would be required to ensure carbon breakthrough does not occur, to maintain required chemical supplies, and to manage wastes produced by the systems as appropriate.

Ground water extraction and discharge of treated ground water to local surface waters will require compliance with New York Pollutant Discharge Elimination System requirements and New York Freshwater Wetlands Permit
Requirements. Because ground water extraction could potentially impact the water levels of adjacent wetlands, implementation of this alternative could be difficult administratively. Discharge of treated ground water to the wetland areas could be considered as a means of minimizing these impacts.

<u>Reduction of Toxicity, Mobility or Volume Through Treatment</u> - Alternative 4 provides a reduction of ground water toxicity through treatment and a reduction of contaminant mobility through pumping, in addition to the containment features previously described for Alternative 2. Chemical precipitation and carbon adsorption will treat currently existing ground water contaminants. In chemical precipitation, the metals are precipitated out of the ground water as a solid filter cake which, when stabilized, is expected to pass TCLP limits. In carbon adsorption, the organics are adsorbed onto the carbon. Due to the relatively small volumes of carbon required for this application, on-site thermal regeneration is not practical and the spent columns will be transported off-site for disposal.

<u>Compliance with ARARs</u> - For ground water treatment, chemical precipitation and carbon adsorption will be used to reduce contaminant levels below Ambient Water Quality Criteria prior to discharge at the site. The cleanup goals for ground water contaminant levels will consist of federal and state MCLs. Effluent from the treatment process must meet the NY State Pollutant Discharge Elimination System (6NYCRR 750-757) requirements before discharge to area surface waters. The effluent must also meet federal regulations under the Clean Water Act. Ground water extraction and subsequent treated effluent discharge would be conducted in compliance with applicable wetlands requirements.

TAMS/TRC

The effectiveness of this alternative in maintaining ground water contaminant levels below federal and state MCLs after the operation of the pump and treat system is discontinued will require monitoring, since the waste will remain in-place.

<u>Overall Protection of Human Health and the Environment</u> - This alternative provides protection of human health and the environment by minimizing ground water contaminant migration through pumping and contaminant toxicity through treatment using chemical precipitation and carbon adsorption. The long-term effectiveness and permanence are expected to be good, with little long-term maintenance. Long-term monitoring will be required to verify the continued protection offered by the alternative once ground water treatment activities are discontinued.

<u>Cost</u> - The cost associated with Alternative 4 includes site clearing, grading, cap construction and maintenance, ground water extraction and treatment, and long-term monitoring costs. These costs include \$3,439,000 in direct capital costs, \$550,000 in indirect capital costs and \$316,000 in annual operation and maintenance costs (\$3,322,000 net present value). The present worth value of this alternative, including contingency, is estimated at \$8,774,000. A detailed cost estimate is presented in Appendix A.

5.3.5 <u>Alternative 4A - Site Use Restrictions</u>, Capping, and Ground Water Extraction with On-Site Innovative Treatment

5.3.5.1 Description

For Alternative 4A, site use restrictions and capping would be as described for Alternative 2 in Section 5.3.2. Therefore, the description presented herein addresses the proposed on-site innovative ground water

treatment system. Figure 5-7 presents the proposed site layout for this alternative. Implementation of the following common elements, as described in Section 5.2, would also be required: Mobilization, Equipment Decontamination, Clearing, Run-on/Run-off Controls, Access Road Reconstruction, Ground Water Monitoring, Ground Water Extraction/Discharge, and Compliance with Wetlands Requirements.

Ground water would be extracted via ground water extraction wells, as described in Section 5.2, at an estimated combined rate of 10 gallons per minute. The extracted ground water would be pumped to an on-site treatment system consisting of a membrane microfiltration unit for inorganics removal and UV oxidation for organics removal. The treatment systems would be operated to reduce contaminant levels in extracted ground water to federal or state criteria in accordance with New York State Pollutant Discharge Elimination System (SPDES) requirements (NYCRR Parts 750 through 758) prior to discharge to surface water.

The membrane microfiltration treatment system used as a basis for the evaluation of this alternative is an innovative treatment system being developed by E.I. DuPont de Nemours & Company (Du Pont). The treatment system is currently included in the U.S. EPA's Superfund Innovative Technology Evaluation (SITE) program. The system is designed to remove solid particles from liquid wastes, forming filter cakes typically ranging from 40 to 60 percent solids. It consists of an automatic pressure filter (Oberlin) combined with Du Pont's special Tyvek filter material (Tyvek T-980) made of spun-bonded olefin. The filter material has 0.1 micron openings, therefore allowing the removal of smaller particles which may have passed through the 0.45 micron openings of the filters used in the field sampling program. A schematic of the system is presented in Figure 5-10.

TAMS/TRC

The microfiltration unit operates in a cyclical manner. The waste feed enters an upper chamber and is pumped through the filter fabric. The fabric allows water and solids less than about one ten-millionth of a meter in diameter to pass through the openings in the fabric. Filtered solids accumulate on the fabric, forming a filter cake, while the filtrate accumulates in a lower chamber. Air is fed into the upper chamber at about 45 pounds per square inch and used to further dry the cake and remove any remaining liquid. When the cake has been dried, the upper chamber is lifted and the filter cake discharged. The entire system is enclosed and therefore can be used to treat wastestreams containing volatile organics.

Pilot tests using this technology have been conducted at the Palmerton Zinc Superfund site in Palmerton, Pennsylvania, where ground water is contaminated with dissolved heavy metals such as cadmium, lead and zinc. The tests produced a 35 to 45 percent-solids filter cake, and a filtrate with non-detectable levels of heavy metals (U.S. EPA, 1989). The filter cake also passed TCLP analysis to render it a non-hazardous waste (E.I. duPont de Nemours & Co., Inc., 1991).

UV oxidation would follow the membrane microfiltration unit. UV oxidation is a process in which UV light and hydrogen peroxide chemically oxidize organic contaminants dissolved in water. A layout of the proposed system is provided in Figure 5-11. Hydrogen peroxide is converted in the presence of UV light to hydroxyl radicals, which are powerful oxidizers. Concurrently, organic molecules absorb energy from the UV light, making them more receptive to the hydroxyl radicals. The combined UV light and hydroxy radicals promote rapid breakdown of organics into carbon dioxide and water without the creation of air emissions or residual waste streams. The oxidation unit will be operated to reduce the contaminant levels in ground water to federal or state

TAMS/TRC

criteria in accordance with state discharge requirements. Self-contained units for the destruction of organics are manufactured by Peroxidation Systems, Inc., and are available in various configurations. For costing purposes, a unit similar to the Peroxidation Systems Model SSB-30 oxidation unit has been assumed.

Operation and maintenance of the unit consists of UV lamp replacement every four months and occasional replenishment of the hydrogen peroxide supply.

5.3.5.2 Criteria Assessment

The following criteria assessment addresses only the evaluation of the innovative ground water extraction and treatment components of Alternative 4A; the remaining components (capping and site use restrictions) were addressed during the assessment of Alternative 2, as presented in Section 5.3.2.2.

<u>Short-Term Effectiveness</u> - Short-term risks to workers under this alternative are not expected to differ significantly from those described for Alternative 4. No significant added risks to the adjacent community or the environment are anticipated as a result of extraction well or treatment system installation. Protection against contaminated ground water migration would be provided upon initiation of extraction and control of ground water gradients.

For Alternative 4A, the time required to meet remedial response objectives will depend on the time required to meet ground water quality standards through the operation of a pump and treat system. For the purposes of this evaluation, it has been assumed that the treatment of three pore volumes of ground water will provide adequate treatment. At the assumed extraction rate

TAMS/TRC

of 14,400 gallons per day, ground water treatment would continue for a period of 12 years.

Long-Term Effectiveness and Permanence – Residual risks associated with long-term future use of the site are not expected to be significant based on restricted future use of the site. That is, implementation of on-site ground water treatment restricts future use of the site for other purposes.

The long-term risks associated with the residuals of ground water extraction and treatment by UV oxidation and membrane microfiltration are relatively small. Ground water treatment by UV oxidation results in the breakdown of organics into carbon dioxide and water. Therefore, risks associated with UV oxidation treatment residuals are expected to be minimal. Ground water treatment by membrane microfiltration results in the filtration of heavy metal particles from the ground water. Therefore, dissolved metals would not be removed by this treatment method. The filtered solids form a filter cake which, in past tests, has passed TCLP limits for disposal in non-hazardous landfills. The tests produced a filtrate also with non-detectable levels of heavy metals. Therefore, risks associated with membrane microfiltration treatment residuals are expected to be minimal. Because this remedial alternative does not permanently eliminate the contact between waste materials and the ground water table, the effectiveness of this alternative is difficult to predict once operation of the pump and treat system is discontinued.

Ground water treatment for Alternative 4A requires regular sampling and analysis of treatment system discharges, in accordance with New York SPDES requirements. Following discontinuation of operation of the pump and treat system, continued ground water monitoring for a period of five years has been

TAMS/TRC

assumed. Because waste materials would remain in-place under this alternative, a five-year review of the action would be required (see Section 5.3.1.2).

<u>Implementability</u> - The construction of a ground water extraction system would be fairly easily implemented. The implementation of UV oxidation and membrane microfiltration on-site would be relatively easy, requiring transport and set-up of commercially offered systems. The availability of a membrane microfiltration system may be limited because of the limited number of vendors offering this technology. Because both treatment systems are relatively new, minor technical problems could be expected. Neither treatment system is expected to impact future remedial actions or to present exposure pathways which are not easily monitored.

Implementation of ground water extraction and discharge to area surface waters would require compliance with the substantive requirements of federal and state discharge requirements and wetlands regulations. Treatability studies would be required to ensure that the proposed remedial technologies could meet effluent limitations. Because ground water extraction could potentially impact the water levels of adjacent wetlands, implementation of this alternative could be difficult administratively.

Reduction of Toxicity, Mobility or Volume Through Treatment - Membrane microfiltration and UV oxidation will treat currently existing ground water contaminants. In membrane microfiltration, the metal particles are filtered out of the ground water as a solid filter cake, which has been shown to pass TCLP limits in some instances. However, no treatment of dissolved metals is provided. In UV oxidation, the organics are broken down into carbon dioxide

TAMS/TRC

and water. Therefore, the overall toxicity of ground water contaminants is reduced. The mobility of the ground water is also reduced through pumping and through the containment features previously described for Alternative 2. The permanence of the reduction in ground water toxicity would have to be monitored upon discontinuation of the pump and treat system. Because the ground water would remain in contact with the waste materials, the potential for additional leaching of contaminants into the ground water would exist.

Compliance with ARARs - For ground water treatment, membrane microfiltration and UV oxidation will be used to reduce contaminant levels to below federal and state surface water quality criteria prior to discharge at the While membrane microfiltration will result in inorganic levels less site. than those observed in the filtered sample analyses during the RI (due to smaller filter openings in the microfilter), the ability of the treatment unit to meet cleanup criteria cannot be confirmed without conducting treatability The cleanup goals for ground water contaminant levels will consist studies. of federal and state MCLs. Effluent from the treatment process must meet the NY State Pollutant Discharge Elimination System (6NYCRR 750-757) requirements before discharge to area surface waters. The effluent must also meet federal regulations under the Clean Water Act. Ground water extraction and subsequent treated effluent discharge would be conducted in compliance with applicable wetlands requirements.

The effectiveness of this alternative in maintaining contaminant levels below federal and state MCLs once operation of the pump and treat system is discontinued will require monitoring, since the waste materials will remain in-place.

<u>Overall Protection of Human Health and the Environment</u> - This alternative provides control of risks posed by ground water contamination through extraction and treatment by membrane microfiltration and UV oxidation. The long-term effectiveness and permanence are expected to be good, with little long-term maintenance and monitoring, although the potential for additional ground water contamination once the extraction/treatment system is turned off exists.

<u>Cost</u> - The cost associated with Alternative 4A includes site clearing, grading, cap construction and maintenance, ground water extraction and innovative treatment, and long-term monitoring costs. These costs include \$3,410,000 in direct capital costs, \$545,000 in indirect capital costs and \$267,000 in annual operation and maintenance costs (\$2,884,000 net present value). The present worth value of this alternative, including contingency, is estimated at \$8,207,000. A detailed cost estimate is presented in Appendix A.

5.4 Comparative Analysis of Alternatives

In this section, the strengths and weaknesses of the alternatives relative to one another are discussed for each of the analysis criteria. In each discussion, the alternative which provides the best overall performance in that category is discussed first, followed by the other alternatives discussed in the relative order in which they perform. These comparisons of alternatives are also presented in summary form in Tables 5-10 through 5-16.

While no recommendation for selection of an individual alternative is made within the Feasibility Study, a discussion of the basis for the final decision is appropriate. In selecting the remedy for the site, overall protection of

TAMS/TRC

human health and the environment and compliance with ARARS (unless grounds for invoking a waiver are provided) are threshold criteria that must be satisfied by the selected alternative. Long-term effectiveness and permanence, reduction of toxicity, mobility or volume through treatment, short-term effectiveness, implementability and cost are primary balancing criteria, with the NCP placing special emphasis on long-term effectiveness and reduction of toxicity, mobility or volume through treatment. State and community acceptance, to be determined after comments on the completed RI/FS are received, are modifying criteria that may have significant input in the final remedy selection.

The comparative analyses for each of the evaluation criteria follow.

5.4.1 Short-Term Effectiveness

Those alternatives which provide short-term achievement of remedial response objectives while minimizing short-term risks and environmental impacts are considered to be the most effective. In general, all alternatives except the no action alternative require clearing of vegetation from the landfill area, road improvements or other activities involving disturbance of contaminated soils. These alternatives pose, at a minimum, non-cancer risks which exceed acceptable risk ranges to on-site remedial workers due to inhalation of contaminants adsorbed to fugitive dust. This pathway of exposure can be minimized through the use of personal protection equipment. Once remedial activities are completed, this exposure pathway ceases to exist for these alternatives. Therefore, the alternatives which offer reductions in the current risks posed by the site within the shortest time frame offer the greatest short-term protection against the principle threats.

The no action alternative can be considered to be the most effective alternative in the short-term. This alternative poses risks as described in the baseline risk assessment (Section 1.5). Because no remediation is proposed under this alternative, no disturbance of existing contamination occurs and no short-term risks are realized. It should be emphasized, however, that while no increases in risks result in the short-term, no protection against the principle site threats is achieved.

For alternatives that involve site remediation, Alternatives 2 and 2A provide the greatest short-term effectiveness. They pose the least amount of risk to on-site remedial workers and achieve protection against dermal contact risks within the shortest time frame. Alternative 2A is considered to be more protective than Alternative 2 because exposures to soil/waste contaminants are mitigated through capping and potential exposures to ground water contamination are reduced through the construction of a slurry wall. The slurry wall will lower the water table, potentially below the base of the waste materials, without requiring any handling of the contaminated ground water. Alternative 2 offers similar short-term risks but does not provide the same degree of protection against ground water contaminant migration.

Alternatives 4 and 4A also provide good short-term effectiveness. They pose additional risk to on-site workers due to the installation of ground water extraction wells within contaminated areas, but they also meet remedial response objectives within a limited time frame, with exposures to ground water contamination reduced through ground water pumping and on-site treatment. The additional handling of contaminated ground water and required discharge to surface water increases the potential risks and environmental impacts associated with remediation, and makes these alternatives less

effective in the short-term than Alternative 2A. These alternatives also have longer remedial time frames associated with achievement of cleanup goals.

5.4.2 Long-Term Effectiveness

The alternative which poses the least residual risk due to untreated waste or treatment residues, or the greatest capability for controlling these risks, is considered to provide the greatest long-term effectiveness and permanence. None of the alternatives actively address remediation of contaminants currently detected in surface water or sediment (other than contamination associated with leachate seeps). Therefore, all alternatives present some residual risk based on incidental ingestion and dermal contact with sediments under a recreational use scenario. These calculated risks, however, are within the acceptable risk ranges and are not considered to seriously impact the long-term effectiveness of the alternatives, especially with respect to those alternatives for which site access will be limited for an extended period based on the long-term operation of on-site remedial systems.

Alternative 2A provides minimal residual risk through the containment rather than treatment of on-site contaminants. The combination of the cap and slurry wall minimize contact with soil contaminants and potential exposure pathways associated with ground water contamination, although potential exposure to surface water/sediment will exist if access to the site is not fully controlled (e.g., if the site is used as a recreational area following capping). The slurry wall will minimize contact of the ground water table with in-place waste materials, thereby minimizing future contamination of ground water. These containment features are expected to be highly reliable with minor maintenance or monitoring; if they should fail, replacement or repair would not be exceptionally difficult. The fact that the slurry wall

TAMS/TRC

will be located upgradient of the contaminated ground water area, and therefore will not be in contact with ground water contaminants, limits the potential for failure due to chemical degradation of the slurry wall materials.

Alternatives 4 and 4A provide comparable levels of long-term protectiveness. While treating the ground water and reducing dermal exposure risks through containment features, these alternatives do not provide for source of contamination. Therefore the long-term the treatment of effectiveness of these alternatives in maintaining reduced ground water contaminant levels following discontinuation of the pump and treat system operation is not guaranteed. The water table can be expected to return to a level within the waste materials when pumping is discontinued, thereby contamination. future ground water These potentially allowing for alternatives also require long-term management in the form of cap maintenance and ground water treatment system monitoring and operation. Because of the ongoing operation of the ground water treatment system, use of the site for recreation and the associated potential exposures are not considered to apply to these alternatives.

Alternative 2 would not treat the source of contamination or the contaminated ground water on-site, although it would provide protection against dermal exposures to soil contaminants through its capping containment This alternative requires minimal long-term management in the form feature. and monitoring. Potential exposure to surface maintenance of cap water/sediment contaminants will exist under this alternative if access to the site is not fully controlled (e.g., if the site is used as a recreational area following capping).

Alternative 1, the no action alternative, offers no long-term effectiveness in terms of protection against current risks associated with

TAMS/TRC

dermal contact with soil contaminants or future ground water ingestion scenarios.

5.4.3 Implementability

Those alternatives which offer the greatest technical feasibility, administrative feasibility, and service and material availability are considered to be most implementable. Wetlands regulations will impact the implementation of all alternatives except the no action alternative to varying degrees. Alternatives involving ground water extraction and discharge to surface water (Alternatives 4 & 4A) will require compliance with regulatory requirements for surface water discharges. Alternatives 2, 2A, 4, and 4A would each require site use and ground water use restrictions. Implementation of such restrictions appears to be feasible in the State of New York.

Alternative 1, the no action alternative, is the most implementable because it requires only the installation of additional monitoring wells.

Alternatives 2 and 2A follow Alternative 1 in implementability, respectively. Capping construction methods are well developed and easily implemented. The construction of a slurry wall under Alternative 2A would also be relatively easy to implement, although existing site conditions could hamper construction.

Alternatives 4 and 4A are similar to Alternative 2, involving the construction of a cap, but also include the construction of a ground water extraction and treatment system. The construction of such a system would be relatively easy. Minimal technical problems would be expected in the implementation of Alternative 4. The innovative ground water treatment technologies included in Alternative 4A could pose additional technical problems. The lack of general availability of the innovative treatment

TAMS/TRC

technologies could also limit the availability of treatment systems and experienced operational personnel.

5.4.4 Reduction of Toxicity, Mobility and Volume Through Treatment

Alternative 2A provides a reduction of contaminant mobility through its containment features. The alternative utilizes a cap and slurry wall to isolate in-place waste materials from exposure via direct contact and from infiltration and ground water migration. While the waste materials are not treated, their isolation limits the potential risks they pose.

Alternatives 4 and 4A reduce the toxicity of ground water through treatment and reduce the mobility of soil contaminants through containment. Alternative 4 may provide greater toxicity reduction through the treatment of dissolved inorganics. The reduction in ground water toxicity may not be permanent, however, due to the lack of treatment of the soil/waste matrix and the ability of the ground water table to return to a level within the waste materials upon discontinuation of operation of the pump and treat system.

Alternative 2 only reduces the mobility of the soil contaminants through containment measures. It does not address ground water contamination or limit additional contamination of ground water due to continued contact of waste materials with the water table.

Alternative 1 provides no reduction in toxicity, mobility of volume of contaminants of any media through treatment. Residual risks are identical to those identified by the baseline risk assessment. Future risks posed by the site will depend on future site usage.

5.4.5 Compliance with ARARs

Those alternatives which offer the greatest overall compliance with potential chemical-specific, location-specific and action-specific ARARs are

FS REPORT

considered to offer the best performance under this criterion. A locationspecific ARAR which may directly impact the implementation of alternatives at the site is the location of wetlands, which cover approximately 13 percent of the site. For all alternatives involving construction within or adjacent to wetland areas or ground water discharge to surface waters, compliance with the appropriate wetland and discharge regulations will be required.

It should be noted that under certain cases, compliance with ARARs may be waived. Several of the specific cases under which the waiver of ARARs could potentially be applicable to the Hertel Landfill site were previously noted in Section 5.3.1.2.

Alternatives 4 and 4A are expected to meet chemical-specific ARARs for the ground water. Maintenance of ARARs within the ground water once pump and treat operations are discontinued may not be possible due to the resumption of contact between the soil/waste matrix and the ground water. The innovative technologies employed under Alternative 4A may not be as effective in reaching ARAR-based cleanup levels due to their currently unproven overall capabilities. Action-specific ARARs will be achieved if, as proposed, the final cap and surface drainage features are constructed in accordance with New York Solid Waste Management Facility landfill closure regulations.

Alternatives 2 and 2A are not expected to attain all chemical-specific ARARs for contaminants detected in the ground water due to the lack of ground water treatment. For Alternative 2A, the elimination of ground water flow through the in-place waste materials may eventually result in reduced ground water contaminant levels but treatment of the currently detected contaminant levels will not be provided. Action-specific ARARs will be achieved if, as proposed, the final cap and surface drainage features are constructed in accordance with New York Solid Waste Management Facility landfill closure regulations.

TAMS/TRC

Alternative 1, the no action alternative, is not expected to attain chemical-specific ARARs for the ground water matrix. No location-specific or action-specific ARARs would be applicable under the no action alternative.

5.4.6 Overall Protection of Human Health and the Environment

This criterion considers the previous criteria, especially long-term effectiveness and permanence, short-term effectiveness, and compliance with ARARs, and provides a final overall assessment of whether the alternative provides adequate protection of human health and the environment.

Alternative 2A provides short-term effectiveness through its limited remedial time frame and limited disturbance of contaminated areas, resulting in minimal short-term risks as compared to other alternatives which involve remedial actions. Long-term effectiveness is expected to be good due to the minimization of contact between the waste materials and the ground water table offered by the containment features of the remedial design. While ground water will not be treated to meet chemical-specific ARARs, the invocation of a waiver to ARAR compliance may be appropriate for this site. Through containment rather than treatment, this alternative is protective of human health and the environment.

Alternatives 4, 4A, and 2 follow Alternative 2A in overall protection to human health and the environment. All of these alternatives present minimal short-term risks during implementation and provide protection against the principle threats within a relatively short time frame. The alternatives utilize proven remediation technologies, with the exception of Alternative 4A, and are relatively uncomplicated in their design and implementation. While Alternatives 4 and 4A provide additional protection in the short-term through the pumping and treatment of ground water, the long-term effectiveness of

TAMS/TRC

these alternatives is similar to Alternative 2, due to the ultimate resumption of contact between the soil/waste and ground water table. While these alternatives may not maintain compliance with ARARs, they will be protective of human health and the environment through limited site access, continued ground water monitoring and the required five-year review of the remedial decision.

Alternative 1, the no action alternative, is the least protective of human health and the environment. This alternative does not limit site access or future site development and, therefore, does not address the principle threats posed by the site.

5.4.7 Cost

A comprehensive analysis of present worth cost of the alternatives is presented below, followed by a cost sensitivity analysis.

5.4.7.1 Present Worth Comparative Analysis

Total present worth cost estimates for alternative implementation range from \$2,509,000 to \$13,238,000. The costs vary greatly between alternatives for both capital costs (\$58,000 to \$8,406,000) and, to a lesser degree, for present worth O&M costs (\$2,033,000 to \$3,322,000).

The lowest cost alternative is the no action alternative, which has an estimated total present worth cost of \$2,509,000. The major cost component of this alternative is the net O&M cost (\$2,033,000) which consists of thirty years of ground water sampling and laboratory analysis.

There are three alternatives with present worth costs ranging from \$7,000,000 to \$9,000,000. Alternative 2 can be implemented at a total cost of \$7,182,000 with cap construction, a direct cost, as the major cost component.

Alternatives 4 and 4A are relatively comparable in cost, at \$8,774,000 and \$8,207,000, respectively. The major cost component for both Alternatives 4 and 4A is the net O&M cost, which consists of ground water treatment system maintenance, ground water sampling and laboratory analysis, and cap maintenance costs. Alternative 2A can be implemented at a total cost of \$13,238,000, with slurry wall construction, a direct cost, as the major cost component.

5.4.7.2 Sensitivity Analysis

A sensitivity analysis was conducted to assess the effect that variations in specific assumptions made during alternative development and assessment can have on the total estimated remedial cost. The Hertel Landfill site remedial alternatives are impacted by uncertainties regarding discount factors over the life of the remedies, landfill cap construction cost, slurry wall construction cost, and ground water extraction rates. The assumed variations in each of these factors used to conduct the sensitivity analysis are described below. The resultant impacts to remedial costs are summarized in Table 5-17.

The discount rate can vary from the 5% rate used in the cost evaluation. Alternatives with large O&M cost components and extended remedial periods can be significantly impacted by a variation in the discount rate. The sensitivity analysis has been conducted assuming a variation in the annual discount rate, with total present worth costs estimated for each alternative at annual discount rates of 3% and 10%.

Landfill cap construction cost uncertainties are largely due to potential site-specific impacts on vendor quotes and potential variations in volumes of material required to construct the cap, due to the estimated areal extent of the landfill based on the RI investigations. The sensitivity analysis has

TAMS/TRC

assumed landfill cap constructions costs vary by $\pm 20\%$ for Alternatives 2, 2A, 4 and 4A. The variation in cost impacted the cost of each alternative fairly equally, with the minimum and maximum costs falling within approximately \$850,000 of each other.

Slurry wall construction cost uncertainties are also largely due to potential site-specific impacts on vendor quotes. The presence of subsurface boulders, the sloping topography of the site and the expense associated with a cement-bentonite slurry wall could result in variations to the estimated cost. The sensitivity analysis has assumed slurry wall construction costs vary by $\pm 20\%$. This variation has been applied to the only alternative which includes a slurry wall, Alternative 2A. The minimum and maximum costs fall within a range of approximately \$2,000,000.

Ground water extraction rate uncertainties exist due to variabilities in subsurface conditions which could affect the estimated extraction rate, and the ability of the alternatives to meet treatment goals within the estimated time frame. The sensitivity analysis has assumed that the ground water extraction rate could vary from 5 gpm to 20 gpm, and that the associated remedial timeframes could vary from twenty to six years, respectively. This variation has been applied to the alternatives for which ground water extraction and treatment is a component, namely Alternatives 4 and 4A.

6.0 REFERENCES

- Burgher, B., Culpepper, M., and Zieger, W., 1987. <u>Remedial Action Costing</u> Procedures Manual, JRB Associates, PB88-113691, October 1987.
- Canter, L.W., and Knox, R.C., 1986. <u>Ground Water Pollution Control</u>, Lewis Publishers, Inc., 1986.
- Case International Company, 1982. Case Slurry Wall Notebook, Manufacturers Data. Case International Co., Houston, TX.
- Cheremisinoff, Paul N., and Ellerbusch, Fred, 1978. <u>Carbon Adsorption</u> Handbook, Ann Arbor Science Publishers, 1978.
- Cudahy, James J., and Eicher, Anthony R., <u>Thermal Remediation Industry –</u> <u>Markets, Technologies, Companies</u>, Pollution Engineering, Vol. XXI, No. 12, November 1989, pp. 76-80.
- E.I. duPont de Nemours & Company, Inc., Personal Communication, March 6, 1991.
- Environmental Information, Ltd., 1990. Environmental Services Directory, 1991, Environmental Information, Ltd., 1990.
- Frank, J.F., M.E. Dinkel, and D.M. Chari, 1987, Use of Mobile Incineration to Remediate the Lenz Oil Site: Superfund '87, Proceedings of the 8th National Conference, November 16-18, Washington, D.C.
- Geosafe Corporation, 1989. Application and Evaluation Considerations for In Situ Vitrification Technology: A Treatment Process for Destruction and/or Permanent Immobilization of Hazardous Materials, Geosafe Corporation, GSC 1901, April 1989.
- Glebs, Robert T., 1988. Landfill Costs Continue to Rise, Waste Age, March 1988.
- Hazardous Materials Control Research Institute, 1991. Capitol Hill Questions EPA's Endorsement of Vitrification Technique, Proponents and Opponents Challenge Data, Focus, February 1991.
- Johnson, N.P., and Cosmos, M.G., 1989. <u>Thermal Treatment Technologies for Haz</u> Waste Remediation, Pollution Engineering, Vol.XXI, No. 11, October 1989.
- McDonald, M.G., and Harbaugh, A.W., 1988. A Modular Three-Dimensional Finite-Difference Ground-Water Flow Model. U.S.G.S. Open-File Report 83-875, Chapter Al.
- McCoy and Associates, 1987, ENSCO's Mobile Incinerators, Configuration and Trial Burn Performance: Hazardous Waste Consultant, Vol. 5, Issue 6, pp. 1-1 1-4.
- Miller, S.P., 1979. <u>Geotechnical Containment Alternatives for Industrial</u> <u>Waste Basin F, Rocky Mountain Arsenal, Denver, Colorado; A Quantitative</u> <u>Evaluation</u>. U.S. Army Corps of Engineers Waterways Experiment Station, Vicksburg, MS, Technical Report GL-79-23. September 1979.

- Palmer, S.A.K., et.al., 1988. <u>Metal/Cyanide Containing Wastes Treatment</u> <u>Technologies</u>, Pollution Technology Review No. 158, Noyes Data Corporation, 1988.
- Stacy, Gregory L., and James, Stephen C., 1991. <u>The Superfund Innovative</u> <u>Technology Evaluation (SITE) Program</u>, Hazardous Materials Control, Vol. 4, No. 1, January/February 1991.
- TAMS, 1989. Final RI/FS Work Plan for the Hertel Landfill Site, September 1989.
- TAMS, 1989a. Final RI/FS Field Operations Plan for the Hertel Landfill Site, October 1989.
- Tchobanoglous, G., Theisen, H., and Eliassen, R., 1977. <u>Solid Wastes:</u> Engineering Principles and Management Issues. McGraw-Hill Book Company.
- Town of Plattekill, Personal Communication, December, 1990.
- USEPA, 1991. Integrated Risk Information Service (IRIS).
- USEPA, 1989. Health Effects Assessment Summary Tables (HEAST), Fourth Quarter FY-1990, NTIS No. PB90-921104.
- USEPA, 1989a. <u>Risk Assessment Guidance for Superfund. Volume I. Human Health</u> Evaluation Manual (Part A). EPA/540/1-89/002.
- USEPA, 1989b. <u>Innovative Technology In Situ Vitrification</u>, Directive 9200.5-251FS, November 1989.
- USEPA, 1989c. <u>The Superfund Innovative Technology Evaluation Program</u>: Technology Profiles, EPA/540/5-89/013, November 1989.
- USEPA, 1988. <u>Guidance for Conducting Remedial Investigations and Feasibility</u> <u>Studies Under CERCLA</u>, Interim Final, October 1988.
- USEPA, 1988a. CERCLA Compliance with Other Laws Manual: Draft Guidance, EPA/540/G-89/006, August 1988.
- USEPA, 1985. <u>Compilation of Air Pollutant Emission Factors</u>. Volume I: <u>Stationary Point and Area Sources</u>. OAQPS. AP-42, 4th Edition.
- USEPA, 1985a. <u>Handbook Remedial Action at Waste Disposal Sites</u> (Revised), EPA/625/6-85/006, October 1985.
- USEPA, 1984. <u>Slurry Trench Construction for Pollution Migration Control</u>, EPA-540/2-84-001, February 1984.
- Weber, Walter J., Jr., 1972. <u>Physicochemical Processes for Water Quality</u> Control, John Wiley & Sons, Inc., 1972.

	PRELEMINARY	Y IDENTIFICATION OF FEDERAL CHEMICAL-SPECIFIC ARU Hertel Landfill Feasibility Study	ARS AND TBCS
FEDERAL STATUTE	REGULATION/GUIDANCE	SISGONAS	APPLICABILITY TO SITE CONDITIONS
Ground Water Safe Drinking Water Act (40 CFR 141.6063)	Max Contaminant Levels (MCL's)	MCL's directly apply to "public water systems". defined as systems with at least 15 connections which service a minimum of 25 persons.	Potential ARARs which address organic compounds and metals in ground water. Potentially applicable due to the location of a population of approximately 1.350 within 3 miles of the site who obtain their water supply from individual wells.
Safe Drinking Water Act (40 CFR 141.5051)	Max Contaminant Level Goals (MCLGs)	Non-enforceable health goals for public water systems, set at levels resulting in no known or anticipated adverse health effects.	Potential ARARs which address organics and metals in ground water. Potentially appli- cable due to the location of a population of approximately 1.350 within 3 miles of the site who obtain their water supply from individual wells.
iurface Water Clean Water Act (Section 3D4)	Ambient Water Quality Criteria (AMQC)	Won-enforceable guidelines established for the protection of human health and/or aquatic organisms.	Potential ARARs which will affect the imple- mentability of remedial actions involving effluent discharge.
H⊤ Clean Air Act (40 CFR 50)	National Ambient Air Quality Stardards (NAADS)	Establishes maximum concentrations for particulates and fugitive dust emissions.	Potential ARARs for alternatives involving remedial actions which impact ambient air (i.e. incinerators, soil venting. etc.).

ATE STATUTE	REGULAT TON/GUIDANCE	SISdONAS	APPLICABILITY TO SITE CONDITIONS
ound Water NY State Water Quality Standards (6 NYCRR 703)	Ground Water Quality Criteria (GWOC)	Establishes ground water quality standards for various classes of GW in the State of NY.	Potential ARARs which address organic compounds and metals in ground water to prevent pollution of ground water and to protect the ground water for use as potable water.
NY State Safe Drinking Water Act (10 NCYRR 5)	NY Max Contaminant Levels (NY MCLs)	NY MCLs directly apply to all "public water systems" in NY.	Potential ARARs which address organic compounds and metals in ground water. Applicable due to the location of a population of approximately 1.350 within 3 miles of the site who obtain their water supply from individual wells.
NY State Drinking Water Supplies (Public Health Law 225- Subpart 5-1)	NY Max Contaminant Levels (NY MCLS)	NY MCLs directly apply to all "public water systems" in NY.	Potential ARARs which address organic compounds and metals in ground water, and the location of a population of approximately 1.350 within 3 miles of the site who obtain their water supply from individual wells.
/arious	NY Technical and operational Guidance Series (TOGs) Mater Quality Standards and Guidance Values (1.1.1)	Compilation of ambient water quality standards and guidance values for toxic and non-conventional pollutants for use in NYDEC programs. including SPDES permit programs.	Potential ARARs/TBCs which address organic and inorganic compounds and which could affect tree ments which discharge to area surface waters.
rface Water NY Surface Water Juality Standards (6 NYCRR 701)	NY Surface Water Quality Standards	Establishes water quality standards for various classes of surface water.	Potential ARARs will affect treatments which discharge to area surface waters.

	PRELIMINARY ID	TABLE 2-2 (CONT.) ENTIFICATION OF POTENTIAL STATE CHEMICAL-SPECIFIC AR Hertel Landfill Feasibility Study	RARS AND TBCS
ATE STATUTE	REGULATION/GUIDANCE	SISGONAS	APPLICABILITY TO SITE CONDITIONS
r (6 NYCRR 256 and 257)	Standards	cation system and air quality standards.	remedial activities.
	Sediment Criteria - December 1989	Guidance document used by the Bureau of Environmental Protection. Division of Fish and Wildlife. for evaluating contaminant levels in sediments.	Potential TBCs which address sediment contamine levels and evaluation of these levels for risk management decisions.

		HEKIEL LANUFILL FEASIDILIT STUUT	
FEDERAL STATUTE	REGULATION/GUIDANCE	SISdonys	APPLICABILITY TO SITE CONDITIONS
Wetlands Executive Order 11990	Protection of Wetlands	Regulates activities conducted in a wetland area to minimize the destruction, loss or degradation of the wetlands.	Potential ARAR as approximately 15 to 20 per cent of the total site has been identified as potential wetlands.
Wetlands Construction and Management Procedures (40 CFR 6. Appendix A)	Protection of Wetlands	Sets forth EPA policy for carrying out the provisions of Executive Order 11900 (see above).	Potential ARAR as approximately 15 to 20 per cent of the total site has been identified as potential wetlands.
Clean Water Act. Section 404 (40 CFR 230; 33 CFR 320-330)	Prohibition of Wetland Filling	Prohibits the discharge of dredged or fill material to a wetland without a permit issued by the Corp of Engineers.	Potential ARAR as approximately 15 to 20 per cent of the total site has been identified as potential wetlands.
Wildlife Fish and Wildlife Coordination Act (16 USC 661 et seq.)	Protection of Wildlife Habitats	Requires actions to protect fish or wildlife when diverting. channeling or modifying a stream.	Potential ARAR as a small stream crosses the site in a northeasterly direction. exits along the eastern property line. and discharges into Pancake Hollow Creek. 1.000 feet east of the site
Farmlands Farmland Protection Policy Act (7 USC 4201 et seq.)	Protection of Significant/ Important Agricultural Lands	Requires evaluation of direct and indirect effects of actions on remaining farms and farm support sources.	Potential ARAR as area contains significant agricultural lands: no farming is currently conducted on-site or on land bordering the site. however.
Cultural Protection and Enhancement of the Cultural Environment (Executive Order 11593)	Protection of Property with Cultural Value	Regulates activities which affect properties of cultural value.	Potential ARAR - a cultural survey would be required to determine applicability to site.

	PRELIMINARY IDENTIFICATION Heri	TABLE 2-4 Of Potential State Location-Specific Arars and T El Landfill Feasibility Study	BCs
STATE STATUTE	REGULATION/GUIDANCE	SISONAS	APPLICABILITY TO SITE CONDITIONS
Wetlands			
NY State Freshwater Wetlands Law (ECL Article 24 & 71 in Title 23)	Protection of Wetlands	Regulates activities conducted in a wetlands area to minimize the destruction, loss or degradation of the wetlands.	Potential ARAR as approximately 15 to 20 percent of the total site has been identified as potential wetlands.
NY State Freshwater Wetlands Permit Requirements Regulations (6 NYCRR Part 663)	Protection of Wetlands	Regulates the procedural requirements to be followed in undertaking different activities in wetlands and in areas adjacent to wetlands.	Potential ARAR as approximately 15 to 20 percent of the total site has been identified as potential wetlands.
Wildlife and Rivers			
Endangered and Threatened Species of Fish and Wildlife Requirements (6 NYCRR 182)	Protection of Endangered Species	Restricts activities in areas inhabited by endangered species.	Potential ARAR as many fish and wildlife species inhabit the site.

	PRELIMINARY IDEN	TABLE 2-5 Itification of Potential Federal Action-Specific AF Hertel Landfill Feasibility Study	RARS AND TBCS
FEDERAL STATUTE	REGULATION/GUIDANCE	SISGONAS	APPLICABILITY TO SITE CONDITIONS
CERCLA (Title I Section 101.111)	National Contingency Plan (40 CFR 300)	Establishes funding and provisions for the clean-up of hazardous waste sites.	ARAR based on the site's placement on the National Priorities List.
Superfund Amendments and Reauthorization Act (42 U.S.C. 9601)	Clean-up standards/ Response Action	Treatments must provide permanent reductions in volume, toxicity and mobility of wastes and satisfy ARARs.	ARAR based on the site's placement on the National Priorities List.
Uniform Relocation Assistance & Real Property Aquisition Act (40 CFR 4.2)	General Relocation Requirements	Requirements for relocation payments and assistance.	Potential ARAR for alternatives which create need for temporary/permanent relocation of area residents.
Resource Conservation and Recovery Act (40 CFR 264 and 265)	Requirements for Hazardous Waste Treatment Facility Design and Operating Standards for Treatment and Disposal Systems	Outlines specifications and standards for design, operation. closure and monitoring of performance for hazardous waste storage. treatment and disposal facilities.	Potential ARARs for alternatives which utilize a surface impoundment. waste pile. landfill. land treatment or incineration for on-site disposal/treatment of wastes.
RCRA (40 CFR 257)	Criteria for Classification of Solid Waste Disposal Facilities and Practices	Presents criteria for use under RCRA in determining which solid waste disposal facilities pose a reasonable probability of adverse effects on health or the environment. Criteria specifically refer to floodplains. endangered species. pollution of surface water and/or ground water. disease. air and safety.	Potential ARAR applicable to the classification of solid waste disposal facilities.
RCRA (40 CFR 264) Subpart G	Closure/Post Closure Requirements	Establishes requirements for the closure and long-term management of a hazardous disposal facility.	Potential ARARs for alternatives which utiliz(a landfill for the ultimate disposal of hazardous waste materials and/or free liquids.

	PRELIMINARY IDEN	TABLE 2-5 (CONT.) Tification of Potential federal action-specific ar Hertel Landfill feasibility study	ARs AND TBCs
FEDERAL STATUTE	REGULATION/GUIDANCE	SISGONS	APPLICABILITY TO SITE CONDITIONS
RCRA (40 CFR 264) Subpart F	Ground Water Protection	Ground water monitoring/corrective action requirements: dictates adherence to MCLs and establishes points of compliance.	Potential ARARs for alternatives which utilize a landfill for the ultimate disposal of hazardous waste materials and/or free liquids.
RCRA (40 CFR 268)	Land Disposal Restrictions	Identifies hazardous wastes that are restricted from land disposal and sets treatment standards for restricted wastes.	Potential ARARs which may limit the use of land disposal in remediating certain hazardous wastes.
Safe Drinking Water Act (40 CFR 144 and 146)	Underground Injection Control Requirements	Provides the general requirements. technical criteria and standards for underground injection wells. including prohibitions of unauthorized injection. prohibition of movement of fluid into underground sources of drinking water. and requirements for the discharge of hazardous wastes.	Potential ARARs for alternatives which utilize underground injection as a remedial method.
Clean Water Act (40 CFR 122-125)	National Pollutant Discharge Elimination System (NPDES) Permit Requirements	Permits contain applicable effluent standards (1.e., technology-based and/or water quality-based), monitoring requirements, and standards and special conditions for discharge.	ARARs for alternatives involving treatment methods which discharge effluents to area water bodies.
Clean Water Act (40 CFR 403)	Discharge to Publicly-Owned Treatment Works (POTW)	A national pretreatment program designed to protect municipal wastewater treatment plants and the environment from damage that may occur when hazardous. toxic or other nondomestic wastes are dis- charged into a sewer system.	ARARs for alternatives involving treatment methods which discharge effluents to POTWs.

		HERTEL LANDFILL FEASIBILITY SIUDY	
FEDERAL STATUTE	REGULATION/GUIDANCE	SISONAS	APPLICABILITY TO SITE CONDITIONS
Clean Air Act (40 CFR 50)	National Ambient Air Quality Standards (NAAQS)-Particulates	Establishes maximum concentrations for particulates and fugitive dust emissions.	ARARs for alternatives involving treatment methods which impact ambient air (i.e. incineration, soil venting. etc.).
Clean Air Act (40 CFR 50)	New Source Performance Standards (NSPS)	Requires Best Available Control Technology (BACT) for new sources, and sets emissions limitations.	ARARs for alternatives involving treatment methods which impact ambient air (i.e incineration, soil venting. etc.).
Clean Air Act (40 CFR 61)	Emissions Standards for Hazardous Pollutants (NESHAPS)	Establishes emissions limitations for hazardous air pollutants.	Potential ARARs for alternatives using treatments (i.e incineration. etc.) which result in emissions to the air.
Occupational Safety and Health Act (29 CFR 1904)	Recordkeeping. Reporting and Related Regulations	Outlines recordkeeping and reporting requirements.	ARAR for all contractors/subcontractors involved in hazardous activities.
Occupational Safety and Health Act (29 CFR 1910)	General Industry Standards	Establishes requirement for 40-hour training and medical surveillance of hazardous waste workers.	ARAR for workers and the workplace throughout the implementation of hazardous activities.
Occupational Safety and Health Act (29 CFR 1926)	Safety and Health Standards	Regulations specify the type of safety equipment and procedures for site remediation/excavation.	ARAR for workers and the workplace throughout the implementation of hazardous activities.

STATE STATIFF	PRELIMINARY II Regulation/guildance	TABLE 2-6 Dentification of Potential State Action-Specifi Hertel Landfill Feasibility Study Strudy Study	C ARARS AND TBCs Applications of the conditions
NY State Pollutant Discharge Elimination	NYSPDES Mater Quality Toxic Effluent Standards.	Establishes water quality standards. effluent limitations. standards of	ARARs for alternatives involving treatments which discharge effluents to area surface
System System	Limitations and Treatment	performance, toxic effluent standards	or ground waters.
(/G/-DG/ XX11MG)	vequi rements	and promibitions, and pretreatment standards.	
NY State RCRA	Standards for the Design	Outlines design specifications and	Potential ARARs for alternatives which
Hazardous Naste	and Operation of Hazardous	standards of performance for disposal	utilize landfill. incineration. tanks.
Regulations (6 NYCRR 37D-372)	Waste Treatment Facilities (Minimum Technology Require- ments)	facilities and treatments.	containers, etc. for disposal of wastes.
NY State RCRA	Closure/Post Closure	Establishes requirements for the	Potential ARARs for alternatives which
Hazardous Waste	Requirements	closure (clean closure and waste-in-	utilize a landfill for the ultimate
Regulations		place closure) and long-term	disposal of hazardous waste materials
(6 NYCRR 373)		management of a hazardous disposal facility.	and/or free liquids.
NY State Solid	Solid Maste Management	Requirements for landfill operation	Potential ARARs for capping alternatives and
Waste Regulations (6 NYCRR 360-361)	Requirements and Siting Restrictions	and closure. Incineration. and other solid waste management activities.	alternatives involving incineration of solid waste.
NY State Air Pollution Control Regulations (6 NYCRR 200-221)	General Air Quality and Air Emission Requirements	Establishes maximum ambient levels for criteria pollutants and establishes emissions limitations for	Potential ARARs for alternatives involving remedial actions which impact ambient air.
		sources which emit VOCs into the air.	

	MAXIMUM CONCI Exceeding Arai	ENTRATION Rs/TBCs in:	FEDI	ERAL ARARs/TB	3	NY ARARS/T 4	BCs	و
					Amblent	Ground Water		RISK-BASED
	MONITORING	LEACHATE	1	2	Water Quality	Quality	2	CLEANUP
	MELLS	SEEP	MCL	MCLG	Criteria	Criteria	NYMCL	LEVEL
PARAMETER	(ddd)	(qdd)	(qdd)	(qdd)	(qdd)	(ddd)	(dqq)	(dqd)
Benzene	9		J.C.	0	0.66	QN	-0	
Ch1 orobenzene	24	68	100	100	488	20	2	
Et hy 1 benzene	5		700	700	2,400	50	2	
Tol uene	33		1.000	1.000	15,000	50	5	
Xy] enes	240		10.000	10.000		50	5	
1.4-D1chlorobenzene	10		75	75	470	4.7	LO LO	
D1ethy1phtha1ate	006	_			434.000	50	50	
2.4-Dimethylphenol	82	_			400	0.3	ŝ	
01 - n - octyl phthal ate	69					50	50	
Naphthalene	39					10	50	
Phenol	72	21			3,500	1	50	
Arsenic	44.1/33.8		50	0	0.0022	25	50	0.486
Barium	1.980/732	3.580	1.000 (2.000)	2.000		1000	1000	
Cadmium		178	9	5	10	10	10	
Chromium	538/ND	316	100	100	50		50	112
Copper	B46/ND	370	(1.300)	(1,300)	1000	1000	200	
Iron	893,800/116,000	526.000				300	300	
Lead	313/5.9	454	50 (5)	<u>(</u> 0)	50	25	50	
Magnesium	133.000/55.500	83,600				35,000		
Manganese	121,000/27,900	25.300				300	300	2
Mercury		4.1	2		10	(2)	(2)	
Stlver	266/ND		50		50	50	50	
Sod1um	115,000/122,000						20.000	
Zinc	2,880/91.6	11.200					300	
1. MCL - Max Contamin	ant Level. Natl. Primary	v Drinking Water	· Regulations.	4.	NYSDEC 6NYCRR	Part 703, Regula	tions for	ground water (1/9/89
Final Rule A	mendments to SDMA. U.S.	EPA. 7/1/90. (4	0 CFR 141):	5.	WYSDOH TONYCR	R Part 5. Regulat	ions for c	drinking water
Phase II MCL	s. 56 FR 3526 (Proposed	MCL)	:	,	supplies (1/9	(68).		
Z. MCLG- Proposed Max	Contaminant Level Goal	, based on healt	ch considerations	9	Risk-based cl	eanup level deriv	ed from a	future use residenti

TABLE 3-1 HERTEL LANDFILL SITE - SUMMARY OF GROUND WATER CONTAMINANTS DETECTED IN GROUND WATER SAMPLES ABOVE APPLICABLE OR RELEVANT AND

only. Amendments to SDMA. U.S. EPA. 7/1/90 (Proposed MCLG) Derived from published EPA Ambient Water Quality Criteria (drinking water only). 45FR 79318-79379. November 28. 1980

ч.

producing a compound specific 1E-05 risk level. For inorganics. unfiltered / filtered analytical results are shown.

scemario for the most sensitive receptor (adult)

TABLE 3-2 HERTEL LANDFILL SITE - SUMMARY OF SOIL/SEDIMENT CONTAMINANTS DETECTED ABOVE APPLICABLE OR RELEVANT AND APPROPRIATE REQUIREMENTS (ARARs) AND TO-BE-CONSIDERED REQUIREMENTS (TBCs) AND ASSOCIATED RISK-BASED CLEAN-UP LEVELS

REACECTING ARAINON CONCENTION I I NETALS I EXCECTING ARRAYTES, IN: 1	M HATE Aquatic Human EP Toxicity Health MENT Basis Criteria m) (ppm) (ppm) (ppm)	TALS 1 Limits 3			
1 1 1 1 SURFACE & STREAM' LEACHATE Aquatic Human Lifnits 3 SUBSURF. WETLANO SEEP Toxicity Health 2 of Barzo(cla)Purene 1.5 1 0.77 0.26 A A Berzo(cla)Purene 1.7 0.93 0.26 A A A A A A A A A A A A B A A A A A A A A A B A A A A A A A A A A A A A A A A A A <t< th=""><th> 1 1 HATE Aquatic Human EP Toxicity Health MENT Basis Basis Criteri m) (ppm) (ppm) (ppm)</th><th>Limits 3</th><th></th><th>5</th><th></th></t<>	1 1 HATE Aquatic Human EP Toxicity Health MENT Basis Basis Criteri m) (ppm) (ppm) (ppm)	Limits 3		5	
Benzo(a) Anthracene 1.5 1 0.26 Benzo(b) Fluoranthene 0.77 0.26 Benzo(b) Fluoranthene 0.77 0.26 Benzo(b) Fluoranthene 1.2 0.77 0.26 Benzo(b) Fluoranthene 1.2 0.77 0.26 Benzo(b) Fluoranthene 1.7 0.93 0.26 Benzo(a) Pyrene 0.87 0.26 0.26 Benzo(1.2.3-CD) Pyrene 1.7 0.93 0.26 Indeno(1.2.3-CD) Pyrene 0.39 0.26 5 Arsenic 109 30 14.5 0.26 Barium 4.490 6.230 302 0.26 Codmium 113 16.5 17.4 0.8 Cober 319 67.8 64.8 19 Iron 137,000 83.300 24.000 40.000		2 of Tolerance (ppm)	4 NJOEP Soil Action Level (ppm)	Observed On-Site Background Levels (ppm)	6 SOIL RISK-BASED CLEANUP LEVELS (ppm)
Berzo(b)Fluoranthene 0.77 0.26 Berzo(k)Fluoranthene 1.2 0.26 Berzo(a)Pyrene 1.7 0.93 0.26 Berzo(a)Pyrene 1.7 0.93 0.26 Chrysene 1.7 0.93 0.26 Chrysene 1.7 0.93 0.26 Indeno(1.2.3-C0)Pyrene 0.39 0.26 5 Arsenic 109 30 14.5 5 Barium 4.490 6.230 302 5 33 Cadmium 113 16.5 17.4 0.8 10 Copper 319 64.8 13 26 111 Copper 137,000 83.300 24.000 40.000	1 0.26				
Berzo(k)Fluoranthene 1.2 0.26 Berzo(a)Pyrene 0.87 0.26 Berzo(a)Pyrene 0.87 0.26 Chrysene 1.7 0.93 0.26 Chrysene 0.39 0.26 5 Indeno(1.2.3-C0)Pyrene 0.39 0.26 5 Arsenfc 109 30 14.5 5 Barfum 4.490 6.230 302 0.26 Cadmfum 113 16.5 17.4 0.8 Cober 319 67.8 64.8 19 Coper 3137,000 83.300 24,000 40.000	0.77 0.26				
Berzo(a)Pyrene 0.87 0.26 Chrysene 1.7 0.93 0.26 Chrysene 1.7 0.93 0.26 Indeno(1.2.3-CD)Pyrene 0.39 14.5 5 33 Arsenic 109 30 14.5 5 33 Arsenic 113 16.5 17.4 0.8 10 Barium 30.9 64.4 26 111 Cober 319 67.8 64.8 19 114 Iron 137,000 83.300 24,000 40.000	0.26				
Chrysene 1.7 0.93 0.26 Indeno(1.2.3-C0)Pyrene 0.39 0.26 5 Arsen1c 0.30 14.5 0.26 Arsen1c 109 30 14.5 5 Bartum 4.490 6.230 302 5 33 Cadm1um 113 16.5 17.4 0.8 10 Copper 319 67.8 64.8 19 114 Copper 3137,000 83.300 24.000 40.000	0.26				
Indeno(1,2,3-C0)Pyrene 0.39 0.26 Arsen1c 109 30 14.5 5 33 Barium 4,490 6,230 302 5 33 Cadmium 113 16.5 17.4 0.8 10 Cadmium 113 16.5 17.4 0.8 10 Cooper 319 67.8 64.8 19 114 Cooper 3137,000 83.300 24.000 40.000	0.93 0.26				
Arsenic 109 30 14.5 5 33 Barium 4.490 6.230 302 5 10 Barium 113 16.5 17.4 0.8 10 Cadmium 113 16.5 17.4 0.8 10 Chromium 30.9 64.4 26 111 Copper 319 67.8 64.8 19 114 Iron 137,000 83.300 24,000 40.000	0.26				
Barlum 4.490 6.230 302 Cadmlum 113 16.5 17.4 0.8 10 Cadmlum 30.9 64.4 26 111 Copper 319 67.8 64.8 19 114 Copper 137,000 83.300 24,000 40.000	14.5	5 33	20	4.8-109	0.06
Cadm ¹ um 113 16.5 17.4 0.8 10 Chrom ¹ um 30.9 64.4 26 111 Copper 319 67.8 64.8 19 114 Copper 319 67.8 64.8 19 114 Iron 137,000 83.300 24,000 40.000	302			37.3-1.300	
Chromium 30.9 64.4 26 111 Copper 319 67.8 64.8 19 114 Copper 317,000 83.300 24.000 40.000 Iron 137,000 83.300 24.000 40.000	17.4 (.8 10	e	0.78-2.7	
Copper 319 67.8 64.8 19 114 Iron 137,000 83,300 24,000 40,000 Iron 137,000 83,300 24,000 40,000	64.4	26 111	100	8.8-16	25
Iron 137,000 83,300 24,000 40,000	64 . B	19 114	170	26	
	.300 24.0	00 40.000		12,000-69,800	
read T.1.1/0 93./ 250		27 250	250-1,000	10.6-44.4	
Manganese 68.100 1.620 428 1.100	.620	28 1,100		782-38.100	2
Mercury 1.6 0.7 0.11 2		11 2	1	<0.11	
Nickel 347 29 31.7 22 90	31.7	22 90	100	17.2-2.509	
21nc 615 372 640 85 800	640	85 800	350	76.7-177	

Sediment Criteria for metals based on geometric mean of no effect levels and lowest effect levels (Persand. 1989). 2 - Sediment Uniteria for metals based on significant portions of the ecosytem. It is highly likely 3 - "Limits of Tolerance"- if exceeded in significant portions of the ecosytem. It is highly likely

that blota are impaired and remediation should be considered necessary.

4 - Due to the lack of applicable New York soil criteria. NJDEP soil Action Levels are provided for comparative purposes.

5 - Reference Table 4-1. Remedial Investigation Report. Hertel Landfill Site. TAMS/TRC. February 1991.

6 - Risk based cleanup level derived from a future use residential scenario (children) producing a compound specific hazard quotient of 1 or less.

TABLE 3-3 HERTEL LANDFILL SITE - SUMMARY OF SURFACE WATER CONTAMINANTS DETECTED ABOVE APPLICABLE OR RELEVANT AND APPROPRIATE REQUIREMENTS (ARARs) AND TO-BE-CONSIDERED REQUIREMENTS (TBCs), AND ASSOCIATED RISK-BASED CLEAN-UP LEVELS

	MAXIMUM COI Exceeding Ai	NCENTRATION Rars/TBCs IN:	FEOERAL	. ARARs/TBCs	£	9	NY ARARS 6	/TBCs	
PARAMETER	SURFACE MATER (ppb)	LEACHATE SEEP (ppb)	1 MCL (ppb)	2 MCLG (ppb)	Amblent Water Quality Criteria (ppb)	Water Nater Supply (ppb)	rishing and fish Propagation (ppb)	risning and Fish Survival (ppb)	
Chlorobenzene		- 8	100	100	488	20	ц л	50	
Bis(2-ethylhexyl)phthalate		- 2		0	21.000	4*	0.6		
Phenol		21			3.500	1	1	1	
A) um f num	4.280	20.400					100*		
Barium		3,580	1,000 (2,000)	2.000		1.000			
Cadmilum	37.1	178	5	5	10	10			
Chromium		316	100	100	50	50			
Copper		370	(1,300)	(1.300)	1.000	200			
Iron	190,000	526.000				300	300	300	
Lead	54.9	454	50 (5)	(0)	50	50			
Magnes 1 um	37,300	836.000				35,000			
Manganese	11,800	25,300				300			
Mercury	1.0	4.1	2		10	2	0.2*	0.2*	
Vanadium		54.6					14	190	
Zinc	347	11.200					30		
Cyanide		85.3					5.2**	22**	

Amendments to SDWA. U.S. EPA. 7/1/90. (40 CFR 141); Phase II MCLs, 56 FR 3526 (Proposed MCL) MCL - Max Contaminant Level, Natl. Primary Drinking Water Regulations. Final Rule :

MCLG- Proposed Max Contaminant Level Goal. based on health considerations only. Amendments to SDWA. U.S. EPA. 7/1/90 (Proposed MCLG) ۶.

Derived from published EPA Ambient Water Quality Criteria (drinking water only). 45FR 79318-79379, November 28, 1980 ы.

NYSDEC 6NYCRR Part 703. Regulations for ground water (1/9/89). 4.

NYSDOH 10NYCRR Part 5. Regulations for drinking water supplies (1/9/89). ۍ. New York State Ambient Water Quality Standards and Guidance Values. NYSDEC 6 NYCRR Part 701 and 702. Regulations for Surface Water. . 9

Guidance value : .

Standard for free cyanide

TABLE 3-4

GENERAL RESPONSE ACTIONS BY ENVIRONMENTAL MEDIA HERTEL LANDFILL FEASIBILTY STUDY

GROUND WATER

- No Action
- Institutional Controls
- Containment
- Collection/Treatment/Discharge

SOIL/WASTE

- No Action
- Institutional Controls
- Containment
- Excavation/Treatment
- In Situ Treatment

		TABLE	3-5	
		Soil/Waste Matrix Tec Hertel Landfill Fi	HNOLOGY SCREENING EASIBILITY STUDY	TECHNICAL IMPLEMENTABILITY
ENERAL RESPONSE ACTION	TECHNOLOGY	PROCESS OPTION	DESCRIPTION	COMMENTS
NO ACTION	NONE	NOT APPLICABLE	No action.	Required for consideration under the NCP.
			 Deed for site would be revised to include restrictions on future site use or development. 	— — — — — — — — — — — — — — — — — — —
CONIHUL	HESINCIUNS	ACCESS RESTRICTIONS	Fencing of entire site and posting of warning signs to limit public access.	Potentially applicable.
		CLAY	Placement of compacted clay over contaminated soils and wastes.	Potentially viable.
		SYNTHETIC	Placement of low permeability synthetic membrane over contaminated soils and wastes.	Potentially viable.
	CAPPING	ASPHALT	Paving of contaminated soils and wastes with bituminous material.	Potentially viable.
		CONCRETE	Paving of contaminated soils and wastes with concrete.	Potentially viable.
		MULTI-LAYER CAP	Multilayer cap of compacted clay, synthetic membrane, drainage layer, topsoil, and vegetative cover.	Potentially viable.
ONTAINMENT		OFF-SITE LANDFILL	Excavation of contaminated soils and wastes with disposal at an off-site landfill licensed to accept such wastes.	Would require excavation of surface soils; waste materials; and subsurface soils.
1		ON-SITE LANDFILL	Construction of landfill on-site, in accordance with current landfill construction requirements, for contaminated soil and waste disposal.	Would require excavation of surface soils; waste materials; and subsurface soils.
	STABILIZATION	CEMENT-BASED	Soils are mixed with Portland cement, forming rigid concrete matrix.	Most suitable for immobilizing metals; may require secondary containment of end product. Not suitable for treatment of waste materials.
I	SOLUPHCATION (CHEMICAL FIXATION)	POZZÓLANIC	Silaceous material, lime and water mixed with contaminated soils to form rigid matrix.	May require secondary containment due to subsequent leaching of process water. Not suitable for treatment of waste materials.
GENERAL RESPONSE ACTION ACTION TREATMENT THEATMENT CHEMICAUPHYSICAU	TABLE 3-5 (Contin Soll/WASTE MATRIX TECHNOL HERTEL LANDFILL FEASIE	ued) LOGY SCREENING BILITY STUDY		
--	--	--	---	
	PROCESS OPTION	DESCRIPTION	COMMENTS	
	ROTARY KILN INCINERATION	Soils and wastes pass through a refractory-lined rotating cylinder. Afterburner for exhaust gases.	Effective for destruction of organics. Does not treat inorganics.	
	FLUIDIZED BED INCINERATION	Soil and waste contaminants thermally destroyed in bed of hot inert granular material fluidized by flow of air.	Effective for destruction of organics. Does not not treat inorganics	
	INFRARED INCINERATION	Flux of near infrared radiation initiates and sustains pyrolysis of feed materials.	Effective for destruction of organics. Does not treat inorganics.	
	MECHANICALTHERMAL	Volatile compounds are removed from soils through heating and/or mechanical aeration.	Effective for destruction of volatile organics. Not suitable for treatment of waste materials.	
	SOIL WASHING	Use of an extractant solution to remove contaminants. Solutions used include water, surfactants, acids, bases, oxidizing or reducing agents.	Several different washing solutions would be required to treat all types of soil contaminants. Not applicable to treatment of waste materials.	
	LANDFARMING	Aerobic biodegradation of contaminants in soils applied to the ground surface, with nutrient addition.	Effective for destruction of volatile organics. Ineffective for inorganic contaminants. Not applicable to treatment of waste materials.	
	DECHLORINATION	Alkali metals or alkali metal/ polyethylene glycol used to strip chlorine atoms from hazardous halogenated hydrocarbons.	Effective for destruction of chlorinated organics. Ineffective for inorganic contaminants. Not applicable to treatment of waste materials.	
	OXIDATION	Raising or lowering of molecular oxidation state to render contaminants less toxic.	May form more toxic or soluble degradation products, especially with inorganic contaminants. Not applicable to treatment of waste materials.	
OFF-SITE TREATMENT	VARIOUS	Soils and wastes transported off-site for treatment at a permitted facility (such as a hazardous waste incinerator) in accordance with current waste disposal regulations.	Effectiveness depends on treatment system used.	

	COMMENTS	 Effective for destruction of organics, especially for fuel spill contaminants, Ineffective for inorganics. Not applicable to treatment of waste materials. 	se of Ineffective for inorganic contaminants. contaminants. Not applicable to treatment of waste materials.	A May be effective for organic contaminants. Not demonstrated on a large scale or on in situ municipal waste materials.	Effective for destruction of chlorinated organics. Ineffective for inorganic contaminants. Not applicable to treatment of waste materials.	May form more toxic or soluble degradation products, especially with inorganic contaminants. Not applicable to treatment of waste materials.	Applicable to medium solubility organics. Ineffective for inorganics. Not applicable to treatment of waste materials.	rials Applicable for soils contaminated with a mixture of contaminants; in loosely packed rubbish, potential for underground fire exists.
inued) OLOGY SCREENING EASIBILITY STUDY	DESCRIPTION	Stimulation of indigenous bacteria o introduced strains, with nutrient addition.	Soil aerated or vacuumed through u air ''wells* to remove volatile contaminants.	Radio frequency waves heat soil anc thermally decompose, vaporize, and distill hazardous constituents.	Alkali metals or alkali metal/ polyethylene glycol used to strip chlorine atoms from hazardous halogenated hydrocarbons.	Raising or lowering of molecular oxidation state to render contaminan less toxic.	Contaminated soils are flooded with water and the elutriated solution is collected.	Contaminated soils and waste mater are melted via an electric current, resulting in a glassy crystalline monolith.
TABLE 3-5 (Cont SOIL/WASTE MATRIX TECHNO HERTEL LANDFILL FE	PROCESS OPTION	BIODEGRADATION	SOIL VENTING	RADIO FREGUENCY HEATING	DECHLORINATION	OXIDATION	SOIL FLUSHING	VITRIFICATION
	ECHNOLOGY				IN SITU TREATMENT			
	GENERAL RESPONSE ACTION							

SCREENED ON BASIS OF TECHNICAL IMPLEMENTABILITY	COMMENTS	Fulfills NCP requirement for consideration of no action alternative.	Fuffills NCP requirement for consideration of no action alternative; also identifies off-site contaminant migration.	Would prevent potential usage of contaminated water for potable water supply.	Private wells are currently unaffected by contaminant migration; potential further migration could be tracked through continued ground water monitoring.	Potentially viable for limiting leachate production.	 Potentially viable for limiting leachate ts production. 	Potentially viable for limiting leachate production.	Potentially viable for limiting leachate production.	r Potentially viable for limiting leachate production.	Depth to bedrock and fractured nature of bedrock limit effectiveness of barrier. Rocky nature of soil limits technical feasibility. Could minimize ground water flow through fill material in a containment scenario.	Unsuitable for rocky soils.
OGY SCREENING BILITY STUDY	DESCRIPTION	No action.	Continued ground water monitoring.	Legal restictions on ground water use in the vicinity of the contaminated area.	Provision of potable water to adjacent landowners currently using private wells.	Placement of compacted clay over contaminated soils limits infiltration.	Placement of low permeability synthetic membrane over contaminated soils limi infiltration.	Paving of contaminated soils with bituminous material limits infiltration.	Paving of contaminated soils with concrete limits infiltration.	Multilayer cap of compacted clay and/c synthetic membrane, drainage layer, topsoil, and vegetative cover limits infiltration.	Vertical trench is excavated under a soil/bentonite slurry and backfilled with a low permeability material.	Sheet piling is driven into soil to form barrier wall.
TABLE 3-6 GROUND WATER TECHNOLO HERTEL LANDFILL FEASIBI	TECHNOLOGY PROCESS OPTION	NONE NOT APPLICABLE	CONTINUED GROUND NOT APPLICABLE WATER MONITORING			CLAY	SYNTHETIC	CAPPING ASPHALT	CONCRETE	MULTI-LAYER CAP	VERTICAL BARRIER	SHEET PRING
	GENERAL RESPONSE ACTION	NO ACTION			CONTROL					CONTAINMENT		

	COMMENTS	Potentially viable, proven technology.	Reinjection of ground water into fill materials may mobilize other contaminants within waste.	Potentially viable, proven technology.	Potentially viable, proven technology.	No sanitary sewers at or near site limit feasibility.	Potentially viable.	Proven effective for organics.	Applicable to volatile organic contaminants.	Applicable to volatile organic contaminants.	Can be effective for organic removal; easily combined with ion exchange.	Used mainly for dissolved solids; metals and low level organics can cause clogging.	 Effective for metals; can be used in conjunction with resin adsorption for organics treatment. 	Applicable to metals.
d) ^ SCREENING TY STUDY	DESCRIPTION	Wells and pumping systems placed within contaminant plume for collection.	Injection wells or infiltration galleries placed upgradient of contamination and extraction wells for greater hydrodynamic control.	Manifolded system of extraction points connected to common collection source.	Placement of trench with high permeability materials, used to divert ground water flow.	Extracted ground water discharged to local POTW for treatment.	Extracted ground water discharged to licensed commercial facility for treatment and/or disposal.	Activated sludge process utilizes acclimated bacteria for aerobic degradation of contaminants.	Transfer of volatile organic compounds to gaseous faction through mixing with large volumes of air in a packed column.	Contaminants adsorbed to activated carbon by internal pores of carbon granules.	Similar to carbon adsorption but synthetic resins are used.	Removal of solutes from solution by a semi-permeable membrane under a high pressure gradient.	Contaminants removed from aqueous phas by exchanging places with ions held by ion exchange material.	Contaminants removed by decreasing solubility.
TABLE 3-6 (Continue Ground Water Technolog) Hertel I Andell I FFASIBIL	PROCESS OPTION	EXTRACTION WELLS	EXTRACTION INJECTION WELLS	WELL POINTS	FRENCH DRAINS	WLDG	COMMERCIAL TREATMENT FACILITY	BIOREACTOR	AIR STRIPPING	CARBON ADSORPTION	RESIN ADSORPTION	HEVERSE OSMOSIS	ION EXCHANGE	PRECIPITATION
	TECHNOLOGY		EXTRACTION					BIOLOGICAL						
	GENERAL RESPONSE ACTION		COLLECTION									TREATMENT		

	COMMENTS	SITE program technology: applicable to ground water contaminated with heavy metals and landfill leachates. Proven for treatment of chlorinated VOCs, semivolatiles & pesticides/PCBs in EPA SITE testing. Primarily used for PCB transformer oils. Does not treat non-chlorinated hydrocarbons. Potential for incomplete oxidation and production of toxic oxidation and production of toxic oxidation and production swhile chlorinated compounds require anaerobic conditions. Potentially viable. Potentially viable. Potentially viable at site. Storm sewers not available at site.
Continued) NOLOGY SCREENING FEASIBILITY STUDY	DESCRIPTION	Solid particles removed from liquids using pressure filter. An oxidizing agent such as hydrogen peroxide is mixed with the waste stream and exposed to ultraviolet light to oxidize contaminants. Chemical agent is mixed with waste and exposed to ultraviolet light to oxidize contaminants. Chemical agent is mixed with waste stream to remove halogen atoms from chlorinated hydrocarbons. Raising or lowering of molecular oxidation state to render contaminants less toxic. Stimulation of indigenous bacteria or introduced strains to degrade organics by means of nutrient addition. Treated water is reinjected imto water table via wells. Treated water is reinjected via subsurface water is discharged directly into surface water body via storm sewer. Treated water is discharged indirectly to surface water body via sanitany sewer and POTW.
TABLE 3-6 (0 GROUND WATER TECH HERTEL LANDFILL F	PROCESS OPTION	MEMBRANE MICROFILTRATION UV OXIDATION UV OXIDATION DEHALOGIENATION REDUCTION BIODEGRADATION BIODEGRADATION GALLERIES SURFACE WATER SURFACE WATER SURFACE WATER SURFACE WATER SURFACE WATER SANTARY SEWER
	TECHNOLOGY	CHEMICAL IN SITU TREATMENT OFF SITE DISCHARGE
	GENERAL RESPONSE ACTION	DISCHARGE

ROCESS OPTION	COST	No cost.	Low capital cost. Low capital cost, low maintenance cost.	Moderate capital; low maintenance. Moderate capital; moderate maintenance.	Moderate capital; high maintenance	Moderate capital; moderate maintenance.	Moderate capital; moderate maintenance.	Very high capital; low maintenance. Very high capital; moderate O&M.	High capital; moderate O&M. High capital; moderate O&M.	High capitai; moderate O&M.	High capital: Iow O&M. High capital: Iow O&M.
* CHOSEN	IMPLEMENTABILITY	sity implemented.	quires appropriate legal authority. sily implemented.	sily implimented; requires future land = restrictions. riy easily implemented; requires ure land use restrictions.	sily implemented ; requires future d use restrictions.	sily Implemented; requires future d use restrictions.	rly easily implemented; requires ure land use restrictions.	Rcult to Identify an existing RCRA lifty with sufficient capacity to cept contaminated materials. Rcult to handle excavation of RCRA dfill facility simultaneously.	ble units evaluable; numerous mmercial suppliers. Nted number of commercial suppliers.	rco brand no longer manufactured but wiously purchased units available ough hazardous waste remediation ns. Pequires adequate electrical wer source.	aliability of certain treatment units 0. incinerators) limited. 9. incinerators) limited. 9. incinerability complicated by 1. inited 1. in
E 3-7 LE 3-7 CESS OPTION SCREENING FEASIBILITY STUDY	EFFECTIVENESS	Allows contaminated soils and waste materials to remain as potential sources of ground water and surface water contamination.	Limits further disturbance of existing contamination or introduction of additional contaminated materials. Limits human exposure to site, exposure to wildlife partially limited.	Susceptible to cracking. Ea Susceptible to surface water ponding. Fa	Susceptible to weathering and cracking.	Susceptible to weathering.	Least susceptible to cracking or Fa degradation; easily supports vegetative fut cover.	Removes soil contaminants and waste Di materials as a furture source of tax continued contamination. ac Soil contaminants and waste materials Di are not treated but are contained mu on-site in a more protective state. lar	Effective for organic destruction; MA applicable to solls and waste materials. co Effective for organic treatment; best suited for slurries and sludges.	Effective for organic destruction; Sh applicable to soils and waste materials. pr fin	Treatment method determined by vendor Av on basis of contaminants and current (e. regulations. Im Effective for mixed contaminants. Pri nu
TABI SOIL/ WASTE MATRIX PRO HERTEL LANDFILL	PROCESS OPTION	NOT APPLICABLE	DEED RESTRICTIONS *	CLAY SYNTHETIC	ASPHALT	CONCRETE	MULTI-LAYER CAP	OFF-SITE LANDFILL	ROTARY KILN ROTARY KILN INCINERATION FLUIDIZED BED INCINERATION	INFRARED INCINERATION	VARIOUS *
1 1 1	TECHNOLOGY	NONE	SITE USE RESITRICTIONS		CAPPING	<u> </u>	<u> </u>	LANDFILL DISPOSAL			OFF-SITE TREATMENT
I I	GENERAL RESPONSE ACTION	NO ACTION	INSTITUTIONAL CONTROL				CONTAINMENT			TREATMENT	

İ

1

		HOSEN PROCESS OPTION	COST	No cost	Low capital; moderate O&M.		High capital; low O&M	Moderate capital; low maintenance.	Moderate capital; moderate maintenance.	Moderate capital; high maintenance.	Moderate capital; moderate maintenance.	Moderate capital; moderate maintenance.	Moderate capital; kw maintenance.	Moderate capital; moderate Q&M.	Moderate capital; moderate O&M.	Moderate capital; moderate O&M.	Moderate capital; highO&M.	Moderate capital; low to moderate O&M.
		ф * У	IMPLEMENTABILITY	No Implementation required.	Easily Implemented.	Requires legal authoffy.	Requires legal authority and provision of an alternate water supply to adjacent homes with private wells.	Easily Implemented; future land use restrictions.	Fairly easily implemented; future land use restrictions.	Easily implemented; future land use restrictions.	Easily implemented; future land use restrictions.	Fairly easily implemented; future land use restrictions.	Implementation complicated by presence of subsurface boulders, Imfred availability of water on-site, and lack of flat areas for backfill mixing and slurry preparation.	Easily implemented; discharge permit required.	Easily implemented; discharge permit required.	Easily implemented; discharge permit required.	Comparing the second seco	Easily implemented.
ier Ier	ABLE 3-8	OCESS OPTION SCREENII ILL FEASIBILITY STUDY	EFFECTIVENESS	Not effective in prohibiting or monitoring contaminant migration.	Would provide Indication of off-site contaminant migration prior to impact of private wells.	Effective in limiting public ingestion of ground water contaminants, by eliminating installation of potable wells in contaminated areas.	Effective in preventing public ingestion of ground water contaminants, should private wells ever be impacted by contaminant migration.	Limits infitration; susceptible to cracking:	Limits infitration; susceptible to surface water ponding.	Limits Infitration; susceptible to weathering and cracking.	Limits Infitration; susceptible to weathering.	Limits infitration; least susceptible to cracking or degradation; provides means of methane control.	Effective in minimizing volume of ground water which comes in contact with waste materials under containment scenario.	Effective; best suited for steep hydraulic gradients and miscible contaminants.	Effective; best suited to shallow aquifers.	Effective; best suited to shallow aquifiers.	Effective with identification of an appropriate treatment faacility.	Proven effective for removal of organics, phenols.
	T	GROUND WATER PR HERTEL LANDF	PROCESS OPTION	NOT APPLICABLE			ALTERNATE WATER SUPPLY	GLAY	SYNTHETIC	ASPHALT	CONCRETE			EXTRACTION WELLS	WELL POINTS	FRENCH DRAINS	COMMERCIAL	BIOREACTOR
			TECHNOLOGY	NONE	CONTINUED GROUND WATER MONITORING		RESTRICTIONS			CAPPING			VERTICAL BARRIER		EXTRACTION			BIOLOGICAL
*** ***			GENERAL RESPONSE ACTION		NO ACIEON		CONTROL			L		CONTAINMENT			COLLECTION			

	COST	Mo derat e capital; O&M.	Moderate capital; moderate O&M.	High capital; moderate O&M.	Moderate capital; moderate O&M.	Low capital; moderate O&M.	Not well defined, due to innovative nature.	High capital; moderate O&M.	Moderate capital; low O&M.	Moderate capital; low O&M.	Low capital; low O&M.
D	I MPLEMENTABILITY	Readily implemented; may require treatment of off-gases.	Readily implemented; requires on- or off-site regeneration of carbon.	Prior to implementation, identification of resin applicable to contaminants in ground water is required.	Readily implemented.	Readily implemented.	Can be manufactured as mobile system.	Readily implemented.	Readily implemented; requires compliance with discharge criteria.	Readity Implemented; requires compliance with discharge criteria.	Readily implemented; requires compliance with discharge criteria.
ROCESS OPTION SCREENIN JFILL FEASIBILITY STUDY	EFFECTIVENESS	Generally effective for volatile organics.	Effective for low solubility organics.	Effective for organic removal.	Effective for Inorganic removal.	Effective for inorganic removal; precipitate must be disposed of.	Effective in removing heavy metals, based on pilot tests.	Effective for treatment of volatiles and semi-volatiles; no emissions or waste by-products produced.	Effective with permeable soils and relatively low flow rates.	Effective with permeable near-surface soils and at higher discharge rates.	Effective for discharge of treated ground water.
GROUND WATER P HERTEL LAND	LOGY PROCESS OPTION	AIR STRIPPING		RESIN ADSORPTION	ION EXCHANGE	PRECIPITATION *				ARGE INFILTRATION GALLERIES	SURFACE WATER
	VERAL RESPONSE ACTION TECHNOI			TREATMENT]			CHEMICAL		DISCHARGE ON-SITE DISCH	

TABLE 3-8 (Continued)

TABLE 3-9

REPRESENTATIVE PROCESS OPTION SUMMARY HERTEL LANDFILL FEASIBILITY STUDY

MEDIA	TECHNOLOGY	REPRESENTATIVE PROCESS OPTION
SOIL/WASTE MATRIX	No Action	Not applicable
	Institutional Control	Deed and access restrictions
	Capping	Multi-layer cap
	Landfill Disposal	Off-site landfill
	On-site Thermal Treatment	Rotary kiln incineration
	Off-Site Treatment	Various
	In Situ Treatment	In situ vitrification
GROUND WATER	No Action	Continued ground water monitoring
	Institutional Control	Deed Restrictions
	Capping	Multi-layer cap
	Vertical Barrier	Slurry wall
	Extraction	Extraction wells
	Off-Site Treatment	Commercial treatment facility
	Biological Treatment	Bioreactor
	Physical Treatment	Carbon adsorption Precipitation Membrane microfiltration
	Chemical Treatment	UV Oxidation
	On-Site Discharge	Discharge to surface water

TABLE 4-1

PRELIMINARY REMEDIAL ALTERNATIVE SUMMARY HERTEL LANDFILL FEASIBILITY STUDY

<u>Alternative 1</u> - No action, contin	ued ground water monitoring
<u>Alternative 2</u> - Site use restrict	ions, capping
<u>Alternative 2A</u> - Site use restrict	ions, capping, slurry wall
<u>Alternative 3</u> – Excavation of co off-site disposa treatment	ontaminated soil and waste materials with l, ground water collection with off-site
<u>Alternative 4</u> - Site use restric on-site physical	tions, capping, ground water extraction with treatment
<u>Alternative 4A</u> - Site use restric on-site treatment	tions, capping, ground water extraction with using UV oxidation/membrane microfiltration
<u>Alternative 5</u> - Excavation and ground water extr	on-site incineration of soils and wastes, action and on-site physical treatment
<u>Alternative 6</u> - Excavation and of water extraction	f-site treatment of soils and wastes, ground and on-site biological treatment
<u>Alternative 7</u> – In situ vitrifi on-site treatment	cation, and ground water extraction and using UV oxidation/membrane microfiltration

	NCP/GUIDANCE CRITERIA:	No Action	T T	tainment h Little or No eatment	Minimize Long-Term Management		Treat Primar	tment as ry Compon	ent	Innovative
T ECHNOLOGY TYPE	DESCRIPTION	-	2	2A	e	4	44	2	9	7
Ц	No action	*								
e Restrictions	Use of deed restrictions and fencing to limit access and use of site		•	*			٠			
	RCRA multimedia cap over landfill area		٠	*		٠	٠			
fon/Landfill Disposal	Disposal of contaminated soils and waste materials at off-site landfill				*					
Thermal Treatment	Rotary Kiln							٠		
e Treatment	Various								*	
Treatment	In Situ Vitrification									•
Б	No Action									
	RCRA multimedia cap over landfill area		•	٠		٠				
l Barrier	Slurry wall			•						
lon	Extraction wells				٠	٠	٠	٠	٠	٠
e Treatment	RCRA facility				٠					
cal Treatment	Bioreactors								Ŧ	
l Treatment	Carbon adsorption/precipitation					+		•		
l Treatment	UV oxidation/membrane microfiltration						•			•
	al Barrier tion te Treatment ical Treatment al Treatment al Treatment	al Barrier Slurry wall al Barrier Slurry wall tion Extraction wells te Treatment RCRA facility fcal Treatment Bioreactors al Treatment Carbon adsorption/precipitation al Treatment UV oxidation/membrane microfiltration	al Barrier Slurry wall tion Extraction wells te Treatment RCRA facility ical Treatment Bioreactors al Treatment UV oxidation/membrane al Treatment UV oxidation/membrane	al Barrier landfill area * al Barrier Slurry wall tion Extraction wells te Treatment RCRA facility fcal Treatment Bioreactors al Treatment UV oxidation/membrane al Treatment UV oxidation/membrane	al Barrier landfill area * * * * * * * * * * * * * * * * * * *	al Barrier landfill area * * * * * * * * * * * * * * * * * * *	al Barrier Slurry wall area * * * * * * * * * * * * * * * * * * *	al Barrier Slurry wall area * * * * * * * * * * * * * * * * * * *	al Barrier Slurry wall area * * * * * * * * * * * * * * * * * * *	al Barrier Slurry wall * * * * * * * * * * * * * * * * * *

TABLE 4-2

TABLE 4-3

PRELIMINARY REMEDIAL COST ESTIMATES HERTEL LANDFILL FEASIBILITY STUDY

	TOTAL PRESENT WORTH
	(MIIIIONS)
Alternative 1	
 No action, continued ground water monitoring 	\$2.5
Alternative 2	
 Site use restrictions, capping 	\$7.0
Alternative 2A	
 Site use restrictions, capping, slurry wall 	\$15
Alternative 3	
 Excavation of contaminated soil and waste materials with off-site disposal, ground water collection with off-site treatment 	\$190
Alternative 4	
 Site use restrictions, capping, ground water extraction with on-site physical treatment 	\$8.2
Alternative 4A	
 Site use restrictions, capping, ground water extraction with on-site treatment using UV oxidation/membrane microfiltration 	\$7.9
<u>Alternative 5</u>	
 Excavation and on-site incineration of soils and wastes, ground water extraction and on-site physical treatment 	\$ 120
Alternative 6	
 Excavation and off-site treatment of soils and wastes, ground water extraction and on-site biological treatment 	\$510
Alternative 7	
- In situ vitrification, and ground water extraction and on-site treatment using UV oxidation/membrane microfiltra	ation \$110

REMEDIAL ALTERNATIVES UNDERGOING DETAILED ANALYSIS HERTEL LANDFILL FEASIBILITY STUDY

<u>Alternative 1</u> - No action, continued ground water monitoring

<u>Alternative 2</u> - Site use restrictions, capping

Alternative 2A - Site use restrictions, capping, slurry wall

<u>Alternative 4</u> - Site use restrictions, capping, ground water extraction with on-site physical treatment

<u>Alternative 4A</u> - Site use restrictions, capping, ground water extraction with on-site innovative treatment (UV oxidation/membrane microfiltration)

					11	HOLE 5-2					
	Summary	of can	KCER RISK I	ESTINATES Herte	- ROAD IN	Provenen Feasibii	IS AND CLEARING VEGETAT .ITY STUDY	ion (Alternat	IVES 2, 2A,	4 and 4a)
11 1	CHRONIC	DAILY	COI	1				1 1	CHEMICAL	TOTAL	TOTAL
CHEMICAL	INTAKE((ICD	ADJUSTED I	FOR	SF H	IGHT OF	TYPE OF	SF BASIS/	SPECIFIC	PATHNAY	EXPOSURE
	(ng/kg/	(day)	ABSORPTIC)N ¦(∎g/k	g/day)-1;E	IDENCE	CANCER	SOURCE	RISK	RISK	RISK
* * * * * * * * * * * * * * * * * * * *			****		* * * * * * * * * * * *						
f I f I											
EXPOSURE PATHMAY: INHALATI	ON OF AI	RBORNE	CHEMICALS	S ADSORBE	d to dust					1E-06	2E-06
INORGANICS				1							
Arsenic	8.04	E-09	No		50	A	Lung	Air/HEAST	4E-07		
Chromium	2.13	E-08	No		41	A	Lung	OCCUPA/HEAST	9E-07		
		11111									

.

TABLE 5-3 SUMMARY OF CHRONIC MAZARD INDEX ESTIMATES - ROAD IMPROVEMENTS AND CLEARING VEGETATION (ALTERNATIVES 2, 2A, 4 AND 4A)

				,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	HERTEL LANDFILL FEASIBILITY STUDY		*****		
- 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	SUBCHRONIC	; CDI	1010101011111		***************************************		RFD ¦)) \	PATHNAY TOTAL
CHEMICAL	DAILY INTAKE	ADJUSTED FOR	RFD ;	CONFIDENCE	CRITICAL	SOURCE/	UNCERTAINTY ! NO	ODIFYING; HAZARD	HAZARD EXPOSURE
11	(ag/kg/day)) ABSORPTION	{(mg/kg/day)}	LEVEL	EFFECT	BASIS	ADJUSTHENTS; F	FACTORS QUOTIENT	(INDEX (HI); HI ;
								4 4 1 1 1 1 1 1 1 1 1 4 1 1 4 1 4 4 7 1 1 1 1	, , , , , , , , , , , , , , , , , , ,
14									
EXPOSURE PATHWAY: INHA	LATION OF AIRBOR	NE CHEMICALS A	osorbed to du	ST					1E+01 1E+01
- 1 1 1 1 1 0 0 1 0 1 0 1 0 1 0 1 0			, 3 4 1 6 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		·)				
INORGANICS	1	1				4	1 1		1
Chromium	3.6E-05	5¦ No	6E-06	NA	Nasal mucosa atrophy	HEAST	300	6E+00	1
Manganese	2.5E-03	B No	48-04	Medium	Increased prevalence of	IRIS	300	3 66+00	1
			****	1111111111111111		••••••••••••••••		• • • • • • • • • • • • • • • • • • •)
EXPOSURE PATHNAY: DERM	WAL CONTACT WITH	CHEMICALS IN S	OIL						9 E-0 1
- 4 1 4 1 1 4 4 4 1 3 1 4 1 4 1 4 1 4 1 5 1 4 1 4 1 4 1 4 1 4			1	1 1 1 1 1 1 1 1 1 4 4 1 1 1 4 4 4 1 1 1 1	·	* • • • • • • • • • • • • • • • • • • •)	1
INORGANICS	1	1	: :	1		;	1 1	1	
Arsenic	9.0E-05	0.10	1E-03	NA	Keratosis and hyperpigmentation	NA/HEAST	1	9E-01	1
	1							1 C 1 S 1 I F 1 A F 1 S 1 S 1 S 1 A F 1 A	1

		JUNINARI UT C	HERTEL LANDFILL F	EASIBILITY STUDY)

11	CHRONIC DAILY	CDI	! !	}	CHENICAL	TOTAL TOTAL
CHENTCAL	INTAKE(CDI)	ADJUSTED FOR	SF VEIG	T OF TYPE OF	SF BASIS/ SPECIFIC	PATHIAY EXPOSURE
	(sq/kq/day)	ABSORPTION	(mg/kg/day)-1 EVIDE	NCE CANCER	SOURCE RISK	RISK RISK
				,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		1
EXPOSURE PATHWAY: INHALATI	ION OF AIRBORN	e chenicals a	dsorbed to dust			7E-06 1E-05
INORGANICS		4 1		}		
Arsenic	4.29E-08	No	50 A	Lung	Air/HEAST 2E-06	:
			1 1 1 4 4 4 4 1 1 1 1 1 1 1 1 1 1 1 4	* * * * * * * * * * * * * * * * * * * *		\
LEXPOSURE PATHNAY: DERMAL C	CONTACT WITH C	HEMICALS IN S	OIL			5E-06 ¦
INORGANICS		1 1 1		1		1
Arsenic	2.82E-07	0.10	1.75	Skin	Water/IRIS 5E-06	1
iin maan maan maan maa			<u>inn minn</u>	*****		

•

•

TABLE 5-4 SUMMARY OF CANCER RISK ESTIMATES - CAP CONSTRUCTION (ALTERNATIVES 2, 2A, 4, AND 4A)

								SU	mmary of	CHRO	NIC HAZARD I	NDEX EST	INATES - CAP	CONS	TRUCTION (ALTERN	ATIVES 2	2, 2A	4 AND 4A)			
					111111						HERIEL LAN	UP ILL FEA	ASTRICTA ST	J UY								
[[]]]]]]]]]]]]]]]]]]		11111111	SUBCH	ONIC	: 0	DI	!	171311	1		********	11111111			8FD	RFT)		1 \$ 1 + 1 1 2 1 1	: Pathway	. TOTAL	!!
	CHENICAL		DAILY	NTAKE	ADJUS	ted for	i F	FD	CONFIDE	3OK		CRITICA	L	i	SOURCE/	UNCERT	INTY	MODIFYING	HAZARD	HAZARD	EXPOSURE	
11			(n g/kg	/day)	ABSO	RPTION	(mg/)	g/day)	LEVE	. 1		EFFECT		i	BASIS	ADJUSTI	ENTS	FACTORS	QUOTIENT	INDEX (HI), HI	
		111111																				1
					-																	11
TEAPUSUK	E PAIMAT: *********		101710176 		E UNER 111111	ILALSA HHHHH			USI 111111111						,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,					1 1E+01	; 1E+01	
	TNORGANTOS		!	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	! !		11111))))))))))))))))))))))))))))))))))))))		1 4 1 2 4 4 4 1 1 1 4 4 4					• • • • • • • • • • • • • • • • • • •	11111	******		41111111111		11
¦¦Chr om iu			3.	6E-05	: .	No		6E-06	. NA		Nasal	SUCOSA a	trophy		HEAST		300		6E+00			
Mangane	se		2.	5E-03	; ;	No	i –	4E-04	Mediu	ne ¦	Increas	ed preval	ence of	Ì	IRIS		300	3	6E+00	1		
							11111															
EXPOSUR	e páthmay:	dernal	CONTACT	WITH C	HEMICA	S IN S	OIL													9E-01	L I	
							888															
	INORGANICS						1			1				-						l .		
Arsenic		DUUD	9.	OE-OS		0.10	; (1 + 1 +	1E-03	; NA		Keratosis	and hyper	pigmentation	1 ;	NA/HEAST		1		9€-01	1		
	1111111111111	111111		111111	itiiii			111111		iiiii		111111111						1111111111	1111111111	i		

SUMWARY OF CANCER RISK ESTIMATES: CHILDREN (ALTERNATIVES 2 AND 2A) HERTEL LANDFILL FEASIBILITY STUDY

		HERIEL DANAFILL FERSION		
11 11	CHRONIC DAILY CDI		1	CHENICAL TOTAL
CHENICAL	INTAKE(CDI) ADJUSTE	FOR SF WEIGHT	OF TYPE OF SF BASIS/	SPECIFIC PATHWAY TOTAL
8	(mg/kg/day) ABSORP	ION (mg/kg/day)-1 EVIDE	ice cancer source	RISK RISK RISK
		I I <th></th> <th>1 4 4 4 4 4 5 4 5 4 6 4 6 4 6 4 6 7 4 7 4 6 6 7 7 4 6 6 7 7 4 6 7 7 4 6 7 7 4 6 7 7 4 6 7 7 4 7 7 4 7 7 7 7</th>		1 4 4 4 4 4 5 4 5 4 6 4 6 4 6 4 6 7 4 7 4 6 6 7 7 4 6 6 7 7 4 6 7 7 4 6 7 7 4 6 7 7 4 6 7 7 4 7 7 4 7 7 7 7
EXPOSURE PATHWAY: DERNAL (CONTACT WITH SOILS - CHILDREN			2E-05 ¦ 2E-05
			· · · · · · · · · · · · · · · · · · ·	\
INORGANICS	: :			1 1
Arsenic	1.1E-06	0.10 1.8E+00 A	Bladder,Liver,Lung Water/IRIS	2E-05
				1 1 1 4 1 4 1 4 4 4 1 1 1 4 5 4 1 4 4 4 4 1

							HERT	iel l and fill fea	SIBILITY STUDY											
						(: + 1 1 4 1 1 1 1 8 1 3 1 1 + 3 3 4 1 7 1 1 6 1 4 1 1 1					1111		****		
11	CHRO	NIC DAILY	r¦ CDI				1			1	RFD	1 8	¥FD ¦				PATHMAY		total	
H CHEMICAL	INT	KE(COI)	ADJUSTED	OR F	RFD ¦(CONFIDENCE	;	CRIT	CAL	1	SOURCE/	UNCER	TAINTY	NODIFYING	HAZARD		HAZARD		hazard	11
11	(mg.	/kg/day)	ABSORPTI	DN ¦{mg/i	kg/day)¦	LEVEL	t	EFF	CT	1	BASIS	ADJU	THENTS	FACTORS	QUOTIEN	T ¦I	INDEX (HI);IN	DEX (H)	1)]
										::::::::								1111		
EXPOSURE PATHWAY:	DERMAL CONTACT I	ITH SOIL	- CHILDRE	١												-	9€-0	2 1	1E-0	11
100000000000000000000000000000000000000									1111111111111							HİI		нiн		
INORGANIC	S ;		1	1	1		1			1		1	1		1	1				
Arsenic		8.7E-06	0.	10	1E-03		K	eratosis and hy	erpigmentatio	n !	NA/IRIS	1	1		9E-0	2 !				
111111111111111111111111111111111111111			innnii					1111010101010101		min					innin	HI.				

TABLE 5-7 SUMMARY OF CHRONIC HAZARD INDEX ESTIMATES: CHILDREN (ALTERNATIVES 2 AND 2A)

i

			Sunnakt up lan	rek kisk e	STIMATES - WELL CONSTRU	LITON (ALTERN	AIIVES 4 AND)4R)
				HERTEL LA	OFILL FEASIBILITY STUD	1		
	***********		1					
!!	CHRONIC DATLY	1 001	1	!			CHENICAL	TOTAL TOTAL
CHENTCAL	THTAKE(COT)	ADTINSTED FOR	SE SE	LETCHT OF		I SE BASTS /	COFFICIENCE	DATILLAY FYDOCIDE
	(ma/ka/day)	ADCODOTION	; J (n= /k= /day _1	CUIDENCE				DICK I DICK
	1 (119/ K9/ 02 y)	1 RESURPTION	(E9/ k9/ 0ay)-1	EVIDENCE	LANUER	SUURLE	i KISK i	RISK RISK
								
EXPOSURE PATHWAY: INHALAT	ION OF AIRBORN	e chentcals a	dsorbed to dus	T				2E-05 4E-05
	1111111111111111		*****	innnn				
	1111111111111111	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	1111111111			1 1 I I I I I I I I I I I I I I I I I I	
INUKGANICS	1			į			i i	
Arsenic	1.34E-07	No	50	A I	Lung	Air/HEAST	7E-06	
3 2 6 5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	111111111111111111111111111111111111111	1 1 1 5 1 5 5 5 5 5 5 4 5 4 5 5 5 5 5 5	1	111111111111 11111111111				111414144
EXPOSURE PATHWAY: DERMAN	CONTACT WITH C	HENTCALS IN SE	OTI					2E-05 !
	1111111111111111	11111111111111111						
	*****	**********	* * * * * * * * * * * * * * * * * * * *	11111111111			**********	1111111
; INORGANICS	i	i	i	i			; ;	
Arcenic	0 015-07	· 0.10	1 75	1 1	chin	Untor /TOTC	1 25-05	
1 m Sente	, 0.01C-V/	, 0.10	1.75		SKLII	Mareilikis	1 2C-V3 I	

							SUM	MRY OF (HROM	HERTEL LANDE	OEX EST	inates - Hell Structy Stud	CONSTI	RUCTION (ALTE	RNATIV	e 4 and	4A)				
													inn		I, IIII						
;; ;; a	HENICAL	DAILY	NTAKE	ADJUSTE	FOR	RFD	lo	ONFIDEN	ε		CRITICA	L		SOURCE/	UNCE	RTAINTY	MODIFYING	HAZARD	HAZARD	EXPOS	JRE ¦
		(ag/kg	/day)	ABSORPT	TION ¦(ng/kg/ (day)¦	LEVEL	;;,,		EFFECT			BASIS	ADJU	STHENTS	FACTORS	QUOTIENT	INDEX (HI)	HI HI	1
11111111111 11 11			(111)																	111114	
; ;EXPOSURE	Pathiay: Inhalat	ION OF A	IRBORN	e chemica	nls ads	Sorbed 1	io dus	ат 1111111								111111			1E+01	; 1E+	-01
II IN	ORGANICS		(F . AF						-	Need			1	IFACT	1	200		1	1		
Manganese		2	62-05 5E-03	No		62. 4E-	-04	nea Medium	ł	Increase	d preva	acropny lence of		IRIS	i I	300 300	3	6E+00	1		
FYDOSIDE (DATHLAY: DEPMA	11111111 CONTACT	нни нги с	HFNTCALS	IN SOT								::::::						9 F- 01	1	
									111				nin.		ų nu			111111		:	
IN Arsenic	ORGANICS	9.	0E-05).10	1E-	-03 ¦	NA		Keratosis a	ind hyper	rpigmentation		NA/HEAST		1		9E-01			
					HHİ		ШĤ		HH		HUH		min								

		TABLE 5-10 HERTEL LANDFILL FEASIBILITY STUDY COMPARISON AMONG ALTERNATIVES SHORT-TERM EFFECTIVENESS
Alternative 1:	No action	No remedial activities conducted: therefore no short-term risks other than existing baseline risks result. Remedial response objectives not achieved
Alternative 2:	Site use restrictions. capping	Non-cancer risks to on-site workers exceed acceptable levels: personal protective equipment required Effective in limiting soil exposure risks within a short time frame: no protection against contaminated ground water migration
Alternative 2A:	Site use restrictions. capping and slurry wall construction	Non-cancer risks to on-site workers exceed acceptable levels: personal protective equipment required Effective in limiting soil exposure risks and continued contamination of ground water to a limited time frame
Alternative 4:	Site use restrictions, capping: Ground water extraction with on-site physical treatment	Non-cancer risks to pn-site workers exceed acceptable levels: personal protective equipment required Effective in limiting soil exposure risks and migration of contaminated ground water to a limited time frame
Alternative 4A:	Site use restrictions, capping: Ground water extraction with on-site innovative treatment using UV oxidation and membrane microfiltration	Won-cancer risks to on-site workers exceed acceptable levels: personal protective equipment required Effective in limiting soil exposure risks and migration of contaminated ground water to a limited time frame

`

		TABLE 5-11 HERTEL LANDFILL FEASIBILITY STUDY COMPARISON AMONG ALTERNATIVES LONG-TERM EFFECTIVENESS AND PERMANENCE
ternative 1:	No action	Baseline risks remain unchanged
ternative 2:	Site use restrictions. capping	Surface and subsurface soil/waste contaminants are untreated but contained: Long-term monitoring of containment area and ground water required: Potential long-term risks associated with contaminated ground water migration not addressed
ernative 2A:	Site use restrictions. capping and slurry wall construction	Surface and subsurface soil/waste contaminants are untreated but contained: Long-term contact between ground water and waste materials minimized or eliminated. thereby limiting potential for contaminated ground water migration: Long-term monitoring of containment area and ground water required
ernative 4:	Site use restrictions. capping: Ground water extraction with on-site physical treatment	Surface and subsurface soil/waste contaminants are untreated but contained: Contaminated ground water migration minimized through pumping and on-site treatment: Long-term monitoring of containment areas and ground water required: may not be effective in maintaining cleanup levels in ground water once pump and treat operations are discontinued
ernative 4A:	Site use restrictions. capping: Ground water extraction with on-site innovative treatment using UV oxidation and membrane microfiltration	Surface and subsurface soil/waste contaminants are untreated but contained: Contaminated ground water migration minimized through pumping and on-site treatment: Long-term monitoring of contairment areas and ground water required: may not be effective in maintaining cleanup levels in ground water once pump and treat operations are discontinued

	AVAILABILITY OF SERVICES AND MATERIALS	Suppliers of services and equipment readily available	Suppliers of services and equipment readily available	Suppliers of services and equipment readily available	Suppliers of services and equipment readily available	Suppliers of services and equipment for ground water treatment system limited
ILITY STUDY ERNATIVES TY	ADMINISTRATIVE FEASIBILITY	No inter-agency coordination required	Requires compliance with wetlands protection regulations	Requires compifance with wetlands protection regulations	Requires compliance with wetlands protection regulations and authorization to discharge treated ground water to surface water	Requires compliance with wetlands protection regulations and authorization to discharge treated ground water to surface water
TABLE 5-12 Hertel Landfill Feasib Comparison among alt Implementabili	TECHNICAL FEASIBILITY	Installation of additional monitoring weils is feasible	Construction easily implemented: Cap could inhibit future remedial activites in capped area or could enhance activities such as ground water extraction	Construction fairly easily implemented. although presence of boulders and slopes could complicate slurry wall construction: Cap could inhibit future remedial activites in capped area	Construction easily implemented: Ground water treatment technologies easily implemented: Cap could inhibit future remedial activites in capped area	Construction easily implemented: Technical problems possible with ground water treatment system: Cap could inhibit future remedial activities in capped area
	R- VE DESCRIPTION OF ALTERNATIVE	No action	Site use restrictions, capping	: Site use restrictions, capping and slurry wall construction	Site use restrictions. capping: Ground water extraction with on-site physical treatment	Site use restrictions, capping: Ground water extraction with on-site innovative treatment using UV oxidation and membrane microfiltration

	REDUCTION	TABLE 5-13 HERTEL LANDFILL FEASIBILITY STUDY Comparison among Alternatives of Toxicity (T). Mobility (M) or volume (V) Through treatment
Niternative 1:	No action	No reductions in T. M or V: Site conditions remain unchanged
Viternative 2:	Site use restrictions, capping	No reduction in T or V: M of ground water contamination is decreased through containment and prevention of leachate production: ground water table remains within waste materials
Viternative 2A:	Site use restrictions. capping and slurry wall construction	No reduction in T or V: M of ground water contamination is decreased through prevention of leachate production and minimization of contact between waste materials and ground water table
lternative 4:	Site use restrictions, capping: Ground water extraction with on-site physical treatment	T of ground water reduced through treatment: M of ground water reduced through pumping: No reduction of T or V of soil/waste: M of soil/waste is decreased through containment: Permanence of ground water T reductions not well-defined since ground water table resumes contact with wastes upon discontinuation of pump and treat system
Niternative 4A:	Site use restrictions. capping: Ground water extraction with on-site innovative treatment	T of ground water reduced through treatment: M of ground water reduced through pumping: No reduction of T or V of soil/waste: M of soil/waste is decreased through containment: Permanence of ground water T reductions not well-defined since ground water table resumes contact with wastes upon discontinuation of pump and treat system

		COMPLIANCE WITH ARARS		
NUMBE	R AND DESCRIPTION OF ALTERNATIVE	CHEMICAL -SPECIFIC	LOCATION - SPECIFIC	ACTION-SPECIFIC
:	lo action	ARARs not attained for sediment, ground water or surface water: No applicable soil ARARs	Not applicable	Not applicable
2: 5	ilte use restrictions, capping	ARARs not attained for sediment or ground water: Wo appiicable soil ARARs: Cap designed to prevent leachate seeps. thereby reducing surface water contaminant levels	Construction activities to comply with wetlands requirements	Cap design and surface water drainage design to comply with landfill closure requirements
2A: 5	ite use restrictions, capping and slurry wall construction	ARARs not attained for sediment or ground water although minimization of contact between the ground water table and soil/waste may result in reduced ground water contaminant levels; No applicable soil ARARs; Cap designed to prevent leachate seeps, thereby reducing surface water contaminant levels	Construction activities to comply with wetlands requirements	Cap design and surface water drainage design to comply with landfill closure requirements
4 	ite use restrictions, capping: round water extraction with on-site physical treatment	ARARs not attained for sediment: No applicable soil ARARs: Ground water treatment to attain ARARs: Maintenance of ARARs in ground water following discontinuation of pump and treat operations is not guaranteed: Cap designed to prevent leachate seeps. thereby reducing surface water contaminant levels	Construction activities to comply with wetlands requirements	Cap design. surface water drainage design and ground water discharge standards to comply with landfill closure requirements and NPDES requirements
4 A : 6	ite use restrictions, capping; round water extraction with on-site innovative treatment using UV oxidation and membrane microfiltration	ARARs not attained for sediment: No applicable soli ARARs; Ground water treatment expected to attain ARARs although innovative technologies not well proven; Maintenance of ARARs in ground water following discontinuation of pump and treat operations is not guaranteed: Cap designed to prevent leachate seeps, thereby	Construction activities to comply with wetlands requirements	Cap design, surface water drainage design and ground water discharge standards to comply with landfill closure requirements and NPDES requirements

.

reducing water contaminant levels

~•

TABLE 5-14 HERTEL LANDFILL FEASIBILITY STUDY COMPARISON AMONG ALTERNATIVES COMPLIANCE WITH ARARS

	ΝO	TABLE 5-15 Hertel Landfill Feasibility Study Comparison Among Alternatives Call Protection of Human Health and The Environment
Alternative 1:	No action	Baseline risks remain unchanged: Least protective alternative
Alternative 2:	Site use restrictions. capping	Provides soil/waste contaminant containment but risks associated with ground water contaminant migration not addressed: Effective in the short-term: Does not attain chemical-specific ARARs
Alternative 2A:	Site use restrictions. capping and slurry wall construction	Provides soil/waste contaminant containment: Long-term risks associated with ground water contaminant migration minimized through elimination of contact between the soil/waste and ground water table: Effective in the short-term: Does not attain chemical-specific ARARs
Niternative 4:	Site use restrictions, capping: Ground water extraction with on-site physical treatment	Provides soil/waste contaminant containment and treatment of contaminated ground water; Good short-term effectiveness: Long-term maintenance of ground water ARARs following discontinuation of pump and treat system may not be possible
ulternative 4A:	Site use restrictions. capping: Ground water extraction with on-site innovative treatment using UV oxidation and membrane microfiltration	Provides soil/waste contaminant containment and treatment of contaminated ground water; Short-term effectiveness dependent on effectiveness of innovative ground water treatment methods; Long-term maintenance of ground water ARARs following discontinuation of pump and treat system may not be possible

		HERTEL LANDFILL I COMPARISON AMON CO	FEASIBILITY STUDY NG ALTERNATIVES OST			
	Alternative	Duration	TOTAL CAPITAL COST	ANNUAL O&M COST	* Present Worth - O&M Cost	** TOTAL PRESENT WORTH
-	No action	30 yrs - G.W. mon. (1)	\$ 58,000 \$	132,000	\$ 2,033,000	\$ 2,509,000
7	Site use restrictions, capping	30 yrs - G.W. mon. (1) 30 yrs - cap maint.	3,482,000	163,000	2,503,000	7,182,000
2A	Site use restrictions, capping and slurry wall construction	30 yrs - G.W. mon. (1) 30 yrs - cap maint. 30 yrs - slurry wall maint.	8,406,000	171,000	2,626,000	13,238,000
4	Site use restrictions, capping; Ground water extraction with on-site physical treatment	17 yrs - G.W. mon. (1) 30 yrs - cap maint. 12 yrs - G.W. extr. & trtmt	3, 989, 000	316,000	3, 322, 000	8,774,000
4 A	Site use restrictions, capping; Ground water extraction with on-site innovative treatment using UV oxidation and membra microfiltration	17 yrs - G.W. mon. (1) 30 yrs - cap maint. 12 yrs - G.W. extr. & trtmt ne	3,955,000	267,000	2,884,000	8,207,000

* Based on 5% discount rate ** Includes 20% contingency on all components (1) Includes 4 rounds of samples per year

TABLE 5-17 COST SENSITIVITY ANALYSIS HERTEL LANDFILL FEASIBILITY STUDY

Item	Varied	Alternative	Total Pre	sent Worth
(Minimum	- Maximum)		Minimum Cost	Maximum Cost
Discount	Factor	1	\$ 1 565 000	¢ 3 180 000
(10% -	247	1	\$ 1,500,000 6 020 000	\$ 3,180,000
(10% -	32)	2	12 019 000	14 105 000
		4	7 385 000	9 026 000
		4A	6,866,000	8,237,000
Landfill	Cap Construction Cost	2	6,755,000	7,608,000
(-20%)	+20%)	2A	12,804,000	13,672,000
		4	8,340,000	9,208,000
		4A	7,773,000	8,641,000
Slurry Wa (-20% -	all Construction Cost - +20%)	2A	12.078.000	14,398,000
Ground Wa	ater Remediation Pumping			
Rates		4	8,188,000	9,176,000
(20 ga)	/min to 5 gal/min)	4A	7,424,000	8,913,000
































APPENDIX A

DETAILED COST ESTIMATES

THIS PAGE INTENTIONALLY LEFT BLANK

Item	Quantity U	ults I	Unit Price	Liternative 1: No Action Basis Year	Reference	Escalation	1991 Unit Costs	1991 Costs	Years (O&M)	(1) Present Value (0&M)
CAPITAL COSTS - DIRECT										
Monitoring Well Installation (4 60-ft. bedrock wells - 2° diam 2 30-ft. deep overburden wells - 2°)										
-Well Construction & Matls.	300 ft		\$125.00	1991	1	1.00	\$125.00	\$37.500.00		
-Health & Safety (17%) -Mobilization	1 tl	a	\$8,000.00	1991	89 -	1.00	\$8,000.00	\$6.375.00 \$8.000.00		
Total Monitoring Well Cost										\$51,875.00
Direct Capital Cost Subtotal										\$51.875.00
CAPITAL COSTS - INDIRECT										
Engineering and Design (10%) Legal and Administrative (2%)					~ ~			\$5,187.50 \$1,037.50		
Indirect Capital Cost Total										\$6.225.00
TOTAL CAPITAL COSTS										\$58,100.00

A-1.1

			•	Viternative 1						
				No Action						
				(cont1nued)						(1)
							1991	1991	Years	Present
Item	Quantity	Units	Unit Price	Basis Year	Reference	Escalation	Unit Costs	Costs	(08M)	Value (O&M)
OPERATION AND MAINTENANCE COSTS					6 9 4 6 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	, , , , , , , , , , , , , , , , , , ,				•
-Ground Water Monitoring										
Annual Sampling (15 Hells)	15	samples	\$200.00	1991	13	1.00	\$200.00	\$3,000.00	30	\$46.116.00
Quarterly Sampling (12 Wells)	48	samples	\$200.00	1991	13	1.00	\$200.00	\$9.600.00	30	\$147,571.20
Analysis:										
TCL	63	samples	\$1.800.00	1988	4	1.055	\$1,899.00	\$119,637.00	30	\$1,839,059.96
TOTAL NET PRESENT VALUE OF D & M								\$132,237.00		\$2,032,747.16
SUBTOTAL COST										\$2,090,847.16
CONTINGENCY (20%)										\$418,169.43
TOTAL PRESENT VALUE COST FOR ALTERNATIVE	1									\$2,509,016.60
(1) - Calculated based on an assumed 5% 1	Interest rat									

Å

A-1

			Alternative 2						
		Site Use Resi	trictions. Mu	lti-Layer Ca	٩				3
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	1991 Unit costs	1991 Costs	Years (OGM)	Present Value (0&M)
CAPITAL COSTS - DIRECT									
Monitoring Well Installation (4 60-ft. bedrock wells - 2° diam 2 30-ft. deep overburden wells - 2°)					1				
-Well Construction & Matls. (Tubex)	300 ft	\$125.00	1661	-	1.00	\$ 125.00	\$37,500.00		
-Health & Safety (17%) -Mobilization	1 time	\$8,000.00	1991	8 7	1.00	\$8.000.00	\$6.375.00 \$8.000.00		
Total Monitoring Well Cost									\$51.875.00
factorial to									
security -Perm. Chain Link Fence	6.250 linear ft	\$11.65	1661	ŝ	1.00	\$11.65	\$72,812.50		
-Warning Signs	20 signs	\$42.00	1991	2	1.00	\$42.00	\$840.00		
Total Security Cost									\$73,652.50
Site Preparation Mob/Demob.	1 time	\$590.00	1991	5	1.00	\$590.00	\$590.00		
Clearing	12.2 acres	\$3,675.00	1661	9	1.00	\$3,675.00	\$44 .835.00		
Grading	80.000 cu.yd.	\$3.53	1987	9	1.083	\$3.82	\$305,839.20		
Fill Material	80.000 cu.yd.	\$11.03	1661	2	1.00	\$11.03	\$882,400.00		
Access Road Reconstruction	7.200 sq.ft.	\$15.20	1991	ŝ	1.00	\$15.20	\$109,440.00		
Total Site Preparation									\$1.343.104.20
 Calculated based on an assumed 5% 1; 	nterest rate.								

A-2.1

		Cita Ilea Bae	Alternative 2: Prictions Mul	: +1_ avar Ca					
			(continued)	I LI - LEJEI LE	1				(1)
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	1991 Unit costs	1991 Costs	Years (O&M)	Present Value (O&M)
Run-on/Run-off Controls -Ditching -Sedimentation Basin	2100 1.ft. 1 each	\$1.70 \$10.000.00	1988 1988	7	1.055	\$1.79 \$10.550.00	\$3,766.35 \$10,550.00		
Total Run-On/Run-Off Controls									\$14.316.35
Multi-Layer Cap Construction			Ĭ	5	-	55	431E 000 00		
-12 ⁻ Gas Yent Layer -40-m1] HOPE Liner	20.000 сц.уд. 530.000 сq.ft.	62.714 80.80	1661 1661	51 EI	1.00	08.08	\$424,000.00		
-Filter Fabric (2 layers)	1.060.000 sq.ft.	\$0.17	1991	13	1.00	\$0.17	\$180.200.00		
-24" Barrier Protection Layer	40.000 cu.yd.	\$2.50	1988	1	1.055	\$2.64	\$105.500.00		
-6" Topsoil Layer	530 msf	\$400.00	1991	S	1.00	\$400.00	\$212.000.00		
-Seed. Fertilizer. Mulch	530 msf	\$43.00	1991	ŝ	1.00	\$43.00	\$22.790.00		
-Vertical Gas Vent Pipes	15 each	\$500.00	1988	2	1.055	\$527.50	\$7.912.50		
-Lateral Gas Vent Pipe	6,500 ft.	\$6.00	1988	7	1.055	\$6.33	\$41.145.00		
-Health and Safety(17%)				8			\$220,558.43		
Total Cap Construction Costs									\$1.559.105.93
Equipment Decontamination									
-Rental of steam cleaner	4 months	\$390.00	1991	5	1.00	\$390.00	\$1,560.00		
-Construct Decon Pit									
Excevate Pit	100 cu.yd.	\$2.24	1987	5	1.083	\$2.43	\$242.59		
Polyethylene Tarpaulin	1200 sq.ft.	\$0.31	1987	2	1.083	\$0.34	\$402.88		
-Tanker rental	1 each	\$800.00	1989	0	1.036	\$828.80	\$829.90		
- D1 spose1	l each	\$1,100.00	1989	6	1.036	\$1,139.60	\$1,139.60		
Total Equipment Decon Costs									\$4.173.87
	:				:				
Engineering Mgmt. Mob/Demob (1 Trailer)	6 months	\$ 430.00	1991	2	1.00	54 30-00	5 2,580.00		52,580.00
(1) - Calculated based on an assumed 5\$	i interest rate.								

. 111

A

	•	A Site Use Rest	lternative 2: rictions. Mul	ti-Layer Cap				-	
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	Unit costs	Costs	(MBO)	Value (O&M)
Dust Control - Water Tank Sprayer	800 hours	\$ 7.10	1991	مر	1.00	\$7.10	\$5,680.00		\$5,680.00
Direct Capital Cost Subtotal									\$3,054,487.84
CAPITAL COSTS - INDIRECT									
Engineering and Design(11%) Legal and Administrative(3%)				2 2					\$335,993.66 \$91,634.64
TOTAL CAPITAL COSTS									\$3,482,116.14
OPERATION AND MAINTENANCE COSTS									
-Ground Water Monitoring				:				8	
Annual Sampling (12 Weils) Quarterly Sampling (12 Weils)	15 samples 48 samples	\$200.00 \$200.00	1661	13	1.00	\$200.00	\$9,600.00	2 R	\$46.116.00 \$147.571.20
Analysis TCL	63 samples	\$1,800.00	1988	4	1.055	\$1,899.00	\$119,637.00	30	\$1,839,059.96
-Cap Maintenance	-	** 000 00	0001		1.065	4E 97E 00	46 976 DO	ç	01 100 104
Mowing/Revertation	1 EBCH 530.000 \$0.ft.	\$0.04	1991	0	1.00	\$0.04	\$21.200.00	2 P	\$325.886.40
Erosion Control	12.2 acres	\$200.00	1982	12	1.247	\$249.40	\$3.042.68	30	\$46.772.08
Repairs(total for 1 year)	1 each	\$1.000.00	1988	1	1.055	\$1,055.00	\$1,055.00	30	\$16.217.46
TOTAL NET PRESENT VALUE OF D & M				×			\$162,809.68		\$2.502.710.40

A-2.3

Inflag Inflag 191 191 Team 101 Unflag Unflag Inflag 191 191 Team Team Unflag Unflag Inflag Inflag Team Team Team Unflag Unflag Inflag Inflag Team Team Team Unflag Inflag Inflag Inflag Team Team Team Inflag I	Initial 191 191 191 191 191 191 101 1	Site	A te Use Resti	lternative 2: rictions. Mul	: t1-Layer Ca	ē				ŧ
Units Unit Price Basis year Reference Exalision Value (0M) Value (0M) 13:106 96:107 13:106 96:107 13:106 96:101	Units Unit Price Basis year Reference E calatition Unit costs Code Value (0M) 1:96:56:3:1 1:96:75:1 1:96:75:1 1:96:45:1 1:96:56:3:1 1:96:56:3:1 1:96:56:3:1 1:96:56:3:1 1:96:56:3:1 1:96:56:3:1 1:96:56:3:1 1:96:56:3:1 1:96:56:3:1 1:96:56:3:1 1:96:56:3:1 1:96:56:3:1 1:96:56:3:1 1:91:79:16			(כסחנו חשפט)			1991	1991	Years	(1) Present
12:366.362.31 13:196.362.31 14:191.791.85	97 92 1967 19 17 1997 96 17 16 17 19 19 10 10 10 10 10 10 10 10 10 10 10 10 10	Units Unit	it Price	Basis year	Reference	Escalation	Unit costs	Costs	(MBO)	Value (O&M)
	13.181.791.85									\$5.984.826.54 \$1.196.965.31
										\$7,181,791.85
		ite.								

	Site	: Use Restricti	Alternative 2 ons. Multi-La	A: yer Cap. Slu	lla H all				
ltem	Quantity Units	Unit Price	Basis year	Reference	Escalation	1991 Unit costs	1991 Costs	Years (OLM)	(1) Present Value (O&M)
APITAL COSTS - DIRECT									
<pre>40nitoring Well Installation (4 60-ft. bedrock wells - 2" diam 2 30-ft. deep overburden wells - 2 ") </pre>	:		Ĩ		1				
-Meil Construction & Matis. (Tubex)	300 ft	00.4218	1661	-	1.00	00.6218	5 3/,500.00		
-Health & Safety (17%) -Mobilization	1 time	\$8,000.00	1661	8 1	1.00	\$8.000.00	\$6,375.00 \$8,000.00		
fotal Monitoring Well Cost									\$51,875.00
ecurity									
-Perm. Chain Link Fence	6.250 linear ft	\$11.65	1661	5	1.00	\$11.65	\$72.812.50		
-Warning Signs	20 signs	\$42.00	1661	LC	1.00	\$42.00	\$840.00		
otal Security Cost									\$73,652.50
ite Preparation									
Mob/Demob.	1 time	\$590.00	1991	ŝ	1.00	\$590.00	\$590.00		
Clearing	12.2 acres	\$3.675.00	1661	5	1.00	\$3.675.00	\$44.835.00		
Grading	B0.000 cu.yd.	\$3.53	1987	9	1.083	\$3.82	\$305,839.20		
Fill Material	B0.000 cu.yd.	\$11.03	1661	ŝ	1.00	\$11.03	\$882.400.00		
Access Road Reconstruction	7.200 sq.ft.	\$15.20	1661	ŝ	1.00	\$15.20	\$109.440.00		
otal Site Preparation									61.343.104.20
 Calculated based on an assumed 5% in 	nterest rate.								

A-2A.1

	+5	A A llea Daetrictio	<pre>//ternative 2/ //ternative 2/ /</pre>	l: ∆er Can Slu	[[e] vaa				
	2		(continued)						Ð
						1991	1991	Years	Present
ten	Quantity Units	Unit Price	Basis year	Reference	Escalation	Unit costs	Costs	(0&M)	Value (O&M)
un-on/Run-off Controls -Ditching	2100 1 ft	1 ZD	1988	7	1_055	\$ 1,70	43 766 36		
-Sedimentation Basin	1 each	\$10,000.00	1988		1.055	\$10.550.00	\$10.550.00		
otal Run-On/Run-Off Controls									\$14.316.35
ultf-Layer Cap Construction									
-12" Gas Vent Layer	20.000 cu.yd.	\$17.25	1991	13	1.00	\$17.25	\$345,000.00		
-40-m11 HDPE Liner	530,000 sq.ft.	\$0.80	1661	13	1.00	\$0.80	\$424,000.00		
-Filter Fabric (2 layers)	1.060.000 sq.ft.	\$0.17	1661	13	1.00	\$0.17	\$180.200.00		
-24" Barrier Protection Layer	40.000 cu.yd.	\$2.50	1988	7	1.055	\$2.64	\$105,500.00		
-6" Topsoil Layer	530 msf	400	1991	5	1.00	\$400.00	\$212,000.00		
-Seed. Fertilizer, Mulch	530 msf	43	1991	5	1.00	\$43.00	\$22.790.00		
-Vertical Gas Vent Pipes	15 each	500	1988	7	1.055	\$527.50	\$7.912.50		
-Lateral Gas Vent Pipe	6.500 ft.	\$6.00	1988	7	1.055	\$6.33	\$41,145.00		
-Health and Safety(17%)				8			\$220.558.43		
otal Cap Construction Costs									\$1.559.105.93
lurry Wall Construction									
(1800 ft x 3 ft x 40 ft)									
- Mob11./Demob.	1 time	\$590.00	1991	5	1.00	\$590.00	\$590.00		
-Backhoe (2 1/2 cu.yd. bucket)	B.000 cu.yd.	\$2.69	1991	5	1.00	\$2.69	\$21.520.00		
-Bulldozer (300 H.P.)	B.000 cu.yd.	\$2.82	1661	ŝ	1.00	\$2.82	\$22,560.00		
-Cement-Bentonite Trench	72.000 sq.ft.	\$30.00	1979	14	1.651	\$49.53	\$3.566.160.00		
-Health and Safety (14%)				8			\$505,433.60		
-Water tank rental	2 months	\$100.00	1991	5	1.00	\$100.00	\$200.00		
-Temporary road construction	B.B00 sq.ft .	\$3.29	1991	5	1.00	\$3.29	\$28.952.00		
 Pumping/Mixing equipment 	2 months	\$8.745.00	1991	ŝ	1.00	\$8,745.00	\$17.490.00		
- Survey	12.2 acres	\$225.00	1991	5	1.00	\$225.00	\$2,745.00		
otal Slurry Wall Construction Costs									\$4 ,165,650.60
1) - Calculated based on an assumed 51	f interest rate.								

A=2¤.2

		-	Alternative 2	ï					
	Site	e Use Restricti	ons. Multi-La (continued)	yer Cap. Slu	rry Wall				8
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	1991 Unit costs	1991 Costs	Years (OBM)	Present Value (O&M)
Observation Well Installation (8 30-ft. deep overburden wells - 2 ⁻) -Well Contruction & Matls. (Tubex) -Health & Safety (17%)	240 ft	\$114.00	1991	- 0	1.00	\$114.00	\$27,360.00 \$4,651.20		
Total Observation Well Cost									\$32,011.20
Equipment Decontamination -Rental of steam cleaner	4 months	00.028	1991	م	1.00	00.005	\$1,560.00		
-Construct Decon Pit	WI	12 21	1007	u	1 002	57 F3	4242 ED		
Polyethylene Tarpaulin	1200 sq.ft.	\$0.31	1987) ц а	1.083	\$0.34	\$402.88		
-Tanker rental	1 each	\$800.00	1989	6	1.036	\$828.80	\$828.80		
-Disposal	1 each	\$1,100.00	1989	9	1.036	\$1,139.60	\$1,139.60		
Total Equipment Decon Costs									\$4.173.87
Engineering Mgmt. Mob/Demob (1 Trailer)	6 months	\$430.00	1991	S.	1.00	\$430.00	\$2,580.00		\$2,580.00
Dust Control - Water Tank Sprayer	800 hours	\$7.10	1991	5	1.00	\$7.10	\$5.680.00		\$5,680.00
Direct Capital Cost Subtotal								\$	7.246.469.64
 Calculated based on an assumed 5% fi 	nterest rate.								

A-2A.3

	Site	e Use Restrictic	Alternative 2A ons. Multi-Lay (continued)	ı: 'er Cap. Slur	ry Wall				8
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	1991 Unit costs	1991 Costs	Years (O&M)	Present Value (O&M)
CAPITAL COSTS - INDIRECT									
Engineering and Design(131)				~ ~					\$942,041.05
LEVEL AUTITION AUTITION AUTOL									60.96C,1126 60 Ant DA
DPERATION AND MAINTENANCE COSTS									n
-Ground Water Monitoring Annual Sampling (15 Wells)	15 samples	\$200.00	1991	13	1.00	\$200.00	53,000.00	30	\$46.116.00
Quarterly Sampling (12 Wells)	48 samples	\$200.00	1991	13	1.00	\$200.00	\$9,600.00	30	\$147.571.20
Analysis TCL	63 samples	\$1,800.00	1988	4	1.055	\$1,899.00	\$119,637.00	30	\$1,839,059.96
-Cap Maintenance									
Annual Inspection	1 each	\$5,000.00	1988	2	1.055	\$5,275.00	\$5.275.00	30	\$81,087.30
Mowing/Revegetation Erosion Control	530.000 sq.ft. 12.2 acres	\$0.04 \$200.00	1991 1982	12	1.247	\$0.04 \$249.40	\$21.200.00 \$3.042.68	30	\$325.886.40 \$46.772_08
Repairs(total for 1 year)	1 each	\$1,000.00	1988	1	1.055	\$1.055.00	\$1.055.00	30	\$16.217.46
-Annual Slurry Wall Maintenance/ Monitoring	B0 hours	\$100.00	1991	13	1.00	\$100.00	\$8,000.00	30	\$122.976.00
TOTAL NET PRESENT VALUE OF O & M							\$170.809.68		\$2.625.686.40
SUBTOTAL Sontingency(20%)									\$11.031.591.19 \$2.206.318.24
TOTAL PRESENT VALUE COST FOR ALTERNATIVE	ZA								\$13,237,909.42
(1) - Calculated based on an assumed 5%	interest rate.								
	-	-	-	. .4			-		

	Ground Water Extra	Site Use Res ction. On-Site	Alternative 4 trictions. Mu Physical Trea	: 1t1-Layer Ca tment and D1	p. scharge to Su	Irface Mater			
					ı	1991	1991	Years	(1) Present
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	Unit costs	Costs	(D&M)	Value (O&M)
CAPITAL COSTS - DIRECT									
Monitoring Well Installation									
(4 60-ft. bedrock wells - 2 ⁻ diam.									
z 3U-TT. deep overburgen weits - Z 7 -Well Construction & Matls.) 300 ft	\$125.00	1991	1	1.00	\$125.00	\$37,500.00		
(Tubex)									
-Health & Safety (17%)				8			\$6,375.00		
-Mobilization	1 time	\$8.000.00	1991	1	1.00	\$8,000.00	\$8,000.00		
Total Monitoring Well Cost									\$51,875.00
Security		10 114			-	911 CE	470 010 ED		
-Perm. Chain Link Fence	6,250 linear ft	3 11.65	1991	. .	1.00	GG.118	09.218.2/8		
-Marning Signs	20 signs	54 2.00	1991	un.	1.00	54 2.00	584 0.00		
Total Security Cost									\$73.652.50
Site Preparation									
Mob/Demob.	1 time	\$590.00	1991	S	1.00	\$590.00	\$590.00		
Clearing	12.2 acres	\$3.675.00	1991	2	1.00	\$3.675.00	\$44,835.00		
Grading	B0.000 cu.yd.	\$3.53	1987	9	1.083	\$3.82	\$305,839.20		
Fill Material	B0.000 cu.yd .	\$11.03	1991	5	1.00	\$11.03	\$882.400.00		
Access Road Reconstruction	7200 sq.ft.	\$15.20	1991	ŝ	1.00	\$15.20	\$109.440.00		
Total Site Preparation									1.343.104.20
(1) - Calculated based on an assumed 52 1	Interest rate.								

A-4.1

		-	Alternative 4						
		Site Use Resi	trictions, Mu	lti-Layer Ca					
	Ground Water Extra	action, On-Site F	Physical Trea (continued)	tment and D1:	scharge to Su	irface Water			(1)
						1991	1991	Years	Present
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	Unit costs	Costs	(MBO)	Yalue (O&M)
Run-on/Run-off Controls		¢, 11	900	·	- DEF	5			4 4 4 1 1 1 1 1 1 1 1 1 1 1 1
-Sedimentation Basin	1 each	\$10.000.00	1988		1.055	\$10,550.00	\$10,550.00		
Total Run-On/Run-Off Controls									\$14,316.35
Multi-Layer Cap Construction									
-12" Gas Vent Layer	20.000 cu.yd.	\$17.25	1991	13	1.00	\$17.25	\$345,000.00		
-40-mil HOPE Liner	530,000 sq.ft.	\$0.80	1991	13	1.00	\$0.80	\$424,000.00		
-Filter Fabric (2 layers)	1.060.000 sq.ft.	\$0.17	1991	13	1.00	\$0.17	\$180.200.00		
-6" Topsoil Layer	530 msf	\$400.00	1991	5	1.00	\$400.00	\$212,000.00		
-24" Barrier Protection Layer	40.000 cu.yd.	\$2.50	1988	7	1.055	\$2.64	\$105,500.00		
-Seed. Fertilizer. Mulch	530 msf	\$43.00	1991	2	1.00	\$43.00	\$22.790.00		
-Vertical Gas Vent Pipes	15 each	\$500.00	1988	7	1.055	\$527.50	\$7.912.50		
-Lateral Gas Vent Pipe	6.500 ft.	\$6.00	1988	7	1.055	\$6.33	\$41,145.00		
-Health and Safety(17%)				8			\$220,558.43		
Total Cap Construction Costs									\$1.559,105.93
Ground Water Extraction									
(22 30-ft. deep overburden wells - 6")									
-Well Construction and Materials	660 ft	\$124.00	1991	1	1.00	\$124.00	\$81.840.00		
(Tubex)									
-Health and Safety(17%)				8			\$13.912.80		
-Ejector Pumps	22 pumps	\$4.264.00	1991	24	1.00	\$4.264.00	\$93,808.00		
Total Extraction Cost									\$189,560.80
Piping To and From Treatment System									
-(2" diam. PVC in Trench)	2450 ft	\$5.81	1991	5	1.00	\$5.81	\$14.234.50		\$14.234.50
 Calculated based on an assumed 5% ini 	terest rate.								
	-			14 C			-		
						Û	Ű	1	Ĩ

			A] ternative 4						
		Site Use Res	trictions. Mu	lti-Layer Ca	.	:			
	Ground Water Extra	iction. On-Site	Physical Trea (continued)	tment and Oi	scharge to S	urface Water			Ð
						1991	1991	Years	Present
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	Unit costs	Costs	(NRI)	Value (O&M)
Ground Water Treatment Svetem									
-Neutralization/Precipitation/	1 each	\$126.520.00	1984	16	1.15	\$145.498.00	\$145.498.00		
-Piping	500 1 Ft	\$2 ED	1 988	7	1 055	11 A	¢1 371 60		
-Equalization Tank	1 each	\$12.500.00	1988		1.055	\$13.187.50	\$13.187.50		
-Filter Press	1 each	\$20,800.00	1985	16	1.138	\$23,670.40	\$23.670.40		
Total Ground Water Treatment System Cost	1								\$183,727.40
Eurliment Decontaction									
-Rental of steam riestor	A monthe	00 00.÷	1001	4		4300 00	¢1 560 00		
-Construct Decon Pit			1661	•		nn. nere	00.000.10		
Excavate Pit	100 cu.yd.	\$27.00	1991	4	1.00	\$27.00	\$2.700.00		
Polyethylene Tarpaulin	1200 sq.ft.	\$0.31	1991	5	1.00	\$0.31	\$372.00		
-Tanker rental	1 each	\$800.00	1989	6	1.036	\$828.80	\$828.80		
- Disposal	1 each	\$1.100.00	1989	0	1.036	\$1.139.60	\$1,139.60		
Total Equipment Decon Costs									\$6,600.40
engineering Mgmt. Mob/Demob (1 Trailer)	6 months	\$4 30.00	1991	L.	1.00	\$430.00	\$2.580.00		\$2,580.00
vast vontrol - Water Tank Sprayer	800 hours	\$7.10	1991	ۍ	1.00	\$7.10	\$5,680.00		\$5,680.00
 Calculated based on an assumed 5% 	interest rate.								

A-4.3

			Alternative 4						
		Site Use Res	trictions, Mu	lti-Layer Cap					
	Ground Water Extra	ction, On-Site	Physical Trea	tment and Dis	charge to Su	irface Water			
			(continued)						(1)
						1991	1991	Years	Present
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	Unit costs	Costs	(M30)	Value (O&M)
	· · · · · · · · · · · · · · · · · · ·			, , , , , , , , , , , , , , , , , , ,					
Direct Capital Cost Subtotal									\$3,438,757.08
CAPITAL COSTS - INDIRECT									
Engineering and Design(13%)				2					\$447,038.42
Legal and Administrative(3%)				2					\$103,162.71
TOTAL CAPITAL COSTS									\$3,988,958.21
OPERATION AND MAINTENANCE COSTS									
-Ground Water Monitoring									
Annual Sampiing (15 Wells)	15 samples	\$200.00	1991	13	1.00	\$200.00	\$3,000.00	17	\$33,822.00
Quarterly Sampling (12 Wells)	48 samples	\$200.00	1991	13	1.00	\$200.00	\$9,600.00	17	\$108.230.40
TCL Analysis	63 samples	\$1,800.00	1988	4	1.055	\$1,899.00	\$119.637.00	17	\$1,348,787.54
-Cap Maintenance									
Annual Inspection	1 each	\$5,000.00	1988	7	1.055	\$5,275.00	\$5,275.00	30	\$81.087.30
Mowing/Revegetation	530.000 sq.ft.	\$0.04	1991	5	1.00	\$0.04	\$21.200.00	30	\$325,886.40
Erosion Control	12.2 acres	\$200.00	1982	12	1.247	\$249.40	\$3.042.68	30	\$46,772.08
Repairs(total for 1 year)	1 each	\$1,000.00	1988	7	1.055	\$1,055.00	\$1.055.00	30	\$16.217.46

ر انتقار A-4

أتعد

		Site Use Rest	Alternative 4 trictions. Mu	: lt1-Layer Ca					
	Ground Water Extrac	tion. On-Site	Physical Trea (continued)	tment and D1	scharge to Si	urface Water			(1)
						1991	1991	Years	Present
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	Unit costs	Costs	(MBO)	Value (O&M)
OPERATION AND MAINTENANCE COSTS (contin	nued)								
-Precipitation O&M	l year	\$26,500.00	1984	16	1.15	\$30,475.00	\$30,475.00	12	\$270,099.92
-Chemical Costs(per gallon treated)	5.256.000 gal	\$0.01266	1985	16	1.138	\$0.01441	\$75,723.61	12	\$671,138.38
-Carbon Treatment O&M									
Carbon Supply	14 canisters	\$450.00	1991	15	1.00	\$450.00	\$6.300.00	12	\$55,836.90
0el1very	4 each	\$200.00	1991	15	1.00	\$200.00	\$800.00	12	\$7.090.40
Spent Carbon Disposal	14 drums	\$400.00	1991	15	1.00	\$400.00	\$5.600.00	12	\$49,632.80
-Precipitate Transportation &									
D1sposal	4 tons	\$1.300.00	1991	22	1.00	\$1,300.00	\$5.200.00	12	\$46.087.60
-Discharge to Surface Water Sampling									
And Analysis	24 samples	\$850.00	1661	13	1.00	\$850.00	\$20.400.00	12	\$180.805.20
-Filter Press O&M	l year	\$7,030.00	1985	16	1.138	\$8,000.14	\$8,000.14	12	\$70.905.24
-Stabilization	4 tons	\$250.00	1985	16	1.138	\$284.50	\$1,138.00	12	\$10.086.09
TOTAL NET PRESENT VALUE OF O & M							\$316,446.43		\$3,322,485.71
SUBTOTAL Contingency(20%)									\$7,311,443.92 \$1,462,288.78

TOTAL PRESENT VALUE COST FOR ALTERNATIVE 4

(1) - Calculated based on an assumed 5% interest rate.

\$8.773.732.70

A-4.5

THIS PAGE INTENTIONALLY LEFT BLANK

.

	Ground Mater Extract	Site Use Resi tion, On-Site I	Alternative 4. trictions, Mui nnovative Tree	A: t -Layer Ca atment and D	p. ischarge to S	urface Mater			
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	1991 Unit costs	1991 Costs	Years (OGM)	(1) Present Value (O&M)
CAPITAL COSTS - DIRECT									
Monitoring Well Installation (4 60-ft. bedrock wells - 2" diam 2 30-ft. deep overburden wells - 2 ")									
-Well Construction & Matls.	300 ft	\$125.00	1991	1	1.00	\$125.00	\$37,500.00		
() Health & Safety (17%) -Mobilization	1 time	\$8 ,000.00	1991	8 1	1.00	\$8,000.00	\$6.375.00 \$8.000.00		
Total Monitoring Well Cost									\$51.875.00
Security									
-Perm. Chain Link Fence	6.250 linear ft	\$11.65	1991	2	1.00	\$11.65	\$72,812.50		
-Warning Signs	20 signs	\$4 2.00	1661	ŝ	1.00	\$42.00	\$840.00		
Total Security Cost									\$73.652.50
Site Preparation									
Clearing	12.2 acres	\$3.675.00	1991	5	1.00	\$3.675.00	\$44.835.00		
Grading	B0.000 cu.yd.	\$3.53	1987	9	1.083	\$3.82	\$305,839.20		
Fill Material	B0,000 cu.yd.	\$11.03	1991	5	1.00	\$11.03	\$882.400.00		
Access Road Reconstruction	7,200 sq.ft.	\$15.20	1661	5	1.00	\$15.20	\$109,440.00		
Total Site Preparation									\$1.342.514.20
(1) - Calculated based on an assumed 5% 1	interest rate.								

A-4A.1

		A Sita llea Daet	lternative 4/	1: +1-1 aver Ca					
	Ground Water Extract	tion. On-Site In	novative Trea	itment and 0	ischarge to	surface Water			:
			(continued)			1991	1991	Years	(1) Present
ltem	Quantity Units	Unit Price	Basis year	Reference	Escalation	Unit costs	Costs	(NBO)	Value (O&M)
Run-on/Run-off Controls	6 7 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8								
-Ditching	2100 l.ft.	\$1.70	1988	7	1.055	\$1.79	\$3,766.35		
-Sedimentation Basin	1 each	\$10,000.00	1988	1	1.055	\$10,550.00	\$10,550.00		
Total Run-On/Run-Off Controls									\$14,316.35
Little and fan fanteinieten									
		417 9E	1001	13	1	¢17 96	4345 NON NO		
-16 das veil Layer -10-mil LADE iinan	Ean nnn en ft	40 BU	1001	3 5	1.00	¢0 BU	424 NON NO		
-Filter Fabric (2 layers)	1.060.000 sq.ft.	\$0.17	1991	: 1	1.00	\$0.17	\$180.200.00		
-24" Barrier Protection Layer	40.000 cu.yd.	\$2.50	1988	7	1.055	\$2.64	\$105,500.00		
-6" Topsoil Layer	530 msf	\$400.00	1991	ŝ	1.00	\$400.00	\$212,000.00		
-Seed. Fertilizer. Mulch	530 msf	\$43.00	1991	S	1.00	\$4 3.00	\$22,790.00		
-Vertical Gas Vent Pipes	15 each	\$500.00	1988	7	1.055	\$527.50	\$7.912.50		
-Lateral Gas Vent Pipe	6.500 ft	\$6.00	1988	7	1.055	\$6.33	\$41.145.00		
-Health and Safety(17%)				89			\$220.558.43		
Total Cap Construction Costs									\$1.559,105.93
Ground Mater Extraction									
(22 30-ft. deep overburden wells - 6")									
-Well Construction and Materials	660 ft	\$124.00	1991	1	1.00	\$124.00	\$81,840.00		
(Tubex)									
-Health and Safety(17%)				8			\$13,912.80		
- Ejector Pumps	22 pumps	\$4,264.00	1991	24	1.00	\$4.264.00	\$93,808.00		
Total Extraction Cost									\$189,560.80
Piping To and From Treatment System -(2" diam. PVC in Trench)	2450 ft	\$5.81	1991	5	1.00	\$5.81	\$14.234.50		\$14,234.50
 Calculated based on an assumed 5% 1n¹ 	terest rate.								
				·		W			

			Alternative 4,	ž					
		Site Use Res	trictions. Mu	lt1-Layer Ca	<u>6</u>				
	Ground Water Extrac	tion. On-Site I	nnovative Tre	atment and D	Mischarge to 9	Surface Water			Ę
			(continued)			1991	1991	Years	(I) Present
Item	Quantity Units	Unit Price	Basis year	Reference	Escalation	Unit costs	Costs	(NBO)	Value (O&M)
Ground Water Treatment System							•		
-Membrane Microfiltration Unit	1 each	\$50,000.00	1991	25	1.00	\$50,000.00	\$50,000.00		
-Filter Aid System	1 each	\$20,000.00	1991	25	1.00	\$20,000.00	\$20,000.00		
-UV Oxidation Unit	1 each	\$59,950.00	1991	18	1.00	\$59,950.00	\$59,950.00		
-UV Oxidation Service Connection	1 time	\$5,000.00	1991	18	1.00	\$5,000.00	\$5,000.00		
-Piping	500 l.ft.	\$2.60	1988	7	1.055	\$2.74	\$1.371.50		
-Equalization Tank	1 each	\$12,500.00	1988	7	1.055	\$13.187.50	\$13.187.50		
Total Ground Water Treatment System Cost	2								\$149.509.00
Fuultment Decontamination									
	:								
-Rental of steam cleaner -fourtmint Daron Dit	4 months	\$390.00	1991	LO LO	1.00	\$390.00	\$1,560.00		
FURCH STORE TIC		00 101	1001			10 104	00 00L L4		
EXCAVALE FIL	IUU CU.Ya.	00.128	1661	n	1.UU	00.12\$	\$2./00.00		
Polyethylene Tarpaulin	1200 sq.ft.	\$0.31	1991	5	1.00	\$0.31	\$372.00		
-Tanker rental	1 each	\$800.00	1989	6	1.036	\$828.80	\$828.80		
- Disposal	1 each	\$1,100.00	1989	0	1.036	\$1,139.60	\$1.139.60		
fotal Equipment Decon Costs									\$6,600.40
engineering Mgmt. Mob/Demob (1 Trailer)	6 months	\$430.00	1991	Ð	1.00	\$430.00	\$2.580.00		\$2,580.00
Just Control		:		1	:				
- Water Tank Sprayer	800 hours	\$7.10	1991	n I	1.00	\$7.10	\$5,680.00		\$5.680.00
Latetting trad Latetting									03 063 004 64
Virect Lapital Lost Subtoral									89.879,604.64
The second second second second second second second second second second second second second second second se									

A-4A.3

,

			Alternative 4/	A:					
		Site Use Res	trictions, Mu	lti-Layer Ca	ь.				
	Ground Water Extract	tion. On-Site I	nnovative Tre	atment and D	ischarge to	Surface Water			
			(continued)						(1)
						1991	1991	Years	Present
tem	Quantity Units	Unit Price	Basis year	Reference	Escalation	Unit costs	Costs	(NBO)	Value (O&M)
APITAL COSTS - INDIRECT									
ingineering and Design(13%)				2					\$443.251.73
egal and Administrative(3%)				2					\$102,288.86
OTAL CAPITAL COSTS									\$3,955,169.26
PERATION AND MAINTENANCE COSTS									
-Ground Water Monitoring Annual Sampling (15 Wells)	15 samples	\$200.00	1991	13	1.00	\$200.00	\$3,000.00	17	\$33,822.00
Quarterly Sampling (12 Wells)	48 samples	\$200.00	1991	13	1.00	\$200.00	\$9,600.00	17	\$108,230.40
TCL Analysis	63 samples	\$1,800.00	1988	4	1.055	\$1.899.00	\$119,637.00	17	\$1,348,787.54
-Cap Maintenance									
Annual Inspection	1 each	\$5,000.00	1988	1	1.055	\$5,275.00	\$5.275.00	30	\$81,087.30
Mowing/Revegetation	530.000 sq.ft.	\$0.04	1991	5	1.00	\$0.04	\$21.200.00	30	\$325,886.40
Erosion Control	12.2 acres	\$200.00	1982	12	1.247	\$249.40	\$3,042.68	30	\$46,772.08
Repairs(total for 1 year)	1 each	\$1,000.00	1988	1	1.055	\$1,055.00	\$1,055.00	30	\$16,217.46
-Membrane Microfiltration O&M	5.256 1000 gal	\$4.00	1991	26	1.00	\$4.00	\$21,024.00	12	\$186,335.71
-UV Oxidation O&M Cost	12 months	\$4,796.00	1991	18	1.00	\$4,796.00	\$57,552.00	12	\$510,083.38
-Filter Cake Transportation &									
Disposal	4 tons	\$1,300.00	1991	22	1.00	\$1,300.00	\$5,200.00	12	\$46.087.60
-Discharge to Surface Water Sampling									
And Analysis	24 samples	\$850.00	1991	13	1.00	\$850.00	\$20.400.00	12	\$180.805.20
OTAL NET PRESENT VALUE OF O & M							\$266,985.68		\$2,884.115.06
 Calculated based on an assumed 5% in 	iterest rate.								

<u>iii</u>

У-Ч

				Alternative 4A						
			Site Use Rest	trictions. Muì	t1-Layer Ca	Å.				
	Ground Wat	er Extracti	on. On-Site Ir	nnovative Trea	tment and D	ischarge to Si	urface Water			
				(continued)						(1)
							1991	1991	Years	Present
Item	Quant1ty	Units	Unit Price	Basis year	Reference	Escalation	Unit costs	Costs	(MBO)	Value (O&M)
SUBTOTAL										66, 839, 284. 33
CONTINGENCY(20%)										\$1.367.856.87
TOTAL PRESENT VALUE COST FOR ALTERNATIVE 4	4									8,207,141.19

HERTEL LANDFILL REFERENCES FOR COST ASSUMPTIONS

- 1. Empire Soils Investigation Inc.; February 1991.
- Remedial Action Costing Procedures Manual; JRB Associates, October 1987. EPA/600/8-87/049.
- 3. Costs of Remedial Response Actions at Uncontrolled Hazardous Waste Sites; Howard J. Rishel; 1981.
- 4. CompuChem Analytical Laboratories, March 1991.
- 5. Means Site Work Cost Data, 1991.
- 6. Means Building Construction Cost Data, 1987.
- 7. Waste Age; March 1988.
- Compendium of Costs of Remedial Technologies at Hazardous Waste Sites; Environmental Law Institute; October 1987. EPA/600/2-87/087.
- 9. TRC Environmental Consultants, Inc.; December 1989.
- 10. Means Site Work Cost Data; 1988.
- 11. PEDCO Environmental, Inc. Fugitive Dust Control Techniques at Hazardous Waste Sites. Draft Report to USEPA Municiple Environmental Research Lab; Cincinnati, Ohio; September 1984.
- 12. Compendium of Costs of Remedial Technologies at Hazardous Waste Sites; Environmental Law Institute; September 1983.
- 13. TRC Environmental Consultants, Inc.; 1991.
- Slurry Trench Construction for Pollution Migration Control; February 1984. EPA/540/2-84/001.
- 15. Personal Communication, WesStates Carbon; March 12, 1991.
- Remedial Action at Waste Disposal Sites; October 1985. EPA/625/6-85/006.
- 17. Underground Storage Tank Corrective Action Technologies; January 1987. EPA/625/6-87/015.
- Personal Communication, Peroxidation Systems, Inc.; March 15, 1991.
- 19. Carbon Adsorption Handbook; Paul N. Cheremisinoff and Fred Ellerbusch; 1978.
- 20. Pollution Engineering; November 1989.

- 21. CECOS, 1988.
- 22. Personal Communication, Clean Harbors; February 1991.
- 23. Geosafe Corporation. Application and Evaluation Considerations for InSitu Vitrification Technology: A Treatment Process For Destruction and/or Permanent Immobilization of Hazardous Materials; April 1989.
 - 24. Personal Communication, Ejector Systems; March 8, 1991.
- 25. Personal Communication, Oberlin Filter Company; March 8, 1991.
 - 26. Du Pont Company; March, 1991.

THIS PAGE INTENTIONALLY LEFT BLANK

APPENDIX B

A

EXPOSURE EQUATIONS, FUGITIVE DUST MODELING AND DOSE CALCULATIONS

.
THIS PAGE INTENTIONALLY LEFT BLANK

7

EXPOSURE EQUATIONS, FUGITIVE DUST MODELING AND DOSE CALCULATIONS

FUTURE RECREATIONAL USE: CHILDREN PLAYING ON SITE (Alternatives 2 and 2A)

Dermal Contact with Chemicals in Sediment

Equation:

Absorbed Dose (mg/kg-day) = CS x CF x SA x AF x ABS x EF x ED BW x AT

where:

CS = Chemical of Concentration in Soil (mg/kg) $CF = Conversion Factor (10^{-6} kg/mg)$ SA = Skin Surface Area Available for Contact (cm²/event)AF = Soil to Skin Adherence Factor (mg/cm²)ABS = Absorption Factor (unitless) EF = Exposure Frequency (events/year) ED = Exposure Duration (years) BW = Body Weight (kg) AT = Averaging Time (period over which exposure is averaged - days) Specific Parameter Values: CS = Concentration of chemicals in soil $SA = 6,800 \text{ cm}^2$, based upon exposed arms, hands and legs $AF = 1.45 \text{ mg/cm}^2$, based upon commercial potting soil adherence to hands ABS = 0.1 (organics), 0.01 (inorganics), except arsenic (0.1) EF = 12 days/year, based upon weekly visits during the summer months ED = 9 years BW = 49 kgAT = 3,285 days for non-cancer risks 25,550 days for cancer risks

Ingestion of Chemicals in Sediment

Equation:

Intake (mg/kg-day) = CS x IR x CF x FI x EF x ED

BW x AT

where:

```
CS = Chemical Concentration in Soil (mg/kg)
IR = Ingestion Rate (mg soil/day)
CF = Conversion Factor (10<sup>-6</sup> kg/mg)
FI = Fraction Ingested from Contaminated Source (unitless)
EF = Exposure Frequency (days/years)
ED = Exposure Duration (years)
BW = Body Weight (kg)
AT = Averaging Time (period over which exposure is averaged - days)
Specific Parameter Values:
CS = Chemical concentration in sediments
IR = 100 mg/day, which is typical for this age group.
FI = 1.0, assuming 100% of sediment ingestion occurs on site
EF = 12 days/year
ED = 9 years
BW = 49 kg
```

AT = 3,285 days for non-cancer risks 25,550 days for cancer risks COMMON REMEDIAL ELEMENTS: ROAD IMPROVEMENTS AND CLEARING VEGETATION (Alternatives 2, 2A, 4 and 4A) Dermal Contact with Chemicals in Soil Equation: Absorbed Dose (mg/kg-day) = $CS \times CF \times SA \times AF \times ABS \times EF \times ED$ _____ BW x AT where: CS = Chemical Concentration in Soil (mg/kg) $CF = Conversion Factor (10^{-6} kg/mg)$ SA = Skin Surface Area Available for Contact (cm²/event)AF = Soil to Skin Adherence Factor (mg/cm²)ABS = Absorption Factor (unitless) EF = Exposure Frequency (events/year) ED = Exposure Duration (years) BW = Body Weight (kg) AT = Averaging Time (period over which exposure is averaged - days) Specific Parameter Values: SA = 6,300 cm² for hands, forearms, upper arms, head, neck and a portion of the trunk $AF = 1.45 \text{ mg/cm}^2$ ABS = 0.1 (organics), 0.01 (inorganics), except arsenic (0.1) EF = 15 days/year ED = 1 year BW = 70 kgAT = 15 days for non-cancer risks 25,550 days for cancer risks Ingestion of Chemicals in Soil Equation: Intake (mg/kg-day) = CS x IR x CF x FI x EF x ED BW x AT where: CS = Chemical Concentration in Soil (mg/kg)IR = Ingestion Rate (mg soil/day) $CF = Conversion Factor (10^{-6} kg/mg)$ FI = Fraction Ingested from Contaminated Source (unitless) EF = Exposure Frequency (days/years) ED = Exposure Duration (years) BW = Body Weight (kg) AT = Averaging Time (period over which exposure is averaged - days)

Specific Parameter Values: IR = 100 mg/day, based upon extensive contact with soil FI = 1.0; all soil ingested comes from on-site sources EF = 15 days/year ED = 1 year BW = 70 kg AT = 15 days for non-cancer risks 25,550 days for cancer risks

Inhalation of Airborne Chemicals Adsorbed to Dust

Equation:

Intake (mg/kg-day) = CD x CS x IR x ET x EF x ED BW x AT

where:

CD = Ambient Dust Concentration CS = Contaminant Concentration in Soil (mg/kg) IR = Inhalation Rate (m³/hour) ET = Exposure Time (hours/day) EF = Exposure Frequency (days/year) ED = Exposure Duration (years) BW = Body Weight (kg) AT = Averaging Time (period over which exposure is averaged -- days) Specific Parameter Values:

 $IR = 2 m^3$ /hour for adults under moderate exertion

- ET = 8 hours/day
- EF = 15 days/year
- ED = 1 year
- BW = 70 kg
- AT = 15 days for non-cancer risks 25,550 days for cancer risks

CONSTRUCTION OF THE LANDFILL CAP (Alternatives 2, 2A, 4 and 4A)

Dermal Contact with Chemicals in Soil Equation: Absorbed Dose (mg/kg-day) = CS x CF x SA x AF x ABS x EF x ED BW x AT where: CS = Chemical Concentration in Soil (mg/kg) $CF = Conversion Factor (10^{-6} kg/mg)$ SA = Skin Surface Area Available for Contact (cm²/event)AF = Soil to Skin Adherence Factor (mg/cm²)ABS = Absorption Factor (unitless) EF = Exposure Frequency (events/year) ED = Exposure Duration (years) BW = Body Weight (kg) AT = Averaging Time (period over which exposure is averaged - days) Specific Parameter Values: $SA = 6,300 \text{ cm}^2$ for hands, forearms, upper arms, head, neck and a portion of the trunk $AF = 1.45 \text{ mg/cm}^2$ ABS = 0.1 (organics), 0.01 (inorganics), except arsenic (0.1) EF = 80 days/yearED = 1 year BW = 70 kgAT = 80 days for non-cancer risks 25,550 days for cancer risks Ingestion of Chemicals in Soil Equation: Intake (mg/kg-day) = CS x IR x CF x FI x EF x ED BW x AT where: CS = Chemical Concentration in Soil (mg/kg) IR = Ingestion Rate (mg soil/day) $CF = Conversion Factor (10^{-6} kg/mg)$ FI = Fraction Ingested from Contaminated Source (unitless) EF = Exposure Frequency (days/years) ED = Exposure Duration (years) BW = Body Weight (kg) AT = Averaging Time (period over which exposure is averaged - days)

Specific Parameter Values: IR = 100 mg/day, based upon extensive contact with soil FI = 1.0; all soil ingested comes from on-site sources EF = 80 days/year ED = 1 year BW = 70 kg AT = 80 days for non-cancer risks 25,550 days for cancer risks

Inhalation of Airborne Chemicals Adsorbed to Dust

Equation:

Intake (mg/kg-day) = CD x CS x IR x ET x EF x ED BW x AT

where:

CD = Ambient Dust Concentration CS = Contaminant Concentration in Soil (mg/kg) IR = Inhalation Rate (m³/hour) ET = Exposure Time (hours/day) EF = Exposure Frequency (days/year) ED = Exposure Duration (years) BW = Body Weight (kg) AT = Averaging Time (period over which exposure is averaged -- days)

Specific Parameter Values:

IR = 2 m³/hour for adults under moderate exertion ET = 8 hours/day EF = 80 days/year ED = 1 year BW = 70 kg AT = 80 days for non-cancer risks 25,550 days for cancer risks INSTALLATION OF EXTRACTION WELLS (Alternatives 4 and 4A)

Dermal Contact with Chemicals in Soil Equation: Absorbed Dose (mg/kg-day) = CS x CF x SA x AF x ABS x EF x ED BW x AT where: CS = Chemical Concentration in Soil (mg/kg) $CF = Conversion Factor (10^{-6} kg/mg)$ SA = Skin Surface Area Available for Contact (cm²/event)AF = Soil to Skin Adherence Factor (mg/cm²)ABS = Absorption Factor (unitless) EF = Exposure Frequency (events/year) ED = Exposure Duration (years) BW = Body Weight (kg) AT = Averaging Time (period over which exposure is averaged - days) Specific Parameter Values: SA = 6,300 cm² for hands, forearms, upper arms, head, neck and a portion of the trunk $AF = 1.45 \text{ mg/cm}^2$ ABS = 0.1 (organics), 0.01 (inorganics), except arsenic (0.1) EF = 250 days/yearED = 1 year BW = 70 kgAT = 250 days for non-cancer risks 25,550 days for cancer risks Ingestion of Chemicals in Soil Equation: Intake (mg/kg-day) = CS x IR x CF x FI x EF x ED BW x AT where: CS = Chemical Concentration in Soil (mg/kg)IR = Ingestion Rate (mg soil/day) $CF = Conversion Factor (10^{-6} kg/mg)$ FI = Fraction Ingested from Contaminated Source (unitless) EF = Exposure Frequency (days/years) ED = Exposure Duration (years) BW = Body Weight (kg)AT = Averaging Time (period over which exposure is averaged - days)

Specific Parameter Values:

IR = 100 mg/day, based upon extensive contact with soil FI = 1.0; all soil ingested comes from on-site sources EF = 250 days/year ED = 1 year BW = 70 kg AT = 250 days for non-cancer risks 25,550 days for cancer risks

Inhalation of Airborne Chemicals Adsorbed to Dust

Equation:

Intake (mg/kg-day) = CD x CS x IR x ET x EF x ED BW x AT

where:

CD = Ambient Dust Concentration CS = Contaminant Concentration in Soil (mg/kg) IR = Inhalation Rate (m³/hour) ET = Exposure Time (hours/day) EF = Exposure Frequency (days/year) ED = Exposure Duration (years) BW = Body Weight (kg) AT = Averaging Time (period over which exposure is averaged -- days)

Specific Parameter Values:

 $IR = 2 m^3$ /hour for adults under moderate exertion

- ET = 8 hours/day
- EF = 250 days/year
- ED = 1 year
- BW = 70 kg
- AT = 250 days for non-cancer risks 25,550 days for cancer risks

MODEL ESTIMATES OF FUGITIVE DUST GENERATION

Emissions estimates were calculated for activities resulting in soil disturbance, such as heavy equipment operation and wind erosion which may occur over the site during the construction scenario.

The potentially significant components of fugitive dust at this site are:

- wind erosion of dust from surfaces without vegetative cover,
- 2) dust from loading/unloading of excavated soil, and
- 3) emissions from vehicular traffic on unpaved roads.

Fugitive dust from wind erosion over exposed soil and from loading/unloading activities was calculated using (EPA, AP-42, 1985). Fugitive dust generation tables showing all model inputs, are presented in Tables C-1 through C-3. The data are summarized in Table C-4. The models are described below.

 $E = a \bullet I \bullet K \bullet C \bullet L \bullet V \bullet A \bullet T$

where:

- E = Emission rate (kg/day)
- a = Fraction of total wind losses (wind erosion of soil)
 that remain suspended
- I = Soil erodibility
- C = Climatic factor
- K = Soil roughness factor
- L = Field length factor
- V = Vegetative cover factor
- A = Area of the site
- T = Time conversion factor

Most of these values are specified in USEPA (1985) for worst-case treatments. The climatic factor is read from a map and multiplied by .01 as specified. The variables a and I are determined based on site soil characteristics. The following values were used to estimate the emission rates during road improvements and clearing vegetation across the landfill area:

> a = 0.01 I = 134 tons acre⁻¹ yr⁻¹ K = 1 (worst-case for flat terrain) V = 1 (no vegetative cover-worst case) L = .7 C = 0.02 A = 13.2 acres T = 1 yr/365 days

The wind erosion emission rate is presented in Table C-1.

The second component is due to loading/unloading of soils (excavating the landfill prior to incineration) due to excavation activities and can be accounted for by:

$$E = \frac{k \cdot (.0016) \cdot (U/2.2)^{1.3}}{(M/2)^{1.4}}$$

and

 $E_{ed} = V \cdot D \cdot E/T$

where:

E = Emission factor due to loading/dumping (kg/Mg)
k = Particle size multiplier
U = Mean wind speed (m/s)
M = Soil moisture (%)
Eed = Emission rate due to loading/dumping (kg/day)
V = Volume of soil excavated (m³)
D = Density of soil (Mg/m³)
T = Time conversion factor (days of excavation)

Using conservative assumptions and appropriate guidelines (EPA, AP-42, 1985):

k = .74 U = 4.56 m/sM = 5% V = 229,365 m³ D = 1.5 Mg/m³ T = 1,250 days

The emissions due to loading/dumping are presented in Table C-2.

Emissions due to vehicular traffic are also estimated by (EPA, 1988):

$$E_{VT} = \left[1.7 \bullet K \bullet \left(\frac{s}{12} \right) \bullet \left(\frac{s_p}{48} \right) \bullet \left(\frac{w}{2.7} \right)^{.7} \bullet \left(\frac{M_w}{4} \right)^{.5} \bullet \frac{(365 - D_p)}{365} \right] \bullet D_{vt}$$

where:

EvT = Emissions due to vehicular traffic (kg yr⁻¹) K = Particle size multiplier s = Silt content of soil (%) Sp = Vehicle speed (km/hr) W = Vehicle weight (Mg) Mw = Mean number of wheels Dp = Day with >.254 mm of precipitation Dvt = Vehicle miles (km yr) Using conservative assumptions and suggested values (EPA, 1988):

K = 1.0 s = 28.5% Sp = 10 km/hr $M_W = 6 (trucks)$ Dp = 140 days W = 10.0 Mg (truck) $D_{vt} = 407.4 \text{ km/yr (truck)}$

Table C-3 contains fugitive dust emission rates for vehicular activity over unpaved roads.

Total fugitive emissions are presented in Table C-4.

The dust concentration on site is calculated by:

$$Cs = \frac{E}{W \cdot W \cdot H} \cdot C_{f}$$

where:

Cs = Dust concentration on site (mg/m³) E = Total emission rate (kg/day) w = Wind speed = 4.56 m/s W = Width (entire site) = 198 m H = Breathing height = 2 m Cf = Factors for converting from days to seconds and from kg to mg

Total fugitive dust concentrations on site are shown in Table C-4.

The concentration of contaminant suspended in air is estimated by a simple ratio of contaminant concentration in soil to fugitive dust emissions:

 $A_{c} = CC \cdot C_{s} \cdot C_{f}$

where:

 A_c = Concentration of suspended contaminant (mg/m³) CC - Contaminant concentration in soil (mg/kg) C_s = Dust concentration on site (mg/m³) C_f = Conversion factor (kg/mg)

	uind Erosion: Emission Raite (kg/day)	6.07E-01
	AREA (acres)	13.2
	TIME CONSTANT (year/day)	0.0027
	EMISSION FACTOR (E) kg/acre/year)	1.7E+01
	CONVERSION FACTOR (kg/ton) [(907.18
ITY STUDY	EMISSION FACTOR (E) (ton/acre/yr	1.95-02
DFILL FEASIBILI	PORTION AS SUSPENDED PARTICULATES (a)	0.010
HERTEL LAN	501L ERODIBILITY (1) ton/acre/year);	134
	SURFACE ROUGHNESS FACTOR (K)	-
	CL IMATIC FACTOR (c)	0.02
	FIELD WIDTH FACTOR (L ')	<i>C</i> .0
	VEGETATIVE VEGETATIVE COVER FACTOR (V')	

TAELE B-1 NIND EROSION - DUST EMISSION RATE

	CONDING/DUPTING	(kg/day)	2.29€-01	
	VOLUNE	EXTAVATED	229365	
	0606117	(D) (10) (10)	1.5	
	######################################	(days)	1250	
y study	EMISSION	(E) (kg/Mg)	8.3E-04	
ofill feasibility	PARTICLE SIZE	WULIATEK (k)	0.74	
LUMUTING & UN HERTEL LAN	PARTICLE SIZE	CONSIANI	1.60E-03	
	HEAN UIND	SPEED (U) (m/s)	4.5	
	NINO SPEED	CONSTANT	2.2	
	MATERIAL	MOISTURE CONTENT (M) (2)	Ś	
	MOISTURE CONTEN	CONSTANT	2	

TABLE B-2

IABLE B-3 UNPAVED ROAD EMISSIONS HERTEL LANDFILL FEASIBILITY STUD>

Ē	NOI	::	::	===	==	:::	=	Ē
	PITAT	VSTANT			ľ	5		Ē
I	PREC	8 					_	Ē
	ъ	[TATIO	_				Ē	Ē
	DAYS	RECIPI	٩					Ē
		<u>×</u>				 ?	- [Ē
i	E	INI				•		Ē
	±	SNOC						Ē
	 e		'		<u> </u>	•	- [Ē
I	NUNB	HEELS	Ē				-	Ē
I	ЖW	5	-				Ē	Ξ
							-	-
I	IGHT	STANT				•	-	-
	불	ŝ					-	Ē
	ICLE!					• •	- 1	Ē
I	N VEH	ue 1 GH	(n)	(6H)	-	-		Ī
	ME.						- [-
=	រ រ រ					1.04		-
	y;	ۍ ۲					Ē	Ē
	E		~ -	÷.		2	Ē	Ē
	I VEHI	SPEED	(2)	noų/w			-	Ξ
	HEA			=				Ē
		Ŧ				12.0	Ī	Ī
	SILT	ONSTAN						Ē
					ļ			-
	FACE	RENT	_	_	1	C. 67	-	Ē
	NO SU	C0	S	~				Ē
	8	: SI			 		-	Ī
	ICLE	TANT				-	-	=
	PARI	CONS						Ē
						•		Ē
	SIZE	A IER	_					
	RIICLE	MULTIF	*					
	3				ļ.			
	8	æ	_	Ē		595-02		
	FNISS1	FACTO	5	(kg/V		-		
		:::	: = :	:::	==	= = :	:=	

		DUST 10	ON SITE	(£∎/6∎)	1.02E-02 8.68E-03	
		UIND	A (()	(a/s)	4.5	
		SITE		•	98 138	
		BREATHING		•	2 2	
	ITRATIONS	CONVERSION	LHCION	(day/sec)	1.16E-05 1.16E-05	
TABLE 8-4	TIVE DUST CONCEN	CONVERSION		(19/kg)	1.00E+06 1.00E+06	
	TOTAL FUGI	TOTAL DATE		(kg/day)	1.35E+00 1.33E+00	
		UNDAVED		(kg/day)	7.28E-01 7.28E-01	
		LOADING/DUMPING		(kg/day)	2.29E-01 0.00E+00	
		UIND EROSION		(kg/day)	6.07E-01 6.07E-01	

VEHICULAR DAYS OF VEHICULAR EMISSION VEHICLE RAFTC EVENT ON EMISSION FACTOR KILORETR RAFT CONTANTHATED FCR EVENT (E) TRAVELED RAFT CONTANTHATED FCR EVENT (E) TRAVELED RAFT SOIL (Ag) (Kg/WT) (WT) (WT) 7.2RE-01 5 3.64E+00 8.93E-03 407.40

UNPAVED ROAD EMISSIONS

INGESTION OF CHEMICALS IN SOIL ROAD IMPROVEMENTS AND CLEARING VEGETATION (ALTERNATIVES 2, 2A, 4 AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

		11111111111111 11111111111111111111111	11111111111111111111111111111111111111	CONC IN	INGESTION	CONVERSION	FRACTION	FYDOSLIDE	EXPOSIDE	RODY	AVG TIME!	AVG TIME!	
1	CHEM1CA	NONCANCER	CANCER	SOTI	RATE	FACTOR	INGESTED	FREDUENCY	DIRATION!	UFIGHT	NONCANCER!	CANCER	ł
ł		(no/ko/day)	(mo/ko/day)	(mo/ka)	(ac soil/day)	(10F-6 kg/mg)	(unitless)	(days/year)	(years)	(ka)	(days)	(days)	1
į	!	1(113) 13, 04/ /								(***)		!	į
ì	SEMIVOLATILES	!		1			1		: !			i	i.
i	Benzo(a)anthracene	6.16E-07	3.62E-10	0.4311	100	1.0E-06	1	15	1	70	15	25550	i
1	Chrysene	6.15E-07	3.61E-10	0.4302	100	1.0E-06	1	15	1	70	15	25550	į.
į	Benzo(b)fluoranthene	9.79E-07	5.758-10	0.6856	100	1.0E-06	1	15	1	70	15	25550	į.
į	Benzo(k)fluoranthene	5.58F-07	3.28E-10	0.3908	100	1.0E-06	1	15	1	70	15	25550	į.
i	Benzo(a) ovrene	3 43E-07	2 01F-10	0.2400	100	1.0E-06	1	15	1	70	15	25550	į.
ł	Indepo(123cd)ovrene	6.13E-06	3.60F-09	4.2918	100	1.0E-06	1	15	1	70	15	25550	į
i	Bis(2ethylberyl)ohthalate	9.12E-06	5.36E-09	6.3869	100	1.0F-06	1	15	1	70	15	25550	į.
i		,											i.
į					1			,					į.
i	Naphthalene	9.42E-07	5.53E-10	0.6593	100	1.0E-06	1	15	1	70	15	25550	i.
į	Fluorene	8.60E-07	5.05E-10	0.6017	100	1.0E-06	1	15	1	70	15	25550	į.
i	Phenanthrene	1.20E-06	7.02E-10	0.8366	100	1.0E-06	1	15	1	70	15	25550	i.
i	Anthracene	2.14E-07	1.26E-10	0.1500	100	1.0E-06	1	15	1	70	15	25550	į.
ì	Fluoranthene	1.93E-06	1.13E-09	1.3511	100	1.0E-06	1	15	1	70	15	25550	į.
i	Pyrene	1.73E-06	1.01E-09	1.2083	100	1.0E-06	1	15	1	70	15	25550	i
i	Benzo(ghi)perylene	6.13E-06	3.60E-09	4.2918	100	1.0E-06	1	15	1	70	15	25550	į.
i	Diethylphthalate	1.57E-07	9.23E-11	0.1100	100	1.0E-06	1	15	1	70	15	25550	i.
i	Di-n-butylphthalate	B.02E-06	4.71E-09	5.6157	100	1.0E-06	1	15	1	70	15	25550	i
i	Di-n-octylphthalate	2.86E-07	1.68E-10	0.2000	100	1.0E-06	1	15	1	70	15	25550	i
i	Butylbenzylphthalate	3.43E-07	2.01E-10	0.2400	100	1.0E-06	1	15	1	70	15	25550	i
į	P-Cresol	4.86E-07	2.85E-10	0.3400	100	1.0E-06	1	15	1	70	15	25550	i
i	Phenol	6.13E-06	3.60E-09	4.2918	100	1.0E-06	1	15	1	70	15	25550	i
i					••••••								
į	INORGANICS	:	:	;	1		:	1	:		:	: :	
į		4	1	1									i
1	Aluminium	1.89E-02	1.11E-05	13255.5	100	1.0E-06	1	15	1	70	15	25550	
ł	Antimony	3.26E-05	1.91E-08	22.8	100	1.0E-06	1	15	1	70	15	25550	
ł	Arsenic	9.86E-06	5.79E-09	6.9	100	1.0E-06	1	15	5. 1	70	15	25550	
ł	Barium	1.70E-04	9.96E-08	118.8	100	1.0E-06	1	15	1	, 70	15	25550	
ł	Beryllium	1.14E-06	6.71E-10	0.80	100	1.0E-06	- 1	15	1	70	15	25550	
ł	Cadaium	1.14E-06	6.71E-10	0.80	100	1.0E-06	1	15	1	70	15	25550	
1	Calcium	2.46E-03	1.44E-06	1722.8	100	1.0E-06	1	15	1	70	15	25550	
ł	Chromium	2.61E-05	1.53E-08	18.3	100	1.0E-06	1	15	1	70	15	25550	
Ì	Cobalt	1.79E-05	1.05E-08	12.5	100	1.0E-06	1	15	1	70	15	25550	1
ľ	Copper	5.16E-05	3.03E-08	36.1	100	1.0E-06	; 1	15	1	70	15	25550	1
ł	Iron	3.47E-02	2.04E-05	24278.1	100	1.0E-06	1	15	1	70	15	25550	
ł	Lead	8.59E-05	5.04E-08	60.1	100	1.0E-06	1	15	1	70	15	25550	
ì	Magnesium	7.08E-03	4.16E-06	4954.3	100	1.0E-06	; 1	15	1	70	15	25550	11
ł	Manganese	1.79E-03	1.05E-06	1250.2	100	1.02-06	1	15	: 1	70	15	25550	
į	Hercury	0.00E+00	0.00E+00	;	100	1.0E-06	1	15	1	70	15	25550	
į	Nickel	3.01E-05	1.77E-08	21.1	100	1.0E-06	1	15	1	70	15	25550	1
į	Potassium	1.91E-03	1.12E-06	1334.6	100	1.0E-06	1	15	1	70	15	25550	ii.
į	Selenium	5.71E-07	3.35E-10	0.4	100	1.0E-06	1	15	1	70	15	25550	
į	Sodium	2.74E-04	1.61E-07	191.5	100	1.0E-06	1	15	1	70	15	25550	ii.
1	Thallium	0.00E+00	0.00E+00		100	1.0E-06	1	15	1	70	15	25550	ii.
ļ	Vandium	2.40E-05	1.41E-08	16.8	100	1.0E-06	1	15	1	70	15	25550	ii.
ļ	Zinc .	1.92E-04	1.12E-07	134.1	100	1.0E-06	; 1	15	; 1	70	15	25550	ii
ł			1				·			 1			
i	PESTICIDES/PCBS				1		1	1	1		1		11
i		1 C 075 04	1 0 455 11			1				1 70	1		
i	4,4-00t	j 5.8/E-08	3.452-11	0.0411	100	1.01-06	1	15	1	1 70	15	25550	ii.
i	14,4°UUI	1 5.8/E-V8	3.45E~11	1 0.0411	1 100	, 1.0L-06	1 1	1 15	1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1	, /0	1 15	1 20000	H
í	181471111111111111111111111111111111111	mannann	11111111111111	1111111111	1111111111111111111	1111111111111111	11111111111	1111111111111			11111111111		11

Ű,

٠Ż

'n

đ.

Ľ,

DERMAL CONTACT WITH CHEMICALS IN SOIL ROAD IMPROVEMENTS AND CLEARING VEGETATION (ALTERNATIVES 2, 2A, 4 AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

													1
	ABS. DOSE	ABS. DOSE	CONC. IN	CONVERSION	SURFACE	ADHERENCE	ABSORPTION	EXPOSURE	EXPOSURE	BODY	AVG. TIME	AVG. TIME	ł
CHEMICAL	NONCANCER	; CANCER	SOIL	FACTOR	AREA SKIN	FACTOR	FACTOR	FREQUENCY	DURATION	WEIGHT	NONCANCER	CANCER	i
	(199/Kg/day)	(mg/kg/day)	;(Mg/kg)	i (kg/mg)	{ Ch 2/event }	(mg/cm2)	(unitless)	;(days/year)	(years)	; (kg)	(days)	(days)	i
		h r I	1				1		-			1	ł
		0 205 00	1 0 1011		(į
Cenzo(a januni acene	1 5.03E-00	3.305-09	0.4311	1.0E-06	6300	1.45	0.10	15		/0	; 15	25550	i
unrysene Doorst half and have	5.612-06	3.302-09	0.4302	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	i
Benzo(b) (100ranthene	8.95E-06	5.252-09	0.6856	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	į
Benzo(k)fluoranthene	5.102-06	2.991-09	0.3908	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	ł
Benzo(a)pyrene	; 3.13E-06	1.64E-09	0.2400	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	į
Incered 123cd pyrene	5.602-05	3.29E-08	4.2918	1.06-06	6300	1.45	0.10	15	1	70	15	25550	i
BIS(Zethy Inexy I)phthalate	8.332-05	4.891-08	6.3869	1.0E-06	6300	1.45	0.10	15	1	. 70	15	25550	i
UI-N-OCLYIPHINAIALE	2.612-06	1.53E-09	0.2000	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	į
BULYDENZYIPHLNAIALE	3.13E-06	; 1.84E-09	0.2400	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	i
P-CL6201	4.441-06	2.60E-09	0.3400	1.0E-06	6300	1.45	0.10	15	: 1	70	15	25550	ł
l Marthtalana													i
Naphinalene	8.60E-06	5.05E-09	0.6593	1.0E-06	6300	1.45	0.10	15	1	; 70	15	25550	į
Fluorene	7.85E-06	4.61E-09	0.6017	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	ł
Phenanthrene	1.092-05	6.41E-09	0.8366	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	ļ
Anthracene	1.96E-06	1.15E-09	0.1500	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	ł
Fluoranthene	1.76E-05	1.04E-08	1.3511	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	ł
Pyrene	1.58E-05	9.26E-09	1.2083	1.0E-06	6300	1.45	0.10	15	; 1	70	15	25550	ł
Benzo(ghi)perylene	5.60E-05	3.29E-08	4.2918	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	i
Diethylphthalate	1.44E-06	8.43E-10	0.1100	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	ł
D1-n-butylphthalate	7.33E-05	4.30E-08	5.6167	1.0E-06	6300	1.45	0.10	15	1	70	, 15	25550	ļ
Pheno]	5.60E-05	3.29E-08	4.2918	; 1.0E-06	6300	1.45	0.10	15	1	70	15	25550	ł
INORGANICS	1	1	:	1		!	1		:	1	 	1	-
			;						•		1	1	1
Aluminium	1.73E-02	1.02E-05	13255.5	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	ł
Antimony	2.98E-05	1.75E-08	22.8	1.0E-06	6300	1.45	• 0.01	15	1	70	15	25550	ł
Arsenic	9.00E-05	5.29E-08	6.9	1.0E-06	6300	1.45	0.10	15	1	70	15	25550	ł
Barium	1.55E-04	9.10E-08	118.8	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	ł
Beryllium	1.04E-06	6.13E-10	0.80	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	ł
Cadmium	1.04E-06	6.13E-10	0.80	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	ł
Calcium	2.25E-03	1.32E-06	1722.8	1.0E-06	6300	1.45	0.01	15	; 1	70	15	25550	ł
Chromium	2.39E-05	1.40E-08	18.3	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	ł
Cobalt	1.63E-05	9.58E-09	12.5	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	ł
Copper	4.71E-05	2.77E-08	36.1	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	ţ
lron	3.17E-02	1.86E-05	24278.1	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	i
Lead	7.84E-05	4.60E-08	60.1	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	į
Magnesium	6.47E-03	3.80E-06	4954.3	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	i
Manganese	1.63E-03	9.58E-07	1250.2	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	į
Nickel	2.75E-05	1.62E-08	21.1	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	i
Potassium	1.74E-03	1.02E-06	1334.6	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	į
5elenium	5.22E-07	3.06E-10	0.4	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	i
Sodium	2.50E-04	1.47E-07	191.5	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	i
Vandium	2.19E-05	1.29E-08	16.8	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	ł
Zinc	1.75E-04	1.03E-07	134.1	1.0E-06	6300	1.45	0.01	15	1	70	15	25550	1
PESTICIDES/PCBS													
A A-DDE	5 345-07	3 155-10	0.0411	1 05-04	(200								1
4 4-DOT	5.30E-V/	3.100-10	0.0411	1.05-00	6300	1 1.45	0.10	15		70	15	25550	i
ועע דרד	0.30E-V/	3.100-10	, V.V4II I	, I.VE-V6	6300	1.45	0.10	15	1	70	15	25550	i
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1						1 1			; :::::::::::	i 110000			i

APPENDIX 8

.

INHALATION OF CHEMICALS BORN ON DUST PARTICLES ROAD IMPROVEMENTS AND CLEARING VEGETATION (ALTERNATIVES 2, 2A, 4 AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

DATA DATA <thdata< th=""> DATA DATA <thd< th=""><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th></thd<></thdata<>												
Checker I Image Part of th		NONCANCED	CANCED	DUSI CUNU.	CUNC. IN		EXPUSURE	EXPOSURE	EXPOSURE	BODY	AVG. LINE	AVG. TIME;
INCREMUES Interview Processing		(NUNUANUER)	(LAMLER (ma/ka/day)	(ka/=2)	(SUIL	KAIL	j ll⊓t: !(hourno/dau)	; FREQUENCT	(DURATION)	WEIGHI	NONCANCER	
INDEX-VIDES Image: Second	 	(E9/K9/04)	(∎9/k9/0ay)	(Kg/a3)	(RG/K9)	(13/nour)	(nours/oay)	(days/year)	(years)	(Kg)	((ays)	(days) ;;
DREGNITICS 2.6.5 2 8 15 1 70 15 25500 Aluaining 2.6.5 2.6.6 6.6.6 9 2 6 15 1 70 15 25500 Arsenic 1.377-05 8.6.67-06 8.667-06 0.8 2 6 155 1 70 15 25500 Barlin 1.576-06 3.327-10 8.667-06 0.8 2 6 155 1 70 15 25500 Cachium 3.467-03 2.017-06 8.667-06 122.5 2 6 155 1 70 15 25500 Cobain 3.467-03 2.017-06 8.667-06 122.5 2 6 155 1 70 15 25500 Cobain 2.467-03 1.467-06 8.647-06 12.5 2 6 155 1 70 15 25500 Cobain 9.016-06 8.647-06 12.5 2 <td< td=""><td>1</td><td>1</td><td>!</td><td>!</td><td>!</td><td>!</td><td>!</td><td>1</td><td>, ,</td><td>1</td><td></td><td></td></td<>	1	1	!	!	!	!	!	1	, ,	1		
Atuanina 2.6.2 1.6.2-05 1.6.2-05 1.6.2-05 2.2.6 15 1 70 15 25550 Antisony 4.52-05 2.66-06 8.667-06 22.8 2 8 155 1 70 15 25550 Aresuic 2.362-04 1.362-07 8.667-06 0.8 2 8 155 1 70 15 25550 Cabria 1.567-06 9.327-10 8.667-06 0.8 2 8 155 1 70 15 25550 Cabria 3.457-05 1.457-06 8.667-06 0.8 2 8 155 1 70 15 25550 Chosin 3.457-05 1.457-06 8.667-06 38.1 2 8 155 1 70 15 25550 Chosin 3.457-05 1.426-06 8.667-06 38.1 2 8 155 1 70 15 25550 Chosin 1.467-05 2.467-06 8.667-06 38.1 2 8 155 1 70 15 <td>INORGANICS</td> <td></td>	INORGANICS											
Altiony 4.52-05 2.06-06 22.8 2 8 15 1 70 15 25500 Barian 2.36-04 1.36-07 8.66-06 10.8 2 8 15 1 70 15 25500 Barian 2.36-04 1.36-07 8.66-06 10.8 2 8 15 1 70 15 25500 Cabius 3.47-03 2.01-06 8.66-06 1722.8 2 8 15 1 70 15 25500 Cabius 3.47-03 2.01-06 8.66-06 122.8 2 8 15 1 70 15 25500 Cober 7.146-05 1.46-06 8.66-06 122.5 2 8 15 1 70 15 25500 Cober 7.146-05 1.46-06 8.66-06 122.5 2 8 15 1 70 15 25500 Coper 7.146 8.66-06 122.4 0 15 15 15 15 15 15 15 25500	Aluminium	2.63E-02	1.54E-05	8.68E-06	13255.5	2	8	15	1	70	15	25550
Arsenic 1.37E-05 8.04E-07 8.86E-06 6.9 2 8 15 1 70 115 2550 Berryllium 1.58E-06 9.32E-10 8.68E-06 0.8 2 8 15 1 70 155 2550 Cabium 1.59E-06 9.32E-10 8.68E-06 0.8 2 8 15 1 70 155 2550 Choraium 3.42E-05 2.13E-06 8.68E-06 18.3 2 8 15 1 70 155 2550 Cobelt 2.46E-05 8.64E-06 2.84E-06 2.84E-06 2.84E-06 1.81 2 8 15 1 70 155 2550 Cober 1.46E-06 1.64E-06 8.64E-06 6.01 2 8 15 1 70 155 2550 Magnesian 9.85E-03 1.5E-06 8.64E-06 2.11 2 8 15 1 70 155 2550 Magnesian 2.5E-00 1.5E-07 8.64E-06 1.91 2 8	Antimony	4.52E-05	2.66E-08	8.68E-06	22.8	2	8	15	1	70	15	25550
Barrian 2.94C-04 1.38C-07 8.48C-05 118.8 2 8 15 1 70 115 25500 Cadaiua 1.59T-06 9.32C-10 8.48E-05 0.8 2 8 15 1 70 115 25500 Cadaiua 1.59T-06 9.32C-10 8.48E-05 10.8 2 8 15 1 70 115 25500 Choolin 3.42C-03 2.01F-06 8.48E-05 12.5 2 8 15 1 70 115 25500 Cooper 7.44E-05 4.24C-00 8.48E-06 12.5 2 8 15 1 70 115 25500 Itad 1.970-07 2.44E-00 1.44E-06 8.48E-06 120.2 2 8 15 1 70 115 25500 Magnesian 9.43E-03 5.277-08 8.48E-06 120.2 2 8 15 1 70 155 25500 So	Arsenic	1.37E-05	8.04E-09	8.68E-06	6.9	2	8	15	1	70	15	25550
Bery Jina 1.5% -06 9.32:-10 8.48:-06 0.8 2 8 15 1 70 15 2550 Calcium 3.47:-03 2.01:-06 8.486:-06 1722.8 2 8 15 1 70 15 2550 Chronium 3.58'-05 2.13:-08 8.486:-06 18.3 2 8 15 1 70 15 2550 Coper 7.16:-05 4.26:-06 8.486'-06 38.1 2 8 15 1 70 15 2550 Lead 1.15'C-02 2.86'C-06 8.486'-06 40.1 2 8 15 1 70 15 2550 Lead 1.15'C-02 2.86'C-06 8.486'C-06 121.1 2 8 15 1 70 155 2550 Magnesian 2.36C'-01 1.55'C-06 8.486'C-06 121.1 2 8 15 1 70 155 25550 Sodien 3	Barium	2.36E-04	1.38E-07	8.68E-06	118.8	2	8	15	1	70	15	25550
Cadaium 1.587-66 9.327-10 8.687-66 0.8 2 8 15 1 70 15 2550 Chronium 3.637-65 2.137-08 8.687-66 112.5 2 8 15 1 70 15 2550 Cobalt 2.468-65 1.467-06 8.687-66 6.81 2 8 15 1 70 15 2550 Cobalt 2.468-65 1.467-66 8.687-66 6.81 2 8 15 1 70 15 2550 Tron 4.827-62 2.087-65 8.687-66 6.61 2 8 15 1 70 15 2550 Magnesian 9.817-03 5.776-66 8.687-66 125.2 2 8 15 1 70 15 2550 Wickel 4.197-65 2.467-66 8.687-66 125.02 2 8 15 1 70 15 2550 Sodium 3.667-44 2.327-70 8.687-66 134.1 2 8 15 1 70	Beryllium	1.59E-06	9.32E-10	8.68E-06	0.8	2	8	15	1	70	15	25550
Calcium 3.427-03 2.01E-06 8.68E-06 122.8 2 8 15 1 70 15 25500 Cooper 7.16E-05 1.46E-08 8.68E-06 38.1 2 8 15 1 70 15 25500 Cooper 7.16E-05 1.46E-08 8.68E-06 38.1 2 8 15 1 70 15 25500 Cooper 7.16E-05 1.46E-08 8.68E-06 24278.1 2 8 15 1 70 15 25500 Lead 1.19E-04 7.06E-05 8.68E-06 620.1 2 8 15 1 70 15 25500 Magensim 9.68E-06 8.68E-06 125.2 2 8 15 1 70 15 25500 Mickal 4.19E-05 2.46E-00 8.68E-06 191.5 2 8 15 1 70 15 25500 Yonadium 3.36E-04 1.56E-07 8.68E-06 10.070 2 8 15 1 70 15	Cadmium	1.59E-06	9.32E-10	8.68E-06	0.8	2	8	15	1	70	15	25550
Chronium 3.4.8° - 63 2.1.8° - 68 2.1.8° - 68 1.2.5 2 8 15 1 70 15 25550 Cooper 7.1.6° - 63 4.02° - 60 8.08° - 60 2.2.5 2 8 15 1 70 15 25550 Cooper 7.1.6° - 63 0.40° - 60 0.40° - 60 0.4278.1 2 8 15 1 70 15 25550 Maganese 2.44° - 31 1.46° - 60 0.40° - 60 6.48° - 60 215.1 2 8 15 1 70 15 25550 Mickel 4.19° - 64 2.46° - 60 8.48° - 60 213.1 2 8 15 1 70 15 25550 Mickel 3.08° - 64 2.25° - 07 8.48° - 66 133.4.6 2 8 15 1 70 15 25550 Valadium 3.38° - 65 1.95° - 60 1.48° - 60 10.41 2 8 15 1 70 15 25550 Valadium 3.38° - 67 2.48° - 66 0.400° 2	Calcium	3.42E-03	2.01E-06	8.68E-06	1722.8	2	8	15	1	70	15	25550
LC0011 2.48C-00 1.44C-08 8.68C-06 12.5 2 8 15 1 70 15 25500 Lron 4.42C-02 2.83E-05 8.68E-06 427811 2 8 15 1 70 15 25550 Lrad 1.19E-04 7.00E-08 8.68E-06 427811 2 8 15 1 70 15 25550 Magnesium 9.010-05 7.74E-06 8.68E-06 42781.3 2 8 15 1 70 15 25550 Magnesium 2.64E-03 1.44E-06 8.64E-06 121.1 2 8 15 1 70 15 25550 Sodium 3.66E-04 1.55E-06 8.64E-06 134.6 2 8 15 1 70 15 25550 Zinc 2.64E-04 1.56E-07 8.64E-06 0.0030 2 8 15 1 70 15 25550 Zinc 2.64E-04 1.56E-07 8.64E-06 0.0030 2 8 15 1 70	(Chromium)	3.63€-05	2.13E-08	8.68E-06	18.3	2	8	15	1	70	15	25550
Copper 7.10e To 4.02e To 8.02e To <	LODAIL	2.48E-05	1.46E-08	8.68E-06	12.5	2	8	15	1	70	15	25550
100 1.02 2.02.02 2.02.02 2.02.02 8 15 1 70 15 2.2550 Magnesium 9.031-00 5.772-06 8.062-06 4543.3 2 8 15 1 70 15 2.2550 Magnesium 9.031-00 5.772-06 8.062-06 1250.2 2 8 15 1 70 15 2.2550 Magnesium 2.042-03 1.462-08 8.062-06 121.1 2 8 15 1 70 15 25550 Sodium 3.062-04 2.232-07 8.062-06 191.5 2 8 15 1 70 15 25550 Vanadium 3.324-05 1.964-08 8.062-06 134.1 2 8 15 1 70 15 25550 Zinc 2.062-04 1.562-07 8.062-06 0.0030 2 8 15 1 70 15 25550 Chiorobenzene 1.396-06 8.042-06 0.0276 2 8 15 1 70 15	Lopper	1.166-05	4.20E-08	; 8.68E-06	36.1	; 2	8	15	1	70	15	25550
1.18.4.00 1.18.4.00 0.000000 0.000000 0.000000 0.000000 0.000000 0.000000 0.0000000 0.00000000 0.0000000000 0.000000000000 0.0000000000000000000000 0.00000000000000000000000000000000000	11101	1 10C-04	2.03E-05	, 8.08E-V6	242/8.1		; 8	15	1	70	15	25550
Imagestate 7.01 (2) 2.01 (2) 0.01 (2) <th0.01 (2)<="" th=""> <th0.01 (2)<="" th=""> <th< td=""><td>Wagnesium</td><td>0 005-00</td><td>, 7.00E-00 E.77E-04</td><td>0.000-00</td><td>1 00.1</td><td>2</td><td>1 8</td><td>15</td><td>1</td><td>70</td><td>15</td><td>25550</td></th<></th0.01></th0.01>	Wagnesium	0 005-00	, 7.00E-00 E.77E-04	0.000-00	1 00.1	2	1 8	15	1	70	15	25550
International 2.100.03 1.000.04 0.000.00 2 0 1 <th1< th=""> 1 1</th1<>	Mandanes	1 7.03E-03	1 445-04	0.00E-V0	1 1250 2	2	1 0	1 15		i 70	15	25550
Protexts: Protexts:	[Nicke]	1 19E-05	1.40C-00	1 8 68E-06	1 1250.2	1 2	, 0 , 0	1 ID		, /U , 70	1 15	20000
Sodium 3.80E-06 2.22E-07 8.68E-06 15.1 1 70 15 22550 Vanadium 3.32E-05 1.96E-00 8.68E-06 16.8 2 8 15 1 70 15 22550 Zinc 2.66E-04 1.56E-07 8.68E-06 134.1 2 8 15 1 70 15 22550 Zinc 2.66E-04 1.56E-17 8.68E-06 0.0030 2 8 15 1 70 15 22550 Carbon Disulfide 5.95E-08 3.49E-12 8.68E-06 0.0070 2 8 15 1 70 15 25550 Chlorobenzene 1.39E-10 8.68E-06 0.0236 2 8 15 1 70 15 25550 Xrlenes 8.37E-07 4.91E-10 8.68E-06 0.0236 2 8 15 1 70 15 25550 SEHIVOLATILES 8.38E-07 2.30E-08 8.68E-06 0.2200 2 8 15 1 70 15 25550	Potassium	2 655-03	1 55E-06	8 68E-06	1324.6	1 2	, 0 I 0	1 15		1 70	+ 15 1 15	20000
Vanadium 3.33E-05 1.96E-08 0.68E-06 16.8 2 8 15 1 70 15 25500 Zinc 2.66E-04 1.56E-07 8.68E-06 134.1 2 8 15 1 70 15 25550 VULATILE5	Sodium	3.80E-04	2 23F-07	8 68F-06	191 5	2 2	1 8	15	1 1	70	1 15	2000011
Zinc Z.66E-04 1.56E-07 8.68E-06 134.1 Z 8 15 1 70 15 25550 Carbon Disulfide 5.95E-09 3.49E-12 8.68E-06 0.0030 2 8 15 1 70 15 25550 Chloroberzene 1.39E-08 8.15E-12 8.68E-06 0.0070 2 8 15 1 70 15 25550 Toluene 2.30E-08 1.35E-11 8.68E-06 0.0276 2 8 15 1 70 15 25550 Strivense 8.37E-07 4.91E-10 8.68E-06 0.0200 2 8 15 1 70 15 25550 Strivensylphthalate 3.7E-07 2.32E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Strivensylphthalate 1.27E-05 7.44E-07 2.86E-06 0.2000 2 8 15 1 70 15 25550 <t< td=""><td>Vanadium</td><td>3.338-05</td><td>1 96F-08</td><td>8 68E-06</td><td>16.8</td><td>2</td><td>1 8</td><td>1 15</td><td>1</td><td>70</td><td>1 15</td><td>2000 1</td></t<>	Vanadium	3.338-05	1 96F-08	8 68E-06	16.8	2	1 8	1 15	1	70	1 15	2000 1
VOLATILES Add to be an analysis Add to be and to be analysis Add	Zinc	2.66E-04	1.56E-07	8.68E-06	134.1	2	1 8	1 15	1	2 70	15	25550
VQLATILES VQLATILES Carbon Disulfide 5.95E-09 3.49E-12 8.68E-06 0.0030 2 8 15 1 70 15 25550 Chlorobenzene 1.39E-08 8.15E-12 8.68E-06 0.00236 2 8 15 1 70 15 25550 Toluene 2.30E-08 1.35E-11 8.68E-06 0.0236 2 8 15 1 70 15 25550 Toluene 2.30E-08 1.35E-11 8.68E-06 0.4218 2 8 15 1 70 15 25550 SEHIVOLATILES 8.37E-07 4.91E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 SEHIVOLATILES 8.68E-06 0.2000 2 8 15 1 70 15 25550 Bis(2-ethylhexylphthalate 1.27E-05 7.44E-09 8.68E-06 0.2000 2 8 15 1 70 15			, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,					1			1.	
Carbon Disulfide 5.95E-09 3.49E-12 8.68E-06 0.0030 2 8 15 1 70 15 25550 Chlorobenzene 1.39E-08 8.15E-12 8.68E-06 0.0236 2 8 15 1 70 15 25550 Ethylbenzene 4.68E-08 2.75E-11 8.68E-06 0.0236 2 8 15 1 70 15 25550 Toluene 2.30E-08 1.35E-11 8.68E-06 0.0216 2 8 15 1 70 15 25550 SHIVOLATILES 8.37E-07 4.91E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Bis(2-ethylhexyl)phthalate 1.27E-07 2.33E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Bis(2-ethylhexyl)phthalate 1.27E-07 2.33E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Phenanthrene 1.66E-06 9.74E-10 8.68E-06 0.2000	VOLATILES				1							
Chlorobenzene 1.39E-08 8.15E-12 8.68E-06 0.0070 2 8 15 1 70 15 25550 Ethylbenzene 4.68E-08 2.73E-11 8.68E-06 0.0236 2 8 15 1 70 15 25550 Toluene 2.30E-08 1.35E-11 8.68E-06 0.0216 2 8 15 1 70 15 25550 Valenes 8.37E-07 4.91E-10 8.68E-06 0.4218 2 8 15 1 70 15 25550 Bis(2-ethylhexyl)phthalate 1.27E-05 7.44E-09 8.68E-06 0.2000 2 8 15 1 70 15 25550 Bis(2-ethylhexyl)phthalate 4.76E-07 2.80E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Butybenzylphthalate 1.66E-06 9.74E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Phenathrene 1.66E-06 9.74E-10 8.68E-06 1.3511 <th< td=""><td>Carbon Disulfide</td><td>5.95E-09</td><td>3.49E-12</td><td>8.68E-06</td><td>0.0030</td><td>2</td><td>8</td><td>15</td><td>1</td><td>70</td><td>15</td><td>25550</td></th<>	Carbon Disulfide	5.95E-09	3.49E-12	8.68E-06	0.0030	2	8	15	1	70	15	25550
Ethylbenzene 4.68E-08 2.75E-11 8.68E-06 0.0236 2 8 15 1 70 15 25550 Toluene 2.30E-08 1.35E-11 8.66E-06 0.0216 2 8 15 1 70 15 25550 Xylenes 8.37E-07 4.91E-10 8.68E-06 0.4218 2 8 15 1 70 15 25550 SEMIVOLATILES 8 3.7E-07 2.33E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Di-N-Octylphthalate 3.97E-07 2.33E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Bis(2-ethylphthalate 3.97E-07 2.33E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Phenanthrene 1.66E-06 9.74E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Pyrene 2.40E-06 1.57E-09 8.68E-06 1.2083 2	Chlorobenzene	1.39E-08	8.15E-12	8.68E-06	0.0070	2	8	¦ 15	1	70	15	25550
Toluene 2.30E-08 1.35E-11 8.66E-06 0.0116 2 8 15 1 70 15 25550 Xylenes 8.37E-07 4.91E-10 8.68E-06 0.4218 2 8 15 1 70 15 25550 SEMIVOLATILES Bis(2-ethylhexyl)phthalate 1.27E-05 7.44E-09 8.68E-06 0.2000 2 8 15 1 70 15 25550 Di-N-Octylphthalate 3.97E-07 2.33E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Butybenzylphthalate 4.76E-07 2.30E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Phenanthrene 1.66E-06 9.74E-10 8.68E-06 1.3511 2 8 15 1 70 15 25550 Pyrene 2.40E-06 1.41E-09 8.68E-06 1.2083 2 8 15 1 70 15 25550 Pyrene 2.40E-07 2.65E-07 8.68E-06 1.2083	Ethylbenzene	4.68E-08	2.75E-11	8.68E-06	0.0236	2	8	15	1	70	15	25550
Xylenes 8.37E-07 4.91E-10 8.68E-06 0.4218 2 8 15 1 70 15 2550 SEHIV0LATILES Bis(2-ethylhexyl)phthalate 1.27E-05 7.44E-09 8.68E-06 6.3869 2 8 15 1 70 15 25550 Di-N-Octylphthalate 3.97E-07 2.33E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Butybenzylphthalate 4.76E-07 2.80E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Phenathrene 1.66E-06 9.74E-10 8.68E-06 0.2400 2 8 15 1 70 15 25550 Phrene 2.68E-06 1.57E-09 8.68E-06 1.2083 2 8 15 1 70 15 25550 Butylphthalate 1.11E-05 6.54E-09 8.68E-06 0.2000 2 8 15 1 70 15 25550 Butylphthalate 1.11E-05 6.54E-09 8.68E-06 <t< td=""><td>Toluene</td><td>2.30E-08</td><td>1.35E-11</td><td>8.68E-06</td><td>0.0116</td><td>2</td><td>8</td><td>15</td><td>1</td><td>70</td><td>15</td><td>25550</td></t<>	Toluene	2.30E-08	1.35E-11	8.68E-06	0.0116	2	8	15	1	70	15	25550
SEMIVOLATILES Bis(2-ethylhexyl)phthalate 1.27E-05 7.44E-09 8.68E-06 6.3869 2 8 15 1 70 15 25550 Di-W-Octylphthalate 3.97E-07 2.33E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Butybenzylphthalate 4.76E-07 2.80E-10 8.68E-06 0.2400 2 8 15 1 70 15 25550 Phenanthrene 1.66E-06 9.74E-10 8.68E-06 1.3511 2 8 15 1 70 15 25550 Pyrene 2.40E-06 1.41E-09 8.68E-06 1.2083 2 8 15 1 70 15 25550 Benzoic Acid 4.36E-07 2.56E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Benzoic Acid 4.36E-07 3.66E-06 0.2000 2 8 15 1 70 15	Xylenes	8.37E-07	4.91E-10	8.68E-06	0.4218	2	8	15	1	70	15	25550
Bis(2-ethylhexyl)phthalate 1.27E-05 7.44E-09 8.68E-06 6.3869 2 8 15 1 70 15 25550 Di-N-Octylphthalate 3.97E-07 2.38E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Butybenzylphthalate 4.76E-07 2.80E-10 8.68E-06 0.2400 2 8 15 1 70 15 25550 Phenanthrene 1.66E-06 9.74E-10 8.68E-06 0.8366 2 8 15 1 70 15 25550 Fluoranthene 2.66E-06 1.57E-09 8.68E-06 1.3511 2 8 15 1 70 15 25550 Pyrene 2.40E-06 1.41E-09 8.68E-06 1.2083 2 8 15 1 70 15 25550 Benzoic Acid 4.36E-07 2.56E-10 8.68E-06 0.2200 2 8 15 1 70 15 25550 Phenol 8.51E-06 5.00E-09 8.68E-06 0.3400 2	5EMIVOLATILES			1 1						1		
Bis(2-ethylhexyl)phthalate 1.27E-05 7.44E-09 8.68E-06 6.3869 2 8 15 1 70 15 25550 Di-N-Octylphthalate 3.97E-07 2.33E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Butybenzylphthalate 4.76E-07 2.80E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Phenanthrene 1.66E-06 9.74E-10 8.68E-06 0.8366 2 8 15 1 70 15 25550 Pyrene 2.68E-06 1.57E-09 8.68E-06 1.3511 2 8 15 1 70 15 25550 IN-Woutplothalate 1.11E-05 6.54E-09 8.68E-06 1.2083 2 8 15 1 70 15 25550 Benzoic Acid 4.36E-07 2.56E-10 8.68E-06 0.2200 2 8 15 1 70 15 25550 Poresol 6.75E-07 3.96E-10 8.68E-06 0.100 2 </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>1</td> <td>1</td> <td>1</td> <td></td> <td>1</td> <td></td>							1	1	1		1	
Di-N-Octylphthalate 3.97E-07 2.33E-10 8.68E-06 0.2000 2 8 15 1 70 15 25550 Butybenzylphthalate 4.76E-07 2.80E-10 8.68E-06 0.2400 2 8 15 1 70 15 25550 Phenanthrene 1.66E-06 9.74E-10 8.68E-06 0.8366 2 8 15 1 70 15 25550 Fluoranthene 2.66E-06 1.57E-09 8.68E-06 1.3511 2 8 15 1 70 15 25550 Pyrene 2.40E-06 1.41E-09 8.68E-06 1.2083 2 8 15 1 70 15 25550 Benzoic Acid 4.36E-07 2.56E-10 8.68E-06 0.2200 2 8 15 1 70 15 25550 Percesol 6.75E-07 3.96E-10 8.68E-06 0.2200 2 8 15 1 70 15 25550 Ip-cresol 6.75E-07 3.96E-10 8.68E-06 0.1100 2 8	Bis(2-ethylhexyl)phthalate	1.27E-05	7.44E-09	8.68E-06	6.3869	2	8	15	1	70	15	25550
Butybenzylphthalate 4.76E-07 2.80E-10 8.68E-06 0.2400 2 8 15 1 70 15 25550 Phenanthrene 1.66E-06 9.74E-10 8.68E-06 0.8366 2 8 15 1 70 15 25550 Fluoranthene 2.68E-06 1.57E-09 8.68E-06 1.3511 2 8 15 1 70 15 25550 Pyrene 2.40E-06 1.41E-09 8.68E-06 1.2083 2 8 15 1 70 15 25550 Benzoic Acid 4.36E-07 2.56E-10 8.68E-06 0.2200 2 8 15 1 70 15 25550 Phenol 8.51E-06 5.00E-09 8.68E-06 0.2200 2 8 15 1 70 15 25550 p-cresol 6.75E-07 3.96E-10 8.68E-06 0.3400 2 8 15 1 70 15 25550 Fluoranthene 2.68E-06 1.55E-07 3.96E-10 8.68E-06 0.100 2	Di-N-Octylphthalate	3.97E-07	2.33E-10	8.68E-06	0.2000	2	8	15	1	70	15	25550
Prehanktriene 1.662-06 9.742-10 8.682-06 0.8366 2 8 15 1 70 15 25550 Fluoranthene 2.662-06 1.57E-09 8.68E-06 1.3511 2 8 15 1 70 15 25550 Pyrene 2.40E-06 1.41E-09 8.68E-06 1.2083 2 8 15 1 70 15 25550 Di-N-Butylphthalate 1.11E-05 6.54E-09 8.68E-06 5.6167 2 8 15 1 70 15 25550 Benzoic Acid 4.36E-07 2.56C-10 8.68E-06 0.2200 2 8 15 1 70 15 25550 Porcesol 6.75E-07 3.96E-10 8.68E-06 0.2200 2 8 15 1 70 15 25550 Iberversol 6.75E-07 3.96E-10 8.68E-06 0.3400 2 8 15 1 70 15 25550 Iberversol 6.75E-07 1.26E-10 8.68E-06 0.1100 2 8	Butybenzylphthalate	4.76E-07	2.80E-10	8.68E-06	0.2400	2	8	15	1	70	15	25550
r100rantheme 2.662-06 1.5/E-09 8.68E-06 1.3511 2 8 15 1 70 15 25550 Pyrene 2.40E-06 1.41E-09 8.68E-06 1.2083 2 8 15 1 70 15 25550 Di-N-Butylphthalate 1.11E-05 6.54E-09 8.68E-06 5.6167 2 8 15 1 70 15 25550 Benzoic Acid 4.36E-07 2.56E-10 8.68E-06 0.2200 2 8 15 1 70 15 25550 Phenol 8.51E-06 5.00E-09 8.68E-06 0.2200 2 8 15 1 70 15 25550 p-cresol 6.75E-07 3.96E-10 8.68E-06 0.3400 2 8 15 1 70 15 25550 Joiethylphthalate 2.18E-07 1.28E-10 8.68E-06 0.1100 2 8 15 1 70 15 25550 Fluoranthene 2.66E-06 1.57E-09 8.68E-06 0.1500 2 8	Phenanthrene	1.662-06	9./4E-10	8.681-06	0.8366	2	8	15	1	70	15	25550
Prime 2.40E-00 1.41E-09 6.66E-06 1.2083 2 8 15 1 70 15 25550 Di-N-Butylphthalate 1.11E-05 6.54E-09 8.68E-06 2.200 2 8 15 1 70 15 25550 Benzoic Acid 4.36E-07 2.56E-10 8.68E-06 0.2200 2 8 15 1 70 15 25550 Phenol 8.51E-06 5.00E-09 8.68E-06 0.2200 2 8 15 1 70 15 25550 p-cresol 6.75E-07 3.96E-10 8.68E-06 0.3400 2 8 15 1 70 15 25550 Diothylphthalate 2.18E-07 1.28E-10 8.68E-06 0.1100 2 8 15 1 70 15 25550 Fluoranthene 2.66E-06 1.57E-09 8.68E-06 0.1500 2 8 15 1 70 15 25550 Fluoranthene 1.96-06 7.68E-10 8.68E-06 0.6017 2 8 <t< td=""><td>In 1001 anchene</td><td>2.600-00</td><td>1.5/6-09</td><td>8.68E-V6</td><td>1.3511</td><td>2</td><td>; 8</td><td>15</td><td>1</td><td>, 70</td><td>15</td><td>25550</td></t<>	In 1001 anchene	2.600-00	1.5/6-09	8.68E-V6	1.3511	2	; 8	15	1	, 70	15	25550
In Heider Philipartic 1.11E-03 0.34E-05 0.36E-05 0.368F-06 2 8 15 1 70 15 25550 Benzoic Acid 4.36E-07 2.56E-10 8.68E-06 0.2200 2 8 15 1 70 15 25550 Phenol 8.51E-06 5.00E-09 8.68E-06 0.2200 2 8 15 1 70 15 25550 p-cresol 6.75E-07 3.96E-10 8.68E-06 0.3400 2 8 15 1 70 15 25550 Diethylphthalate 2.18E-07 1.28E-10 8.68E-06 0.1100 2 8 15 1 70 15 25550 Fluoranthene 2.68E-06 1.57E-09 8.68E-06 0.1500 2 8 15 1 70 15 25550 Fluoranthene 2.98E-07 1.75E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Fluoranthene 1.98E-06 7.68E-10 8.68E-06 0.6017 2 8	Pyrene Di-N-Butylobtbalata	1 115-05	1.41C-V7	0.000-00	1.2003	. 2	1 8	15		70	15	25550
Dencol APIG 4.562 6.500E-09 8.68E-06 4.2918 2 8 15 1 70 15 25550 p-cresol 6.75E-07 3.96E-10 8.68E-06 4.2918 2 8 15 1 70 15 25550 Diethylphthalate 2.18E-07 1.28E-10 8.68E-06 0.1100 2 8 15 1 70 15 25550 Fluoranthene 2.68E-06 1.57E-09 8.68E-06 0.1100 2 8 15 1 70 15 25550 Anthracene 2.99E-07 1.75E-10 8.68E-06 0.1500 2 8 15 1 70 15 25550 Fluoranthene 1.19E-06 7.01E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Fluoranthene 1.31E-06 7.68E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Renzo(a Janthracene 1.31E-06 7.68E-10 8.68E-06 0.6017 2 8 <td>Renzoic Acid</td> <td>1.11C-0J</td> <td>2 545-10</td> <td>1 8 68F-06</td> <td>0 2200</td> <td>1 2</td> <td>· 0</td> <td>1 15</td> <td>1 1</td> <td>, /0 ' 70</td> <td>1 15</td> <td>25550 1</td>	Renzoic Acid	1.11C-0J	2 545-10	1 8 68F-06	0 2200	1 2	· 0	1 15	1 1	, /0 ' 70	1 15	25550 1
p-cresol 6.75E-07 3.96E-10 8.68E-06 0.3400 2 8 15 1 70 15 25550 Diethylphthalate 2.18E-07 1.28E-10 8.68E-06 0.1100 2 8 15 1 70 15 25550 Fluoranthene 2.68E-06 1.57E-09 8.68E-06 1.3511 2 8 15 1 70 15 25550 Anthracene 2.99E-07 1.75E-10 8.68E-06 0.1500 2 8 15 1 70 15 25550 Fluoranthene 1.19E-06 7.01E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Fluoranthene 1.31E-06 7.68E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Naphthalene 1.31E-06 7.68E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Benzo(a)anthracene 8.55E-07 5.02E-10 8.68E-06 0.4311 2 8 </td <td>Phenol</td> <td>8.51E-06</td> <td>5.00E-09</td> <td>8.68F-06</td> <td>4 2918</td> <td>2</td> <td>! 8</td> <td>15</td> <td>1</td> <td>70</td> <td>15</td> <td>25550</td>	Phenol	8.51E-06	5.00E-09	8.68F-06	4 2918	2	! 8	15	1	70	15	25550
Diethylphthalate 2.18E-07 1.28E-10 8.68E-06 0.1100 2 8 15 1 70 15 25550 Fluoranthene 2.68E-06 1.57E-09 8.68E-06 1.3511 2 8 15 1 70 15 25550 Anthracene 2.96E-07 1.75E-10 8.68E-06 0.1500 2 8 15 1 70 15 25550 Fluoranthene 1.95-06 7.01E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Naphthalene 1.31E-06 7.68E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Benzo(a)anthracene 8.55E-07 5.02E-10 8.68E-06 0.4311 2 8 15 1 70 15 25550 Benzo(a)anthracene 4.76E-07 2.80E-10 8.68E-06 0.4302 2 8 15 1 70 15 25550 <	p-cresol	6.75E-07	3.96E-10	8.68E-06	0.3400	2	1 8	15	1	70	15	25550
Fluoranthene 2.68E-06 1.57E-09 8.68E-06 1.3511 2 8 15 1 70 15 25550 Anthracene 2.96E-07 1.75E-10 8.68E-06 0.1500 2 8 15 1 70 15 25550 Fluorene 1.19E-06 7.01E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Naphthalene 1.31E-06 7.68E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Benzo(a)anthracene 8.55E-07 5.02E-10 8.68E-06 0.4311 2 8 15 1 70 15 25550 Benzo(a)ayorene 4.76E-07 2.80E-10 8.68E-06 0.2400 2 8 15 1 70 15 25550 Chrysene 8.54E-07 5.01E-10 8.68E-06 0.4302 2 8 15 1 70 15 25550 Benzo(b) fluoranthene 1.36E-06 7.99E-10 8.68E-06 0.6856 2 <t< td=""><td>Diethylphthalate</td><td>2.18E-07</td><td>1.28E-10</td><td>8.68E-06</td><td>0.1100</td><td>2</td><td>8</td><td>15</td><td>1</td><td>70</td><td>15</td><td>25550</td></t<>	Diethylphthalate	2.18E-07	1.28E-10	8.68E-06	0.1100	2	8	15	1	70	15	25550
Anthracene 2.96E-07 1.75E-10 8.68E-06 0.1500 2 8 15 1 70 15 25550 Fluorene 1.19E-06 7.01E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Naphthalene 1.31E-06 7.68E-10 8.68E-06 0.6017 2 8 15 1 70 15 25550 Benzo(a)anthracene 8.55E-07 5.02E-10 8.68E-06 0.4311 2 8 15 1 70 15 25550 Benzo(a)ayorene 4.76E-07 2.80E-10 8.68E-06 0.2400 2 8 15 1 70 15 25550 Chrysene 8.54E-07 5.01E-10 8.68E-06 0.4302 2 8 15 1 70 15 25550 Benzo(b)fluoranthene 1.36E-06 7.99E-10 8.68E-06 0.6856 2 8 15 1 70 15 25550	Fluoranthene	2.68E-06	1.57E-09	8.68E-06	1.3511	2	8	15	1	70	15	25550
Fluorene 1.19E-06 7.01E-10 8.66E-06 0.6017 2 8 15 1 70 15 25550 Naphthalene 1.31E-06 7.68E-10 8.68E-06 0.6593 2 8 15 1 70 15 25550 Benzo(a)anthracene 8.55E-07 5.02E-10 8.68E-06 0.4311 2 8 15 1 70 15 25550 Benzo(a)ayorene 4.76E-07 2.80E-10 8.68E-06 0.2400 2 8 15 1 70 15 25550 Chrysene 8.54E-07 5.01E-10 8.68E-06 0.4302 2 8 15 1 70 15 25550 Benzo(a)fluoranthene 1.36E-06 7.99E-10 8.68E-06 0.6856 2 8 15 1 70 15 25550	Anthracene	2.98E-07	1.75E-10	8.68E-06	0.1500	2	8	15	1	70	15	25550
Naphthalene 1.31E-06 7.68E-10 8.68E-06 0.6593 2 8 15 1 70 15 25550 Benzo(a)anthracene 8.55E-07 5.02E-10 8.68E-06 0.4311 2 8 15 1 70 15 25550 Benzo(a)ayrene 4.76E-07 2.80E-10 8.68E-06 0.2400 2 8 15 1 70 15 25550 Chrysene 8.54E-07 5.01E-10 8.68E-06 0.4302 2 8 15 1 70 15 25550 Benzo(b)fluoranthene 1.36E-06 7.99E-10 8.68E-06 0.6856 2 8 15 1 70 15 25550	Fluorene	1.19E-06	7.01E-10	8.68E-06	0.6017	2	8	15	1	70	15	25550
Benzo(a)anthracene 8.55E-07 5.02E-10 8.68E-06 0.4311 2 8 15 1 70 15 25550 Benzo(a)pyrene 4.76E-07 2.80E-10 8.68E-06 0.2400 2 8 15 1 70 15 25550 Chrysene 8.54E-07 5.01E-10 8.68E-06 0.4302 2 8 15 1 70 15 25550 Benzo(b)fluoranthene 1.36E-06 7.99E-10 8.68E-06 0.6856 2 8 15 1 70 15 25550	Naphthalene	1.31E-06	7.68E-10	8.68E-06	0.6593	2	8	15	1	70	15	25550
Benzola joyrene 4.76E-07 2.80E-10 8.68E-06 0.2400 2 8 15 1 70 15 25550 Chrysene 8.54E-07 5.01E-10 8.68E-06 0.4302 2 8 15 1 70 15 25550 Benzolb /f luoranthene 1.36E-06 7.99E-10 8.68E-06 0.6856 2 8 15 1 70 15 25550	Benzo(a)anthracene	8.55E-07	5.02E-10	8.68E-06	0.4311	2	8	15	1	70	15	25550
Lnrysene 8.54E-07 5.01E-10 8.66E-06 0.4302 2 8 15 1 70 15 25550 Benzo(b)fluoranthene 1.36E-06 7.99E-10 8.66E-06 0.6856 2 8 15 1 70 15 25550	Benzo(a)pyrene	4.76E-07	2.80E-10	8.68E-06	0.2400	2	8	15	1	70	15	25550
jbenzo(D)/luoranthene 1.366-06 7.99E-10 8.68E-06 0.6856 2 8 15 1 70 15 25550	Chrysene	8.54E-07	5.01E-10	8.68E-06	0.4302	2	8	15	1	70	15	25550
	Benzo(D) Tuoranthene	1.36E-06	7.99E-10	8.68E-06	0.6856	2	8	15	1	70	15	25550
jbenzo(xjriuoraninene ; /./5k-0/; 4.55k-10; 8.68E-06; 0.3908; 2; 8; 15; 1; 70; 15; 25550;	; Benzol K JT LUOT ANT hene	7./SE-07	4.55E-10	8.68E-06	0.3908	2	; 8	15	$\begin{bmatrix} 1 \\ 1 \end{bmatrix}$; 70	15	25550

SUMMARY OF CANCER RISK ESTIMATES - ROAD IMPROVEMENTS AND CLEARING VEGETATION (ALTERNATIVES 2, 2A, 4 AND 4A)

		1	HERTEL LANDFIL	L FEASIBIL	ITY STUDY			
		*****						1 5 1 (7 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
	CHRONIC DAILY	CDI					CHENICAL	IOTAL ; IOTA
CHEMICAL	INTAKE(CDI)	ADJUSTED FOR	SF	WEIGHT OF	TYPE OF	SF BASIS/	SPECIFIC	PATHWAY EXPOS
	(mg/kg/day)	ABSORPTION	(mg/kg/day)-1	EVIDENCE	CANCER	SOURCE	RISK	RISK RIS
		111111111111	11111111111					
								1
EXPOSURE PATHWAY: INNALAT	10N OF ALRBORNE	CHEMICALS AD	SORBED TO DUS	I				1E-06 2E-
INDRGANICS	1			1				
Arsenic	8.04E-09	No	50	A	Lung	AIT/HEAST	4E-07	
Beryllium	9.32E-10	Ko	8.4				8E-09	
Cadnium	9.328-10	No	6.1	81		OCCUPA/HEAST	6E-09	
Chronium	2.13E-08	No	41	A	Lung	OCCUPA/HEAST	9E-07	
Nickel	2.46E-08	No	8,4E-01	A	Respiratory tract	OCCUPA/HEAST	2E-08	
								1
SEMIVOLATILES		:		1	1	i 54		
Bis(2ethylhexyl)phthalati	e¦ 7.44E-09	No	1.4E-02	1		1	18-10	1
Butylbenzylphthalate	NA	No	ND	C C	· ·	NF/IRIS	NA	1
	Ì		1	1		1	1	
8enzo(a)anthracene	5.02E-10	No	6.1	82		NA/IRIS	3E-09	1
Benzo(a)pyrene	2.80E-10	No	6.1	B2	Stomach	Diet/IRIS	2E-09	1
Chrysene	5.018-10	No	6.1	82	Respiratory tract	NA/1RIS	3E-09	
Benzo(b)fluoranthene	7.99E-10	No	6.1	82		NA/IRIS	5E-09	
Benzo(k)fluoranthene	4.S5E-10	No	6.1	B2		NA/IRIS	3E-09	F
Dibenzo(a,h)anthracene	NA	No	6.1	82		NA/IRIS	NA NA	-
Indeno(123cd)pyrene	NA	No	6.1	B2		NA/IRIS	NA	•
	1	1	1	1	1	1	:	1
1,4-Dichlorobenzene	NA	l No	2.4E-02	1		1	NA	1
							·¦	1
PESTICIDES/PCBS	1	1	1		1	1	1	1
4,4'-DDD	NA	No	2.4E-01	82	Lung, Liver, Thyroid	Diet/IRIS	NA	
4,4'-ODE	NA	No	3.4E-01	B2	Liver, Thyroid	Diet/IRIS	NA	1
4,4'-DDT	NA	No	3.4E-01	82	; Liver	Diet/IRIS	NA	1
			1111111111111111	* * * * * * * * * * * *		1		

APPENDIX B

SUMMARY OF CANCER RISK	ESTIMATES - RO	AD IMPROVEMEN	TS AND CLEARIN	IG VEGETATI	ON (ALTERNATIVES 2, 2A	, 4 AND 4A)		
			HERIEL LANDFI	L FEASIBIL	ITY STUDY			
		4144111114491	7			.		
	CHRONIC DAILY	CD1				1 I 1 I	CHEMICAL	101
CHENICAL	INTAKE(CDI)	AOJUSTED FOR	SF	WEIGHT OF	TYPE OF .	SF BASIS/	SPECIFIC ;	PATH
1	(mg/kg/day) ¦	ABSORPTION	(mg/kg/day)-1	EVIDENCE	CANCER	SOURCE	RISK	R
8					1 1 1 1 1 1 1 1 1 1	1		
EXPOSURE PATHWAY: DERMAL (CONTACT WITH CH	HEHICALS IN SC	IL					9
\$ 1								
INORGANICS				: :	1	1 1		
Arsenic	5.29E-08	0.10	1.75	A	Skin	Water/IRIS	9E-07	
SEKIVOLATILES								
Bis(2ethylhexyl)phthalate	4.89E-08	0.10	1.4E-02	B2	Liver	Diet/IRIS	7E-09	
Butylbenzylphthalate	NA	0.10	NO	C		NF/IRIS	NA	
Benzo(a)anthracene	3.30E-09	0.10	ND	82	1	NA/IRIS	NA	1
Benzo(a)pyrene	1.84E-09	0.10	ND	B2	Stomach	Diet/IRTS	NA	í.
Chrysene	3.30E-09	0.10	ND	82	1	NA/IRIS	NA	1
Benzo(b)fluoranthene	5.25E-09	0.10	ND	B2		NA/IRIS	NA	i i
Benzo(k)fluoranthene	2.99E-09	0.10	ND	82	!	NA/IR1S	NA	1
Dibenzo(a,h)anthracene	NA	0.10	ND	B2	1	NA/IRIS	NA	1
Indeno(123cd)pyrene	NA	0.10	ND	B2	1	NA/IRIS	NA	:
11	1		1 1	1		1	1	!
1,4-Dichlorobenzene	NA	0.10	2.4E-02	82	Liver	¦Gavage/HEAST	4	
PESTICIDES/PCBS		!	!	!	1	!		i t
4.4'-DDD	NA	0.10	2.4E-01	B2	Lung, Liver, Thyroid	Diet/IRIS	NA	
4,4'-DDE	NA	0.10	3.4E-01	82	Liver Thyroid	Diet/IRIS	NA	i
4.4'-DDT	NA	0.10	3.4E-01	B2	Liver	Diet/IRIS	NA	
	111111111111	immin		innin	in non in non no		in num	i

.

SUNVARY OF CANCER RISK ESTIMATES - ROAD IMPROVEMENTS AND CLEARING VEGETATION (ALTERNATIVES 2, 2A, 4 AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

		111111111111			111111111111111111111111111111111111111	11111111111	11111111111	
1)	CHRONIC DAILY	CDI					CHE HICAL	101AL
CHEMICAL	INTAKE(CD1)	ADJUSTED FOR	SF	WE LOUIT OF	TYPE OF	SF BASIS/	SPECIFIC	PATHWAY
11	(mg/kg/day)	ABSORPTION	(mg/kg/day)-1	EVIDENCE	CANCER	SOURCE	RISK	RISK
L'EXPOSURE PATHWAY: INGESTI	ON OF CHEMICAL	5 IN SOIL						3E-08
			111111111111					
!! INORGANICS	1							
Arsenic	5.79E-09	No	1.75	A	Skin	Water/IRIS	1E-08	
Beryllium	6.71E-10	No	4.3	B2	Skin	Water/IRIS	3E-09	
Lead	5.04E-08	No	CH1	B2	Renal	Oral/IRIS	NA	
!!								
SENIVOLATILES		!!!						į
Bis(2ethylhexy) whthalate	5.36E-09	No	1.4E-02	B2	Liver	Diet/1Ri5	7E-11	
Butylbenzylphthalate		No	ND	C		NF/IRIS	NA	
11								
Benzo(a)anthracene	3,62E-10	No	11.50	B2		NA/IRIS	4E-09	
Benzo(a)pyrene	2.01E-10	No	11.50	B2	Stonach	Diet/1RIS	2E-09	
Chrysene	3.61E-10	No	11.50	B2		NA/IRIS	4E-09	
Benzo(b)fluoranthene	5.75E-10	No	11.50	82	1	NA/IRIS	7E-09	
Benzo(k)fluoranthene	3.28E-10	No	11.50	82	1	NA/IRIS	4E-09	
Dibenzo(a,h)anthracene	NA	No	11.50	B2		NA/IRIS	NA	i i
Indeno(123cd)pyrene	NA	No	11.50	82		NA/IRIS	NA	Ì
	1	1						1
1,4-Dichlorobenzene	NA	No	2.4E-02	B2	Liver	Gavage/HEAST	NA	i
11								i i
PESTICIDES/PCBS	1	1	1	;		1		į
4,4'-DDD	NA	No	2.4E-01	82	Lung, Liver, Thyroid	Diet/1R1S	NA	1
4,4'-DDE	NA	No	3.4E-01	B2	Liver, Thyroid	Diet/IRIS	NA	1
4,4°-001	¦ NA	No	3.4E-01	B2	Liver	Diet/1815	NA	
							1010000	1

.

SUMMARY OF CHRONIC HAZARD INDEX ESTIMATES - ROAD IMPROVEMENTS AND CLEARING VEGETATION (ALTERNATIVES 2, 2A, 4 AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

		SUBCHRONIC	CDI				RFD	RFD			Pathnay
		(mo/ko/day)	ADJUSTED FUR	i NFU { no/ko/day }	LONG TOENCE	EFFECT	SUURLE/			HAZARD	HAZARD ;;
1	1 1 1 2 3 1 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1 (mg/ kg/ ud/)		((mg/ kg/ 00/)			61CHG 1				10057 (UT)**
i	EXPOSURE PATHNAY: INGESTIC	ON OF CHEMICAL	S IN SOIL				,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,				1E-01
- 1	INORGANICS										
į	Alumanum	1.9E-02	NO NO	DI	l au	taxamity bland alware and				NA) 2 1
1	i i i i i i i i i i i i i i i i i i i	1 3.3L-03	1 10	46-04		chalesteral	6 1			8C-02	1
	Arsenic	9.98-06	No	1E-03	NA	Keratosis and hyperpignentation				1E-02	
Ì	Barius	1.7E-04	No	5E-02	Medium	Increased blood pressure				3E-03	
-	Beryllium	1.18-06	No	5E-03	LOW	None observed				2E-04	
į	Cadmium	1.1E-06	No	ND	, Kigh	Proteinuria				NA	
n¦	Chronium VI	2.6E-05	. No	2F-02	l Iow	No effects reported				15-02	
	Cobalt	1.8E-05	No	NF						NA	
1	Copper	5.2E-05	No	4.0E-02	NA	Local GI irritation				1E-03	
	Lead	8.6E-05	No	NA	NA	Neurobehaioral effects	1			NA	1
Pi	Manganese Mercury	; 1.8E-03	i No	1E-01	Medium	CNS effects				2E-02	
1	Nickel	3.0E-05	No	2E-02	. Medium	Decreased body and organ weight				NA 25-03	1
į	Selenium	NA	No	NF			1			NA	1
	Vanadium	2.4E-05	No	7E-03	, NA	None observed				3E-03	
	Zinc	1.98-04	No	2E-01	NA	Anemia				1E-03	1
-	Cyanice	i NA	NO	2E-02	Medium	Weight Loss, thyroid effects,				NA	1
			1 				•				
I.	SENIVOLATILES						:				1
		ł	•	1			1			NA	1
i	Bis(2ethylhexyl)phthalate	9.1E-06	No	2E-02	Medium	Increased relative liver weight	Diet/IRIS	1000.00	1.00	5E-04	
1	Bucyibenzyiphchatace	3.4L-U/	I NO	22+00	LOW	i Effects on body; weight gain, testes liver kidnevs	Diet/IRIS	100.00	1.00	2E-07	,
,	Benzo(a)anthracene	6.2E-07	No	NF			HEAST, IRIS			NA	
1	Benzo(a)pyrene	3.4E-07	No	NF			HEAST, IRIS			NA	
ł	Chrysene*	6.1E-07	No	DI			HEAST			NA	
i	Benzo(b) fluoranthene	5.6E-07	NO No	i N7 1 N F	1		HEAST, IRIS			NA .	1
	Dibenzo(a,h)anthracene	NA NA	No	NF			HEAST, IRIS			NA	
1	Indeno(1,2,3-c,d)pyrene	NA	No	NF	1		HEAST, IRIS			NA	
1			 No	1 1 1							1
	1,4-Dichiorobenzene	NB NB	NO	, NU	1		HEAST, IRIS			NA	,
i	Diethylphthalate	1.6E-07	No	8E+00	LOW	Decreased growth rate, food	Diet/IR1S	1000.00	1.00	2E-08	4 1 6
-	4			5		consumption rate, altered					
j	i Di a hutulahthalata			15.00		organ weights					(] {
•	Di-n-octylphthalate	1 NH 2 9F-07	I NO	1E+00	LOW NA	Increased mortality Flevated liver and kidney weights	UIEL/IKIS	1000.00	1.00	N9 15-05	
						increased SGOT and SPGT	I DICUTERST	1000.00		12-03	
1		1		1			1			NA	1
-	Acenaphthene	NA	No	62-01	Low	Hepatotoxicity	Oral/IRIS			NA	
	Anthracene Benzo(g.h.i)oervlene	2.1E-07	NO NG	3E100	Low	No observed effects	Gavage/IRI5			7E-08	
ì	Fluoranthene	1.9E-06	No	4E-01	Low	Neuropathy, liver weight changes	Gavage/IRIS			5E-06	
-			1	1		hematological alterations and					
	Eluorene	P 45-07	No	45-01	1.011	clinical effects	i como itore			NA	
	1 1001.616	0.00-07		42-01	LOW	volume and hemoolchin	Gavage/IRIS				
1	p-Cresol	9.4E-07	No	NF			HEAST, IRIS			NA	
1	Naphthalene	9.4E-07	No	4E-02	NA	Ocular and internal lesions	Diet/HEAST			2E-05	
1	Phenanthrene	1./E-06 1.2E-04	NO No	31:-01	LOW	Kidney effects	Gavage/IRIS			6E-06	
	Phenol	6.12-06	No	0.60		Reduced fetal body weight	Gavage/HEAST	100	i	NA	
1											
,	PESTICIDES/PCBS	NA							1		
i	4,4 -DUU	NA NA	NO No	NF NF			HEAST, IRIS			NA	i.
	4,4'-00T	NA	No	5E-04	Medium	Liver lesions	Diet/IRIS	100	1 00	NA	
Ì											1

SUMMARY OF CHRONIC HAZARD INDEX ESTIMATES - ROAD IMPROVEMENTS AND CLEARING VEGETATION (ALTERNATIVES 2, 2A, 4AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

1.940-0000 0.900 1.900 1.900 1.900 1.900 1.900 1.900-000											
UPENIX UMENIX UPENIX UPENIX<		SUBLIKUTIL		000	CONCIDENCE	011104	FFU i	KEU (100-15-05-001		PAIHWAT
Construction Construction<	ii UNENIUAL	DAILE INIAKE	AUJUSILU FOR	NTU j	LURP THENCE		SUURLE/	UNCERTAINIT	HOUTETING;	HAZARD	HALARD ;;
LEMODOLE PARLAN: OVERATION OF MALLAN: OVERATION OF	14 11 (11 11 11 11 11 11 11	(mg/kg/day)	ABSURPTION 1	(109/x9/0ay)	111111111111		Rep12	RUJUSINENIS	FACTORS ;	QUOLIERI ;	IRUEX (HI);;
Zeroser 9.62-05 0.10 IE-00 NA Keratosis and hyperpigentation NA/(EAS) 1 92-01 INSOGANCS 9.62-05 0.10 IE-00 NA Keratosis and hyperpigentation NA/(EAS) 1 92-01 SENVOLATILES	HELDER DATE AND DEDUAL										
INKKAWLS Arsenic 9.62-65 0.10 IE-0 NA Keratosis and hyperpigentation NA/LEAST 1 92-01 SERVIGUALILES SERVIGUALILES Bis/2et/Viperpibilibalize B.22-05 O.10 ZE-00 Hedian Increased relative liver wight Effects on body: wight gain, bis/Viperpibilibalize 1000 1 4E-01 Bis/2et/Viperpibilibalize B.22-05 O.10 NF Effects on body: wight gain, bis/Viperpibilibalize Dis/VIPES 1000 1 4E-01 Benold Janthracene Benold Janthracen S.62-06 O.10 NF HEAST, RISS NA Clower S.1E-06 O.10 NF HEAST, RISS NA Discredu Jiponathene Benold Vipotathene Nichologenzene NA O.10 NF HEAST, RISS NA In-butyiphthalate 1.4E-06 O.10 NF Excertailor of any wights onsamption rate, food consamption rate, aiterd organ wights increased Statility Dist//RIS 1000 NA Distrovipiphthalate NA O.10	PERFOSORE PHINWRI - DERNHE (100104031030	/1L 							· 9E-01 ;
Instantics 9.62-05 0.10 II-00 NA Retactors and hyperpresentation NA/LENST 1 95-01 SENVOLATILES Senvolatiles 0.10 22-00 Median Increased relative liver weight Diet/IRIS 1000 1 4E-02 Bistylbenrylphthalate 3.1E-06 0.10 22-00 Low Effects on body: weight gain, tests, IKIS Diet/IRIS 1000 1 22-05 Benck of low relative S.6E-06 0.10 NF Effects on body: weight gain, tests, IKIS NA NA Benck of low relative S.6E-06 0.10 NF KEAST, IKIS NA Benck of low relative S.7E-06 0.10 NF KEAST, IKIS NA Benck of low relative S.7E-06 0.10 NF KEAST, IKIS NA Index (1, 2, 3-c, d) prese NA 0.10 NF KEAST, IKIS NA Index (1, 2, 3-c, d) prese NA 0.10 NF KEAST, IKIS NA Index (1, 2, 3-c, d) preter NA 0.10 K		1 8 8 8 8 8 8 8 1 8 1 8 1 8 1 8 1 8 1 8	*****			*****					nemun
Anstruct File of a Out It of a Metabols and myerygenetic (0) Multisity It Ye of a Sendout Jack Sendout (11) Sen		0.05-05	0.10	15-02	NA	Koratoric and humanniamontation	NAUSACT			05 01	1
SENIVULATILES Johnstate 8.34-05 0.10 22-02 Predian Increased relative liver weight Effects on body, weight spin, testes, liver, kidneys Diet/IRIS 1000 1 42-02 Benold Jahrhacene 5.45-06 0.10 NF Effects on body, weight spin, testes, liver, kidneys Diet/IRIS 100 1 42-02 Benold Jahrhacene 5.45-06 0.10 NF HA NA NA NA Chrysne 5.45-06 0.10 NF HA NA NA NA Chrysne 8.45-06 0.10 NF HA NA	11HC5001C	9.02-05	, 0.10	, IE-03 ,	пн	Netatosis and hyperpigmentation ,	101/12/051	1	· i	92-01	
Bis(2eth/hey/by/bhthalate B.3C-05 0.10 ZE-02 Hedium Increased relative liver weight bow Diet/IRIS 1000 1 4E-02 Bis(2eth/hey/bhthalate J.E-06 0.10 NF Low Effects on body, weight gain, testes, liver, kidneys Diet/IRIS 1000 1 2E-02 Bennols/anthracene S.E-06 0.10 NF Hest, IRIS NA Bennols/barratene S.E-06 0.10 NF Hest, IRIS NA Bennols/Houranthene S.E-06 0.10 NF Hest, IRIS NA Bennols/Houranthene S.E-06 0.10 NF Hest, IRIS NA Ibennols/Houranthene S.E-06 0.10 NF Hest, IRIS NA I.4-0ichlorobenzene NA 0.10 NF Hest, IRIS NA I.4-20/Houranthene NA 0.10 Effects Increased growth rate, food consymmetry weights Diet/IRIS 1000 1 Zt-06 I.4-20/Houranthene NA 0.10 Effects Increased growth rate	SENIVOLATILES	1	!	!							1
Dist Dist <thdis< th=""> Dist Dist D</thdis<>											
Butylbenzylphthalate 3.1E-06 0.10 ZE 100 Low Effects on body; weight gsin, testes, liver, kidneys Diet/[R15 100 1 ZE-05 Benzola Jantracene 3.6E-06 0.10 NF HEAST, IRIS HA Chryssene 3.1E-06 0.10 NF HEAST, IRIS HA Benzola Jihuwanthene 5.4E-06 0.10 NF HEAST, IRIS HA Benzola Jihuwanthene 5.1E-06 0.10 NF HEAST, IRIS HA Benzola Jihuwanthene 5.1E-06 0.10 NF HEAST, IRIS HA Ibenzola Jihuwanthene 5.1E-06 0.10 NF HEAST, IRIS HA Idenxola Jihuwanthene 1.4E-06 0.10 NF HEAST, IRIS HA Idenxola Jihuwanthene 1.4E-06 0.10 RE Decreased growth rate, food consemption rate, altered organ weights Diet/IRIS IO I Identifylphthalate 1.4E-06 0.10 ZE-02 M Heatotoricity Diet/IRIS IO I IC-03 <	Bis(2ethylhexyl)phthalate	8.3E-05	0.10	2E-02	Medium	Increased relative liver weight	Diet/IR1S	1000	1	45-02	
Benzol a Janthracene Benzol A Janthracene Stef-66 0.10 NF testes, liver, kidneys HEAST, IRIS NA Benzol A Jayrene 3.11-06 0.10 NF HEAST, IRIS NA Idenzol A Jayrene 5.61-06 0.10 NF HEAST, IRIS NA Benzol M Jouranthene 5.61-06 0.10 NF HEAST, IRIS NA Benzol M Jouranthene 5.16-06 0.10 NF HEAST, IRIS NA Ibiterize A, Jhanthracene NA 0.10 NF HEAST, IRIS NA Ibiterize A, Jhanthracene NA 0.10 NF HEAST, IRIS NA Ibiterize A, Jhanthracene NA 0.10 BE 100 Low Decreased growth rate, food consumption rate, altered organ weights Diet/IRIS IOCO I Ibiterize A, Ibiteree NA 0.10 EE 00 Low Hepatoxicity Diet/IRIS IOCO I EE 00 Ibiterize A, Ibiteree NA 0.10 66-01 Low Hepatoxicity Diet/IRIS IOCO IE 0	Butylbenzylphthalate	3.1E-06	0.10	2E+00	LOW	Effects on body: weight gain.	Diet/IRIS	100	1	2E-05	
Benzol a Janthracene S.4E-66 0.10 NF HEAST, IRIS HA Chryssene S.E-66 0.10 NF HEAST, IRIS NA Denzol (J) Juoranthene B.YE-66 0.10 NF HEAST, IRIS NA Dibenzol (J) Juoranthene B.YE-66 0.10 NF HEAST, IRIS NA Dibenzol (J) Juoranthene S.E-66 0.10 NF HEAST, IRIS NA Dibenzol (J) Juoranthene S.E-66 0.10 NF HEAST, IRIS NA Dibenzol (J) Juoranthene NA 0.10 NF HEAST, IRIS NA I.1.4-Dichlorobenzene NA 0.10 NC Decreased growth rate, food Diet/IRIS 1060 1 22-66 Di-m-butylphthalate NA 0.10 EE-60 Low Increased growth rate, food Diet/IRIS 1060 1 22-66 Stringtone NA 0.10 EE-60 Low Increased growth rate, food Diet/IRIS NA Ni-mextriphthalate NA <t< td=""><td>11</td><td></td><td></td><td></td><td></td><td>testes, liver, kidneys</td><td></td><td></td><td>-</td><td></td><td>!</td></t<>	11					testes, liver, kidneys			-		!
Hencold byrene 3.1E-05 0.00 NF HEAST, IRIS NA Ichrystene 5.6E-06 0.10 NF HEAST, IRIS NA Benzold // Ulouranthene 5.1E-05 0.10 NF HEAST, IRIS NA Benzold // Vilouranthene 5.1E-06 0.10 NF HEAST, IRIS NA Indeno(1,2,3-c,d)pyrene NA 0.10 NF HEAST, IRIS NA Indeno(1,2,3-c,d)pyrene NA 0.10 NF HEAST, IRIS NA Interhylphthalate 1.4E-06 0.10 RE Decreased growth rate, food consumption rate, altered organ weights Diet/IRIS 1000 1 2E-06 Di-n-butylphthalate NA 0.10 IEF00 Low Decreased straitity Diet/IRIS 1000 1 MA In-restriphthalate NA 0.10 IEF00 Low Hepatotoxicity Oil/IRIS 1060 1 E-03 Interace 2.06-06 0.10 XF00 Low Hepatotoxicity Oral/IRIS <	Benzo(a)anthracene	S.6E-06	0.10	NF			HEAST, IRIS			NA	
Ichrsee 5.62-06 0.10 DI Heast Heast Heast Benzold Filouranthene 9.96-06 0.10 NF HEAST, IRIS NA Dibenzold Filouranthene NA 0.10 NF HEAST, IRIS NA Dibenzold Filouranthene NA 0.10 NF HEAST, IRIS NA Lodenol 1.2, 3-c, d) pyrene NA 0.10 NF HEAST, IRIS NA 1, 4-01chlorobenzene NA 0.10 NF HEAST, IRIS NA Dichwidylphthalate 1.4E-06 0.10 BE+00 Low Decreased growth rate, food consignuin rate, altered or gran wights Diet/IRIS 1000 1 MA Di-m-butylphthalate NA 0.10 EF-00 Increased Sol and SFGI Diet/IRIS 1000 1 MA Ini-m-butylphthalate NA 0.10 GE+01 Low Hepatotoxicity Diet/IRIS IAG Inthraceae 2.06-06 0.10 K+00 Low Hepatotoxicity Oral/IRIS HA <td>Benzo(a)pyrene</td> <td>3.1E-06</td> <td>0.10</td> <td>NF</td> <td></td> <td></td> <td>HEAST, IRIS</td> <td>i</td> <td></td> <td>NA</td> <td></td>	Benzo(a)pyrene	3.1E-06	0.10	NF			HEAST, IRIS	i		NA	
Benzo(b)/Juoranthene 8.95-06 0.10 NF HEAST, IRIS HA Denzo(b)/Juoranthene 5.10-06 0.10 NF HEAST, IRIS NA Diebnzo(s, J)Amtracene NA 0.10 NF HEAST, IRIS NA 1.4-Dichlorobenzeno NA 0.10 NF HEAST, IRIS NA Diethylphthalate 1.4E-06 0.10 NF HEAST, IRIS INA Diethylphthalate 1.4E-06 0.10 BE:00 Low Decreased growth rate, food consumption rate, altered organ weights Diet//RIS 1000 1 Di-m-butylphthalate 2.6E-06 0.10 2E-02 NM Elevated liver and kidney weights Diet//RIS 1000 1 MA Di-motylphthalate 2.6E-06 0.10 2E-02 NM Elevated liver and kidney weights Diet//RIS 1060 1E-03 Seenaphthene NA 0.10 6E-01 Low Heaptotxicity Oral/RIS Gavage/RIS 7E-06 Isoranthene 1.8E-05 0.10 3E	Chrysene	5.62-06	0.10	DI			HEAST			NA	
Benzol, K/ Huoranthene 5, IE-06 0.10 NF HA Diberol, K/ Huoranthene NA 0.10 NF HEAST, IRIS HA Indenol, 2, Jord, byprene NA 0.10 NF HEAST, IRIS HA I.4-Dichlorobenzene NA 0.10 NF HEAST, IRIS HA I.4-Dichlorobenzene NA 0.10 NF HEAST, IRIS HA Ibiethylphthalate 1.4E-06 0.10 BE+00 Low Decreased growth rate, food consumption rate, altered organ weights Diet/IRIS 1000 1 Ibi-m-butylphthalate NA 0.10 IE+00 Low Increased social and SFGI Diet/IRIS 1000 1 Ibi-motylphthalate NA 0.10 GE+01 Low Hepatotoxicity Diet/IRIS 10600 1 HA Iberugo, J., jperylene NA 0.10 GE+01 Low Hepatotoxicity Oral/IRIS HA Iberugo, J., jperylene NA 0.10 GE+01 Low Heuropathy, liver weighth tenotosi	Benzo(b)fluoranthene	8.9E-06	0.10	NF	1		HEAST, IRIS		1	NA	1
Dibenzo(a,h)anthracene NA 0.10 NF HEAST, IRIS NA 1.4-Dichlorobenzene NA 0.10 NF HEAST, IRIS HA 1.4-Dichlorobenzene NA 0.10 NO HEAST, IRIS HA 1.4-Dichlorobenzene NA 0.10 NO HEAST, IRIS NA Diethylphthalate 1.4E-06 0.10 8E+00 Low Decreased growth rate, food consumption rate, altered organ weights Diet/IRIS 1000 1 2E-06 Toi-m-butylphthalate 2.6E-06 0.10 2E-02 Nh Elevated liver and kidney weights increased S001 and SPG1 Diet/IRIS 1000 1 NA Nathracene 2.0E-06 0.10 2E-02 Nh Elevated liver and kidney weights Diet/IRIS 1000 1 NA Nathracene NA 0.10 4E-01 Low Hepatotoxicity Oral/IRIS NA Fluoranthene 1.0E-05 0.10 4E-01 Low Heuropathy, liver weight changes Gavage/IRIS 2E-04	Benzo(k)fluoranthene	5.1E-06	0.10	NF	1		HEAST, IR1S			NA	
Indexe(1,2,3-c,d)pyrene NA 0.10 NF HEAST, IRIS HEA 1,4-Dichlorobenzene NA 0.10 NO HEAST, IRIS HA 1,4-Dichlorobenzene NA 0.10 BE 400 Low Decreased growth rate, food consumption rate, altered organ weights Diet/IRIS 1000 1 2E-06 10-im-butylphthalate NA 0.10 IE 400 Low Decreased growth rate, food consumption rate, altered organ weights Diet/IRIS 1000 1 HA 10-im-butylphthalate NA 0.10 IE 400 Low Increased potions Diet/IRIS 1000 1 HA 10-im-outylphthalate 2.6E-06 0.10 2E-02 NA Elevated Uver and kidney weights Diet/IRIS 1000 1 HA Anthracene 2.0E-06 0.10 3E+00 Low Hepatological alterations and clinical effects Gavage/IRIS ZE-04 Fluorenc 7.9E-06 0.10 SE-01 Hedium Decreased body weight, Hourotoxicity IRIS 1000 1 9E-05	Dibenzo(a,h)anthracene	NA	0.10	NF	1		HEAST, IRIS			NA	
NA 0.10 NO PEAST, IRIS NA Diethylphthalate 1.4E-06 0.10 8E400 Low Decreased growth rate, food consumption rate, altered organ weights Diet/IRIS 10000 1 2E-06 Diethylphthalate NA 0.10 1E400 Low Increased portality Diet/IRIS 10000 1 PA Dien-butylphthalate 2.6E-06 0.10 2E-02 NA Elevated liver and Kidney weights Diet/IRIS 1000 1 PA Acenaphthene NA 0.10 6E-01 Low Hepatotoxicity Oral/IRIS HA Inthracene 2.0E-06 0.10 3E+00 Low Hepatotoxicity Oral/IRIS HA Fluorantene 1.8E-05 0.10 4E-01 Low Heuropathy, liver weight changes Gavage/IRIS 4E-04 Fluorene 7.9E-06 0.10 4E-01 Low No Ocular and internal lesions Diet/FRIST 2E-04 Preceol 4.4E-06 0.10 3E-01 Mediun	Indeno(1,2,3-c,d)pyrene	NA	0.10	NF	1		HEAST, IRIS			NA	
II.4-Dichlorobenzene NA 0.10 NO HA Diethylphthalate 1.4E-06 0.10 BE+00 Low Decreased growth rate, food orsam weights Diet/IRIS 1060 1 2E-06 Diethylphthalate NA 0.10 IE+00 Low Decreased growth rate, food orsam weights Diet/IRIS 1060 1 NA Diethylphthalate 2.6E-06 0.10 2E-02 Nh Elevated liver and kidney weights increased SSOT and SPGT Diet/IRIS 1060 1 NA Acenaphthene NA 0.10 6E-01 Low Hepatotoxicity Oral/IRIS NA NA Betato(g.h.i)perylene NA 0.10 3E-00 Low Hepatotoxicity Oral/IRIS NA Fluoranthene 1.8E-05 0.10 3E-01 Low Heuropathy, liver weight changes Gavage/IRIS 2E-04 IP-Crecol 4.4E-06 0.10 SE-01 Hedium Decreased RSC, packed cell volume and heoglobin Gavage/IRIS 2E-04 Pyrene 1.6E-05 0.10 SE-01 Hedium Decreased Rody weight, Reurotoxicity IRIS	11	1	1				1				
Diethylphthalate 1.4E-06 0.10 BE+00 Low Decreased growth rate, food consumption rate, altered organ weights Diet/IRIS 1000 1 2E-06 Di-n-butylphthalate 2.6E-06 0.10 2E-02 NN Elevated liver and kidney weights increased SSOI and SPGI Diet/IRIS 1000 1 NA Acenaphthene NA 0.10 6E-01 Low Hepatotoxicity NN Oral/IRIS NA NA Acenaphthene NA 0.10 6E-01 Low Hepatotoxicity NN Oral/IRIS NA Inthracene 2.06-06 0.10 3E+00 Low Hepatotoxicity No observed effects Gavapr/IRIS 7E-06 Benzol(g,h,i)perylene NA 0.10 4E-01 Low Heuropathy, liver weight changes Gavage/IRIS 4E-04 Fluorene 7.9E-06 0.10 4E-01 Low Decreased RBC, packed cell Gavage/IRIS 2E-03 Pyrene 1.6E-05 0.10 3E-01 Hedium Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05 Pyrene 1.6E-05 0.10 3E-01	1,4-Dichlorobenzene	NA	0.10	NO	i i		HEAST IRIS			NA	•
Diethylphthalate 1.4E-06 0.10 8E+00 Low Decreased growth rate, food consumption rate, altered organ weights Diet/IRIS 1000 1 2E-06 [0i-n-butylphthalate NA 0.10 1E+00 Low Increased bortality Diet/IRIS 1000 1 MA [0i-n-butylphthalate 2.6E-06 0.10 2E-02 NA Elevated liver and kidney weights increased SG01 and SFG1 Diet/IRIS 1000 1 MA [Acenaphthene NA 0.10 6E-01 Low Hepatotoxicity Oral/IRIS NA [Fluoranthene NA 0.10 8E+00 Low Hepatotoxicity Oral/IRIS NA [Fluoranthene NA 0.10 8E+00 Low Hepatotoxicity Oral/IRIS NA [Fluoranthene 1.8E-05 0.10 4E-01 Low No observed effects Gavage/IRIS 2E-04 [Fluorene 7.9E-06 0.10 4E-01 Low Decreased RSC, packed cell Gavage/IRIS 2E-03 [Pyrene		1	1	1		1	1		i		
Di-n-butylphthalate NA 0.10 IE+00 Low Increased sortality Diet/IRIS 1000 1 MA Di-n-butylphthalate 2.6E-06 0.10 2E-02 NA Elevated liver and kidney weights Diet/IRIS 1000 1 MA Acenaphthene NA 0.10 2E-02 NA Elevated liver and kidney weights Diet/IRIS 1060 1 MA Acenaphthene NA 0.10 3E+00 Low Hepatotoxicity Oral/IRIS NA Anthracene 2.0E-06 0.10 3E+00 Low Heuropathy, liver weight changes Gavage/IRIS 7E-06 Benzols,h,i)perylene NA 0.10 4E-01 Low Heuropathy, liver weight changes Gavage/IRIS 4E-04 Fluorene 7.9E-06 0.10 4E-01 Low Decreased RBC, packed cell Gavage/IRIS 2E-04 P-Cresol 4.4E-06 0.10 SE-01 Hedium Decreased body weight Gavage/IRIS 5E-04 Presol 1.6E-05	Diethylphthalate	1.4E-06	0.10	8E+00	Low	Decreased growth rate, food	Diet/JRIS	1000	1	2E-06	
Di-n-butylphthalate NA 0.10 IE+00 Low Increased sortality Diet/IRIS IGO I MA Di-n-outylphthalate 2.6E-06 0.10 2E-02 NA Elevated liver and kidney weights Diet/IRIS IGO I MA Acenaphthene NA 0.10 6E-01 Low Hepatotxicity Oral/IRIS IGO IE-03 Anthracene 2.0E-06 0.10 3E+00 Low Hepatotxicity Oral/IRIS NA Fluoranthene NA 0.10 AE-01 Low Heuropathy, liver weight changes Gavage/IRIS 2E-04 Fluoranthene 1.8E-05 0.10 4E-01 Low Heuropathy, liver weight changes Gavage/IRIS 2E-04 Profesol 4.4E-06 0.10 4E-01 Low Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05 Profesol 4.4E-06 0.10 3E-01 Hed Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05	11	1				consumption rate, altered	1				
IDI-n-butylphthalate NA 0.10 IE+00 Low Increased mortality Diet/IFIS 1000 1 HA IDI-n-octylphthalate 2.6E-06 0.10 2E-02 NA Elevated liver and kidney weights Diet/IEIS 1000 1 HA Acenaphthene NA 0.10 6E-01 Low Hepatotoxicity Oral/IRIS NA Anthracene 2.0E-06 0.10 3E+00 Low Hepatotoxicity Oral/IRIS NA Benzolg.h.iperylone NA 0.10 NF Heuropathy, liver weight changes Gavage//RIS 7E-06 IP-oranthene 1.8E-05 0.10 4E-01 Low Heuropathy, liver weight changes Gavage//RIS 4E-04 Fluoranthene 1.8E-05 0.10 4E-01 Low Decreased REC, packed cell Gavage//RIS 2E-04 P-Cresol 4.4E-06 0.10 5E-01 Medium Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05 Pyrene 1.6E-05 0.10		1	;	1	8	organ weights	1	1	i i		1
Dinnoctylphthalate 2.6E-06 0.10 2E-02 NA Elevated liver and kidney weights increased SGOT and SFGT Diet/NEAST 1660 1E-03 Acenaphthene NA D.10 6E-01 Low Hepatotoxicity Oral/IRIS NA Anthracene 2.0E-06 0.10 3E+00 Low Hepatotoxicity Oral/IRIS NA Benzolgs,h,i)perylene NA 0.10 NF Heuropathy, liver weight changes Gavage/IRIS 4E-04 Fluoranthene 1.8E-05 0.10 4E-01 Low Heuropathy, liver weight changes Gavage/IRIS 4E-04 Fluorene 7.9E-06 0.10 4E-01 Low Decreased RBC, packed cell Gavage/IRIS 2E-04 P-Crecol 4.4E-06 0.10 4E-01 Low Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05 Naphthalene 8.6E-06 0.10 4E-01 Low Kidney effects Gavage/IRIS 5E-04 Pyrene 1.6E-05 0.10 DI Kidney effect	[]Di-n-butylphthalate	NA	0.10	1E+00	Low	Increased mortality	Diet/IRIS	1000	1 1	NA NA	
Acenaphthene NA D.10 6E-01 Low Hepatotoxicity Oral/IRIS NA Anthracene 2.0E-06 0.10 3€100 Low Hepatotoxicity Oral/IRIS 7E-06 Benzo(g.h.i)perylene NA 0.10 NF Kesti/IRIS FLooranthene 1.8E-05 0.10 4E-01 Low Heuropathy, liver weight changes Gavage/IRIS 4E-04 Fluoranthene 1.8E-05 0.10 4E-01 Low Neuropathy, liver weight changes Gavage/IRIS 4E-04 Fluorene 7.9E-06 0.10 4E-01 Low Decreased RBC, parked cell Gavage/IRIS 2E-04 Volume and henoglobin Volume and henoglobin IRIS 1000 1 9E-05 Na 0.60 SE-01 Medium Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05 Naphthalene 8.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS SE-03 Pyrene 1.6E-05 0.10 DI <td< td=""><td>Dí-n-octylphthalate</td><td>2.6E-06</td><td>0.10</td><td>28-02</td><td>, NA</td><td>Elevated liver and kidney weights</td><td>Diet/HEAST</td><td>1000</td><td>1</td><td>1E-03</td><td>1</td></td<>	Dí-n-octylphthalate	2.6E-06	0.10	28-02	, NA	Elevated liver and kidney weights	Diet/HEAST	1000	1	1E-03	1
AcenaphtheneNAD.106E-01LowHepatotoxicityOral/IRISNAAnthracene2.0E-060.103E+00LowNo observed effectsGavage/IRIS7E-06Behzo(g,h,i)peryleneNA0.10NFNANoNFNAFluoranthene1.0E-050.104E-01LowHeuropathy, liver weight changesGavage/IRIS4E-04Fluorene7.9E-060.104E-01LowHeuropathy, liver weight changesGavage/IRIS2E-04Fluorene7.9E-060.104E-01LowDecreased REC, packed cellGavage/IRIS2E-04P-Crecol4.4E-060.105E-01MediumDecreased body weight, NeurotoxicityIRIS100019E-05Naphthalene8.6E-060.103E-01LowKidney effectsGavage/IRIS2E-03Pyrene1.6E-050.103E-01LowKidney effectsGavage/IRIS5E-04Phenol5.6E-050.103E-01LowKidney effectsGavage/IRIS5E-04Phenol5.6E-050.100.60Reduced fetal body weightGavage/IRISNAPestICIDES/PCBS0.10NFHEAST, IRISNA4.4*-00ENA0.10NFHEAST, IRISNA4.4*-00TNA0.10SE-04HediumLiver lesionsDiet/IRIS10011NA0.10SE-04HediumLiver lesionsDiet/IRIS1001N				1	-	increased SGOT and SPGT) 	1	1	1	;
Acemaphthene NA D.10 6E-01 Low Hepatotoxicity Oral/IRIS NA Anthracene 2.0E-06 0.10 3E+00 Low No observed effects Gavage/IRIS 7E-06 Benzo(g,h,i)perylene NA 0.10 NF No observed effects Gavage/IRIS HEAST/IRIS HA Fluoranthene 1.0E-0S 0.10 4E-01 Low Neuropathy, liver weight changes Gavage/IRIS 4E-04 Fluoranthene 1.0E-0S 0.10 4E-01 Low Neuropathy, liver weight changes Gavage/IRIS 4E-04 Fluorene 7.9E-06 0.10 4E-01 Low Decreased REC, packed cell Gavage/IRIS 2E-04 Volume and hemoglobin Internal lesions Diet/HEAST 2E-03 INPyrene 1.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS 5E-04 Phenol 5.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS 5E-04 Phenol 5.6E-05 0.10 0.60 Reduced fetal body weight Gavage/IRIS 100 9E-04 <td></td> <td></td> <td>-</td> <td>i.</td> <td></td> <td></td> <td>1</td> <td>1</td> <td>1</td> <td>1</td> <td>1</td>			-	i.			1	1	1	1	1
Anthracene 2.0E-06 0.10 3E+00 Low No observed effects Gavage/IRIS 7E-06 Benzo(g,h,i)perylene NA 0.10 NF NA Na NA Na Fluoranthene 1.8E-05 0.10 4E-01 Low Neuropathy, liver weight changes Gavage/IRIS 4E-04 Fluoranthene 1.8E-05 0.10 4E-01 Low Neuropathy, liver weight changes Gavage/IRIS 4E-04 Fluorene 7.9E-06 0.10 4E-01 Low Decreased RBC, packed cell Gavage/IRIS 2E-04 P-Cresol 4.4E-06 0.10 5E-01 Medium Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05 Naphthalene 8.6E-06 0.10 4E-02 NA Ocular and internal lesions Diet/HEAST 2E-03 Pyrene 1.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS 5E-04 Phenol 5.6E-05 0.10 0.60 Reduced fetal body weight Gavage/REAST 100 9E-04 Phenol 5.6E-05 0.10	Acenaphthene	NA	D.10	6E-01	Low	Hepatotoxicity	Oral/1R1S	;	i i	NA	1
Idenzo(g.h.1)perylene NA 0.10 NF Heuropathy, liver weight changes HEASI/IRIS HA Fluoranthene 1.8E-05 0.10 4E-01 Low Neuropathy, liver weight changes Gavage/IRIS 4E-04 Fluoranthene 7.9E-06 0.10 4E-01 Low Neuropathy, liver weight changes Gavage/IRIS 2E-04 Fluorene 7.9E-06 0.10 4E-01 Low Decreased REC, packed cell Gavage/IRIS 2E-04 P-Cresol 4.4E-06 0.10 5E-01 Medium Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05 Naphthalene 8.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS 5E-04 Pyrene 1.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS 5E-04 Phenoi 5.6E-05 0.10 0.60 Reduced fetal body weight Gavage/IRIS NA Phenoi 5.6E-05 0.10 0.60 Reduced fetal body weight Gavage/IRIS 100 9E-04 Phenoi 5.6E-05 0.10 NA	Anthracene	2.0E-06	0.10	3E+00	Low	No observed effects	Gavage/IRIS	1	ł	¦ 7E-06	1
Fluoranthene 1.8E-0S 0.10 4E-01 Low Heuropathy, liver weight changes Gavage/IRIS 4E-04 Fluorene 7.9E-06 0.10 4E-01 Low Decreased RBC, packed cell Gavage/IRIS 2E-04 P-Cresol 4.4E-06 0.10 5E-01 Medium Decreased RBC, packed cell Gavage/IRIS 2E-04 Naphthalene 8.6E-06 0.10 4E-02 NA Ocular and internal lesions Diet/HKAST 2E-03 Pyrene 1.6E-05 0.10 4E-01 Low Kidney effects Gavage/IRIS 5E-04 Phenanthrene 1.1E-05 0.10 DI Low Kidney effects Gavage/IRIS 5E-04 Phenoi 5.6E-05 0.10 DI Reduced fetal body weight Gavage/IRAST 100 9E-04 PESTICIDES/PCBS 0.10 0.60 Reduced fetal body weight HEAST, IRIS NA 4,4'-00C NA 0.10 NF HEAST, IRIS NA NA 4,4'-00T NA 0.10 NF Hedium Liver lesions Diet/IRIS 100 1 <td>, Senzo(g,h,1)perylene</td> <td>NA</td> <td>0.10</td> <td>NF</td> <td>1</td> <td></td> <td>HEAST/IRIS</td> <td>1</td> <td></td> <td>NA</td> <td>1</td>	, Senzo(g,h,1)perylene	NA	0.10	NF	1		HEAST/IRIS	1		NA	1
Fluorene 7.9E-06 0.10 4E-01 Low Decreased RBC, packed cell Gavage/IRIS 2E-04 P-Cresol 4.4E-06 0.10 5E-01 Medium Decreased RBC, packed cell Gavage/IRIS 2E-04 Naphthalene 8.6E-06 0.10 5E-01 Medium Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05 Naphthalene 8.6E-05 0.10 4E-02 NA Ocular and internal lesions Diet/HEAST 2E-03 Pyrene 1.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS 5E-04 Phenanthrene 1.1E-05 0.10 DI Reduced fetal body weight Gavage/IRIS SE-04 Phenoi 5.6E-05 0.10 0.60 Reduced fetal body weight Gavage/IEAST 100 SE-04 PESTICIDES/PCBS 0.10 NA 0.10 NF HEAST, IRIS NA 4.4'-00E NA 0.10 NF HEAST, IRIS NA NA 4.4'-00T NA 0.10 NF Hedium Liver lesions Diet/IRIS <td< td=""><td>Fluoranthene</td><td>1.8E-05</td><td>0.10</td><td>4E-01</td><td>LCH</td><td>Neuropathy, liver weight changes</td><td>Gavage/IR1S</td><td></td><td>1</td><td>4E-04</td><td>1</td></td<>	Fluoranthene	1.8E-05	0.10	4E-01	LCH	Neuropathy, liver weight changes	Gavage/IR1S		1	4E-04	1
Fluorene 7.9E-06 0.10 4E-01 Low Decreased RBC, packed cell Gavage/IRIS 2E-04 P-Cresol 4.4E-06 0.10 5E-01 Medium Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05 Naphthalene 8.6E-06 0.10 4E-02 NA Ocular and internal lesions Diet/HGAST 2E-03 Pyrene 1.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS 5E-04 Phenanthrene 1.1E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS 5E-04 Phenanthrene 1.1E-05 0.10 0.60 Reduced fetal body weight Gavage/IRIS 5E-04 Phenoi 5.6E-05 0.10 0.60 Reduced fetal body weight Gavage/IEAST 100 9E-04 PestTICIDES/PCBS 0.10 NA 0.10 NF HEAST, IRIS NA 4,4'-00E NA 0.10 NF Hedium Liver lesions Diet/IAIS 100 1 4,4'-00T NA 0.10 SE-04 Hedium Liver lesion	11				1	hematological alterations and			1		1
IP-Juorene 7.9E-06 0.10 4E-01 Low Decreased RBC, packed cell Gavage/IRIS 2E-04 Volume and hesoglobin volume and hesoglobin volume and hesoglobin IRIS 1000 1 9E-05 Naphthalene 8.6E-06 0.10 4E-02 NA Ocular and internal lesions Diet/HGAST 2E-03 Pyrene 1.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS 5E-04 Phenanthrene 1.1E-05 0.10 DI Reduced fetal body weight Gavage/RAST 100 9E-04 Phenoi 5.6E-05 0.10 DI Reduced fetal body weight Gavage/RAST 100 9E-04 Phenoi 5.6E-05 0.10 NF HEAST, IRIS NA PestTICIDES/PCBS 0.10 NF HEAST, IRIS NA 4,4'-00E NA 0.10 NF HEAST, IRIS NA 4,4'-00T NA 0.10 SE-04 Hedium Liver lesions Diet/IAIS 100 1					i .	clinical effects					1
volume and hesoglobin volume and hesoglobin IP-Cresol 4.4E-06 0.10 5E-01 Medium Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05 Naphthalene 8.6E-06 0.10 4E-02 NA Ocular and internal lesions Diet/HGAST 2E-03 Pyrene 1.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS 5E-04 Phenoi 5.6E-05 0.10 DI Reduced fetal body weight Gavage/HEAST 100 9E-04 Phenoi 5.6E-05 0.10 0.60 Reduced fetal body weight Gavage/HEAST 100 9E-04 Phenoi 5.6E-05 0.10 NE NA 100 9E-04 PESTICIDES/PCBS 0.10 NE HEAST, IRIS NA 4,4'-000 NA 0.10 NE HEAST, IRIS NA 4,4'-001 NA 0.10 SE-04 Hedium Liver lesions Diet/IRIS 100 1 NA	Fluorene	7.9E-06	0.10	4E-01	LOW	Decreased RBC, packed cell	Gavage/IRIS		1	2E-04	
1P-Cresol 4.4E-06 0.10 SE-01 Medium Decreased body weight, Neurotoxicity IRIS 1000 1 9E-05 Haphthalene 8.6E-06 0.10 4E-02 NA Ocular and internal lesions Diet//KAST 2E-03 Pyrene 1.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS SE-04 Phenanthrene 1.1E-05 0.10 DI NA NA NA NA Phenanthrene 1.6E-05 0.10 DI Reduced fetal body weight Gavage/HEAST 100 9E-04 Phenanthrene 5.6E-05 0.10 D.60 Reduced fetal body weight Gavage/HEAST 100 9E-04 PestTICIDES/PCBS 0.10 NA 0.10 NF NA NA 4,4'-000 NA 0.10 NF HEAST, IRIS NA 4,4'-00T NA 0.10 NF Hedium Liver lesions Diet/IRIS 100 1 NA						volume and hemoglobin	1				1
11Aphthalene 8.6E-06 0.10 4E-02 NA Ocular and internal lesions Diet/HEAST 2E-03 Pyrene 1.6E-05 0.10 3E-01 Low Kidney effects Gavage/IRIS 5E-04 Phenanthrene 1.1E-05 0.10 DI 1.4E-05 0.10 DI NA Phenol 5.6E-05 0.10 DI Reduced fetal body weight Gavage/HEAST 100 9E-04 PESTICIDES/PCBS 0.10 NA 0.10 NF NA NA 4,4*-000 NA 0.10 NF HEAST, IRIS NA 4,4*-00T NA 0.10 SE-04 Hedium Liver lesions Diet/IRIS 100 1	P-Cresol	4.4E-06	0.10	5E-01	Medium	Decreased body weight, Neurotoxicity	IRIS	1000	1	9E-05	
IPYTENE I.bt-VD O.10 3E-01 Low Kidney effects Gavage/IRIS SE-04 Phenanthrene I.IE-05 O.10 DI "HEASI NA Phenoi 5.6E-05 O.10 0.60 Reduced fetal body weight Gavage/HEASI 100 9E-04 PESTICIDES/PCBS 0.10 NA 0.10 NE HEASI, IRIS NA 4,4'-0CE NA 0.10 NE HEASI, IRIS NA 4,4'-DDT NA 0.10 SE-04 Hedium Liver lesions Diet/IRIS NA	inaphinalene	8.6E-06	0.10	4E-02	NA	Ocular and internal lesions	Diet/HLAST		1	2E-03	
Impenantmene I.IE-US O.IO DI "HEASI NA Phenoi 5.6E-OS O.IO 0.60 Reduced fetal body weight Gavage/HEASI 100 9E-04 PESTICIDES/PCBS 0.10 NF HEASI, IRIS NA 4,4'-DCE NA 0.10 NF HEASI, IRIS NA 4,4'-DDE NA 0.10 NF HEASI, IRIS NA 4,4'-DDE NA 0.10 SE-04 Hedium Liver lesions Diet/IRIS 100 1 NA	Pyrene	1.6E-05	0.10	3E-01	LOW	Kidney effects	Gavage/IRIS		1	5E-04	
Impendi 5.66-05 0.10 Reduced fetal body weight Gavage/HEAST 100 9E-04 PESTICIDES/PCBS 0.10 NA 0.10 NF NA NA <t< td=""><td>, Phenanthrene</td><td>1.1E-05</td><td>0.10</td><td>DI</td><td>i i</td><td></td><td>HEAST</td><td></td><td>1</td><td>NA</td><td>1</td></t<>	, Phenanthrene	1.1E-05	0.10	DI	i i		HEAST		1	NA	1
PESTICIDES/PCBS 0.10 4,4'-060 NA 0.10 NE 4,4'-00E NA 0.10 NE 4,4'-00E NA 0.10 NE 4,4'-00E NA 0.10 NE 4,4'-00T NA 0.10 SE-04 Medium Liver lesions Diel/IRIS 100	(Phenol	5.6E-05	0.10	; 0.60	i	Reduced fetal body weight	Gavage/HEAST	100		9E-04	
PESTICIDES/PLDS 0.10 4,4'-060 NA 0.10 NE 4,4'-00E NA 0.10 NE 4,4'-00E NA 0.10 NE 4,4'-00E NA 0.10 NE 4,4'-00T NA 0.10 SE-04 Medium Liver lesions Diet/JRIS 100			1 0 10	1							
14,4'-DDE , NA 0.10 NF , M2+51, [N]5 NA 14,4'-DDE , NA 0.10 NF , HE∧51, [N]5 NA 14,4'-DDT NA 0.10 5E-04 Medium Liver Lesions Diet/IRI5 100 1 NA	I PESTICIDES/FCBS	L NA	0.10	1 10	:				1		
14,4'-DDT NA 0.10 5E-04 Medium Liver lesions Diet/IRIS 100 1 NA	14,4 -000		0.10	i Nt Lur	1		HEAST, IRIS		-	NA	i
rine ver μ n μ μ υ.υν σε σνη πουση μ Liver lesions - μιθεί/μιδ ; IW ; 1 ; NA ; Επιστοποτοτιστιστιστιστιστιστιστιστιστιστιστιστιστ	114,4 "DUC	I DA	0.10	I NI I SE_A4	Modium	l liver lesione	HLASI, IRIS	1 100		NA	ļ
			0.10	- 5C-04	I LIGTON		I NIGEAUXE	100		, NA 11111111	

VERIL LORFAUL FRAMELITY SUDT HO APIR LORFAUL FRAMELITY SUDT CORTICE SACEMERAL FOX SACE SACEMERAL FOX SACEMERAL FOX SACE SACEMERAL FOX SACEMERAL FOX SACE SACEMERAL FOX SACE	_	SUNHA	RY OF CHRONIC	HAZARO INDE	X ESTIMATES	AFPENDIX B - ROAD IMPROVEHENTS AND CLEARING VEGET	Tation (Alternat	IVES 2, 2A, 4	AND 4A)		
Later and Lat	 141201156117841511961861819119	(111410101010104				HERTEL LANGFILL FEASIBILITY STUDY		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		,,,,,,,,,,,,	
OCICAC DUDLINE PAGE PTO CONFIRMENT CONFIRMENT DUDLINE PAGE DUDLINE PAGE <thdudline page<="" th=""> <thdudline page<="" th=""> DUDL</thdudline></thdudline>		SUBCHRONIC	CDI			***************************************	RF O	. RFD	1111111111	********	FATIMAY TOTAL
(repleyer) RECRETION OF ADDRESS RECRETION OF ADDRESS RECRETION OF ADDRESS (RECRET ALS ADDRESS 10 UST) Exposed Pointer: INPLATION OF ADDRESS (RECRET ALS ADDRESS 10 UST) International internationa	CHCHICAL	DAILY INTAKE	NOJUSTED FOR	RFD	CONFIDENCE	CRITICAL	SOURCE/	UNCERTAINTY	HCOIFYING	HAZARD	HAZARD EXPORTE
Depose periodul investigione designed designed designed by the designed designe		(mg/kg/day)	ABSORPTION	(mg/kg/day)	LEVEL	EFFECT	BASIS	ADJUSTHENTS	FACTORS	QUOTIENT	INDEX (H])¦ H]
Description Description <thdescription< th=""> <thdescription< th=""></thdescription<></thdescription<>	1999991199119911911939569551109 89					******************************	(11))))))))))))))))))))))))))))))))))))				
JUNCAULTURE JUNCAULTURE <thjuncaulture< th=""> <thjuncaulture< th=""></thjuncaulture<></thjuncaulture<>	EXPOSURE PATHWAY: INHALATI	ON OF AIRBORNE	CHENICALS AD	SORBED TO DU	51						16+01 16+01
INSPANICS Install Installin Install Install											
Hallmöry 4, 24-05 HO 42-04 Ho	INORGANICS							1			
Barlan 1.7.4.0 B0 11.6.50	Antimony	4.5E-05 ;	NO NO	46-04				i		1E-01	
Barylina (16chian) 1.4-56 No 0.01 10reaina 1.4-56 No 45-56 No Head No 16-70 No 16-70 No 16-70 No 16-70 No 16-70 No 16-70 10000 16-70 10000 16-70 10000 16-70 10000 16-70 10000 16-70 10000 16-70 10000 16-70 10000 16-70 10000 16-70 10000 16-70 10000 16-70 10000 16-70 10000 16-70 10000 16-70 10000	Rarium	2.46-04	No	1E-03		1		!		2E-01	
Electronic 1.64-06 No 11-03 No 11-03 No 12-03 No 22-03 No 42-04 No 42-04 No 42-04 No 42-04 No 42-04 No 42-04 No 10-12-04 No 10-12-04 No 22-03 No No 22-03 No No 22-03 No No No 22-03 No	Berylliun	1.6E-06	No	0.01				1		3E-04	
Chronium 3.45-05 No 64-06 MM Messal access attrophy FEASI 300 55-00 Marganese 2.35-00 No 4-04 Median Increasing structures of problem IRIS 300 3 55-00 Intercorr No LE-04 Median No IE-04 No IE-04 No IE-01 No IE-01 </td <td>Cadmium</td> <td>1.6E-06</td> <td>No</td> <td>1E-03</td> <td></td> <td></td> <td></td> <td>1</td> <td></td> <td>2E-03</td> <td></td>	Cadmium	1.6E-06	No	1E-03				1		2E-03	
Nangaterie 2.45-03 No 44-04 Indua Increases prevalence of respiratory systems and psycho respiratory systems and psycho intor disturbances REASI 900 2 64/0 Heroury NA No 12-04 NA Neurotoxisty NevEAST 30 48.0 Heroury NA No 27-04 No 27-03 School 55-02 No Jine 2.77-04 No 27-01 No 27-01 No 55-02 No Jine 2.77-04 No 11-01 No 11-00 No 11-00 No Jackon Biolifide No 11-01 No 11-00 No 11-00 No 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 11-00 12-00 10-00 11-00 12-00 10-00 11-00 12-00 10-00 11-	Chromium	3.68-05	No	65-06	NA	Nasal nucosa atrophy	HEAST	300		6E+00	
Itercury NB ND LE-04 NB NB LE-04 NB NB LE-04 NB NB LE-04 NB NB LE-04 NB NB LE-04 <td>Manganese</td> <td>2.5E-03</td> <td>No</td> <td>4E-04</td> <td>Medium</td> <td>Increased prevalence of</td> <td>IRIS</td> <td>300</td> <td>3</td> <td>6E+00</td> <td></td>	Manganese	2.5E-03	No	4E-04	Medium	Increased prevalence of	IRIS	300	3	6E+00	
Heroury Matchel Vachel Internation NB HE-04 NA Herotoxicty NB/MCAST 20 NB Vacabilin 3.2.4-05 NB 22-00 1 22-00 1 22-00 1 22-00 1 22-00 1 22-00 1 22-00 1 22-00 1 1 22-00 1 1 22-00 1 1 22-00 1 1 1 22-00 1 1 1 0					1	i respiratory syntoms and psycho- i	HEAS1	900	1		
Nuclei 4.22-05 Me 22-02 No 22-03 tradium 3.35-05 Me 72-03 12-03 12-03 lice 2.72-04 Mo 22-02 14-03 12-03 VALILES NA NO 22-02 14-03 14-03 VALUES NA NO 22-02 14-03 14-03 VALUES NA NO 22-02 14-03 14-03 VALUES NA 14-08 NO 22-02 14-03 VALUES NA 14-08 NO 22-02 14-03 Values 2.32-08 NO 22-00 NA Close effects, eves and nose irritation 162-03 Values 2.32-08 NO 22-00 NA Close effects, eves and nose irritation 162-03 100 32-06 Values 3.42-07 NO 22-00 NA Close effects, eves and nose irritation 162-03 100 32-06 Values NA NF NA	Hercury	NA	No	1E-04	NA	Neurotoxicty	NA/HEAST	30		NA	
Anadian 3.8-05 No 77-01 line 2.77-04 No 27-01 Variance NA No 27-01 Variance NA No 27-01 Variance NA No 27-02 Variance NA No 27-02 Variance 1.4-08 No 57-01 No 57-02 NA Liver and kidney effects HEAST Toburos 2.3-06 No 27-00 NA Close effects, eyes and nose invitation HEAST 1000 HE-06 Toburos 2.3-07 No 37-02 NA Close effects, eyes and nose invitation HEAST 100 HE-06 ist 2ethylhevyl johthalate 1.3-05 Ho 25-02 26-02 26-02 26-02 26-02 26-02 26-02 26-02 26-07 NA 100 HEAST 100 HEAST 100 HEAST 100 HEAST 100 HEAST 100 HEAST 100 1	Nickel	4.2E-05	No	2E-02						2E-03	
Line 2.764 No 27-01 11-03 Lipsaide MA No 27-02 NA VALATILES NA No 27-02 NA Intervence 1.408 No 16-01 NA Intervence 1.408 No 16-01 NA Universe 2.37-08 No 16-01 NA CNS effects, eyes and mose irritation HEAST 1000 12-03 SHIVUATILES 162-07 No 22-02 14 CNS effects, eyes and mose irritation HEAST 100 12-03 SHIVUATILES 162-07 No 22-02 14 No 16-03 Universe NA No NF 16-03 16-03 16-03 Heast Alphanthacee NA No NF 16-0	vanadium	3.3E-05	No	7E-03	1	5 · · · · ·	**			5E-03	
I(sando NA No Z-02 NA WARHES No 16-08 16-01 No 56-02 NA Liver and kidney effects HEAST 100001 36-07 Thorosestene 1.4-08 No 55-02 NA Liver and kidney effects HEAST 10000 36-07 Tobusce 2.37-08 No 25-02 NA Cive and kidney effects HEAST 1000 36-07 Tobusce 8.47-07 No 37-01 NA Cive effects, eyes and nose irritation, HEAST 100 X-66 Tistlens/shuthate 1.36-05 No 26-02 NA Cive effects, eyes and nose irritation, HEAST 100 X-66 Tistlens/shuthate 1.46-07 No 26-02 KEAST, IRIS HA KEAST, IRIS HA Thardee NA No MF HEAST HA KEAST, IRIS HA Thardee NA No MF HEAST, IRIS HA HA Thardee NA No <td>inc.</td> <td>2.7E-04</td> <td>No</td> <td>2E-01</td> <td></td> <td></td> <td></td> <td></td> <td>1</td> <td>1E-03</td> <td></td>	inc.	2.7E-04	No	2E-01					1	1E-03	
V0.411LES arbon Disulfide NA No 11-01 bit of bis life NA Liver and kidney effects REAST 10000 32-07 1tbr/brearee 4.7-00 No 32-00 NA Liver and kidney effects REAST 1000 32-07 1br/brearee 4.7-00 No 32-01 NA CKS effects, eyes and nose invitation REAST 100 16-00 1brears 8.42-07 No 32-01 NA CKS effects, eyes and nose invitation REAST 100 32-06 ist 2ethylexylohthalate 1.32-05 No 22-02	Cyanide	NA	No	2E-02	1	1			[NA	
arbon Disulfide NA No 15-01 Thirdence 1.4E-08 No 5E-02 NA Liver and kidney effects HEAS1 10000 3E-07 Tolune 2.3E-08 No 2E-07 NA USE effects, eyes and nose irritation HEAS1 1000 1E-08 Tolune 2.3E-08 NO 2E-07 NA USE effects, eyes and nose irritation HEAS1 100 1E-08 Tolune 2.3E-08 NO 2E-07 NA USE effects, eyes and nose irritation HEAS1 100 3E-06 Termoda janchracene NA NO 2E-02 KEAS1 100 3E-07 Throsene NA NO 2E-02 KEAS1 KEAS1 </td <td>VOLATILES</td> <td></td> <td>!</td> <td>!</td> <td>!</td> <td></td> <td></td> <td></td> <td>!</td> <td></td> <td></td>	VOLATILES		!	!	!				!		
Thirdextence 1.41-08 No SE-02 NA Liver and kidney effects HEAS1 10000 SE-02 'Iburne 2.3E-08 No 2E+00 NA CNS effects, eyes and nose irritation HEAS1 100 HE-00 'Iburne 2.3E-08 No 3E+00 NA CNS effects, eyes and nose irritation HEAS1 100 HE-00 'Iburne 2.3E-08 No 3E+00 NA CNS effects, eyes and nose irritation HEAS1 100 HE-00 'Iburne 2.3E-08 No 2E+02 KA	Carbon Disulfide	NA	No	1E-01						: NA	
Ithlylbarene 4.7E-00 No IE+00 NA (NS effects, eyes and nose irritation) HEAST 100 IE-00 StRIV0LATILES 8.4E-07 No 3E-01 NA (NS effects, eyes and nose irritation) HEAST 100 1E-08 StRIV0LATILES Isi2 activities Isi2 activities <td>Tthlorobenzene</td> <td>1.4E-08</td> <td>No</td> <td>5E-02</td> <td>NA</td> <td>Liver and kidney effects</td> <td>HEAST</td> <td>10000</td> <td></td> <td>3€-07</td> <td></td>	T thlorobenzene	1.4E-08	No	5E-02	NA	Liver and kidney effects	HEAST	10000		3€-0 7	
***Outome 2.4E-09 No 22+00 NA (NS effects, eves and nose irritation) HEAST 100 HE-08 \$EM100_ATILES ISC No 2E-00 NA (NS effects, eves and nose irritation) HEAST 100 EF-08 \$SEM100_ATILES ISC No 2E-02 ISC ISC ISC ISC \$SEM100_AntLes ISC=05 No 2E-02 ISC ISC ISC ISC \$SEM100_Alanthacene NA No VE ISC ISC ISC ISC **Introduction No NF ISC ISC ISC ISC ISC **Introduction No NF ISC ISC <td>Ethylbenzene</td> <td>4.78-08</td> <td>No</td> <td>1E+00</td> <td></td> <td></td> <td></td> <td>1</td> <td>1</td> <td>5E-08</td> <td></td>	Ethylbenzene	4.78-08	No	1E+00				1	1	5E-08	
Jieros 0.4C.0 NO JC.0 NO JC.0 NO JC.0 JC.00 JC.00 <thj< td=""><td>' long</td><td>2.3E-08</td><td>No</td><td>2E+00</td><td>AN IN</td><td>CNS effects, eyes and nose irritation</td><td>HEAST</td><td>100</td><td></td><td>1E-08</td><td></td></thj<>	' long	2.3E-08	No	2E+00	AN IN	CNS effects, eyes and nose irritation	HEAST	100		1E-08	
SERIVOLATILES Id 2E-02 Rtylbertylphthalate Id 2E-02 Rtylbertylphthalate GE-04 Rtylbertylphthalate GE-06 Rtylbertylphthalate GE-06 Rtylbertylp	.) 10105	0.46-0/	I NU	1 3E-01	, MH	Ins effects, eyes and mose iffication	i MERSI		i 	3E-06	
I is(2ethylexyl)ohthalate 1.3E-05 No 2E-02 KE-04 ix(2benxylphthalate 1.3E-05 No 2E+02 KE-04 ix(1benxylphthalate No NF KE-051, IRIS NA Bend(a) byrene NA No NF KE-051, IRIS NA I'thrysene NA No NF KE-051, IRIS NA I'thrysene NA No NF KE-051, IRIS NA ithrysene NA No NF KE-051, IRIS NA ithrysene NA No NF KEA51, IRIS NA itholenthe NA No NF KEA51, IRIS NA Iblento(a, h)shthracene NA No NF KEA51, IRIS NA Iblento(a, h)shthracene NA No NF KEA51, IRIS NA Iblento(a, h)shthrace NA No NF KEA51, IRIS NA Iblento(a, h)shthrace NA No KE KEA51, IRIS NA Iblento(a, h)shthrace NA No KE KEA51, IRIS NA Iblento(a, h)shthrace NA No KE KEA51, IRIS NA Iblento(a, h)shthrace	SEHIVOLATILES		1	:			1		1)
is (2 stryl heryl) (bith late 1. 3::-05 No 2::-02 ix y lonxyl pht halate 4. 6::-07 No 2::00 NF (EAST, 1815 NA 18 mod a) pyrere NA No NF (EAST, 1815 NA 19 mod a) pyrere NA No NF (EAST, 1815 NA mod b) Low anthene NA No NF (EAST, 1815 NA mod b) Low anthene NA No NF (EAST, 1815 NA 10 benzo(a, h) shtracean NA No NF (EAST, 1815 NA 10 benzo(a, h) shtracean NA No NF (EAST, 1815 NA 10 benzo(a, h) shtracean NA No NF (EAST, 1815 NA 10 benzo(a, h) shtracean NA No NF (EAST, 1815 NA 4 - 0 bith low anthene NA No NF (EAST, 1815 NA 4 - 0 bith low anthene NA No NF (EAST, 1815 NA 10 benzo(a, h) shtracean NA No NF (EAST, 1815 NA 4 - 0 bith low anthene NA No NF (EAST, 1815 NA 4 - 0 bith low anthene NA No Z:=01 NA Liver and tidney effects (EAST) 100 NA 10	11		1	:	1		1		1		, , ,
Itylicity ipin Kalate 4.86-07 No 2E-00 Pennod a John Racee NA No NF HEASI, IRIS HA Idenzola John Racee NA No NF HEASI, IRIS HA Introd (a) John Racee NA No NF HEASI, IRIS HA Introd (a) John Racee NA No NF HEASI, IRIS HA Introd (a) John Kracee NA No NF HEASI, IRIS HA Introd (a) John Kracee NA No NF HEASI, IRIS HA Index (J, 2, c, dbyree NA No NF HEASI, IRIS HA Index (J, 2, c, dbyree NA No NF HEASI, IRIS HA Index (J, 2, c, dbyree NA No NF HEASI, IRIS HA Index (J, 2, c, dbyree NA No NF HEASI, IRIS HA Index (J, 2, c, dbyree NA No Effects HEASI, IRIS HA Index (J, 2, c, dbyree NA No Effects HEASI, IRIS HA Index (J, 2, c, dbyree NA No Effects HEASI HA Index (J, a, c, dbyree NA No Effects HEASI	is(2ethylhexyl)phthalate	1.3E-05	No	2E-02			1		}	6E-04	
Mathematical Sector Mathematical Sector	stylbenzylphthalate	4.8E-07	NO NO	2E100	i					28-07	4
1 ^{rh} Hysene NA No DI 1 ^{rh} Hysene NA No N ^r HA HA nzo(k) (fluoranthene NA No N ^r HEAST HA nzo(k) (fluoranthene NA No N ^r HEAST HA Indenc(1, 2, 3 ⁻ c, d) pyrene NA No N ^r HEAST HA Indenc(1, 2, 3 ⁻ c, d) pyrene NA No N ^r HEAST HA Indenc(1, 2, 3 ⁻ c, d) pyrene NA No N ^r HEAST HA Indenc(1, 2, 3 ⁻ c, d) pyrene NA No 22 ⁻ O1 NA Liver and kidney effects HEAST HA Indenc(1, 1, 2, 3 ⁻ c, d) pyrene NA No 22 ⁻ O2 NA HEAST HA Indenc(1, 1, 1) pythalate 4.0E-O7 No 2E ⁻ O2 HA HA Informathene J.2E ⁻ O5 NO 4E ⁻ O1 HEAST HA I/ Fluoranthene J.2E ⁻ O6 NO 4E ⁻ O1 HEAST HA <t< td=""><td>L'Benzo(a)pyrene</td><td>NA NA</td><td>No</td><td>l NF</td><td>1</td><td></td><td>HERST, INIS</td><td></td><td></td><td>; NA ! NA</td><td>1</td></t<>	L'Benzo(a)pyrene	NA NA	No	l NF	1		HERST, INIS			; NA ! NA	1
Into(b)fluoranthene NA NO NF IEAST, IRIS NA Into(k) J/Luoranthene NA NO NF IEAST, IRIS NA Ibbenzo(a,h)anthracene NA NO NF IEAST, IRIS NA 4-Dichlorobenzene NA NO EFOO IEAST, IRIS NA 4-Dichlorobenzene NA NO EFOO IEAST, IRIS NA 4-Dichlorobenzene NA NO EFOO IEAST, IRIS IEAST IOO *n-butyliphthalate 4.06-07 NO 2E-02 IEAST, IRIS IEAST IA I/*reenes 3.0E-07 NO 2E-02 IEAST IA IA I/*loaranthene NA NO AFOO IEAST IA IA <td>!! hrysene</td> <td>NA</td> <td>No</td> <td>DI</td> <td>1</td> <td></td> <td>HEAST</td> <td></td> <td>2</td> <td>NA NA</td> <td></td>	!! hrysene	NA	No	DI	1		HEAST		2	NA NA	
International NA NA	nzo(b)fluoranthene	NA	No	NE	1		HEAST, IRTS	i i		NA	1
Liberzola, Njanthratene NA No NF HEAST, IRIS NA Ilidenc(1,2,3,c,d)pyrene NA No NF HEAST, IRIS NA 4-Dickiblorobenzene NA No 22-01 HA Liver and kidney effects HEAST, IRIS NA 4-Dickiblorobenzene NA No 22-01 HA Liver and kidney effects HEAST, IRIS NA 4-Dickiblorobenzene NA No 22-02 HA HEAST HA **n-butylphthalate 4.0E-07 No 22-02 HA HA **n-acptiphthalate No 62-01 HA HA HA **n-acptiphthalate 4.0E-07 No 32+00 HA HA **recape 3.0E-07 No 32+00 HA HA **recapi 6.7E-07 No 4E-01 HEAST/IRIS HA */Fluoranthene 1.2E-06 No 4E-01 HEAST HEAST HEAST */Flouranthene 1.3E-06 <td< td=""><td>nzo(k)f luor ant hene</td><td>NA</td><td>No</td><td>NF</td><td>-</td><td></td><td>HEAST, IRIS</td><td></td><td></td><td>NA</td><td></td></td<>	nzo(k)f luor ant hene	NA	No	NF	-		HEAST, IRIS			NA	
How Hill, 12, 30 C, 0, prese NA NO NF Heast Hils NA 4 - Dich Tobenzene NA NO 2E-01 NA Liver and kidney effects HEAST 100 NA 4 - Dich Tobenzene NA NO 2E-01 NA Liver and kidney effects HEAST 100 NA 101 - n-octylphthalate A. 0E-07 NO 2E-02 NA NA 102 - n-octylphthalate A. 0E-07 NO 2E-02 NA NA 102 - n-octylphthalate A. 0E-07 NO 2E-02 NA NA 103 - n-octylphthalate A. 0E-07 NO 2E-02 NA NA nod (s, h, i)perylene NA NO NF HEAST/IRIS NA If Loos (h, h) AE-01 NF HEAST/IRIS NA NA If Loos (h, h) AE-01 NE HEAST/IRIS NA NA If Loos (h, h) AE-01 NE HEAST NA NA NA No DI<	; Dicenzol a , n Janunracene	i NA I NA	i No	i NF			HEAST, IRIS			NA	
Interview Interview <thinterview< th=""> Interview <thinterview< th=""> Interview <thinterview< th=""> <thinterview< th=""> <thint< td=""><td>4-Dichlorobenzene</td><td>I NA</td><td>i No</td><td>2F-01</td><td>AIA</td><td>l liver and kidney effecte</td><td>HERSI, IKIS</td><td>100</td><td></td><td>i NA</td><td>1</td></thint<></thinterview<></thinterview<></thinterview<></thinterview<>	4-Dichlorobenzene	I NA	i No	2F-01	AIA	l liver and kidney effecte	HERSI, IKIS	100		i NA	1
Image: NA No 1E+00 NA 101-n-octylphthalate 4.0E-07 No 2E-02 NA 101-n-octylphthalate 4.0E-07 No 2E-02 NA 1/0-n-octylphthalate NA No 6E-01 NA 1/o-naphthene NA No 6E-01 NA 1/nacene 3.0E-07 No 3E+00 NA 1/fluoranthene 2.7E-06 No 4E-01 NA 1/fluoranthene 2.7E-06 No 4E-01 7E-06 1/fluoranthene 1.2E-06 No 4E-01 7E-06 1/fluoranthene 1.3E-06 No 4E-02 3E-05 1/fluoranthene 1.3E-06 No 4E-02 3E-05 1/fluoranthene NA No 01 HEAST NA 1/fluoranthene NA No 01 HEAST NA 1/fluoranthene NA No 01 HEAST NA 1/fluoranthene NA <	ethylphthalate	2.2E-07	No	8E+00		i Elver and kroksy errects		1 100	1	NA NA	
101-n-octylphthalate 4.0E-07 No 2E-02 NA 1^0renaphthene NA No 6E-01 NA 1/brenaphthene 3.0E-07 No 3E+00 NA 1/brenaphthene 3.0E-07 No 3E+00 NA 1/brenaphthene 3.0E-07 No 3E+00 NA 1/brenaphthene 1.0E-07 No 4E-01 NA 1/brenaphthene 2.7E-06 No 4E-01 7E-06 1/fluoranthene 2.7E-06 No 4E-01 7E-06 1/fluoranthene 1.2E-06 No 4E-01 7E-06 1/fluoranthene 1.3E-06 No 4E-02 3E-05 1/friene 1.3E-06 No 3E-01 3E-05 1/friene 2.4E-06 No 3E-01 8E-06 1/friene NA No DI HEAST NA 1/brenanthilding NA No DI HEAST NA 1/brenanthilding NA No DI HEAST NA 1/fluores NA	-n-butylphthalate	NA	No	1E+00	1					NA	
Ma No 6E-01 HA thracene 3.0E-07 No 3E+00 NA nzo(g,h,i)perylene NA No NF HEAST/IRIS NA iFluoranthene 2.7E-06 No 4E-01 7E-06 7E-06 iFluoranthene 2.7E-06 No 4E-01 7E-06 7E-06 iFluorene 1.2E-06 No 4E-01 7E-06 7E-06 iFluorene 1.2E-06 No 4E-01 7E-06 7E-06 iffluorene 1.3E-06 No 4E-02 7E-06 7E-06 iffluorene 1.3E-06 No 4E-02 7E-06 7E-06 iffluorene NA No 0I 7E-06 7E-06 7E-06 iffluorene NA No 0I 7E-06 7E-06 7E-06 7E-06 7E-06 iffluorene 1.3E-06 No 4E-01 7E-06 7E-06 7E-06 7E-06 7E-06 7E-06 7E-06	Di-n-octylphthalate	4.0E-07	¦ No	2E-02	1	1	1		1	NA	
Lhracene 3.0E-07 No 3E+00 NA nzo(g, h, i)perylene NA No NF HEAST/IRIS NA i/Fluoranthene 2.7E-06 No 4E-01 7E-06 7E-06 i/Fluorene 1.2E-06 No 4E-01 7E-06 7E-06 i/Fluorene 1.2E-06 No 4E-01 7E-06 7E-06 i/Fluorene 1.2E-06 No 4E-01 7E-06 7E-06 i/Fluorene 1.3E-06 No 4E-02 7E-06 7E-06 i/Frene 2.4E-06 No 3E-01 7E-06 7E-06 i/Frene NA No DI 7E-06 7E-06 7E-06 i/Frene 2.4E-06 No 3E-01 7E-06 7E-06 7E-06 7E-06 i/Phenalthiene NA No DI 7E-76 7E-06 7E-06 i/A *-000 NA No NF 7E-76 7E-76 7E-76 i/A *-00E <td< td=""><td>, Arenaphthene</td><td>NA</td><td>No</td><td>6E-01</td><td></td><td></td><td></td><td></td><td>1</td><td>NA</td><td></td></td<>	, Arenaphthene	NA	No	6E-01					1	NA	
Provy m, r, per free PR PO PR PO PR 1/Fluoranthene 2.7E-06 No 4E-01 7E-06 7E-06 1/Fluorene 1.2E-06 No 4E-01 7E-06 7E-06 1/Fluorene 1.2E-06 No 4E-01 7E-06 7E-06 2resol 6.7E-07 No 5E-01 1E-06 3E-05 2rene 2.4E-06 No 4E-01 8E-06 8E-06 1/Fhenanthilene NA No 01 8E-06 8E-06 1/Fhenanthilene NA No 01 8E-01 8E-06 1/Fhenanthilene NA No NF 8E-01 8E-01 1/Fhenanthilene NA No NF 8E-01 8E-01 8E-01 1/Filene NA No NF 8E-01 <td>thracene</td> <td>3.0E-07</td> <td>No</td> <td>3E+00</td> <td></td> <td>,</td> <td></td> <td></td> <td></td> <td>NA</td> <td></td>	thracene	3.0E-07	No	3E+00		,				NA	
1.2E-06 No 4E-01 7E-06 VFluorene 1.2E-06 No 4E-01 Cresol 6.7E-07 No 5E-01 Shthalene 1.3E-06 No 4E-02 Vertene 2.4E-06 No 3E-05 Vertene 2.4E-06 No 3E-05 Vertene 2.4E-06 No 3E-01 Vertene 2.4E-06 No 3E-01 Vertene 2.4E-06 No 3E-01 Vertene 2.4E-06 No 01 Vertene 2.4E-06 No 01 Vertene 2.4E-06 No 01 Vertene NA No 01 Vertene NA No NF Vertene	Eluoranthene	1 NS 2 75-04	I NO	1/11 1 /11	F		HEAST/IRIS			NA TE C	1
Cresol 6.7E-07 No 5E-01 1E-06 phthalene 1.3E-06 No 4E-02 3E-05 Prinee 2.4E-06 No 3E-01 3E-05 Phenanthiene NA No 0I HEAST HA * Phenol NA No 0I HEAST HA * PESTICIDES/PCBS FESTICIDES/PCBS HEAST HA * 4,4*-D0E NA No NF HEAST, IRIS HA * '-D0T NA No SE-04 Liver lession HA	1:Fluorene	1.2E-06	No	4E-01			1	1	1	, /E-06	
Dathalene 1.3E-06 No 4E-02 DE-05 Prime 2.4E-06 No 3E-01 BE-06 BE-06 Phenanthiene NA No DI HEAST HA * %enol NA No DI HEAST HA * Kenol NA No DI HEAST HA * FESTICIDES/PCBS HEAST HA * 4,4*-D0E NA No NF HEAST, IRIS HA * '-00T NA No SE-04 Liver lession HA	Cresol	6.7E-07	No	5E-01			5			JE-06	
Transmission 2.4E-06 No 3E-01 BE-06 Phenanthiene NA No DI HEAST HA "Nhenol NA No DI HEAST HA "FESTICIDES/PCBS Image: Construct of the construction of the constr	ohthalene	1.3E-06	No	4E-02					1	3E-05	
NPERanchiene NA No D1 HEAST NA "Ohenol NA No D1 HEAST NA "Ohenol NA No D1 HEAST NA "Ohenol NA No NF HEAST, IRIS NA "4,4"-DÜE NA No NF HEAST, IRIS NA "-OUT NA No SE-04 Liver lession NA	Troff ene	2.4E-06	No	3E-01			1	1		8E-06	
FESTICIDES/PCBS Ref 4,4'-COO NA NO NF HEAST, IRIS NA 4,4'-DRE NA NO NF HEAST, IRIS NA 1'-ODT NA NO SE-04 Liver lession NA	(Phenanthi ene	NA NA	N0 No	10			HEAST HEAST	1	1	NA NA	
PESTICIDES/PCBS HEAST, IRIS NA 14,4*-COD NA NO NF 14,4*-CDE NA NO NF 1*-CDT NA NO SE-04 1*-CDT NA NO SE-04							ICH01			HH	
14,4°-COD NA NO NF HEAST, IRIS NA 14,4°-DEE NA NO NF HEAST, IRIS NA 1°-COT NA NO SE-04 Liver lession NA	PESTICIDES/PCBS		1	1	1		1			1	
14,4°-DUR. NA NO NF HEAST, IRIS NA 1°-DOT NA NO SE-04 Liver lession NA	4,4'-000	NA	No	NF			HEAST, IRIS			NA	
	14,4'-DUE	i NA NA	NO	i NF		liver lession	HEAST, IRIS	1		NA	
				1 DC-04			1			, NA	

-

INGESTION OF CHEMICALS IN SOIL CAP CONSTRUCTION (ALTERNATIVES 2, 2A, 4 AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

, , , , , , , , , , , , , , , , , , ,	11111111111111111111111111111111111111	1									
	INTAKE	CANCED	COTI IN		CACTOR	FRACTION	EXPUSURE	LAPUSURE	BUUT	AVG. LINE	AVG. LINE
UNERICAL	I NUNCHINER	i CANCER	i SUIL	i KALE	¦ PALIUK \/10⊑-(ka/n=)	i INGESIEU j	PREQUENCT	(UNATION	HELUHI	NUNCANCER	(due)
	1 (BU/ NJ/ Ud 7)		1 (B9/K9)	(mg S011/0dy)	(102°0 kg/mg))	(uays/ year)	(years)	(A9)		(oays)
SENIVOLATILES	1	1	:	!				!	!	:	
Benzo(a)anthracene	6.16E-07	1.93E-09	0.4311	100	1.0E-06	1	80	1	70	80	25550
Chrysene	6.15E-07	1.92E-09	0.4302	100	1.0E-06	1	80	1	70	BO	25550
Benzo(b)fluoranthene	9.79E-07	3.07E-09	0.6856	100	1.0E-06	1	80	1	70	80	25550
Benzo(k)fluoranthene	5.582-07	1.75E-09	0.3908	100	1.0E-06	1	80	1	70	80	25550
Benzo(a)ovrene	3.43E-07	1.07E-09	0.2400	100	1.0F-06	1	80	1	70	80	25550
Indeno(123cd)ovrene	6.138-06	1 92F-08	4 2918	100	1.02-06	1	80	1	70	90	25550
Bis(2ethylhexyl)ohthalate	9.12E-06	2.86E-08	6.3869	100	1.0E-06	1	80	1	70	80	25550
			•					-			
Naphthalene	9.42E-07	2.95E-09	0.6593	100	1.0E-06	1	80	1	70	80	25550
Fluorene	8.60E-07	2.69E-09	0.6017	100	1.0E-06	1	80	1	70	80	25550
Phenanthrene	1.20E-06	3.74E-09	0.8366	100	1.0E-06	1	80	1	70	80	25550
Anthracene	2.14E-07	6.71E-10	0.1500	100	1.0E-06	1	80	1	70	80	25550
luoranthene	1.93E-06	6.04E-09	1.3511	100	1.0E-06	1	80	1	70	80	25550
Pyrene	1.73E-06	5.40E-09	1.2083	100	1.0E-06	1	80	1	70	80	25550
Benzo(ghi)perylene	6.13E-06	1.92E-08	4.2918	100	1.0E-06	1	80	1	70	80	25550
Diethylphthalate	1.57E-07	4.92E-10	0,1100	100	1.0E-06	1	80	1	70	80	25550
Di-n-butylphthalate	8.02E-06	2.51E-08	5,6167	100	1.0E-06	1	80	1	70	80	25550
Di-n-octylphthalate	2.86E-07	8.95E-10	0.2000	100	1.0F-06	1	80	1	70	80	25550
Rutvihenzvinhthalate	3 435-07	1 07F-09	0.2400	100	1 0E-06	1	80	1	70	80	25550
P-(reso]	4 86F-07	1 52F-09	0 3400	100	1.0E-06	1	80	1 1	70	1 90 I	25550
Phenol	6.13F-06	1.925-08	4.2918	100	1.0E-06	1	80	1	70	80	25550
INORGANICS		 ! !		1							
Aluniniun	1 995-02	; 5 935-05	12255 5	i 100 l	1 05-06	1	90	i i	70		25550
	1.07C-VZ	1 ACC-07	1 1 2 2 3 . 5	100	1.00-06	 	ov 00		70	00	20000
	0 9(E_N)	1.V2E-V/		100	1.02-06	1 I I	ov 00	1 1	70		2000
el sento	1 705-04	5 215-07	1 0.7	100	1.00-06		0 0		70	80	20000
aryllium	1 1 1 AC_AL	2 595-00		100	1.0E-06	1	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	1 1 I	70	00	20000
ada jua	1 145-04	0,00C-V7	0.00	100	1.02-06	L 	ov 00	(L) / ()	. //	00	20000
	2 445-02	3.30E-V7	1722 0		1.00-06	1	ov 00	1 I I	70	001	25000
broniun	2.400-03	0 105_00	1/22.0	I 100 H	1.00-06		0V 00	1 I I	70	00	20000
an om tur	1 705-05		10.5	100	1.02-06	1	٥٧ ۵۸	1 I I	70	, 00 j ∣ 0∧ l	20000
NODEL		1 415-07	24.1	i 100 j			6V 60		70	00 j	20000
in an	3.170-03	1 000-04	30.1 24270 1	1 100 1	1.05-04		00 60	4 L j 1 4 J	70	00 j	20000
hee	0 LOC_AL	2.05-07	242/0.1	1 100	1.00-00		0V 00	, 1 , 1	70	00	20000
	7 000 00	2.0% ~~ //	1.00				80		70	80	20000
100100100	1 705 40		4704.0		1.05-00		6V		70	80	2000
anyanese		0.000	1250.2			1	80		70	80	20000
rei cully			~ •	100 j	1.02-06		80		/0	80	25550
	3.01E-05	7.442~08	21.1	100	1.02-06	1	80		/0	80	25550
	1.91E-03	5.9/2-06	1334.6	100	1.0E-06	1	80	1	/0	80	25550
etenlum	5./1E-07	1./91-09	0.4	100	1.0E-06	1	80		70	80	25550
OCIUM	2./4E-04	8.5/E-07	191.5	100	1.0E-06	1	80	1	70	80	25550
nallium	0.00E+00	0.00E+00		100	1.0E-06	1	80	1	70	80	25550
andium	2.40E-05	7.51E-08	16.8	100	1.0E-06	1	80	1	70	80	25550
1 NC	1.92E-04	6.00E-07	134.1	100 ;	1.0E-06	1 ;	80	1	70	80 ¦	25550
ESTICIDES/PCBS				1 I							
A-DOF	5 875-09	1 84F-10	0.0411	100	1 05-04	1	80	1	70	80	25550
4_00T	5.0/E-V0	1 945-10	0.0411	1001	1 10-14	1	90	1 L I 1 1 I	70	00 1	20000
Fj47001	J.0/L™V0	1.046-10	V.V411	1	1.00~00	1 1	OV	1 1 1		j V6	20000

DERMAL CONTACT WITH CHEMICALS IN SOIL

CAP CONSTRUCTION (ALTERNATIVES 2, 2A, 4 AND 4A)

	ABS, DOSE	ABS. DOSE	CUNC. IN	LUNVERSION	SURFACE	AUHERENCE	ABSURP 10N	LAPUSURE	LAPUSURE	BUUY	AV6. LINE	AVG. 1
CHEMICAL	NONCANCER	; cancer	SOIL	FACTOR	AREA SKIN	FACTOR	FACTOR	FREQUENCY	DURATION	WEIGHT	NONCANCER	CANC
	(mg/kg/day)	(Mg/kg/day) 	(mg/kg)	(kg/mg)	(cm2/event)	(mg/cm2);	(unitless)	(days/year)	(years)	; (Kg)	; (Oays) ;	; (day
			1	1				1				
SENIVOLATILES												
Benzo(a)anthracene	5.63E-06	1.76E-08	0.4311	1.0E-06	6300	1.45	0.10	80	1	70	80	25
Chrysene	5.61E -0 6	1.76E-06	0.4302	1.0E-06	6300	1.45	0.10	80	1	70	80	25
Benzo(b)fluoranthene	8.95E-06	2.80E-08	0.6856	1.0E-06	6300	1.45	0.10	80	1	70	80	2
Benzo(k)fluoranthene	5.10E-06	1.60E-06	0.3908	1.0E-06	6300	1.45	0.10	80	1 1	70	; 80	1 25
Benzo(a)pyrene	3.13E-06	9.81E-09	0.2400	1.0E-06	6300	1.45	0.10	80	1	70	80	2
Indeno(123cd)pyrene	5.60E-05	1.752-07	4.2918	1.0E-06	6300	1.45	0.10	80	; 1	70	80	2
Bis(2ethylhexyl)phthalate	8.33E-05	2.61E-07	6.3869	1.0E-06	6300	1.45	0.10	80	1	70	80	25
Di-N-Octylphthalate	2.61E-06	8.17E-09	0.2000	1.0E-06	6300	1.45	0.10	80	1	70	80	25
Butybenzylphthalate	3.13E-06	9.81E-09	0.2400	1.0E-06	6300	1.45	0.10	80	; 1	70	80	25
P-Cresol	4.44E-06	1.39E-08	0.3400	1.0E-06	6300	1.45	0.10	80	1	70	80	25
				1				1		1		4 1
Naphthalene	8.60E-06	2.69E-08	0.6593	1.0E-06	6300	1.45	0.10	80	1	70	80	2
Fluorene	7.85E-06	2.46E-08	0.6017	1.0E-06	6300	1.45	0.10	80	1	70	80	2
Phenanthrene	1.09E-05	3.42E-08	0.8366	1.0E-06	6300	1.45	0.10	80	1 1	70	80	2
Anthracene	1.96E-06	6.13E-09	0.1500	1.0E-06	6300	1.45	0.10	80	1	70	80	2
Fluoranthene	1.76E-05	5.52E-08	1.3511	1.0E-06	6300	1.45	0.10	80	1	70	80	2
Pyrene	1.58E-05	4.94E-08	1.2083	1.0E-06	6300	1.45	0.10	80	1	70	80	2!
Benzo(ghi)perylene	5.60E-05	1.75E-07	4.2918	1.0E-06	6300	1.45	0.10	80	1	70	80	2
Diethylphthalate	1.44E-06	4.49E-09	0.1100	1.0E-06	6300	1.45	0.10	80	1	70	80	2
Di-n-butylphthalate	7.33E-05	2.30E-07	5.6167	1.0E-06	6300	1.45	0.10	80	1	70	80	2
Phenol	5.60E-05	1.75E-07	4.2918	1.0E-06	6300	1.45	0.10	80	1	70	80	2
INORGANICS	 	 	 	 1 +			 ! !	 	I 1	 ! !	• • • • • • • • • • • • • • • • • • •	 ! !
			 	1 1								
Aluminium	1.732-02	5.421-05	13255.5	1.0E-06	6300	1.45	0.01	; 80		1 70	80	
Antimony	2.982-05	9.32E-08	22.8	1.05-06	6300	1.45	0.01	i 80		i /V	; 80 j	
Arsenic	9.00E-05	2.82E-07	6.9	1.0E-06	6300	1.45	0.10	80		/0	80	; 2
Barium	1.55E-04	4.85E-07	118.8	1.0E-06	6300	1.45	0.01	80		; /0	; 80;	2
Beryllium	1.04E-06	3.27E-09	0.80	1.0E-06	6300	1.45	0.01	80	1	/0	80	2
Cadeium	1.04E-06	3.27E-09	0.80	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Calcium	2.25E-03	7.04E-06	1722.8	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Chroniun	2.39E-05	7.485-08	18.3	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Cobalt	1.63E-05	5.11E-08	12.5	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Copper	4.71E-05	1.48E-07	36.1	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Iron	3.17E-02	9.92E-05	24278.1	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Lead	7.84E-05	2.46E-07	60.1	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Magnesium	6.47E-03	2.02E-05	4954.3	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Manganese	1.63E-03	5.11E-06	1250.2	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Nickel	2.75E-05	8.62E-08	21.1	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Potassium	1.74E-03	5.45E-06	1334.6	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Selenium	5.22E-07	1.63E-09	0.4	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Sodium	2.50E-04	7.82E-07	191.5	1.0E-06	6300	1.45	0.01	80	1	70	80	2
Vandium	2.19 E-05	6.86E-08	16.8	1.0E-06	6300	1.45	0.01	80	1	70	80	25
Zinc	1.75E-04	5.48E-07	134.1	1.0E-06	6300	1.45	0.01	80	1	70	80	2!
		i		, 					; !	1		; ; ;
PESTICIDES/PCBS		1				1			!			<u>.</u>
PESTICIDES/PCBS	5 315-A7	i i i i t i <u>kor</u> -no -	 0 0411	1 ^2 _^	1 100	1 1 1 1 1	A 1A	<u>م</u>	1 	 70	 	1 1 1
PESTICIDES/PCBS	5.36E-07	1.68E-09	0.0411	1.0E-06	6300	1.45	0.10	80	1	70	80	2

APPENDIX 8

INHALATION OF CHEMICALS BORN ON DUST PARTICLES CAP CONSTRUCTION (ALTERNATIVES 2, 2A, 4 AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

	INTAKE	INTAKE	DUST CONC.	CONC. IN	INHALTION	EXPOSURE	EXPOSURE	EXPOSURE	BODY	AVG. TIME	AVG. TIM
CHENICAL	NONCANCER	CANCER	ON SITE	SOIL	RATE	TIME	FREQUENCY	DURATION	MEIGHT	NONCANCER	CANCER
	(mg/kg/day)	¦(mg/kg/day)	(kg/m3)	¦(∎g/kg)	(ma3/hour)	(hours/day)	(days/year)	¦(years)	(kg)	¦(days)	(days)
THODCANTCS	1	1 1 1	1 1 1	1	1 1 1)] 	1	1 1 1		8	
inininin	1 2 635-02	1 1 8 235-05	1 1 8 695-06	1 12255 5	, ,	i 1		1	I '7∩	1 1 90	25550
	1 A 505-05	1 0.23C-VJ	0.00C-V0	1 10200.0	1 2		0V 00	1 I.	/V ⊐∧	1 0V	2000
RILLINORY	1 4.32E-V3	1.425-0/		1 22.0	1 4 I	1 01 1 01	0V	1 1	1 70	1 OV:	20000
AT SENIC	1.3/E-05	1 4.2% ⁻ 00	0.00E-00	0.7		i 8 i	80		i 70	80	20000
Sar lum	2.365-04	/.381-0/	8.661-06	118.8	i 2	i 8;	80	i 1,	/0	i 80	25550
Seryllium	1.59E-06	4.9/1-09	8.68E-06	0.8	2	8	80	1	70	80	25550
admium	1.59E-06	4.97E-09	8.68E-06	0.8	2	8	80	1	70	80	25550
Calcium	3.42E-03	1.07E-05	8.68E-06	1722.8	2	8	80	1	70	80	25550
Chromium	3.63E-05	1.14E-07	8.68E-06	18.3	2	8	80	1	70	80	25550
Cobalt	2.48E-05	7.77E-08	8.68E-06	12.5	2	8	80	1	70	80	25550
opper	7.16E-05	2.24E-07	8.68E-06	36.1	2	8	80	1	70	80	25550
Iron	4.82E-02	1.51E-04	8.68E-06	24278.1	2	; 8;	80	1	70	80	25550
ead	1.19E-04	3.73E-07	8.68E-06	60.1	2	8	80	1	70	80	25550
lagnesium	9.83E-03	3,08E-05	8.68E-06	4954.3	2	8	80	1	70	80	25550
langanese	2.48 -03	7.77E-06	8.68E-06	1250.2	2	8	80	1	70	80	25550
lickel	4.195-05	1.31F-07	8.68F-04	21 1		R R	80	1	70	80	25550
Intaccium	2 455-03	0 70F-N	1 8 49E-04	1 1334 6	·		80	1	70	1 90	20000
odium	1 2 005-04	1 105-04	1 0.00C 00	101 E	1 2	i 01	۵0 ۵۸	1 A I	70	1 00	25550
lonadiun.	1 0.00C-04	1.17E-VO	0.000-00	1 171.0		1 01 1 01	0V 00	1 I I	1 /V	i 00 i	20000
anaolum	1 3.33E-V3	1.046-0/	0.000-00	10.0 1 10.0		i 0 i	ev 	1 L I	· /V	, 00 j	20000
1nc	2.665-04	8.33E-0/	8.682-06	; <u>1</u> 34.1	2	; 8;	80		/0	80	25550
VOLATILES	 		- 1 } f	1 1						1 1	
Carbon Disulfide	5.95E-09	1.86E-11	8.68E-06	0.0030	2	8	80	1	70	80	25550
hlorobenzene	1.39E-08	4.35E-11	8.68E-06	0.0070	2	8	80	1	70	80	25550
thylbenzene	4.68E-08	1.47E-10	8.68E-06	0.0236	2	8	80	1	70	80	25550
in luene	2 30F-08	7 21F-11	8 68F-06	0 0116	2	8	80	1	70	80	25550
(ylenes	8.37E-07	2.62E-09	8.68E-06	0.4218	2	8	80	1	70	80	25550
SEMIVOLATILES	4 4 1 1										
Ris(2-ethylheyyl)ohthalate	1 27F-05	3 97F-08	8 68E-06	6 3869	2	8	80		70	80	25550
i-N-Octy]ohthalate	3 975-07	1.245-09	8 695-04	0 2000		2 Q I	20	1	70		25550
a noocyphicialace hitvhenzvlohthelate	1 745-A7	1 405-00	8 695-04	0.2000	· · · ·	<u>0</u> 1	90 90	· 1	70		2000
henent hrene	1 645-04	5 205-00	8 695-04	0 8377	i 41	<u>0</u> 1	۵۷ ۵۸	1 A I	70	, ov i	20000
luoranthene	2 495-14	8 205-NO	8 K95-M	1 2611	2 ()	0	۵V ۵۸	i ∔i ! 1	70	00	20000
	1 2.00L VO	7 515-00	6 105 VC	1 2002	1 41 1 31	0 I	۵0 ۵۸	1 A I	70	1 OV 1	20000
ji unu ji Ni-Distvlohthalata	1 110-AC	2 40C-00	0.00C-V0	L 1.2000	1 4 I	0 i	٥٧ ۵۸	1 <u>1</u>	70	1 0V 1	20000
annois Asid	1 1.11E-V3	1 275-00	0.000-00	1 2.010/	1 41 1 51	1 0 j	0U 6A	1 1 i	70	1 0V I 0A	20000
enzoic Ació Nonol	1 4.30C-V/	1.3/2-07	0.0000-00	1 4 2010		. 0; 	5U ^^	i Li	//	i 00'j ì ∧∧ ì	20000
Incirol	+ 0.01E-00	2.0/1-00	0,0002-00	1 4.2710 1 0 0 000		6	80		/0	; 80	20000
TCT CSO1	6./5±-0/	2.11E-09	0.00L-V6	0.3400	2	8	80		/0	80	25550
letnylphthalate	2.18E-07	6.83E-10	8.68E-06	0.1100	2	8	80	1	70	80	25550
luoranthène	2.68E-06	8.39E-09	8.68E-06	1.3511	2	8	80	1	70	80	25550
nthracene	2.98E-07	9.32E-10	8.68E-06	0.1500	2	8	80	1	70	80	25550
luorene	1.196-06	3.74E-09	8.68E-06	0.6017	2	8 ;	80	1	70	80 ;	25550
aphthalene	1.31E-06	4.10E-09	8.68E-06	0.6593	2	8	80	1	70	80	25550
enzo(a)anthracene	8.55E-07	2.68E-09	8.68E-06	0.4311	2	8	80	1	70	80	25550
enzo(a)pyrene	4.768-07	1.49E-09	8.68E-06	0.2400	2	8	80	1	70	80	25550
hrysene	8.54E-07	2.67E-09	8.68E-06	0.4302	2	8	80	1	70	¦ 80 ¦	25550
enzo(b)fluoranthene	1.36E-06	4.26E-09	8.68E-06	0.6856	2	8	80	1	70	80	25550
enzo(k)fluoranthene	7.75E-07	2.43E-09	8.68E-06	0.3908	2	8	80	1	70	80	25550
mainmanna	immini								111111	mmi	

.

•

1

		SUMMER OF G	HERTEL LAN	DFILL FEAS	IBILITY STUDY	hilles 2, 20,	, no - <i>n</i>)	
								* * * * * * * * * * *
1	CHRONIC DAILY	CDI		1	1		CHEMICAL	TOTAL
CHEMICAL	INTAKE(CDI)	ADJUSTED FOR	SF	WEIGHT OF	TYPE OF	SF BASIS/	SPECIFIC	PATHMAY
	(mg/kg/day)	ABSORPTION	(mg/kg/day)-1	EVIDENCE	CANCER	SOURCE	RISK	RISK
						************		****
EXPOSURE PATHWAY: INGESTIC	ON OF CHEMICAL	S IN SOIL						2E-07
INORGANICS		1		1	t I	1		
Arsenic	3.09E-08	No	1.75	A	Skin	Water/IRIS	5E-08	
Beryllium	3.58E-09	No	4.3	¦ 82	Skin	Water/IRIS	2E-08	
Lead	2.69E-07	No	ND	B2	Renal	Oral/IRIS	NA	
SENIVOLATILES		1	 		1	:		
Bis(2ethylhexyl)phthalate	2.86E-08	No	1.4E-02	82	Liver	Diet/IRIS	4E-10	
Butylbenzylphthalate		No	ND	C	6 4 3	NF/IRIS	NA	
Benzo(a)anthracene	1.93E-09	No	11.50	B2	- 	NA/IRIS	2E-08	
Benzo(a)pyrene	1.07E-09	No	11.50	82	Stomach	Diet/IRIS	1E-08	
Chrysene	1.92E-09	No	11.50	B2	 	NA/IRIS	2E-08	
Benzo(b)fluoranthene	3.07E-09	No	11.50	B2	1	NA/IRIS	4E-08	
Benzo(k)fluoranthene	1.75E-09	No	11.50	B2	- 	NA/IRIS	2E-08	
Dibenzo(a,h)anthracene	NA	No	11.50	82	1	NA/IRIS	NA	
Indeno(123cd)pyrene	NA	No	11.50	82	1	NA/IRIS	NA	
1,4-Dichlorobenzene	NA	No	2. 4E-0 2	82	Liver	Gavage/HEAST	NA	
PESTICIDES/PCBS		1	 	! !	1 1 1	 ! ł	1 1	
4,4'-DDD	NA	No	2.4E-01	B2	Lung, Liver, Thyroid	Diet/IRIS	NA	
4,4'-DOE	NA	No	3.4E-01	B2	Liver, Thyroid	Diet/IRIS	NA	
4,4'-DDT	NA	No	3.4E-01	B2	Liver	Diet/IRIS	NA	
		11111111111111					1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	

SUMMARY OF CANCER RISK ESTIMATES - CAP CONSTRUCTION (ALTERNATIVES 2, 2A, 4, AND 4A)

			HERTEL LA	NOFILL FEAS	SIBILITY STUDY		•, •••	
111111111111111111111111111111111111111								
11	CHRONIC DAILY	CDI	}				CHENICAL	TOTAL
CHENICAL	INTAKE(CDI)	ADJUSTED FOR	SF	NEIGHT OF	TYPE OF	SF BASIS/	SPECIFIC	PATHNAY
	(mg/kg/day)	ABSORPTION	(mg/kg/day)-1	EVIDENCE	CANCER	SOURCE	RISK	RISK
LEXPOSURE PATHWAY: DERMAL	CONTACT WITH C	ENICALS IN S	DIL					5E-06
INORGANICS	1		1					
Arsenic	2.82E-07	0.10	1.75	A	Skin	Water/IRIS	5E-06	
11								1
SENIVOLATILES			1					
Bis(2ethylhexyl)phthalate	2.61E-07	0.10	1.4E-02	B2	Liver	Diet/IRIS	4E-08) •
Butylbenzylphthalate	NA	0.10	ND	C I		NF/IRIS	NA	
11				1				j •
Benzo(a)anthracene	1.76E-08	0.10	ND	82		NA/IRIS	NA]
Benzo(a)pyrene	9.81E-09	0.10	ND	B2	Stomach	Diet/IRIS	NA	1 1
Chrysene	1.76E-08	0.10	ND	B2		NA/IRIS	NA	
Benzo(b)fluoranthene	2.80E-08	0.10	ND	82		NA/IRIS	NA]
Benzo(k)fluoranthene	1.60E-08	0.10	ND	82		NA/IRIS	NA	J
Dibenzo(a,h)anthracene	NA	0.10	ND	82		NA/IRIS	NA) •
Indeno(123cd)pyrene	NA	0.10	ND	82		NA/IRIS	NA	
) [1		
1.4-Dichlorobenzene	NA	0.10	2.4E-02	B2	Liver	Gavage/HEAST		
11								
PESTICIDES/PC8S			1	!		!		k
14,4°-000	NA	0.10	2.4E-01	82	Lung, Liver, Thyroid	Diet/IRIS	NA	1
4,4'-DOE	NA	0.10	3.4E-01	82	Liver,Thyroid	Diet/IRIS	NA	
14,4'-DDT	NA	0.10	3.4E-01	B 2	Liver	Diet/IRIS	NA	
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			<u>****</u> ********************************	, , , , , , , , , , , , , , , , , , ,		, , , , , , , , , , , , , , , , , , ,		J

SUMMARY OF CANCER RISK ESTIMATES - CAP CONSTRUCTION (ALTERNATIVES 2, 2A, 4, AND 4A)

				APPENDIX	8				
		summary of c	ANCER RISK EST	INATES - C	AP CONSTRUCTION (ALTERN RUITY STUDY	ATIVES 2, 2A,	4, AND 4A)	I	
						****	* * * * * * * * * * * * *		
	CHRONIC DAILY	CDI	1	1	1	1	CHEMICAL	TOTAL	TOTAL
CHEMICAL	INTAKE(CDI)	ADJUSTED FOR	SF	WEIGHT OF	TYPE OF	SF BASIS/	SPECIFIC	PATHMAY	EXPOSURE !
l l	(mg/kg/day)	ABSORPTION	{(mg/kg/day)-1	EVIDENCE	CANCER	SOURCE	RISK	RISK	RISK
1 6 1 6 1 6 1 6 1 6 1 6 1 6 6 6 6 6 6 6	 	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1 1 4 5 1 1 5 1 5 4 5 5 5 5 5 5 5 5 5 5	****	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1		1 1 1 1 1 4 1 4 1 1 1 1 1 1 1 1 4 1 4 1	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
EXPOSURE PATHWAY: INHALAT	ION OF AIRBORN	e chemicals a	dsorbed to dus	Т				7E-06	1E-05
							, , , , , , , , , , , , , , , , , , ,		
INORGANICS					1				
Arsenic	4.29E-08	No	50	A	Lung	Air/HEAST	2E-06		
Beryllium	4.97E-09	No	8.4	1		1	4E-08		
Cadmium	4.97E-09	No	6.1	81		OCCUPA/HEAST	3E-08		
Chromium	1.14E-07	No	41	A	Lung	OCCUPA/HEAST	5E-06		
Nickel	1.31E-07	No	; 8.4E-01	: A	Respiratory tract	;OCCUPA/HEAST	1E-07		
SEMIVOLATILES			1		1	1	1		
Bis(2ethylhexyl)phthalate	3.97E-08	No	1.4E-02				6E-10		
Butylbenzylphthalate	NA	No	ND	C	1	NF/IRIS	NA	1	
			1	4 1		1	1	1	
Benzo(a)anthracene	2.68E-09	No	6.1	B2	1	NA/IRIS	2E-08	1	
Benzo(a)pyrene	1.49E-09	No	6.1	B2	Stomach	Diet/IRIS	9E-09	:	
Chrysene	2.67E-09	No	6.1	B2	Respiratory tract	NA/IRIS	2E-08	1	
Benzo(b)fluoranthene	4.26E-09	No	6.1	82	1	NA/IRIS	3E-08	1	
Benzo(k)fluoranthene	2.43E-09	No	6.1	82	1	NA/IRIS	1E-08	1 4	
Dibenzo(a,h)anthracene	NA	No	6.1	82	1	NA/IRIS	NA		
Indeno(123cd)pyrene	NA	No	6.1	B 2	1 4	NA/IRIS	NA	1 (
1,4-Dichlorobenzene	NA	No	2.4E-02	; ; ;	; ; ; ; ;	i 	NA	 	
PESTICIDES/PCBS			1	1	1 1 1	1	1 F	1 	
4,4'-000	NA	No	2.4E-01	82	Lung, Liver, Thyroid	Diet/IRIS	NA	:	
4,4'-DOE	NA	No	3.4E-01	82	Liver, Thyroid	Diet/IRIS	NA	1	
4,4'-DDT	NA	No	3.4E-01	82	Liver	Diet/IRIS	NA	l f	

SUMMARY OF CHRONIC HAZARD INDEX ESTIMATES - CAP CONSTRUCTION (ALTERNATIVES 2, 2A, 4 A	ID 4A
---	-------

					HERTEL LANDFILL FEASIBILITY STUDY			,	,	
+ + + + + + + + + + + + + + + + + + +										
	SUBCHRONIC		050			RED	RFD			PATHWAY
	UAILT INTAKE	ADJUSTED FUR	i KrU KrU	LUNP IDENCE		SOURCE/	UNCERTAINTY	HOULFYING	HAZARD	HAZARD
14 1441 444 1241 141 81 81 41 41 434 141 54	[[#9/ kg/udy]		(mg/kg/ddy)	1 LEVEL	, ErrELi	j DA515	AUJUSIMENIS	FALTURS	QUUTEN	INDEX (HL);;
EXPOSURE PATHWAY: INGESTI	ON OF CHEMICAL	5 TN 5011			,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,					15-01
				0.000000			,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			
TNORGANICS	!	1		1	L # 1 # 9 # # # 8 L # 8 L # 8 H L L # # # L L # # E L L # # E L E L # # E L E L	, , , , , , , , , , , , , , , , , , ,	1 / 1 / / / / 1 / 1 / 1 / / / / / / / /			, , , , , , , , , , , , , , , , , , ,
Aluninum	1.9F-02	No	DT			1		1	NA.	
Antimony	3.3E-05	No	4E-04	Low	Longevity, blood glucose and	1			8F-02	
1	1				cholesterol		1			
Arsenic	9.9E-06	No	1E-03	NA	Keratosis and hyperpigmentation		1	•	1F-02	
Barium	1.72-04	No	5E-02	Medium	Increased blood pressure				37-03	
Beryllium	1.1E-06	No	5E-03	Low	None observed	}			28-04	
Cadmium	1.1E-06	No	NO	High	Proteinuria		1		NA	
11	1									
Chromium VI	2.6E-05	No	2E-02	LOW	No effects reported			1	1E-03	
Cobalt	1.8E-05	No	NF			1			NA	
Copper	5.2E-05	No	4.0E-02	NA	Local GI irritation			1	1E-03	
Lead	B.6E-05	No	NA	NA	Neurobehaioral effects	1	1		NA	
Manganese	1.8E-03	No	1E-01	Medium	CNS effects		1	1	2E-02	
Hercury	0.0E+00	No	3E-04	NA	Kidney effects	1	,	•	NA	1
Nickel	3.0E-05	No	2E-02	Medium	Decreased body and organ weight	1	1		2E-03	
Selenium	NA	No	NF	,			1	1	NA	1
Vanadium	2.4E-05	No	7E-03	NA	None observed	1	1	,	3E-03	
Zinc	1.9E-04	No	2E-01	NA	Anexia	;	1	1	1E-03	1
Cyanide	NA	No	2E-02	Medium	Weight loss, thyroid effects,	1			NA	
				1	syelin degeneration	1	1	1	,	1
				 '						•
II SERIVULATILES	1	1				1	í.			
Pic(Onthe Ihave 1 Johtha lata	0 15-04	1 No.	25 42	i Madius	i I terminal a latin the solution				NA	
1 DIS(Zechy Linexy 1 Aprichadate	1 7.10-00	i No	20-02	neolun Leu	Increased relative liver weight	Diet/IRIS	1000.00	1.00	5E-04	
i purvipenzy ipninarate	1 3.42-07	NO	26100	LOW	Effects on body; weight gain,	Diet/IRIS	100.00	1.00	2E-07	
Renzo(a)anthracene	1 1 6 2E-07		NF.		i testes, liver, kidneys				NA	
Renzo(a)ovrene	3 AF-07	i No		•	1 1	HEASI, IKIS	1	1	NA NA	
Chrysenet	6 1F-07	No	DT DT	1	1	INCHOI, IKID	,	i r	i NR	1
Benzo(b)fluoranthene	9.8F-07	No	NF	•	1	I NCHOI I NEAST IDIS	1	4	1 NH 1 NA	
Benzo(k)fluoranthene	5.6E-07	No	NF			HEAST, INIS	1	1	1 NPA	1
Dibenzo(a,h)anthracene	NA	No	NF			HEAST IRIS	1	1	1 10H	l t
Indeno(1,2,3-c,d)pyrene	NA	No	NF			HEAST IRIS	1	1	I 100	1
	1							1	1 100	
1,4-Dichlorobenzene	NA NA	No	ND		- 1 1	HEAST, IRIS			NA	
1	1			1	1		i	,) ;		
Diethylphthalate	1.6E-07	No	8E+00	Low	Decreased growth rate, food	Diet/IRIS	1000.00	1.00	2E-08	
					consumption rate, altered		1) 		
	1			1	organ weights					
Di-n-butylphthalate	NA	No	1E+00	Low	Increased mortality	Diet/IRIS	1000.00	1.00	NA	
Di-n-octylphthalate	2.9E-07	No	2E-02	NA	Elevated liver and kidney weights	Diet/HEAST	1000.00		1E-05	
					increased SGOT and SPGT	1	1	1	;	1
							1		NA	
Acenaphtnene	NA NA	NO	6E-01	LOW	Hepatotoxicity	Oral/IRIS			NA	•
HILDI ACENE	1. 2.15-07	NO	31:+00	LOW	No observed effects	Gavage/IRIS			7E-08	
Benzo(g,n,1)perviene	i NA 1.05.0/	i NO i	Nt AT AL			HEAST/IRIS	1		NA	
r Iuorantnene	1.92-06	NO i	4E-01	LOW	Neuropathy, liver weight changes	Gavage/IRIS			5E-06	
1	•			1	nematological alterations and		i			
Eluorene	8 45-07	No	45-01	1	Clinical effects	Causes (IDIC			NA	
1	1 0.00 0/		46-01		volume and hemoglobin	eavage/1K15	1			
p-Cresol	9 4F-07	No	NF	1	AATOME TIN HERAATOTI	HEAST TOTO			L MA	
Naphthalene	9 45-07	No	45-02	No	Ocular and internal lecions	Diet/UEACT	1		1 10H	
Pyrene	1.7F-06	No	35-01	104	Kidney effecte	Cavage/1010	l 9		45.00	
Phenanthrene	1 2F-04	No	01	LUN	RIDICY CITECIS				0C-06	
Phenol	6 1F-06	No	0.60		Peduced fetal hady usight		100		NH I	
			v.ov		Neurce letal Duy Melgin	aavaye/nCHS)	100		1	
PESTICIDES/PCBS										
4,4'-000	NA	No	NF			HEAST, IRIS			NΔ	
4,4'-DDE	NA	No	NF			HEAST, IRIS			NA	
4,4'-DOT	NA	No	5E-04	Medium	Liver lesions	Diet/IRIS	100	1.00	NA	

SURVARY OF CIRCUIC INVARID THEEX ESTIMATES - CAP CONSTRUCTION (ALTERNATIVES 2, 2A, 4 AND 4A) HERTEL LANGFILL FEASIBILITY STUDY

		, , , , , , , , , , , , , , , , , , ,									
1	: 1	SUBCHROHIC	C01		1		RF0	RFD	1	1	PATHNAY
- i	CIENICAL C	DAILY INTAKE !/	DJUSTED FOR!	RED	CONFIDENCE!	CRITICAL	SOURCE/	REERTAINIT	KOD IFY ING	HAZARD	HAZARD
į		(mo/ko/day) !	ABSORPTION	(po/ko/day)	LEVEL	EFFECT	BASIS	ADJUSTHENTS	FACTORS 10	UOTIENT	INDEX (HI)
1											
1	EXPOSING DATALINY DEPEND OF		111111111111111 ΩΝΤΓΛΙ Ο ΤΝΙ ΟΛ	111111111111 1t							9F-01
- 1	ENFUSIONE PHILIPPIC DENIER CO			1						11111111	
1			**********		munni		*************		******		
÷	INORGANICS		i						į.		
į	Arsenic	9.0E-05	0.10	1E-03	NA	Keralosis and hyperpigmentation	NA/HEAST	1 ;		9E-01	
ł									·;		
1	SENIVOLATILES										
						1	-				1
	Bis(2ethylhexyl)phthalate	8.3E-05	0.10	2 E -02	Hedium ;	Increased relative liver weight	Diet/IRIS	1000	1	4E-02	1
	Butylbenzylphthalate	3.1E-06	0.10	2E+00	LOW	Effects on body; weight gain,	Diet/1R1S	100	1	2E-0S	1
. !					: :	testes, liver, kidneys			1		
•	Benzo(a)anthracene	5.6E-06	0.10	15	1		IEAST, IRIS		1	NA	i i
j	Benzo(a)ovrene	3.1E-06	0.10	NF	i i		HEAST, IRIS			NA	1
į	Chrysene	5 AF-06	0.10	DI			HEAST		1	NA	
	1 Penzo(h) Duoranthene	B 95-06	0.10	' NE	, , , ,		10,01			No	1
	Penzo(b) fluoranthene	E 15-06	0.10	1 10	1 1		WAST IDIS	, I	1 1	105	
	Hotherse(a blotterene	1 J.10-00	0.10	1 10	1 1		NCAST, 1010			NA NA	1
	1, UIDenzo(a ,n Janum acene	1 104	, 0.10	, IV \/	1 1		NEASE INTS			NH NA	1
	11 Indeno(1,2,3-0,0)pyrene	I RH	0.10	1 10	: :		IKHSI, IKIS	1		194	1
				1							i
	, 1,4-Dichioropenzene	ПА	0.10	i NU			MLAST, INIS		1	na	i i
											1
	Diethylphthalate	1.48-06	0.10	8E+00	Lou	Decreased growth rate, food '	Diet/IRIS	1000	1	2E-06	1
	11				1	consumption rate, altered		1		1	
	11	1	1	1		organ weights		1	1	1	
	Di-n-butylphthalate	NA	0.10	1E+00	LOW	Increased portality	Diet/IR1S	1000	1	NA	1
	Di-n-octylphthalate	2.6E-05	0.10	2E-02	ha i	Elevated liver and kidney weights	Diel/HEAST	1000	1	1E-03	
	11	1	0.10	;	1	increased SGOT and EFGT		1	1	;	1
	N	;	0.10	1	1			1	1	1	1
	: Acenaphthene	NA	0.10	6E-01	Low	Hepatotoxicity	Ora]/]R S	1	1	NA	1
	l'Anthracene	2.0E-06	0.10	3E+00	LOW	No observed effects	Gavage/IRIS			7E-06	5
	Benzola h i cervlere	NA	0 10	NF NF		1	HEAST/IRIS		1	NA	
	Fluoranthena	1 RF-05	0.10	4F-01	L ON	. Neuropathy, liver weight changes	Gavage/IRIS	1		45-04	4 !
	()	1.00.00	0.10	1	1	hematological alterations and	1 014030/1010	i	1		
			1	1	1	i clinical offorto	8	1	1	t	
	11 Usharara	1 7 05 64	1 0.10	1 45 61	1	Decreased CC pashed call	1 1 Course (1015		1	1 25-0	
	1 /r ruor ene	1 7.92-05	0.10	1 4 <u>C</u> -U	LCW	Decreased ADL, packed Cell	1 ogragerikis	-	1	1 21-0	1
		1	1		1	voluse and nesoglobin	1 1014	1		1	
	[P-Creso]	4.4E-06	0.10	5E-0	Medium	pecreased body weight, Heurotoxicity	IRIS	1000	1	9E-0	5
	Raphthalene	B.6E-06	0.10	4E-0	NA	Ocular and internal lesions	Diel/HEAST	i		2E-0	3 11
	Fyrene	1.6E-05	0.10	3E-0	Low	Kidney effects	Gavage/IRIS			SE-0	4 ;
	Phenanthrene	1.1E-05	0.10	DI			HEAST			161	· · ·
	Fhenol	5.68-05	0.10	0.0		Reduced fetal body weight	Gavage/IEAST	100	;	9E-0	14
-		-				-					
	FESTICILES/FCBS		0.10) [1		1	1	1	1	Ý
	4 ,4 '-CCO	NA	0.10	11	1	1	HEAST, IRIS	1	;	NA	1
	14,4°-00£	NA	0.10	D NF	1	1	HEAST, IRIS	1	1	NA	ŀ
	14,4'-DOI	i NA	0.1	0 ¦ 5E-0	4 ¦ Medium	Liver lesions	Diet/IRIS	100	1	NA	

· .

. .

•			ទហ	MARY OF CHR	ONIC HAZARD INDEX ESTIMATES - CAP CONST HERIFL LANDFILL FEASIBILITY STUDY	TRUCTION (ALTERN	ATIVES 2, 2A,	4 AND 4A)		
							1			111111111111111111111111111111111111111
	SUBCHRONIC	COI				RFD	RFD		1	PATHWAY TOTAL
CHEMICAL	DAILY INTAKE	ADJUSTED FOR	RFD	CONFIDENCE	CRITICAL	SOURCE/	UNCERTAINTY	HOD IFY ING	HAZASD	HAZARD EXPOSURE
1	(mg/kg/day)	ABSORPTION	(mg/kg/day)	LEVEL	EFFECT	BASIS	ADJUSTHENTS	FACTORS	OUOTIENT 🕴	INDEX (HI); HI ;;
EXPOSURE PATHWAY: INHALAT	ION OF AIRBORN	E CHEMICALS AD	SORBED TO D	UST	1641245454442441221244441244444444444444					1E+01 1E+01
1 INORGANIES	, , , , , , , , , , , , , , , , , , ,	!		!			1			
Antimony	4.5E-05	No	4E-04						18-01	
Arsenic	1.4E-05	No	1E-03						1E-02	
Barium	2.4E-04	No	1E-03				i i		28-01	
Beryllium	1.6E-06	No	0.01				1		3E-04	1
Cadmium	1.6E-06	No	1E-03				1		2E-03	
Chromium	3.6E-05	No	6E-06	NA	Nasal nucosa atrophy	HEAST	300		6E+00	1
Hanganese	2.5E-03	No	42-04	Medium	Increased prevalence of	IRIS	300	3 1	68+00	1
			1		respiratory symtoms and psycho-	HEAST	900			1
1		1			eotor disturbances		1			J I
Hercury	NA	No	1E-04	NA	Neurotoxicty	NA/HEAST	30		NA	1
Nickel	4.2E-05	No	28-02						2E-03	
Vanadium	; 3.3E-05	i No	; /E-03						5E-03	1
1210C	1 Z./E-04	i NO	20-01	1					1E-03	
C7 d 1 UC	1 MH	I NU	, 20-02	,	1 +		1		NA	1
VOLATTIES		!	1	1	, , , , , , , , , , , , , , , , , , ,	1	}			
Carbon Disulfide	NA	No	18-01			•			NA	
Chlorobenzene	1.4E-08	No	5E-02	NA	Liver and kidney effects	HEAST	10000		3¥-07	
Ethylbenzene	4.7E-08	No	1E+00				-		SF-09	
Toluene	2.3E-08	No	25+00	NA	CNS effects, eyes and nose irritation	HEAST	100		18-08	
Xylenes	8.48-07	No	3E-01	NA	CNS effects, eyes and mose irritation	HEAST	100		3E-06	*
					l					1
SENIVOLATILES	1	1	1	1	;	1	1	1	2	1
			1		:	1	1	:	1	1
Bis(2ethy/hexyl)phthalate	1.3E-05	No	2E-02					•	6E-04	1
Butylbenzylphthalate	4.8E-07	oli	2E+00					1	2E-07	
(Benzo(a)anthracene	NA	NO	i NF			HEAST, IRIS			NA	
Libenzol a Jpyrene	NR NA	i Mo		i i	1	HEAST IRIS			INA INA	
Benzo(h)fluoranthene	1 DR	I NO		1	1	i HEASI	i		i Na	1
Renzo(L)fluoranthene	1 NA	No		1	1	I DEADI, INID	1		j NA L NA	1
Dibenzo(a, h)anthracene	NA	1 10	. NF	1	1	1 HEAST 1015		1	1 104 1 NA	1
Indeno(1.2.3-c.d)pyrene	NA	No	NE	1		HEAST, IRIS		1	1 104	
1,4-Dichlorobenzene	NA	No	2E-01	NA	Liver and kidney effects	HEAST	100		No	
Diethylphthalate	2.2E-07	No	82+00		1			1	NA	
Di-n-butylphthalate	NA	No	1E+00	i					NA	1
Di-n-octylphthalate	4.0E-07	¦ No	2E-02	:	1			i	NA	1
Acenaphthene	NA	No	6E-01	1	-	1			NA	1
(Anthracene	3.0E-07	No	3E+00	1		1		1	NA	
Benzo(g,h,i)perylene	NA	No	NF	1		HEAST/IRIS	1	-	NA	1
Fluoranthene	2.7E-06	No	48-01	1		1	1		7E-06	1
; Fluorene	1.2E-06	No	4E-01					1	3E-06	
ip-tresol	6.7E-07	No	5E-01						1E-06	
Indphilalene	1.3E-06	NO	48-02						312-05	
Fryrene L'Chapanthrana	2.41-06	No	3E-01	1					8E-0 6	
Phenal	I NA	I NO	1 01	1	5 1	HLAST			NA	
[] [] [] [] [] [] [] [] [] [] [] [] [] [1	i MLASI		1	NA	1
PESTICIDES/PCBS						1				1
4.4'-DDD	NA	No	NE			HEAST 1D1C			1 MA	1
4 4'-DCE	NA	No	NF		• .	HEAST IRIS			NA	
4.4'-DDT	NA	No	5E-04		Liver lession	1 16491 1413			1 210	

;;4,4⁻²001 ; NA ; No ; 5E-04 ; ; Liver lession ; ; ; ; NA ;

INGESTION OF CHEMICALS IN SEDIMENTS - CHILDREN (ALTERNATIVES 2 AND 2A) HERTEL LANDFILL FEASIBILITY STUDY

CHENICAL	INTAKE (NONCANCER)	INTAKE (CANCER)	CONC. IN SEDIMENTS	INGESTION RATE	CONVERSION Factor	FRACTION	EXPOSURE	EXPOSURE DURATION	BODY WEIGHT	AVERAGING TIME	AVERAGING TI
	(mg/kg/day)	(mg/kg/day)	(∎g/kg)	(mg soil/day)	(1E-6 kg/mg)	(unitless)	(days/year)	(years)	(kg)	(days)	(days)
SENTIVOLATILES									1		
1.4-Dichlorobenzene	0.0E+00	0.0E+00	NA	100	1F-06	1.00	12	9	49	3285	2555
Benzo(a)anthracene	7.7E-08	9.95-09	1.1500	100	15-06	1 00	12	á	49	3285	2555
Benzo(a) pyrene	0.0E+00	0.0E+00	INA	100	1E-06	1.00	12	é	49	1 3285	2000
Benzo(b)fluoranthene	5.4E-08	6.9E-09	0.7990	100	15-06	1.00	12	ģ	49	3285	2555
Benzo(k)f luor ant hene	0.0E+00	0.0E+00	NA	100	1F-06	1.00	12	ģ	49	1 3285	2555
Bis(2ethylhexyl)phthalate	1.5E-07	1.9E-08	2.2300	100	1E-06	1.00	12	9	49	3285	2555
Butylbenzylphthalate	0.0E+00	0.0E+00	NA	100	1E-06	1.00	12	9	49	3285	2555
hrysene	6.8E-08	8.8E-09	1.0200	100	1E-06	1.00	12	9	49	3285	2555
Dibenzoanthracene	0.0E+00	0.0E+00	NA	100	1E-06	1.00	12	9	49	3285	2555
ndeno(123cd)pyrene	0.0E+00	0.0E+00	NA	100	1E-06	1.00	12	9	49	3285	2555
TOTAL CARC PAH		2.6E-08									
,2-Dichlorobenzene	8.1E-09	1.0E-09	0.1200	100	1E-06	1.00	12	9	49	3285	2555
,4-Dimethylphenol	0.0E+00	0.0E+00	NA	100	1E-06	1.00	12	9	49	3285	2555
-Methylphenol	4.0E-09	5.1E-10	0.0590	100	1E-06	1.00	12	9	49	3285	255
cenaphthene	1.9E-08	2.4E-09	0.2800	100	1E-06	1.00	12	9	49	3285	255
inthracene	0.0E+00	0.0E+00	NA	100	1E-06	1.00	12	9	49	3285	255
Benzoic acid	0.0E+00	0.0E+00	NA	100	1E-06	1.00	12	9	49	3285	255
lenzo(g,h,i)perylene	0.0E+00	0.0E+00	NA	100	1E-06	1.00	12	9	49	3285	255
iethylphthalate	0.0E+00	0.0E+00	NA	100	1E-06	1.00	12	9	49	3285	255
i-m-butylphthalate	4.0E-08	5.1E-09	0.5900	100	1E-06	1.00	12	9	49	3285	255
ioctylphthalate	0.0E+00	0.0E+00	NA	100	1E-06	1.00	12	9	49	3285	255
luoranthene	9.7E-08	1.2E-08	1.4400	100	1E-06	1.00	12	9	49	3285	255
luorene	2.5E-08	3.2E-09	0.3700	100	1E-06	1.00	12	9	49	3285	255
aphthalene	7.4E-08	9.6E-09	1.1100	100	1E-06	1.00	12	9	49	3285	255
henanthrene	9.6E-08	1.2E-08	1.4300	100	1E-06	1.00	12	9	49	3285	255
henol	0.0E+00	0.0E+00	NA	100	1E-06	1.00	12	9	49	3285	255
утепе 	9.6E-08	1.2E-08	1.4300	100	1E-06	1.00	12	9	49	3285	255
INORGANICS											
Aluminum	1.25-03	1 6 E-04	18014-80	100	15-06	1.00	1 12	•	40	2205	1 2000
antimony	0.0E+00	0.0E+00	NA	100	15-06	1.00	1 12	,	47	1 3200	1 2000
rsenic	8.8E-07	1.1F-07	13 10	100	15-06	1.00	1 12	,	47	1 3203	1 2000
Bariumo	3 35-05	A 2E-06	486.40	100	15-04	1.00	1 12	,	47	1 3200	2000
Beryllium	1.5E-07	2.05-08	2 30	100	1E-06	1.00	1 12	, ,	40	1 3205	1 2000
admium	6.6E-07	8.5F-08	9.80	100	16-06	1.00	1 12	, ,	40	1 3205	1 200
alcium	1.3F-03	1 7E-04	19684 70	100	15-06	1.00	1 12	, ,	47	1 3203	1 2000
hromium	1.7E-06	2 28-07	25.60	100	15-04	1.00	1 12	,	47	1 3205	1 200
obalt	7.28-07	9 36-08	10.80	100	15-06	1.00	12	, ,	47	I 3265	1 200
OPDET	2.55-06	3.36-07	38.00	100	15-06	1.00	1 12	, ,	47	1 3200	1 2003
ron	7.1E-03	9.15-04	105995.00	100	1E-06	1.00	1 12	, ,	47	1 3200	(2000 1 0555
ead	6.0E-06	7.85-07	90.00	100	15-06	1.00	1 12	,	47	1 3205	· 2003
agnesium	1.7E-04	2.28-05	2547.60	100	1E-06	1.00	12	,	47	1 3205	2000
anganese	6.8E-04	8.85-05	10161.60	100	1E-06	1.00	12	,	49	3205	1 200
ercury	3.4E-08	4.3E-09	0.50	100	1E-06	1.00	12	ģ	49	1 3285	200
ickel	1.5E-06	1.9E-07	21.90	100	1E-06	1.00	12	9	49	3285	255
otassium	8.5E-05	1.1E-05	1271.50	100	1E-06	1.00	12	9	49	3285	255
elenium	3.0E-07	3.9E-08	4.50	100	1E-06	1.00	12	9	49	3285	255
ilver	3.7E-07	4.7E-08	5.50	100	1E-06	1.00	12	9	49	3285	255
odiu	4.6E-05	5.9E-06	688.50	100	1E-06	1.00	12	9	49	3285	255
hallium	1.5E-07	2.0E-08	2.30	100	1E-06	1.00	12	9	49	3285	255
andium	2.1E-06	2.7E-07	30.90	100	1E-06	1.00	12	9	49	3285	255
inc	1.7E-05	2.28-06	259.80	100	1E-06	1.00	12	9	49	3285	255
yanide	4.0E-07	5.2E-08	6.00	100	1E-06	1.00	12	9	49	3285	2555
ESTICIDE5/PCBS										· · · · · · · · · · · · · · · · · · ·	
.4~DDD	2 45-00	3 45-10	0.0200	100	15-04	1 00					
.4-D0E	0.05+00	3.4E-10	NA 0.0370	100	1E-06	1.00	12	9	49	3285	2555
.4-DDT	3 6F-09	4 7F-10	0.0540	100	15-04	1.00	12	9	49	3285	2555
				100 1	16-00	1.00	12	7	49	3285	1 255

,

• ١

DERNAL CONTACT WITH CHEMICALS IN SEDIMENTS - CHILDREN (ALTERNATIVES 2 000 2A)

	ARSORRED DOSE	Arsorred dose		CONVERSION	SKIN SURFACE	ACHERENCE	ARSORPTION	FOPOSLIPE	EXPOSURE		AVERAGING TIME	AVERAGING TIME
CHENICAL	(NONCANCER) (mg/kg/day)	(CANCER) (mg/kg/day)	IN SEDIMENTS (mg/kg)	FACTOR (1E-6 kg/mg)	AREA (cm2/event)	FACTOR (mg/cm2)	FACTOR (unitless)	FREQUENCY (events/year)	DURATION (years)	WE19HT (kg)	(NONCANCER) (days)	(CANCER) (days)
	1		}	1		!		1			·····	
SENIVOLATILES												
1,4-Dichlorobenzene	0.0E+00	0.0E+00	NA 1 1500	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Benzol a jancin acene	0.000-00	0.0E+00	NA 1.1500	1E-06	6000	1.45	0.10	12	9	49	3265	2000
Benzo(b)fluoranthene	5.3E-07	6.8E-08	0.7990	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Benzo(k)fluoranthene	0.0E+00	0.0E+00	NA	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Bis(2ethylhexyl)phthalate	1.50-06	1.9E-07	2.2300	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Butylbenzylphthalate	0.0E+00	0.0E+00	NA	1E-06	6800	1.45	0.10	12	9	49	3285	25550
i jChrysene Dibenzoant bracene	6./E-0/	8./E-08	; 1.0200	112-06	6800 4900	1.45	0.10	12	9	49	3285	25550;
1 Indeno(123cd)ovrene	0.0E+00	0.0E+00	NA	1E-06	6800	1.45	0.10	12	, ,	49	3285	25550
TOTAL CARC PAH		2.5E-07							,		3285	25550
											3285	25550
1,2-Dichlorobenzene	7.9E-08	1.0E-08	0.1200	1E-06	6800	1.45	0.10	12	9	49	3285	25550
1,2,4-Dimethylphenol	3 95-08	5 0E-09	1768	1E-06	6800	1.45	0.10	12	, y	49	3285	25550
Acenaphthene	1.9E-07	2.4E-08	0.2800	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Anthracene	0.0E+00	0.0E+00	NA	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Benzoic acid	0.0E+00	0.0E+00	NA	1E-06	6800	1.45	0.10	12	9	49	3285	25550
(Benzo(g,h,i)perylene	0.0E+00	0.0E+00	, NA	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Diethylphthalate	0.0E100	; 0.0E+00	NA 0.5000	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Dioctylphthalate	0.0F+00	0.0E+00	NA 0.5700	1E-06	0084	1.45	0.10	12		49	3285	25550
Fluoranthene	9.5E-07	1.2E-07	1.4400	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Fluorene	2.4E-07	3.1E-08	0.3700	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Naphthalene	7.3E-07	9.4E-08	1.1100	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Phenanthrene	9.5E-07	1.2E-07	1.4300	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Phenol	0.000	0.0E+00	NA 1 4200	1E-06	6800	1.45	0.10	12	9	49	3285	25550
, Pyrene	9.52-0/	1.22-0/	1.4300	11-06	6800	1.45	0.10	12	i 9	49	3285	25550
			1 1 1 1 1				1 7 4 4 1	1 1 1 1 1		, , , , , ,		
INORGANICS										1		
Aluminum	1.2E-03	1.5E-04	18014.80	1E-06	6800	1.45	0.01	12	9	49	3285	25550
Antimony	0.0E+00	0.0€+00	NA	1E-06	6800	1.45	0.01	12	9	49	3285	25550
Arsenic	8.7E-06	1.1E-06	13.10	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Barlum	3.22-05	4.1E-06	i 486.40 ! 2.20	E-06	6800	1.45	0.01	i 12	9	i 49	3285	25550
Cadaiun	6.5E-07	B.3E-08	9,80	112-06	6800	1.45	0.01	12	, ,	49	3285	25550
Calcium	1.3E-03	1.7E-04	19684.70	1E-06	6800	1.45	0.01	12	9	49	3285	25550
Chronius	1.7E-06	2.25-07	25.60	1E-06	6800	1.45	0.01	12	9	49	3285	25550
Cobalt	7.1E-07	9.22-08	10.80	1E-06	6800	1.45	0.01	12	9	49	3285	25550
Copper	2.51-06	3.26-07	38.00	1E-06	6800	1.45	0.01	12	9	49	3285	25550
1110#	6 0E-06	7 75-07	90.00	112-06	0086 1	1.45	0.01	12	; y	49	3285	25550
Magnesium	1.7E-04	2.25-05	2547.60	1E-06	6800	1.45	0.01	12	9	49	3285	25550
Manganese	6.7E-04	8.6E-05	10161.60	1E-06	6800	1.45	0.01	12	9	49	3285	25550
Hercury	3. 3£-0 8	4.3E-09	0.50	1E-06	6800	1.45	0.01	12	9	49	3285	25550
i Nickel	1.4±-06	1.96-07	21.90	112-06	6800	1.45	0.01	12	9	49	3285	25550
Seleniu a	3.0E-07	3.85-08	4.50	112-06	6800	1.45	0.01	12	9	49	3285	25550
Silver	3.6E-07	4.7E-08	5.50	1E-06	6800	1.45	0.01	12	9	49	3285	25550
Sodium	4.68-05	5.9E-06	688.50	1E-06	6800	1.45	0.01	12	9	49	3285	25550
[Thallium	1.5£-07	2.0E-08	2.30	1E-06	6800	1.45	0.01	12	9	49	3285	25550
Vandium Vinc	2.0E-06	; 2.6E-07	30.90	1E-06	6800	1.45	0.01	12	9	49	3285	25550
Cvanide	4 0F-07	1 2.20-00 5.1E-08	1 259.60 5 6 00	1E-06	, 6800	1.45	0.01	12	9	49	3285	25550
PESTICIDES/PC8S												
4,4-000	2.62-08	3.3E-09	0.0390	i 1E-06	6800	1.45	0.10	12	9	49	3285	25550
ia,a-DOE Sala-DOT	3.65-08	4.65-09	0.0540	1E-06	6800	1.45	0.10	12	9	49	3285	25550
									, 	.		2000
VOLATILES												
1.2-Dichloroethene	0.06+00	0.06+00	NA	1 1E-06	, 6600	1.45	0.10	12	9	49	3285	25550
Benzene	0.0€+00	0.0€+00	NA	16-06	6800	1.45	0.10	12	9	49	3285	20000
Carbon disulfide	2.3E-08	3.0E-09	3.5E-02	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Chlorobenzene	6.7E-08	8.75-09	1.0E-01	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Chloroethane	0.0E+00	0.0E+00	NA	1E-06	6800	1.45	0.10	12	9	49	3285	25550
LUIIOFOFOF	6.65-09	8.58-10	1.0E-02	i 1E-06	6800	1.45	0.10	12	9	49	3285	25550
Toluene	1.85-09	2.35-09	2.75-02	1E-04	60066	1.45	0.10	12	9	49	3285	25550
Trichloroethene	0.0E+00	0.0E+00	NA	1E-06	6800	1.45	0.10	12	9	49	3285	25550
Xylenes	7.0E-08	9.06-09	1.1E-01	iE-06	6800	1.45	0.10	12	9	49	3285	25550
					*************		111111111111111111		1111111111111111			

SUMMARY OF CANCER RISK ESTIMATES: CHILDREN (ALTERNATIVES 2 AND 2A) HERTEL LANDFILL FEASIBILITY STUDY

* * * * * * * * * * * * * * * * * * *		1 1 1 2 3 1 1 1 5 8 1 4 5 3 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		8 8 1 1 1 5 1 1 1 1 8 8 8 9 1 1 4 4 4 1 1 8 8 1 1 1		4 1 1 1 8 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		
11 1	CHRONIC DAILY	CDI		1	I I	1	CHEMICAL	T
CHEMICAL	INTAKE(CDI)	ADJUSTED FOR	SF	WEIGHT OF	TYPE OF	SF BASIS/	SPECIFIC	PA
	(mg/kg/day)	ABSORPTION	(mg/kg/day)-1	EVIDENCE	CANCER	SOURCE	RISK	R
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	\$ 1 0 6 1 1 1 1 1 5 6 1 6 6 1 1 1 1 1 1 1 1 1				* • 1 1 0 0 1 1 0 0 0 1 0 0 0 1 0 0 0 0 0	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	1 1 1 1 1 1 1 1 1 1	
EXPOSURE PATHWAY: INCIDENTAL	INGESTION OF CHEN	MICALS IN SOIL - (CHILDREN					
· • • • • • • • • • • • • • • • • • • •				1		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		
INORGANICS		l I		1	1	1	.	
Arsenic	1.1E-07	No	1.8E+00	A	Bladder,Liver,Lung	Water/IRIS	2E-07	
¦¦Beryllium	2.0E-08	No	4.30	82	Skin	Water/IRIS	9E-08	1
Lead	7.8E-07	No	ND	B2	Renal	Oral/IRIS	NA	
SEMIVOLATILES		1	 ! !		; ; ;	1		1
l'Bis(2ethylhexyl)phthalate	1.9E-08	No	1.4E-02	82	Liver	Diet/IRIS	3E-10	1
Butylbenzylphthalate	NA	No	ND	C	Liver	NF/IRIS	NA	
				i i		i I		1
Benzo(a)anthracene	9.9E-09	No	11.50	82	Liver	NA/IRIS	1E-07	1
Benzo(a)pyrene	0.0E+00	No	11.50	B2	Stomach	Diet/IRIS	0E+00	
Chrysene	8.8E-09	No	11.50	B2	Stomach	NA/IRIS	1E-07	ł
Benzo(b)fluoranthene	6.9E-09	No	11.50	B2	Stomach	NA/IRIS	8E-08	1
Benzo(k)fluoranthene	0.0E+00	No	11.50	82	Stomach	NA/IRIS	0E+00	1
[Dibenzo(a,h)anthracene	NA	No	11.50	82	Stomach	NA/IRIS	NA	1
Indeno(123cd)pyrene	0.0E+00	No	11.50	82	Stomach	NA/IRIS	0E+00	•
1,4-Dichlorobenzene	NA	No	2.4E-02	82	Liver	Gavage/IRIS	NA	
PESTICIDES/PCBS		1	 	1	 	}	1	1
14,4'-DDD	NA	No	2.4E-01	82	Lung,Liver,Thyroid	Diet/IRIS	NA	
14,4'-DDE	0.0E+00	No	3.4E-01	B2	Liver, Thyroid	Diet/IRIS	0E+00	1
11	1 75 10	1 1		1 00	1 1.1	1 0:-+ /7070		1
SUMMARY OF CANCER RISK ESTIMATES: CHILDREN (ALTERNATIVES 2 AND 2A) HERTEL LANDFILL FEASIBILITY STUDY

•

				1	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1				
11	CHRONIC DAILY	CDI					CHEMICAL ;	TOTAL	11
li chemical	INTAKE(CDI)	ADJUSTED FOR	SF	WEIGHT OF	TYPE OF	SF BASIS/	SRECIFIC	Pathway 🖁	TOTAL
	(mg/kg/day)	ABSORPTION	(mg/kg/day)-1	EVIDENCE	CANCER	SOURCE	RISK	RISK	RISK
1 + 1 + 5 + 1 + 1 + 1 + 5 + 1 + 1 + 1 +	1 6 4 1 1 1 1 1 1 F 1 1 1 I I 6 6 6 1 1 6 3 1 6 6 7 1 1 1 6 6 7 6 7 6 7	1 1 1 1 1		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			1	1 1 8 1 1 1 8 1 1 1 1 9 1 1 1 9 1 1 1 9 1 1 1 9 1 1 1 9 1 1 9 1 1 9 1 1 9 1 1 9 1 1 9 1 1 9 1	1 4 1 5 1 1 1 1 1 1 1 1
EXPOSURE PATHWAY: DERMAL CON	TACT WITH SOILS -	CHILDREN						2E-05	2E-05 ¦
					+ 1 1 1 = 1 = 1 = 1 = 1 = 1 = 1 = 1 = 1			1111111111	11111111111
INORGANICS	1	1	1						
Arsenic	1.1E-06	0.10	1.8E+00	A	Bladder,Liver,Lung	Water/IRIS	2E-05		
11									
SEMIVOLATILES	1	1) I	1	1	1			
Bis(2ethylhexyl)phthalate	1.9E-07	0.10	1.4E-02	B2	Liver	Diet/IRIS	3E-08		
Butylbenzylphthalate	NA	0.10	ND	C	Liver	NF/IRIS	NA		
11	1	1	•	1	:	1	i.		
¦Benzo(a)anthracene	NA	0.10	ND ND	B2	Liver	NA/IRIS	NA		
¦¦Benzo(a)pyrene	NA	0.10	ND	B2	Stomach	Diet/IRIS	NA		
¦Chrysene	NA	0.10	ND	B2	Stomach	NA/IRIS	NA		
Benzo(b)fluoranthene	NA	0.10	ND	B2	Stomach	NA/IRIS	NA		
;;Benzo(k)fluoranthene	NA	0.10	ND	B2	Stomach	NA/IRIS	NA		
¦Dibenzo(a,h)anthracene	NA	0.10	ND	B2	Stomach	NA/IRIS	NA		
lIndeno(123cd)pyrene	0.0E+00	0.10	ND	B2	Stomach	NA/IRIS	NA		
11	1		1	1	1	-			
;1,4-Dichlorobenzene	NA	0.10	2.4E-02	B2	Liver	Gavage/IRIS	NA		
 								1	
PESTICIDES/PCBS	L S		1	1		1	1	ł	
4,4'-DOD	NA	0.10	2.4E-01	B2	Lung,Liver,Thyroid	Diet/IRIS	NA		
4,4'-DDE	0.0E+00	0.10	3.4E-01	82	Liver,Thyroid	Diet/IRIS	0E+00	1	
4,4'-001	4.6E-09	0.10	3.4E-01	B2	Liver	Diet/IRIS	2E-08	•	
					*****		1111111111	1	

.

SURMARY OF CHRONIC HAZARD INDEX ESTIMATES: CHILDREN (ALTERNATIVES 2 AND 2A)

****					MERIEL LAMOFILL FLASIBILIT STOOT					
44111114114111141114451114441151 11	CHONTE DATLY	1001	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		, , , , , , , , , , , , , , , , , , ,	PED	PED	*******	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	ΡΔΤΗΠΟΥ
	INTAKE(CDT)	ADTINSTED END	DED	CONFIDENCE		SOURCE/	INCERTAINTY	MODIEVING	HA74R0 1	H474R0
	(mo/ko/day)	ARSORPTION	(mo/ko/day)	IFVE	FFFFCT	RASIS	ADTUSTMENTS	FACTORS	OUNTIENT	INDEX (HT) !
41										1111111111111
"FYDASIDE DATHUAY: INCIDENTAL	INGESTION OF	SOTI - CHIING	27 N							1F-02
10111011111111111111111111111111111111										
				!			?	!	:	
l'Alusinus	1.2E-03	No	DI			HEAST			NA	
Antinony	NA	No	4E-04	Low	Longevity, blood glucose and	Hater/IRIS	1000	1	NA	
					cholesterol			-		
Arsenic	8.8E-07	No	1E-03	NA	Keratosis and hyperpigmentation	NA/IRIS	ERR		9E-04	
Bariun	3.3E-05	No	7E-02	Medium	Increased blood pressure	Water/IRIS	3	I	5E-04	
Beryllium	1.5E-07	No	5E-03	Low	None observed	Water/IRIS	100	1	3E-05	
Cadmium	6.6E-07	No	5.0E-04	High	Proteinuria	Water/IRIS	10	1	1E-03	
Chroniun VI	1.7E-06	No	5E-0 3	Low	No effect reported	Water/IRIS	500	1	3E-04	
Cobalt	7.2E-07	No	NF		1	IRIS/HEAST	1	1	NA	
Copper	2.5E-06	No	6.1E-02	NA	Local GI irritation	NA/HEAST	NA	1	4E-05	
Lead	6.0E-06	No	NA	NA	Neurobehavioral effects	NA/IRIS	NA	NA	NA	r
Manganese	6.8E-04	No	1E-01	Medium	CNS effects	Diet/IRIS	1 .	1	7E-03	
Hercury	3.4E-08	No	3E-04	NA	Kidney effects	NA/HEAST	1	1	1E-04	4
Nickel	1.58-06	No	2E-02	Medium	Decreased body and organ weight	Diet/HEAST	100	3	7E-0S	
Selenium	NA	No	NF	1		IRIS/HEASI		4	NA)
, Vanadium	2.1E-06	No	7E-03	NA	None observed	Water/HEAST	100	P I	; 3€-04	1
Zinc	1.7E-05	No	2E-01	NA	Anemia	NA/HEAST	10		9E-05	
Cyanide	4.0E-07	No	2E-02	Medium	Weight loss, thyroid effects,	Oral/IRIS	500	•	2E-05	
11					ayelin degeneration			}		
							·¦			
5ENIVOLATILES								į		
1) 1) [] [] [] [] [] [] [] [] [] [] [] [] []		i 1 11.		i Frank -	The second and share the state	i b bisk (1010	1 1000			1
i BIS(Zethylnexyl /phthalate	1.5£-0/	i NO	2E-02	neolum	, Increased relative liver weight	Diet/IRIS	1000		/E-06	
BUCYIDENZYIPHTHAIAte	i NA	NO	21-01	LOW	Effects on Dody weight gain,	; Diet/IRIS	1000	i I	i NA	i 1
11 11 Benne (a) anthracana	7 75_00	l No		1	testes, liver, kioney		1	1	1	4
	1 7.72-00		1 NC	1	1		1	;	1 104	
1 Chrysene	1 0.0E100	I NO		1	•			1		1
1 Renzo(h)f)uoranthene	1 5 AE-08	i No		1	1			;		1
1 Banzo(k) fluor anthene	1 0.0E+00	i No	1 IV 1 NS	4	1			1	1 NA	1
! Dihenzo(a, h)anthracene	NA NA	No	NF NF	1			1		NA NA	
Indeno 1 2 3-c diovrene	0.05+00	No.	NF NF		8		1		NA NA	
	1 0.02.00					1 INTONICHOT	!	1		
1.4-Dichlorobenzene	0.02+00	No	2.4E-02	į		IRIS/HEAST	1	1	0E+00	
							1		1	
Diethylphthalate	0.0E+00	No	8E-01	Low	Decreased growth rate, food	Diet/IRIS	1000	1	0E+00	
	1		1	1	consumption rate and altered	:	1	;	1	1
4 4 9 1	1	1	1	1	organ weights	1	1	1	1	1
;Di-n-butylphthalate	4.0E-08	No	1E-01	Low	Increased mortality	Diet/IRIS	1000	1	4E-07	1
¦Di-n-octylphthalate	NA	No	2E-02	NA	; Elevated kidney and liver weights,	Diet/HEAST	1000	1	NA	-
11	1		1	1	increased SGOT and SGPT	1	1		1	1
11						1	1	1	1	
Acenaphthene	1.9E-08	No	6E-02	Low	Hepatotoxicity	Oral/IRIS	3000	1	3E-07	
Anthracene	0.0E+00	No	3E-01	Low	No observed effect	Gravage/IRIS	3000	1	0E+00	
Benzo(g,h,i)perylene	0.0E+00	No	NF			IRIS/HEAST			NA	
Fluoranthene	9.7E-08	No	4E-02	Low	Neuopathy, liver weight changes	Gavage/IRIS	3000	1	2E-06	
ii U		i			hematological alterations and	i		i i	i	
ii Urlaanse		i 1 1.	1 15 10	1	clinical effects		i anna		1 (5.03)	i
i jr luorene	2.56-08	i No	41-02	LON	Uecreased RBC, packed cell	i Gavage/IRIS	3000	1	6E-0/	1
1)	i NA	l I Ma	l ME	1	volume and nemagiodin		1	1	I MA	1
Naphthalana				1	1 Ondex and internet function	1K15/NER51		1	I DH	1
Pyrene	9 45-08	NO NO	46-03	I NA	CULIAR AND INCERNAL LESIONS	Gavage/IRIS	3000	1	21-05	1
. Phenant hrene	9 65-08	l No	DI	LOW	i Riuney errects				1 144	
Phenol	0.0E+00	No	0.60	Low	Reduced fetal body weight	IRIS	100	1	0F+00	
11			,		,					
PESTICIDES/PCBS		1	:	;		1	1	1	1	
14,4'-DOO	NA	No	NF	1		IRIS/HEAST	1	1	NA	1
4,4'-DDE	0.0E+00	No	NF	1	1	IRIS/HEAST	1	1	NA	1
4,4'-DOT	3.6E-09	No	5E-04	Medium	Liver lesions	Diet/IRIS	100	1	7E-06	1
(),,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	111111111111111111	*********							1114411111	

SUMMARY OF CHRONIC HAZARO INDEX ESTIMATES: CHILDREN (ALTERNATIVES 2 AND 2A) HERTEL LANDFILL FEASIBILITY STUDY

ł										*****			
ł		CHRONIC DAILY	CDI	1			RFD	RFD	1	1	Pathhay	TOTAL	
i	CHEMICAL	INTAKE(CDI)	ADJUSTED FOR	RFD	CONFIDENCE	CRITICAL	SOURCE/	UNCERTAINTY	MODIFYING	HAZARD	HAZARD	HAZAR);
1	1	(∎g/kg/day)	ABSORPTION	(mg/kg/day)	LEVEL	EFFECT	BASIS	ADJUSTMENTS	FACTORS	QUOTIENT	INDEX (HI)	INDEX (I	HI);
1						1		**********		1111111111			1111
	EXPOSURE PATHWAY: DERMAL CON	TACT WITH SOIL	- CHILDREN								9E-02	1E-(01
1			***********	, , , , , , , , , , , , , , , , , , ,									Ш
ł	INORGANICS	1	1	1		1	1	1	1	1			
1	Arsenic	8.7E-06	0.10	1E-03		Keratosis and hyperpigmentation	NA/IRIS	ERR	i	9E-02			
;													
	VOLATILES	1	1	1	· ·		!		!	i			
ł	Carbon Disulfide	NA	0.10	1E-01	Medium	Fetal toxicity, Malformation	Inhalation/IRIS	100	1	NA			
	Chlorobenzene	NA	0.10	28-02	Medium	Histopathological changes	Oral/IRIS	1000	1	NA			
1			1	1	1	in the liver			-				
1	Ethylbenzene	NA	0.10	1E-01	Low	Liver and kidney toxicity	Oral/IRIS	1000	1	NA			
1	Toluene	NA	0.10	28-01	Medium	Changes in liver and kidney	- Gavage/IRIS	1000	1	NA			
	Xylenes	NA	0.10	2€+00	Medium	Hyperactivity.decreased body weight	Gavage/IRIS	100	1	NA			
;											1		
1	SEMIVOLATILES		1	1	:		1	1	!				
		1	1							1			
	Bis(2ethylhexyl)phthalate	1.5E-06	0.10	2E-02	Medium	Increased relative liver weight	Diet/IRIS	1000	1	7F-04			
	Butylbenzylphthalate	NA	0.10	2E-01	LOW	Effects on body weight gain.	Diet/IRIS	1000	1	NA NA			
		-	1	i		testes.liver.kidney							
	Benzo(a)anthracene	7.6E-07	0.10	NF			IRIS/HEAST	1	1	NA			
	8enzo(a)pyrene	0.0E+00	0.10	NF			IRIS/HEAST			NA			
Ľ	Chrysene	6.7E-07	0.10	DI			HEAST		1	NA NA	1		
	Benzo(b)fluoranthene	5.3E-07	0.10	NF			IRIS/HEAST		1 1	I NA			
	Benzo(k)fluoranthene	0.0E+00	0.10	₩F			INTS/HEAST	1			• •		
	Dibenzo(a.h)anthracene	NA	0.10	NF		1		1	4 1	I AUH I NA	1		
	Indeno(1.2.3-c.d)pyrene#	0.0F+00	0 10	. NF		1		1	4 1	I NH			
			0 10		1	ł	INTO/UCHO!	1	1	1 DH	1		
	1.4-Dichlorobenzene	0.0E+00	0 10	2 AF-02				1	1	1 05.00			
	-,						I INIS/ILASI	:	1				
	Diethylphthalate	0.0F+00	0.10		104	Decreased growth rate food	Diet/IDIC	1 1000		1 05:00			
						consumption rate and altered	I DIEC/IRIS	1 1000	; 1				
				1	1	organ usights	1	1	1	1			
	Di-n-butylphthalate	3.9F-07	0.10	15-01	104	Increased mortality	Diet /TOTC	1000		1 45.05	1		
	Di-n-octylphthalate	NA	0.10	2E-02	NA NA	Flevated kidney and liver weighte	1 Diet/IRIS	1000		46-05			
						increased SCAT and SCAT	(DIEC/IERS)	1 1000	1	I NH			
								1	1	;			
	Acenapht hene	1.9F-07	0.10	65-02	1 104	Henstotovicity		2000	1	1 25.05	1		
	Anthracene	0.0E+00	0.10	38-01	1.04	No observed effect	Gravace/IDIC	2000	1 1	1 00-100			
ļ	Senzo(g.h.i)pervlene	0.0E+00	0.10	NF				1 3000					
	Fluoranthene	9.55-07	0.10	4F-02	104	Neumathy liver weight changes		2000	1	25-04			
						headopachy, liver weight changes		1 3000	1 I	1 20-04	1		
i					1	linical affacts	1		1	1	1		
	Fluorene	2 45-07	0.10	AF-02	1 1 1 0 4	Decreased DBC packed call		1 2000		1 (5.46			
						uplume and homeolohin	i bavage/1R15	3000	1	66-05	1		
	4-Methylphenol	NA	0.10	NF		t southing and menhagroupill	I IDIS AUTACT	1		1 1 kin			
	Naphthalene	7 37-07	1 0.10	45-02	i NA	I Ocular and internal lenions	INIO/MEHSI	1 10 000	1	i NA			
	Ругеле	9.55-07	0.10	35-02	1	Kidney offecte	Gavage /TDTC	10,000		1 20-03			
	Phenanthrene	9.5E-07	0.10	D1		i niurey etteucs		3000		1 JE-04			
	Phenol	0.0E+00	0.10	0.40	ίου	Reduced fetal body wight		100					1
		·.v.·vv	, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	, v.ov	. LUW	I REGULES FELSI DOUT MEISHL	1 1412	1 100	+ 1	+ UE+00			
	PESTICIDES/PCRS					!		1	1	1			
	4.4'-DDD	2.6E-08	0.10	NF		1		•	1	i NA	1		
	4.4'-DDE	0.0E+00	0.10	NF				1	1	1 10A			1
	4.4'-DDT	3.6E-08	0.10	55-04	Medium	liver licione	1 Diet/IDIC	100		1 146	1		
					(CUTUE		1 016r/1k12	1 100	1 1 111111111111	, /E-04			
			***********								1		

;

J

INGESTION OF CHEMICALS IN SOIL WELL CONSTRUCTION (ALTERNATIVES 4 AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

	INTAKE	INTAKE	CONC. IN	INGESTION	CONVERSION	ERACTION	EXPOSURE	EXPOSURE	RODY	AVG. TIME	AVG. TIM
CHEMICAL	NONCANCER	CANCER	SOIL	RATE	FACTOR	INGESTED	FREQUENCY	DURATION	WEIGHT	NONCANCER	CANCER
, , ,	(mg/kg/day)	¦(mg/kg/day)	(mg/kg)	(mg soil/day)	¦(10E-6 kg/mg)	(unitless)	(days/year)	¦(years)	(kg)	(days)	(days)
SEMIVOLATILES	!		1	1	!	1		1	:	1	
Benzo(a)anthracene	6.16E-07	6.03E-09	0.4311	100	1.0E-06	1	250	1	70	250	25550
Chrysene	6.15E-07	6.01E-09	0.4302	100	1.0E-06	1	250	1	70	250	25550
Benzo(b)fluoranthene	9.79E-07	9.58E-09	0,6856	100	1.0E-06	1	250	1	70	250	25550
Benzo(k)fluoranthene	5.5BE-07	5.46E-09	0.3908	100	1.0E-06	1	250	1	70	250	25550
Benzo(a) pyrene	3.43F-07	3.35F-09	0.2400	100	1 0E-06	1	250	1	1 70	250	25550
Indeno(123cd)pyrene	6.13E-06	6.00F-08	4.2918	100	1 08-06	1	250	1	. 70	250	25550
Bis(2ethylhexyl)phthalate	9.12E-06	8.93E-08	6.3869	100	1.0E-06	1	250	1	. 70	250	25550
Nanhthalene	9 425-07	0.225-00	1 1 1 0 4502	100	1 1 1 0E_04	, ,	1 1 1 250		1		
Fluorene	9 40E-07	1 9.22E-07	0.0070	100	1.02-06		1 200	1 I	1 70	1 250	
Phenanthreno	1 205-06	1 17E_00	0 0244	100	1 1.0E-06		250		1 70	250	25550
Anthracene	1 2 14E=07	1 2 10E-00	0.0300	100			I - 200		1 70	250	25550
Fluoranthone	1 025-04	1 00C-00	1 2511	100	1.0E-06		250		; /0	i 250	25550
Dyrana	1.73E-00	1.075-00	1.3011	100	1.0E-06		i 250	i 1	; /0	250	25550
Ponzo(abi)norulano	1./3E-06	1.092-08	1.2083	100	1.0E-06		250	1	/0	250	25550
Denzo(gni)perylene	1 0.13E-VO	0.00E-08	4.2918	100	1.0E-06	i 1	250	1	70	250	25550
	1.5/E-0/	1.545~09	0.1100	100	1.0E-06		250	1	70	250	25550
DI-N-DULYIPHLNAIALE	8.02E-06	/.85E-08	5.616/	100	1.0E-06	1	250	1	70	250	25550
Di-n-octyiphthalate	2.86E-07	2.80E-09	0.2000	100	1.0E-06	1	250	1	70	250	25550
Butyloenzylphthalate	3.43E-0/	3.35E-09	0.2400	100	1.0E-06	1	250	1	70	250	25550
P-Cresol	4.86E-0/	4./5E-09	0.3400	100	1.0E-06	1	250	1	70	250	25550
Pheno1	; 6.13E-06	; 6.00E-08	4.2918	100	1.0E-06	; 1	250	1	70	250	25550
INORGANICS	1 9 9	1	9 8 9	1	L I 1	1	1	1 1	6	8	
Aluminium	1.89E-02	1.85E-04	13255.5	100	1.0E-06	1	250	1	70	250	25550
Antimony	3.26E-05	3.19E-07	22.8	100	1.0E-06	1	250	1	70	250	25550
Arsenic	9.86E-06	9.64E-08	6.9	100	1.0E-06	1	250	1	70	250	25550
Barium	1.70E-04	1.66E-06	118.8	100	1.0E-06	1	250	1	70	250	2555(
Beryllium	1.14E-06	1.12E-08	0.80	100	1.0E-06	1	250	1	70	250	25550
Cadmium	1.14E-06	1.12E-08	0.80	100	1.0E-06	1	250	1	2 70	250	25550
Calcium	2.46E-03	2.41E-05	1722.8	100	1.0E-06	· ·	250	1	70	250	2555
Chromium	2.61E-05	2.56E-07	18.3	100	1.0E-06	1	250	1	. 70	250	20000
Cobalt	1.79E-05	1.75E-07	12.5	100	1.0E-06	1	250	1	70	250	2555(
Copper	5.16E-05	5.05E-07	36.1	100	1_0E-06	1	250	1	2 70	250	2555
Iron	3.47E-02	3.39E-04	24278.1	100	1.0E-06	1	250	1	70	250	2555
Lead	8.59E-05	8.40E-07	60.1	100	1.0E-06	1	250	1	20	250	2555
Magnesium	7.08E-03	6.93E-05	4954.3	100	1 0E-06	1	250	1 1	2 70	250	2000
Manganese	1.79E-03	1.75E-05	1250.2	100	1 0E-06	1	250	1	1 70 70	250	1 2000 1 2555/
Mercury	0.00E+00	0.00E+00		100	1.0E-06	1	200	1	70	250	2000
Nickel	3.01E-05	2.95E-07	21.1	100	1 0E-06	1	250	1 1	1 70	1 250	2000
Potassium	1.91E-03	1.87E-05	1334.6	100	1 0E-04	1	230	1 1	1 70	1 250	2000
Selenium	5.71E-07	5,59E-09	0.4	100	1 0F-04	1 1	250		1 /V	200	1 2000(1 0000/
Sodium	2.74F-04	2.68E-06	191 5	100	1 05-04	1 1	1 200		1 70	1 250	2005(
Thallium	0 00F+00	0 00E+00	171.5	100	1 05-04		1 200		1 70	250	25550
Vandium	2 40E-05	2 255-07	16.0	1 100	1 05-06	1 1	250	1 1	, /0	250	25550
Zinc	1.92F-04	1 87F-04	13/ 1	100	1.05-06		1 250		1 /0	250	25550
			. 134.1		1 I.VE-V6	· I	. 250	· I	, /0	1 250	25550
PESTICIDES/PCBS	1	1) 	1			1			
4,4-DDE	5.87E-08	5.75E-10	0.0411	100	1.0E-06	1	250	1	70	250	25551
4,4-DOT	5.87E-08	5.75E-10	0.0411	100	1.0E-06	1	250	1	10	1 250	2000
101111111111111111111111111111111111111			111111111			1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	230	1 1 1 1 1 1 1 1 1	1 70	1 200	20000

DERMAL CONTACT WITH CHEMICALS IN SOIL WELL CONSTRUCTION (ALTERNATIVES 4 AND 4A) WERTEL LANDFILL FEASIBILITY STUDY

Ì

׼

ł														
i		AB5. DOSE	AB5. DOSE	CONC. IN	CONVERSION	5URFACE	ADHERENCE	ABSORPTION	EXPOSURE	EXPOSURE	BODY	AVG. TIME	AVG. TIME	i i
i	CHENICAL	NONCANCER	CANCER	SOIL	FACTOR	AREA SKIN	FACTOR	FACTOR	FREQUENCY	DURATION	WEIGHT	NONCANCER	CANCER	i 🖬
i	i i	(Ing/kg/day);	(mg/kg/day)	{ ng/kg };	(kg/ng)	(cm2/event)	(ng/cn2)	(unitless);	(days/year)	(years)	(Kg)	; (days)	(days)	i
i	1									· · · · · · · · · · · · ·		1		
Ì												1		
1	Panzo(a)anthronomo	E COE. AC		1 4211	1 05 00	(200	1 4 4 5		250	1 i	1 70	1 250	25550	
1	Benzo(a Janunracene	5.032-00	5.50E-08	0.4311	1.05-06	6300	1.45	0.10	250		, /V	250	2000	
i	Denze(b)fluerenthene	0.01E-00 1	0.755.00	0.4302	1.05-06	6300	1.45	0.10	250			250	20000	
!	Benzo(D) Tuoranchene	0.70E-V0 1	8./JE-V8	0.0000	1.0E-06	6300	1.45	0.10	250		i 70	250	2000	ii 🚅
1		3.10E-00	4.77E-00	0.3700	1.05-06	6300	1.40	0.10	200	i 1	1 /V	200		14
1	I Delizo(a jpyrene j	5.13E-V0	5.00E-00	1 V.2400	1.05-06	6300	1 1.45	0.10	200	i -Li F Vel	i /0	1 250	2000)
1	Pic(2ethylbery) obtalate	0.00C-V0 1	0.40E-V/	1 4.2710 1 4 2040	1.05-06	6300	L.40	0.10	200	i ui Fai	i /V	200		1
;	Distream and the second s	0.33E-05	2 555-09	0.3007	1.05-06	6300	1.40		200	1 L	1 /0 70	1 250	2000	H —
1	Putyboozylohthalate	2.010-00	2.000-00	0.2000	1.05-06	6300	1 1.40		200	1 I I	1 /V 1 70	200		11
ł		A AAE_04	A 245-00	0.2400	1.05-06	(200	1.40	0.10	200		1 /V	200		11
1	IL CLE207	4,445-00	4.546-00	0.3400	1.02-06	0300	i 1.40	0.10	200		, /0	1 200	2000	
i	Napht hallene	8 605-04	8 425-08	0 6593	1 0F-06	6300	1 45	0.10	250	1	70	250	25550	
ļ	Fluorene	7.855-06	7 68F-08	0.0373	1 0F-06	6300	1 45	0.10	250	1 1	70	250	25550	
ļ	Phenanthrene	1.095-05	1 075-07	0 8366	1 0F-06	6300	1.45	0.10	250	1	70	250	25550	11 11
1	Anthracene	1.96F-06	1.92F-08	0.1500	1 05-04	6300	1.45	0.10	250	1	70	250	25550	
ļ	Fluoranthene	1.76E-05	1.738-07	1.3511	1.0E-06	6300	1 45	0.10	250	1	70	250	25550	
i	Pyrene	1.58E-05	1.54E-07	1 2083	1 0F-06	6300	1 45	0 10	250	1	70	250	25550	11
i	Benzo(ghi)pervlene	5.60F-05	5.48E-07	4,2918	1.0E-06	6300	1 45	0 10	250	1	70	250	25550	
i	Diethylphthalate	1.44E-06	1.40E-08	0.1100	1 0F-06	6300	1 45	0 10	250	1	2 70	250	25550	11
i	Di-n-butylohthalate	7.33E-05	7.17F-07	5.6167	1 OF-06	6300	1 45	0 10	250	1	70	250	25550	
i	Phenol	5.60E-05	5.48E-07	4.2918	1.0E-06	6300	1 45	0 10	250	! 1	70	250	25550	الاسترابا
i	·													
į	INORGANICS		1			1		!	1	!	!	1	1	11
i														
i	Aluminium	1.73E-02	1.69E-04	13255.5	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	ji 🛥
i	Antimony	2.98E-05	2.91E-07	22.8	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
ł	Arsenic	9.00E-05	8.81E-07	6.9	1.0E-06	6300	1.45	0.10	250	1	70	250	25550	8
1	Barium	1.55E-04	1.52E-06	118.8	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
ļ	Beryllium	1.04E-06	1.02E-08	0.80	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	H —
¦	Cadmium	1.04E-06	1.02E-08	0.80	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	11 11
1	Calcium	2.25E-03	2.20E-05	1722.8	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
1	Chronium	2.39E-05	2.34E-07	18.3	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
1	Cobalt	1.63E-05	1.60E-07	12.5	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
i	Copper	4.71E-05	4.61E-07	36.1	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
i	TILION	3.17E-02	3.10E-04	24278.1	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
i	Lead	7.84E-05	7.67E-07	60.1	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
i	Magnesium	6.47E-03	6.33E-05	4954.3	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
i	manganese	1.63E-03	1.60E-05	1250.2	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
i	NICKEL	2.75E-05	2.69E-07	21.1	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	11 ***
į	POLASSIUN	1./4E-03	1./0E-05	1334.6	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
i	;Selenium	5.22E-07	5.11E-09	0.4	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	81
i	5001UM	2.50E-04	2.45E-06	191.5	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
i	vandlum	2.19E-05	2.15E-07	16.8	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	
i	21NC	1./5E-04	1.71E-06	134.1	1.0E-06	6300	1.45	0.01	250	1	70	250	25550	Н.,
i)	,	1						;	;			
į	PESTICIDES/PCB5	1	1	1	1	1		1	5	1	i			i đ
i		1 5 2/5 A7	1 5 255-00			1 (200	1 1 4 4 7		050		1 70	1		
1	14,4-00C	5.30E-07	1 5.20E-09	10.0411	1.02-06	1 0.000 1 4.000	1.40	0.10	250	1	1 /0	250	25550	11 ·
1	14,4°UUI 1	1 3.36E-V/	1 0.20E-VY	1 0.0411	1.VE~V6	0000	1 1.45	0.10	1 250		/0	250	25550	1
1	4 		, , , , , , , , , , , , , , , , , , , ,	1111111	1 		1 11111111111		L 		1	•		::
۱				********					1111111111111	111111111	1111111			11

INHALATION OF CHEMICALS BORN ON DUST PARTICLES WELL CONSTRUCTION (ALTERNATIVES 4 AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

CHEMICAL	i intake Noncancer	INTAKE	DUST CONC.	CONC. IN	INHALTION RATE	EXPOSURE		EXPOSURE	BODY	AVG. TIME	AVG. T
	¦(mg/kg/day)	(mg/kg/day))¦ (kg/m3)	(mg/kg)	(m3/hour)	(hours/day)	(days/year)	(years)	(kg)	(days)	(days
INORGANICS		1	1	1	1	1	 				1
Aluminium	2.63E-02	2.57E-04	8.68E-06	13255.5	. 2	! 8	250	! 1	70	250	25
Antinony	4.52E-05	4.43E-07	8 68F-06	22.8	2	. 8	250	1	70	250	25
Arsenic	1.37E-05	1.34F-07	8 68E-06	6.9	2		250	1	70	250	25
Rarium	2 365-04	2 31F-06	1 8 68E-06	118.8	2		250	1 1	70	250	25
Rerv]]ium	1 59E-06	1 55E-08	8 68E-06	0.8	1 2		1 250	1 1	70	250	1 25
Cadmium	1 59E-06	1 55E-08	8 68E-06	0.0	2	1 8	250	1 1	70	250	1 25
Calcium	3 42E-03	3 34F-05	8 68E-06	1722 8			1 250	1 1	1 70 1 70	250	1 25
Chromium	3 638-05	3 55E-07	8 68E-06	1 1/22.0		1 Q	1 250 1 250	1 1	1 70 70	1 250	1 25
Cobalt	2 48F-05	2 43E-07	1 8 68E-06	1 12.5	1 2	1 0 1 0	1 250	1 I I	1 70 70	1 250	1 25
Copper	7 165-05	7 01E-07	8 68E-06	36.1			1 250		1 70 70	1 250	r 20 I 25
Iron	4 82F-02	A 71E-04	1 8 68E-06	24278 1	1 2	1 O	1 2 <u>9</u> 0 1 250		1 70 70	1 250	1 20 1 25
Lead	1 19F-04	1 17E-06	1 8 68E-06	1 <u>60</u> 1	1 2		250		1 70 1 70	1 250	1 20 1 ac
Magnesium	9 83E-03	9 62E-05	1 8 68E-06	1 A95A 3	1 2	(O	1 250	1 1	70	1 250	1 20 1 25
Manganese	2 48E-03	2 435-05	1 8 68E-06	1250 2	1 2	1 O	1 250		1 70	1 250	↓ 20 I ⊃⊑
Nickel	4 19F-05	1 2.40E 00	1 8 68E-06	1 230.2	1 2		1 200	1 I.	1 VU 1 70	1 200	ι 20 Ι οε
Potassium	2 655-03	2 595-05	1 8 69E-06	1 1224 6	1 2	i 0	1 250		1 70 1 70	1 200	1 20
Sodium	1 2.00L 00	1 2.37C 03	1 0.00L-00	1 101 5	1 2	1 O	1 200		1 /V	250	1 20
Vanadium	1 3.00E V4	1 3.72L 00	1 0.00E-V0	1 171.5	1 2	, O 1 0	1 200	, 1 1 1	1 /V	1 200	1 25
	1 3.33E-VJ	1 3.20E-0/	0.00E-V0	1 124.1	1 2	I 0	1 200		//	1 250	20
VOLATILES Carbon Disulfide	5.95E-09	5.82E-11	8.68E-06	0.0030	2	8	250	1	70	250	25
	1.39E-08	1.36E-10	8.68E-06		2	8	250		70	250	25
Lunyidenzene	4.68E-08	4.58E-10	1 8.68E-06	0.0236	2	8	250		70	250	25
Xylenes	8.37E-07	8.19E-09	8.68E-06	0.0116	2	1 8	250		70 /0	250	i 25
SEMIVOLATILES	-						1 230 1 1			230 	
Bis(2-ethylhexyl)ohthalate	1.27E-05	1 24F-07	1 8 68E-06	6 3869	2	8	1 1 250	¦ ,	, , , , , , , , , , ,	1	1 1 2 25
Di-N-Octylphthalate	3.97F-07	3.88F-09	8 68E-06	0 2000	2	8	250	1 I	1 70	1 250	1 20
Butybenzylphthalate	4.76E-07	4.66E-09	8.68E-06	0 2400	2	! 8	250	1 1	1 70	250	1 25
Phenanthrene	1.66E-06	1.62E-08	8.68E-06	0.8366	2	8	250	1 1	2 70	250	1 25
Fluoranthene	2.68E-06	2.62E-08	8.68E-06	1.3511	2	8	250	1	2 70	250	25
Pyrene	2.40E-06	2.35E-08	8.68E-06	1.2083	2	8	250	1	70	250	25
Di-N-Butylphthalate	1.11E-05	1.09E-07	8.68E-06	5.6167	2	8	250	1	70	250	25
Benzoic Acid	4.36E-07	4.27E-09	8.68E-06	0.2200	2	8	250	1	70	250	25
Phenol	8.51E-06	8.33E-08	8.68E-06	4.2918	2	8	250	1	70	250	25
p-cresol	6.75E-07	6.60E-09	8.68E-06	0.3400	2	8	250	1	70	250	25
Diethylphthalate	2.18E-07	2.14E-09	8.68E-06	0.1100	2	8	250	1	70	250	25
Fluoranthene	2.68E-06	2.62E-08	8.68E-06	1.3511	2	8	250	1	70	250	25
Anthracene	2.98E-07	2.91E-09	8.68E-06	0.1500	2	8	250	1	70	250	1 25
Fluorene	1.19E-06	1.17E-08	8.68E-06	0.6017	2	8	250	1	70	250	25
Naphthalene	1.31E-06	1.28E-08	8.68E-06	0.6593	2	2 8	250	1	70	250	20
Benzo(a)anthracene	8.55F-07	8.37F-09	8.68E-06	0 4311	2	2	250	1	70	1 250	1 20
Benzo(a)pyrene	4.76E-07	4.66E-09	8.68F-06	0.2400	2	ι ο 1 ο	250		70	1 250	1 20
Chrysene	8.54E-07	8.35E-09	8.68E-06	0.4302	2	. р	250	1	70	250	1 20
Benzo(b)fluoranthene	1.36E-06	1.33E-08	8.68E-06	0.6856	2	8	250	1	70	250	1 20
Benzo(k)fluoranthene	7.75F-07	7.59E-09	8.68F-06	0.3908	2	. e	250	1 1	70	1 250	1 05
			111111111111				. 2.00	1 1		1 200	1 23

	SUMMART U	- CANCER RISK	ESTIMATES - W	ELL CONSTRU	CITUM (ALTERNATIVES 4	ANU 4A)		
				HERTEL LAN	IDFILL FEASIBILITY STUD	Y		1
							11111111111	6 7 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8
F1	CHRONIC DAILY	CDI	1	1		1 1	CHEMICAL	TOTAL
CHEMICAL	INTAKE(CDI)	Adjusted for	SF	WEIGHT OF	TYPE OF	SF BASIS/	SPECIFIC	PATHWAY
	(ma/ka/day)	ABSORPTION	(ma/ka/day)-1	EVIDENCE	CANCER	SOURCE	RISK	RISK
				минини			11111111	
L'EXPOSURE PATHWAY: INGESTI	ON OF CHEMICAL	S IN SOTI						6F-07
				unnun				1111111111111
	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		1			•	*******	1
Arsenic	9 64E-08	No.	1 75	۱ ۱	Skin	Uater/IRIS	2F-07	1
[Rerv]]ium	1 12E-08	No	1 43	I B2	l Skin	¹ Hater/IDIC	55-09	۱ ۱
t l cod	1 9 40E-07		1 1.5 1 1.5		Ponal	1 Water/INIS		•
+1			1 110	D4 	UC101		110	1
	1 1 I	I	,	1		1		1 4
Pic(2etby]bery])obtbalate	1 0 02E_00	i Mo	I I 1 45-02	ו ו רפון			15-00	1 1
1 Dis(2000 inc.) inc.) i philipiace	0.752 00	I NO	1.4E ⁻ 02				IE-VY	1
Houry IDenty Iphinalace	1 I		1 110	i し i	1	I NEVIRIO I	ner -	1 +
11 11 Papar (a) anthrosona		l I No	1 11 50	 00			75-00	1
H Benzo(a Januni acene	0.03E-09	I No	i 11.50	1 B2 1	Chanach		/E-V8	1
, Benzo(a)pyrene	3.35E-09	NO	j 11.50	82	Stomacn	Diet/IRIS	4E-08	i
Chrysene	6.01E-09	NO	11.50	BZ i		NA/IRIS	/E-08	i
Benzo(b)fluoranthene	9.58E-09	No	11.50	B2		NA/IRIS	1E-07	-
, Benzo(k)fluoranthene	5.46E-09	No	11.50	B2		NA/IRIS	6E-08	1
, Dibenzo(a , h)anthracene	NA	No	11.50	B 2		NA/IRIS	NA	1
Indeno(123cd)pyrene	NA	No	11.50	B2		NA/IRIS	NA	1 1 1
11	9	1	1	‡ 	1	I 1		1
1,4-Dichlorobenzene	NA	No	2.4E-02	B2	Liver	Gavage/HEAST	NA	1
11) I
PESTICIDES/PCBS	1			4		1		
4,4'-DDD	NA	No	2.4E-01	B2	Lung, Liver, Thyroid	Diet/IRIS	NA	1
4,4'-DDE	NA	No	3.4E-01	B2	Liver,Thyroid	Diet/IRIS	NA	1
4,4'-DDT	NA	No	3.4E-01	B 2	Liver	Diet/IRIS	NA	
1 1 1 1 1 1 1 1 2 1 1 1 5 1 1 5 1 1 5 1 1 5 1 1 5 1 1 5 1 1 5 1 1 5 1 1 5 1 1 5 1 1 5 1 1 5 1		T I I I I I I I I I I I I I I I I I I I			\	5 1 5 C 4 S 8 C 1 C C 1 1		1

- LELL CONCTRUCTION (ALTERNATIVES & AND #A) CUMMARY OF CANCER DICK ECTIMATES

				۰.				
	Ś	summary of can	ICER RISK ESTIM	iates - Wel	L CONSTRUCTION (ALTERNA	TIVES 4 AND 4	A)	
				HERTEL LAN	DFILL FEASIBILITY STUDY	, 		

	CHRONIC DAILY	CDI				1	CHEMICAL	TOTAL
CHEMICAL	INTAKE(CDI)	ADJUSTED FOR	SF	WEIGHT OF	TYPE OF	SF BASIS/	SPECIFIC	PATHWAY
	(mg/kg/day)	ABSORPTION	(mg/kg/day)-1	EVIDENCE	CANCER _	SOURCE	RISK	RISK ;

EXPOSURE PATHWAY: DERMAL (CONTACT WITH C	HEMICALS IN SO)IL					2E-05 ;
							minin	
INUKGANIUS			4 75				05 AF 1	
Arsenic	8.81E-0/	0.10	1./5	A	5810	Water/IRIS	2E-05	
	1							
SEMIVULATILES					. •			
Bis(2ethylhexyl)phthalate	8.16E-07	0.10	1.4E-02	B2	Liver	Diet/IRIS	1E-0/	1
Butylbenzylphthalate	NA	0.10	ND	C		NF/1R1S	NA	i
		1						1
Benzo(a)anthracene	5.50E-08	0.10	NÐ	B2		NA/IRIS	NA	
Benzo(a)pyrene	3.06E-08	0.10	ND	B2	Stomach	Diet/IRIS	NA	1
Chrysene	5.49E-08	0.10	NO	B 2		NA/IRIS	NA	}
Benzo(b)fluoranthene	8.75E-08	0.10	ND	B2		NA/IRIS	NA	۶ 1
Benzo(k)fluoranthene	4.99E-08	0.10	NÐ	82		NA/IRIS	NA	1
¦¦Dibenzo(a,h)anthracene	NA	0.10	ND	B2	1	NA/IRIS	NA	1 1
Indeno(123cd)pyrene	NA	0.10	ND	B 2		NA/IRIS	NA	1
11	1	1	1	1		,	1	,
1.4-Dichlorobenzene	NA	0.10	2.4E-02	<u>B2</u>	Liver	Gavage/HEASI		1
PESTICIDES/PCBS	1	1	1	ł		1	5	1
4,4'-DDD	NA	0.10	2.4E-01	B2	Lung, Liver, Thyroid	Diet/IRIS	NA	1
14,4'-DOE	NA NA	0.10	3.4E-01	B2	Liver,Thyroid	Diet/IRIS	NA	1
14,4'-DDT	NA	0.10	3.4E-01	82	Liver	Diet/IRIS	NA	1
* * * * * * * * * * * * * * * * * * *	******							5

					APPENDIX B					
		(SUMMARY OF CANC	ER RISK ES	STIMATES - WELL CONSTRU	CTION (ALTERNA	ATIVES 4 AND	(4A)		
				HERTEL LAN	DFILL FEASIBILITY STUD	ŕ				
		****								11
	CHRONIC DAILY	CDI	1				CHEMICAL	TOTAL	TOTAL	ii 4
CHEMICAL	INTAKE(CDI)	ADJUSTED FOR	l SF	WEIGHT OF!	TYPE OF	SF BASIS/	SPECIFIC	PATHWAY E	XF0SURE	
	(mg/kg/day)	ABSORPTION	(mg/kg/day)-1	EVIDENCE	CANCER	SOURCE	RISK	RISK	RISK	
								mmi		8.
EXPOSURE PATHWAY: INHALATI	ION OF AIRBORN	e chemicals a	DSORBED TO DUST	T				2E-05	4E-05	
LINORGANICS	1 1 2	!		1						
Arsenic	1.34E-07	No	50	A	lung	Air/HEAST	7E-06			
	1.55E-08	No	8.4		22113	t in the second	1E-07			
.'Cadmium	1.55E-08	No	6.1	B1		, Occupa/Heast	9F-08			l
Chromium	3.55E-07	No	41		luna	OCCUPA/HEAST	1E-05			
Nickel	4.10E-07	No	8.4E-01	A	Respiratory tract	OCCUPA/HEAST	3E-07			
					·			, }		
SENIVOLATILES		1	1	, ,		1	1			
Bis(2ethylhexyl)phthalate	1.24E-07	No	1.4E-02	1		1	2E-09			
Butylbenzylphthalate	NA	No	ND	C		NF/IRIS	NA			
	1		1		1 1					
Benzo(a)anthracene	8.37E-09	No	6.1	B2		NA/IRIS	5E-08	1		
Benzo(a)pyrene	4.66E-09	No	6.1	B2	Stomach	Diet/IRIS	3E-08	, 		
¦Chrysene	8.35E-09	No	6.1	B2	Respiratory tract	NA/IRIS	5E-08	1		1
Benzo(b)fluoranthene	1.33E-08	No	6.1	B2	1	NA/IRIS	8E-08			
Benzo(k)fluoranthene	7.59E-09	No	6.1	B2	, † 1	NA/IRIS	5E-08	•		
Dibenzo(a,h)anthracene	NA	No	6.1	B2	• 1 6	NA/IRIS	NA			
LIndeno(123cd)pyrene	Na	No	6.1	B2	1	NA/IRIS	NA			
	1			1	, } 		1			
1,4-Dichlorobenzene	NA	No	2.4E-02				NA	, 		
 					 	•		1		
PESTICIDES/PCBS	F F	1	1	1	l I	1	1	1		
4,4'-DDD	NA	l No	2.4E-01	B2	Lung, Liver, Thyroid	Diet/IRIS	NA	1		_
4,4'-DDE	NA	l No	3.4E-01	B2	Liver, Thyroid	Diet/IRIS	NA	1		4
14,4'-DDT	NA	No	3.4E-01	B2	Liver	Diet/IRIS	NA	1		
1 1 1 1 1 1 1 1 4 4 5 1 1 1 1 1 1 1 1 5 1 1 4 4 1 6 6 1 1 1 1 1 1 1 1 1 1 1 1 1			11111111111111111111 11111111111111111	1 1 5 5 1 1 4 4 1 1 5 4 4 1 1 5 5 6 6 5 5	1)	I I		

.

1

APPENDIX 8

SUNNARY OF CHRONIC HAZARD INDEX ESTIMATES - WELL CONSTRUCTION (ALTERNATIVE 4 AND 4A.)

					HERTEL LANDFILL FEASIBILITY STUDY			1 7)		
				, , , , , , , , , , , , , , , , , , ,						
	SUBCHRONIC	CDI				RFD	RFD			PATHNAY
CHEHICAL	DAILY INTAKE	ADJUSTED FOR	RFD	CONFIDENCE	CRITICAL	SOURCE/	UNCERTAINTY	MODIFYING	HAZARD	HAZARD
11	(mg/kg/day)	ABSORPTION	(mg/kg/day)	LEYEL	EFFECT	BASIS	ADJUSTNENTS	FACTORS	QUOTIENT	INDEX (HI)

;;EXPOSURE PATHWAY: INGESTIC	on of chemical	S IN SOIL								1E-01
	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			**********						
II INURGANICS	1 05-00	i i	DI DI			1	i		і 1 ма	i
i Antinopy	1.75-02	, אס ו	1 UI AE=04	1	I Landavity blood divesses and	1	1		1 NA 1 95-02	1
					chalesteral	1	1	1	1	1
Arsenic	9.9E-06	No	1E-03	NA	Keratosis and hyperpignentation				1F-02	
Barium	1.7E-04	No	5E-02	Medium	Increased blood pressure	• •			3E-03	1
Beryllium	1.1E-06	No	5E-03	LOW	None observed				2E-04	
Cadmium	1.1E-06	No	ND	High	Proteinuria				NA	
14	1					1	1			
Chromium VI	2.6E-05	No	2E-02	LOW	No effects reported	1		1	1E-03	1
Cobalt	1.8E-05	No	NF		1	1		l .	NA	
Copper	5.2E-05	No	4.0E-02	NA	Local GI irritation		*		1E-03	1
Lead	8.6E-05	No	NA	NA	Neurobehaioral effects	1			NA	
, Manganese	1.8E-03	NO	1E-01	Medium	CNS effects	1			2E-02	1
inercury		i NO i	36-04	i NA Hadium	Kidney effects	1	1	1	i NA	
	1 3.0E-03	ו איי ו	1 25-V2 1 NG		i Decreased body and organ weight	1	1	1	2E-03	1
Vanadium	2.4F-05	No No	7F-03	ΝΔ	None observed	1	1		3E-03	i 1
Zinc	1.9E-04	No	2E-01	NA	Apenia		1	1	1 1E-03	1
Cyanide	NA	No	2E-02	Nedium	Weight loss, thyroid effects.	•			NA	
11	1			•	myelin degeneration	, ,		1		
11 <u></u>										-
SEMIVOLATILES	1			1	1	1	1	!	; 1	1
				•		1	1	;	NA	1 I
Bis(2ethylhexyl)phthalate	9.1E-06	No	2E-02	Mediuma	Increased relative liver weight	Diet/IRIS	1000.00	1.00	5E-04	1 \$
Butylbenzylphthalate	3.4E-07	No	2E+00	LOW	Effects on body; weight gain,	Diet/IRIS	100.00	1.00	2E-07	
II I Ponto(a) anthronom	1 (25 67	i i			testes, liver, kidneys		1	i i	NA	1
, Benzo(a Januni acene	0.2L-U/	i NO	, NP		1	HEAST, IRIS		i	NA	1
fibrizo(a)pyrene	5.4E-07	i No i	I NU I NI	I	1	HEAST, IKIS	1	1	i NA	1
Benzo(b)fluoranthene	9.85-07	No	NE		4 L	HEAST TOTS	4	1	i NA	1
: Benzo(k)fluoranthene	5.6E-07	No	NF	1	8	HEAST IRIS	1	1	NA NA	1
Dibenzo(a,h)anthracene	NA	No	NF			HEAST, IRIS	• 	1	NA NA	1
Indeno(1,2,3-c,d)pyrene	NA	No	NF			HEAST, IR1S		1	NA	
						1		1		
1,4-Dichlorobenzene	NA	No	ND	•	1	HEAST, IRIS	1	i	NA	i 1
						}	1	1	1	i F
Diethylphthalate	1.6E-07	No	8E+00	Low	Decreased growth rate, food	Diet/IR15	1000.00	1.00	2E-08	1
11					consumption rate, altered					5
ii !Ni-a-butylohthalata	i I NA	i i	15.00	1	organ weights					1
'Disp-octylobthalate	1 10H 2 0E-07	, 140 i I No I	25-02	LOW I	Increased portallty	Diet/IRIS	1000.00	1.00	i NA	
11	2.72 0/		20-02		increased SGOT and SOGT	I DIGL/HEADI	1000.00		12-05	1
					i increace currant crur I				NΔ	1
Acenaphthene	NA	No	6E-01	LOW	Hepatotoxicity	Oral/IRIS			NA	1
Anthracene	2.1E-07	No	3E+00	LOW	No observed effects	Gavage/IRIS			7E-08	,
Benzo(g,h,i)perylene	NA	No	NF			HEAST/IRIS		i	NA	1
Fluoranthene	1.9E-06	No	4E-01	LOW	Neuropathy, liver weight changes	Gavage/IRIS	1		5E-06	1
11					hematological alterations and	1	1	1		
	0.45.45	i i	45.45		clinical effects				NA	
11 Iuorene	8.6E-0/	NO	4E-01	LOW	Decreased RBC, packed cell	Gavage/IRIS				
	9 45-07	No	٨E		 volume and nemoglobin 		1		1	
Naphthalene	9.4E-07	No	4F-02	NA	Drular and internal lesions	Diet/MCACT	1		} NA } 2€_0E	1
Pyrene	1.75-06	No	3 - 01	Low	Kidney effecte	Gavana/101C	1	1	1 20-05	1
Phenanthrene	1.2E-06	No	DI		RIVING STIESES	HEAST	1		NA	
Phenol	6.1E-06	No	0.60		Reduced fetal body weight	Gavage/HEAST	100			
PESTICIDES/PCBS				;					1	
4,4'-DDD	NA	No	NF			HEAST, IRIS	•	4	NA	1
14,4'-DOE	NA	No	NF			HEAST, IRIS	1		NA	1
(4,4'-D0	NA	No	5E-04	Medium	Liver lesions	Diet/IRIS	100	1.00	NA	
										1

SUMMARY OF CHRONIC MAZARD INDEX ESTIMATES - WELL CONSTRUCTION (ALTERNATIVE 4 AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

	SUBLIKUNIL	ADJUCTED FOD		I CONTINUE		i Kru	i KEU Linggotatuty			PAINWAT
i Chenical	DAILT INTAKE	ADJUSTED FOR	i KFU V (ka vala)	LUNF IDENCE		SOURCE/	UNCERTAINTY	HOUIFTING;	HAZARD	, HAZAKU ;
	i (∎9/k9/0-ay)	ABSURPTION	((199/Kg/Gay)	1 LEVEL	; E FFECI	; BASIS	AUJUSIMENIS	; FACTORS ;	QUOTIENT	(INDEX (HI);
PERPUSURE PRIMIRIA DEKIME	CONTACT WITH C	HERICHES IN SU	JTC							; 9E-01;
	***************								mm	
INURGANIUS		1 0.10	1 15 00	1					of	
Arsenic	9.0E-05	0.10	12-03	i NA	Keratosis and hyperpigmentation	NA/HEASI	; 1	i i	9E-01	
SENTVOLATTLES	1	!	!	1	1		1			1
	1		1		F 1		1			
Bis(2ethy]hexy])ohthalate	8.3E-05	0.10	2F-02	Kedium	· Increased relative liver weight	Diet/IRIS	1000	1	4E-02	1
Butylbenzylphthalate	3.1E-06	0.10	2E+00	LOW	Effects on body; weight gain.	Diet/IRIS	100	1	2E-05	
		1			testes, liver, kidneys			-		
Benzo(a)anthracene	5.6E-06	0.10	NF		1	HEAST, IRIS			NA	
Benzo(a)pyrene	3.1E-06	0.10	NF	1	1	HEAST, IRIS			NA	
Chrysene	5.6E-06	0.10	DI	1	1	HEAST			NA	:
Benzo(b)fluoranthene	8.9E-06	0.10	NF	1	1	HEAST, IRIS	1	1	NA	
Benzo(k)fluoranthene	5.1E-06	0.10	NF	!	1	HEAST, IRIS			NA	
Dibenzo(a,h)anthracene	NA	0.10	NF	1	1	HEAST, IRIS			NA	
Indeno(1,2,3-c,d)pyrene	NA	0.10	NF	1	1	HEAST, IRIS	1		NA	
1 4-Dichlorobenzene	NA.	0.10	ND ND	ļ	1 1		1		NA.	1
	1 101					I DEH31, 1K15	1		D9H	
Diethylphthalate	1.4E-06	0.10	8E+00	LON	Decreased growth rate, food	Diet/IRIS	1000	1	2E-06	
	1	1		Ì	consumption rate, altered	1				2
	1			1	organ weights					j
Di-n-butylphthalate	NA	0.10	1E+00	LOW	Increased mortality	Diet/IRIS	1000	1	NA	
Di-n-octylphthalate	2.6E-06	0.10	2E-02	NA	Elevated liver and kidney weights	Diet/HEAST	1000		1E-03	
i 4 8 1	1	1		1	increased SGOT and SPGT	1	1			1
 4 }	1	-								!
Anthronge	1 NA 2 0 0 0 0 0	0.10	61-01	LOW	Hepatotoxicity	Oral/IRIS			NA	
Ponto(a h i boryloss	1 2.0E-06	0.10	, 34,100 ! ME	LOW	NO ODSETVED ETTECTS	Wavage/IKIS			7E-06	
Denzo(9,n,1 perviene	1 1 05 05	0.10		1	1 1. Nauvanathu - Huan Intakt daaraa	HEASI/INIS			NA	1
i ir tuoranunene	1.85-02	0.10	45-01	LOW	Neuropathy, liver weight changes	Gavage/IRIS	į	1	4E-04	
	1	1	1	1	nematological alterations and	1				į
Fluorene	7.95-06	0.10	4E-01	100	CITILICAL EFFECTS Decreased DBC _ packed cell	L Gavage /TRTS	i I	:	75-04	1
	1 7.76.00	0.10		LOW	volume and benealobin	1 Odvage/1R15	4 1		20-04	1
P-Cresol	4 4E-06	0 10	5E-01	Kedium	Decreased body weight Neurofovicity	IRIS	1000	; ,	05-04	1
Naphthalene	8 6F-04	0 10	4F-02	NA	! Ocular and internal locione	Diet /HEAST	. 1000		2E-03	
Pyrene	1.6E-05	0.10	38-01	Low	Kidney effects	Gavage/IRIS			5E-0A	
Phenanthrene	1.1E-05	0.10	DI			HEAST	1		NA NA	
Phenol	5.6E-05	0.10	0.60	1	Reduced fetal body weight	Gavage/HEAST	100		9E-04	
		1 0 10				1 1				
1 PESITCIDES/PUBS	NA NA	0.10	NE		:		1	1	1 	;
14.4'-DDF	1 NA	0.10	1 NF	1		HEAST 1010	1			1
4.4'-00T	NA NA	0.10	55-04	Medium	liver lesions	Diet /TDTC	100	¦ , i	, NR NA	1

ເດດ	F N	n t	•	n –
 עעב	ÞΝ	111	1	ĸ
	L I I		Λ.	υ.

•

SUMMARY OF CHRONIC MAZARO INDEX ESTIMATES - WELL CONSTRUCTION (ALTERNATIVE 4 AND 4A) HERTEL LANDFILL FEASIBILITY STUDY

CHENICAL	DAILY INTAKE	ADJUSTED FOR	RFD	CONFIDENCE	CRITICAL	SOURCE/	UNCERTAINTY	MODIFYING:	Hazard	HAZARD	Н
	(mg/kg/day)	ABSORPTION (ng/kg/day)	LEVEL	EFFECT	BASIS	ADJUSTMENTS	FACTORS	QUOTIENT	INDEX (HI)¦
							innin	hunni	REFER		
											1
(POSURE PATHWAY: INHALAT	ION OF AIRBORN	e chemicals ads	orbed to d	UST						1E+01	;
				ļiiiiiiiiii			iiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiii				ij,
INUKGANICS	1 A 55-05	i i	45-04	i i			i i		17 01	l I	
HILLINGINY	i 4.52~05	i No i	45-04	1	5 i				1E-01	J.	
Ar Sellic	1.45-05	I NO I	1E-03	1			1	i i	IE-V2	1 I	
387100	2.4E-04	NO	1E-03	į	1			i i	2E-01	3 1	
Servilium	1.6E-06	NO	0.01	i	1 1		1		3E-04	1	
.80011010	1.65-06	NO	1E-03				1	i i	2E-03	1	
Chromium	3.6E-05	No	6E-06	NA	Nasal mucosa atrophy	HEAST	300	1	6E+00	i i i	
Hanganese	2.5E-03	No	4E-04	Medium	Increased prevalence of	IRIS	; 300	3	6E+00		
	1			1	respiratory symptoms and psycho-	HEAST	900			i	
		; ;		1	motor disturbances		1	; ;		1	
Mercury	l na	l No l	1E-04	NA	Neurotoxicty	NA/HEAST	30		NA		
Nickel	4.2E-05	No	2E-02	1	1		1	; i	2E-03	:	
Vanadium	3.3E-05	No	7E-03	1	1		1	; i	5E-03	1	
Zinc	2.7E-04	No	2E-01	Ì			i		1E-03		
Cyanide	NA	No	2E-02	1					14	i i	
				•	, 		-]			1	
VOLATILES		1		1	1		1	!		1	
Carbon Disulfide	NA	No	1F-01	1			1	1	NΔ	1	
Chlorohenzene	1 4F-08	No	SE-02	No	! "Liver and kidney effects !	HEAST	10000		38-07		
Ethylhenzene	4 7E-08	No	16+00	1		Tiens 1	1 10000	-	SE-09	1	
	2 3E-09	No 1	25+00	I NA	I CNS offects even and ness irritation	UCACT	100	4	15-00	1	
Yvlanac Yvlanac	9 AE-07	1 110 1	25-01	1 114	CNC effects, eyes and nose irritation	ICHOI LICACT	100			1	
~,10165	1 0.4L V/	1 INU 1	JL-01	1 MH		nc#51	100	1	, 31-05 	1	
SEMIVOLATILES				1	1			!			
		1		!	1			1	1	1	
Bis(2ethy]hexy])ohthalate	1.3E-05	No	28-02	i i			į		6F-04	1	
Butylbenzylohthalate	4.8E-07	No	2F+00	i					28-07	1	
Benzo(a)ant hracene	NA NA	No	NF	!	1	HEAST TOTO			1 2C V/	1	
Ranzol a Jourana		I NO I	NE	1	4 i i i	MEAST, INIS		1	1 118 I 118	1 1	
		i ito i	NI DJ	1) I	MEHOI, IRIO	1	1	1 164 } 103	1	
Cill Joche Banzo(h)fluaranthana			NC NC	:	4		1) NA NA	1	
	a 104	, no ,	107 117	1	1 1	HERST, IRIS	1		i 163	i.	
	i nua	140	Nr Nr	1		MEASI, IRIS	1	1	NA	i.	
Dibenzo(a, myanunracene	r NA	1110	NF NF	1		MEASI, IRIS	i.	1	NA	i	
Indeno(1,2,3-C,d)pyrene	i NA	NO	N	i i		HEASI, IRIS	1		NA	į	
1,4-UICNIOTODENZENE	i NA	NO	2E-01	i NA	Liver and kidney effects	HEAST	100	i	NA	1	
Dietnylphthalate	2.2E-07	No	BE+00	į			i		NA	1	
Di-n-butylphthalate	NA	No	1E+00	1			1	is a	lin.		
Di-n-octylphthalate	4.0E-07	No	2E-02	!			1		NA	1	
cenaphthene	NA	No	6E-01	1			ł	1	NG	1	
Anthracene	¦ 3.0E-07	No	3E+00	1			1		NA	1	
Benzo(g,h,i)perylene	, NA	No	NF	1	1	HEAST/IRIS	ł	1	NA	1	
luoranthene	2.7E-06	No	4E-01	1	1		1	1	7E-06	1	
luorene	1.2E-06	No	4E-01	1	:		i.	1	3E-06	i.	
o-Cresol	6.7E-07	No	5E-01	1			į		1E-06	1	
Naphthalene	1.3F-06	No	4F-02	i	1		1	1	35-05	1	
yrene	2 4F-06	No	3F-01	1	• L		1		8F-04	1 1	
henant hrene	! NA	No	01	-		HEAST	4		1 0LV0	F 1	
Phenol	I NA	No 1	D1	1		HEACT	1	1	1 115 1 184	i C	
	11H	1 IV 1			[n£831		!	, 118 	1	
PESTICIDES/PCBS	1			1				1		1	
1.4'-000	NA	No	NF	į			1	1	I NA	1	
4.4'-DDF	NA	No	NE	-		HEACT TOTO	4 1	1	1 194 1 194	1	
4 4'-DOT	NA NA	No 1	55-04		Liver lession	ILMOI INID	1	1) 198 1 106	1	
,,		· ··· ·	~ ~	•	1 11400 1000100		1	1	1 101	•	

THIS PAGE INTENTIONALLY LEFT BLANK

Ĩ.

a di

1

Ű

ő,

100

قند

ŝ.

۲Ì

сń)

, m í

أكان

1

APPENDIX C

GROUND WATER MODELING INFORMATION

APPENDIX C

GROUND WATER MODELING INFORMATION

A preliminary configuration of the proposed ground water extraction wellfield (as described in Section 5.2 of the FS), was determined using an analytical computer model (THWELLS, Version 2.0) based on the Theis equation for uniform non-steady flow in a homogenous, isotropic confined aquifer of uniform thickness and infinite in areal extent. The model further assumes the following:

- The pumping wells are fully penetrating and no water is stored in the well bores;
- Prior to pumping, the piezometric surface is horizontal over the entire area; that is, no regional flow is present;
- Water is released instantaneously from storage by the compaction of the aquifer matrix and by the expansion of the water itself; and
- Unlike unconfined conditions, none of the water stored in the aquifer is released by dewatering of the aquifer.

While the model is based on a confined aquifer, it is not expected to misrepresent the effects in an unconfined aquifer due to the similarity in behavior of confined and unconfined aquifers after pumping has continued for an extended period of time. The pumping response of an unconfined aquifer differs from that of a confined aquifer only during a short period which occurs only hours or, at the most, days after pumping has started. During this intermediate period, a "delayed-yield" recharge from the aquifer pores that were dewatered during the early stage of pumping will occur in a well screened in an unconfined aquifer. After 1,000 days, the time of the THWELLS simulation, an unconfined aquifer of this scale will have passed the delayed yield stage and resumed Theis-type response.

C-1

The model allows for only a horizontal piezometric surface and, therefore, could not be calibrated to existing hydraulic contours.

The effect of ground water extraction on surface water bodies (i.e., adjacent wetlands) was not specifically addressed. In the area of ground water contamination around well MW-13S, the criteria for determining the placement of the well(s) was that the zone of influence should reach to both wetland areas. Any extension of the zone would result in increased recharge from the wetland areas, causing an unnecessary increase in the amount of water extracted and treated, while only minimally increasing the zone of influence. As presently modeled, the drawdown in the area of the wetland west of the access road (wetland C) ranges to a maximum of 1.5 feet.

This model was used to provide a basis for the <u>preliminary</u> design of the ground water extraction system described herein and to provide a basis upon which to develop ground water extraction system costs. Additional evaluation of the ground water system will be required prior to final design of the ground water extraction system.

C-2

CALCULATION OF DRAWDOWN IN A HOMOGENEOUS, ISOTROPIC NONLEAKY CONFINED AQUIFER WITH MULTIPLE PRODUCTION AND INJECTION WELLS

HERTEL LANDFILL FS - EXTRACTION WELLFIELD MODEL _____

TRANSMISSIVITY = 135 [gpd/ft]

STORAGE COEFFICIENT = .2

WELL NO. 1

-

X·	-COORDINATE =	100 [ft]
¥-	-COORDINATE =	100 [ft]
PUMPING/INJ	ECTION RATE =	325 [gpd]
TIME SINCE START PUMPING	G/INJECTION =	1000 [day]

WELL NO. 2

X-COORDINATE	=	200 [ft]
Y-COORDINATE	=	100 [ft]
PUMPING/INJECTION RATE	=	325 [gpd]
TIME SINCE START PUMPING/INJECTION	=	1000 [day]

WELL NO. 3

			X-COORDI	INATE	=	300	[ft]
			Y-COORD:	INATE	=	100	[ft]
		PUMP	ING/INJECTION	RATE	=	325	[gpd]
TIME	SINCE	START	PUMPING/INJEC	CTION	=	1000	[day]

WELL NO. 4

X-COORDINATE	=	400 [ft]
Y-COORDINATE	=	100 [ft]
PUMPING/INJECTION RATE	=	325 [gpd]
TIME SINCE START PUMPING/INJECTION	=	1000 [day]

WELL NO. 5 X-COORDINATE = 500 [ft] Y-COORDINATE = 100 [ft] PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day] WELL NO. 6 X-COORDINATE = 600 [ft]Y-COORDINATE = 100 [ft] PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day] WELL NO. 7 X-COORDINATE = 100 [ft] Y-COORDINATE = 200 [ft]PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day] WELL NO. 8 X-COORDINATE = 200 [ft] Y-COORDINATE = 200 [ft] PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day] WELL NO. 9 X-COORDINATE = 300 [ft] Y-COORDINATE = 200 [ft] PUMPING/INJECTION RATE = 325 [gpd]

TIME SINCE START PUMPING/INJECTION = 1000 [day]

WELL NO. 10 X-COORDINATE = 400 [ft] Y-COORDINATE = 200 [ft]PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day] WELL NO. 11 X-COORDINATE = 500 [ft]Y-COORDINATE = 200 [ft] PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day] WELL NO. 12 X-COORDINATE = 600 [ft] Y-COORDINATE = 200 [ft] PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day] WELL NO. 13 X-COORDINATE = 100 [ft] Y-COORDINATE = 300 [ft] PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day]

WELL NO. 14

X-COORDINATE	=	200 [ft]
Y-COORDINATE	=	300 [ft]
PUMPING/INJECTION RATE	=	325 [gpd]
TIME SINCE START PUMPING/INJECTION	=	1000 [day]

WELL NO. 15 X-COORDINATE = 300 [ft]Y-COORDINATE = 300 [ft] PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day] WELL NO. 16 X-COORDINATE = 400 [ft]Y-COORDINATE = 300 [ft] PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day] WELL NO. 17 X-COORDINATE = 500 [ft]Y-COORDINATE = 300 [ft] PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day] WELL NO. 18 X-COORDINATE = 600 [ft]Y-COORDINATE = 300 [ft]PUMPING/INJECTION RATE = 325 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day] WELL NO. 19 X-COORDINATE = 700 [ft] Y-COORDINATE = 100 [ft] PUMPING/INJECTION RATE = 325 [gpd]

TIME SINCE START PUMPING/INJECTION = 1000 [day]

.

ŧ,

- 20

WELL NO. 20

X-COORDINATE	=	700 [ft]
Y-COORDINATE	=	200 [ft]
PUMPING/INJECTION RATE	=	325 [gpd]
TIME SINCE START PUMPING/INJECTION	=	1000 [day]

WELL NO. 21

--

			X-COORDI	NATE	=	700	[ft]
			Y-COORDI	NATE	=	300	[ft]
		PUMP	ING/INJECTION	RATE	=	325	[gpd]
TIME	SINCE	START	PUMPING/INJEC	TION	=	1000	[day]

		DRAWI	DOWN IN [ft]]		
	x-> -350	-300	-250	-200	-150	-100
¥						
-300	0.2197E+00	0.2761E+00	0.3425E+00	0.4190E+00	0.5054E+00	0.6006E
-250	0.2649E+00	0.3346E+00	0.4172E+00	0.5134E+00	0.6229E+00	0.7448E
-200	0.3144E+00	0.3991E+00	0.5006E+00	0.6199E+00	0.7573E+00	0.9117E
-150	0.3669E+00	0.4684E+00	0.5911E+00	0.7369E+00	0.9070E+00	0.1101E
-100	0.4208E+00	0.5401E+00	0.6859E+00	0.8613E+00	0.1069E+01	0.1309E
-50	0.4736E+00	0.6112E+00	0.7810E+00	0.9880E+00	0.1237E+01	0.1530E
0	0.5227E+00	0.6779E+00	0.8713E+00	0.1110E+01	0.1402E+01	0.1753E
50	0.5651E+00	0.7359E+00	0.9507E+00	0.1219E+01	0.1551E+01	0.1960E
100	0.5979E+00	0.7811E+00	0.1013E+01	0.1305E+01	0.1672E+01	0.2130E
150	0.6187E+00	0.8099E+00	0.1053E+01	0.1361E+01	0.1750E+01	0.2241E
200	0.6258E+00	0.8198E+00	0.1067E+01	0.1380E+01	0.1777E+01	0.2280E
250	0.6187E+00	0.8099E+00	0.1053E+01	0.1361E+01	0.1750E+01	0.2241E
300	0.5979E+00	0.7811E+00	0.1013E+01	0.1305E+01	0.1672E+01	0.2130E
350	0.5651E+00	0.7359E+00	0.9507E+00	0.1219E+01	0.1551E+01	0.1960E
400	0.52275+00	0.67795+00	0.9713E+00	0.1110E+01	0.1402E+01	0.1753E
450	0.32275+00	0.61125+00	0.07135+00	0.00000000	0.12275+01	0.17336
450 500	0.47305+00	0.01126+00	0.78102+00	0.96002+00	0.12575+01	0.13005
500	0.420000+00	0.5401E+00	0.00392700	0.00135+00	0.10092+01	0.13095
550	0.30092+00	0.4004E+00	0.59112+00	0.73092+00	0.90702+00	0.11016
600	0.3144E+00	0.39912+00	0.5008E+00	0.61995+00	0./5/3E+00	0.911/6
650	0.20492+00	0.33462+00	0.41/22+00	0.31345+00	0.02292+00	0.74402
	x-> −50	0	50	100	150	200
v	-					
-						
-300	0.7027E+00	0.8088E+00	0.9153E+00	0.1018E+01	0.1113E+01	0.1197E
-250	0.8765E+00	0.1014E+01	0.1154E+01	0.1288E+01	0.1413E+01	0.1522E
-200	0.1080E+01	0.1259E+01	0.1440E+01	0.1616E+01	0.1779E+01	0.1920E
-150	0.1315E+01	0.1545E+01	0.1780E+01	0.2010E+01	0.2221E+01	0.2404E
-100	0.1580E+01	0.1875E+01	0.2181E+01	0.2480E+01	0.2754E+01	0.2988E
-50	0.1869E+01	0.2247E+01	0.2647E+01	0.3039E+01	0.3394E+01	0.3692E
0	0.2171E+01	0.2654E+01	0.3184E+01	0.3708E+01	0.4163E+01	0.4537E
50	0.2460E+01	0.3067E+01	0.3790E+01	0.4537E+01	0.5082E+01	0.5567E
100	0.2703E+01	0.3423E+01	0.4358E+01	0.7038E+01	0.5974E+01	0.8250E
150	0.2862E+01	0.3652E+01	0.4649E+01	0.5703E+01	0.6395E+01	0.7029E
200	0.2918E+01	0.3732E+01	0.4790E+01	0.7594E+01	0.6621E+01	0.8961E
250	0.2862E+01	0.3652E+01	0.4649E+01	0.5703E+01	0.6395E+01	0.7029E
300	0.2703E+01	0.3423E+01	0.4358E+01	0.7038E+01	0.5974E+01	0.8250E
350	0.2460E+01	0.3067E+01	0.3790E+01	0.4537E+01	0.5082E+01	0.5567E
400	0.2171E+01	0.2654E+01	0.3184E+01	0.3708E+01	0.4163E+01	0.4537E
450	0.1869E+01	0.2247E+01	0.2647E+01	0.3039E+01	0.3394E+01	0.3692E
500	0.1580E+01	0.1875E+01	0.2181E+01	0.2480E+01	0.2754E+01	0.2988E
550	0.1315E+01	0.1545E+01	0.1780 ± 01	0.2010E+01	0.2221 =+01	0.2404
600	0.10805-01	0.12595+01	0.1440=+01	0.1616=+01	0.1770 - 101	0.10205
000	0.100000+01	0.12395701	0.11540.01	0 10000-01	0.14120.01	0.15205
650	0.87658100	0.1014 ± 01	1			

-50

-300	0.1265E+01	0.1316E+01	0.1347E+01	0.1357E+01	0.1347E+01	0.1316E+01
-250	0.1611E+01	0.1677E+01	0.1718E+01	0.1731E+01	0.1718E+01	0.1677E+01
-200	0.2036E+01	0.2121E+01	0.2173E+01	0.2190E+01	0.2173E+01	0.2121E+01
-150	0.2552E+01	0.2660E+01	0.2726E+01	0.2748E+01	0.2726E+01	0.2660E+01
-100	0.3176E+01	0.3311E+01	0.3393E+01	0.3421E+01	0.3393E+01	0.3311E+01
-50	0.3926E+01	0.4093E+01	0.4192E+01	0.4226E+01	0.4192E+01	0.4093E+01
0	0.4821E+01	0.5023E+01	0.5140E+01	0.5181E+01	0.5140E+01	0.5023E+01
50	0.5871E+01	0.6137E+01	0.6241E+01	0.6319E+01	0.6241E+01	0.6137E+01
100	0.6877E+01	0.8892E+01	0.7290E+01	0.9095E+01	0.7290E+01	0.8892E+01
150	0.7375E+01	0.7720E+01	0.7816E+01	0.7936E+01	0.7816E+01	0.7720E+01
200	0.7628E+01	0.9670E+01	0.8080E+01	0 98905+01	0.80802+01	0.9670E+01
250	0 73758+01	0 77205+01	0 78165+01	0.79365+01	0.7816E+01	0 7720 - 01
300	0.68775+01	0 99925+01	0.70102+01	0.00055+01	0.72905+01	0.99925+01
350	0.59718+01	0.00925+01	0.62415+01	0.50555+01	0.62418+01	0.61275+01
350	0.30715+01	0.013/5+01	0.02416+01	0.03192+01	0.02416+01	0.013/5+01
400	0.48215+01	0.50235+01	0.51402+01	0.51816+01	0.51402+01	0.5023E+01
450	0.39265+01	0.4093E+01	0.41928+01	0.42265+01	0.41926+01	0.40938+01
500	0.31/6E+01	0.3311E+01	0.3393E+01	0.3421E+01	0.3393E+01	0.3311E+01
550	0.2552E+01	0.2660E+01	0.2726E+01	0.2748E+01	0.2726E+01	0.2660E+01
600	0.2036E+01	0.2121E+01	0.2173E+01	0.2190E+01	0.2173E+01	0.2121E+01
650	0.1611E+01	0.1677E+01	0.1718E+01	0.1731E+01	0.1718E+01	0.1677E+01
	X-> 550	600	650	700	750	800
Y						
-300	0.1265E+01	0.1197E+01	0.1113E+01	0.1018E+01	0.9153E+00	0.8088E+00
-250	0.1611E+01	0.1522E+01	0.1413E+01	0.1288E+01	0.1154E+01	0.1014E+01
-200	0.2036E+01	0.1920E+01	0.1779E+01	0.1616E+01	0.1440E+01	0.1259E+01
-150	0.2552E+01	0.2404E+01	0.2221E+01	0.2010E+01	0.1780E+01	0.1545E+01
-100	0.3176E+01	0.2988E+01	0.2754E+01	0.2480E+01	0.2181E+01	0.1875E+01
-50	0.3926E+01	0.3692E+01	0.3394E+01	0.3039E+01	0.2647E+01	0.2247E+01
0	0.4821E+01	0.4537E+01	0.4163E+01	0.3708E+01	0.3184E+01	0.2654E+01
50	0.5871E+01	0.5567E+01	0.5082E+01	0.4537E+01	0.3790E+01	0.3067E+01
100	0.6877E+01	0.8250E+01	0.5974E+01	0.7038E+01	0.4358E+01	0.3423E+01
150	0.7375E+01	0.7029E+01	0.6395E+01	0.5703E+01	0.4649E+01	0.3652E+01
200	0.7628E+01	0.8961E+01	0.6621E+01	0.7594E+01	0.4790E+01	0.3732E+01
250	0.7375E+01	0.7029E+01	0.6395E+01	0.5703E+01	0.4649E+01	0.3652E+01
300	0.6877E+01	0.8250E+01	0.5974E+01	0.7038E+01	0.4358E+01	0.3423E+01
350	0.5871E+01	0.5567E+01	0.5082E+01	0.4537E+01	0.3790E+01	0.3067E+01
400	0.4821E+01	0.4537E+01	0.4163E+01	0.3708E+01	0.3184E+01	0.2654E+01
450	0.3926E+01	0.3692E+01	0.3394E+01	0.3039E+01	0.2647E+01	0.2247E+01
500	0.3176E+01	0.2988E+01	0.2754E+01	0.2480E+01	0.2181E+01	0.1875E+01
550	0.2552E+01	0.2404E+01	0.2221E+01	0.2010E+01	0.1780E+01	0.1545E+01
600	0.2036E+01	0.1920E+01	0.1779E+01	0.1616E+01	0.1440E+01	0.1259E+01
650	0.1611E+01	0.1522E+01	0.1413E+01	0.1288E+01	0.1154E+01	0.1014E+01
050	0.10111.01		0.14102.01	0.12002.01	0011042.01	0010142.01
	X-> 850	900	950	1000	1050	1100
		200		2000	2000	~
v						
*						
-300	0.70278+00	0.60068+00	0.5054E+00	0.4190E+00	0.3425E+00	0.2761E+00
-250	0.8765=+00	0.7448=+00	0.6229E+00	0.5134E+00	0.4172E+00	0.3346E+00
-200	0.10805+01	0.9117 ± 00	0.7573E+00	0.61998+00	0.50068-00	0.3991E+00
-150	0.1315 - 101		0 90708+00	0.7369E+00	0.5911=+00	0.46848+00
-100	0.159AF+01	0.13000+01	0 10602+00	0.8613F±00	0.6859710+00	0.54012+00
	0 . I J D V D T V I	しゃ エコロスロテロー				

0.1869E+01 0.1530E+01 0.1237E+01 0.9880E+00 0.7810E+00 0.6112E+00

W.

a de la

1

Ś

١Ż

-

í de la competition

1

Ċ.

	0	0.2171E+01	0.1753E+01	0.1402 ± 01	0.1110 ± 01	0.8713E+00	0.6779E+00
	50	0 24605+01	0 10605:01	0.15517.01	0.10100.01	0.050720.00	0.77507.00
	50	0.24000+01	0.19602+01	0.15516+01	0.1219E+01	0.950/E+00	0./359E+00
	100	0.2703E+01	0.2130E+01	0.1672E+01	0.1305E+01	0.1013E+01	0.7811E+00
	150	0.2862E+01	0.2241E+01	0.1750E+01	0.1361E+01	0.1053E+01	0.8099E+00
	200	0.2918E+01	0.2280E+01	0.1777E+01	0.1380E+01	0.1067E+01	0.8198E+00
	250	0.2862E+01	0.2241E+01	0.1750E+01	0.1361 ± 01	0.1053E+01	0.80995+00
	200	0.27025.01	0.21305.01	0.16700.01	0.13015+01	0.10125.01	0.70115.00
iii	300	0.2/03E+01	0.21302+01	0.16/26+01	0.13056+01	0.1013E+01	0.78112+00
	350	0.2460E+01	0.1960E+01	0.1551E+01	0.1219E+01	0.9507E+00	0.7359E+00
	400	0.2171E+01	0.1753E+01	0.1402E+01	0.1110E+01	0.8713E+00	0.6779E+00
	450	0.1869E+01	0.1530E+01	0.1237E+01	0.9880E+00	0.7810E+00	0.6112E+00
	500	0.1580E+01	0.1309E+01	0.1069E+01	0.8613E+00	0.6859E+00	0.5401E+00
	550	0.1315E+01	0.1101E+01	0.9070E+00	0.7369E+00	0.5911E+00	0.4684E+00
	600	0 10000+01	0.01175.00	0.75735.00	0.61007.00	0.50112000	0.30015.00
	600	0.10802+01	0.911/E+00	0.7573E+00	0.61992+00	0.50062+00	0.39912+00
	650	0.8765E+00	0.7448E+00	0.6229E+00	0.5134E+00	0.4172E+00	0.3346E+00
							

.^

TRANSMISSIVITY = 1400 [gpd/ft]

STORAGE COEFFICIENT = .2

WELL NO. 1

-

X-COORDINATE = 75 [ft] Y-COORDINATE = 75 [ft] PUMPING/INJECTION RATE = 720 [gpd] TIME SINCE START PUMPING/INJECTION = 1000 [day]

********	RESULTS	*****	THWELLS	-	PAGE	2
	1000110					_

		DRAW	DOWN IN [ft]]		
	x-> -50	-45	-40	-35	-30	-25
¥						
-50	0.1726E+00	0.1742E+00	0.1758E+00	0.1774E+00	0.1790E+00	0.1806E+00
45	0.1742E+00	0.1759E+00	0.1776E+00	0.1793E+00	0.1809E+00	0.1826E+00
40	0.1758E+00	0.1776E+00	0.1793E+00	0.1811E+00	0.1829E+00	0.1846E+00
35	0.1774E+00	0.1793E+00	0.1811E+00	0.1830E+00	0.1848E+00	0.1866E+00
30	0.1790E+00	0.1809E+00	0.1829E+00	0.1848E+00	0.1867E+00	0.1887E+00
25	0.1806E+00	0.1826E+00	0.1846E+00	0.1866E+00	0.1887E+00	0.1907E+00
20	0.1822E+00	0.1843E+00	0.1864E+00	0.1885E+00	0.1906E+00	0.1927E+00
15	0.1838E+00	0.1859E+00	0.1881E+00	0.1903E+00	0.1925E+00	0.1948E+00
10	0.1853E+00	0.1875E+00	0.1898E+00	0.1921E+00	0.1944E+00	0.1968E+00
5	0.1868E+00	0.1891E+00	0.1915E+00	0.1939E+00	0.1963E+00	0.1988E+00
)	0.1882E+00	0.1907E+00	0.1931E+00	0.1956E+00	0.1982E+00	0.2008E+00
5	0.1897E+00	0.1922E+00	0.1947E+00	0.1973E+00	0.2000E+00	0.2027E+00
10	0.1910E+00	0.1936E+00	0.1963E+00	0.1990E+00	0.2018E+00	0.2046E+00
.5	0.1923E+00	0.1950E+00	0.1978E+00	0.2006E+00	0.2035E+00	0.2064E+00
0	0.1936E+00	0.1963E+00	0.1992E+00	0.2021E+00	0.2051E+00	0.2082E+00
5	0.1947E+00	0.1976E+00	0.2005E+00	0.2035E+00	0.2067E+00	0.2099E+00
0	0.1958E+00	0.1987E+00	0.2018E+00	0.2049E+00	0.2081E+00	0.2114E+00
5	0.1968E+00	0.1998E+00	0.2029E+00	0.2061E+00	0.2095E+00	0.2129E+00
10	0.1977E+00	0.2008E+00	0.2040E+00	0.2073E+00	0.2107E+00	0.2143E+00
15	0.1985E+00	0.2016E+00	0.2049E+00	0.2083E+00	0.2118E+00	0.2155E+00
50	0.1992E+00	0.2024E+00	0.2057E+00	0.2091E+00	0.2127E+00	0.2165E+00
5	0.1997E+00	0.2030E+00	0.2064E+00	0.2099E+00	0.2135E+00	0.2174E+00
50	0.2002E+00	0.2035E+00	0.2069E+00	0.2104E+00	0.2142E+00	0.2181E+00
65	0.2005E+00	0.2038E+00	0.2073E+00	0.2109E+00	0.2146E+00	0.2186E+00
70	0.2007E+00	0.2040E+00	0.2075E+00	0.2111E+00	0.2149E+00	0.2189E+00
75	0.2008E+00	0.2041E+00	0.2076E+00	0.2112E+00	0.2150E+00	0.2190E+00
80	0.2007E+00	0.2040E+00	0.2075E+00	0.2111E+00	0.2149E+00	0.2189E+00
85	0.2005E+00	0.2038E+00	0.2073E+00	0.2109E+00	0.2146E+00	0.2186E+00
90	0.2002E+00	0.2035E+00	0.2069E+00	0.2104E+00	0.2142E+00	0.2181E+00
95	0.1997E+00	0.2030E+00	0.2064E+00	0.2099E+00	0.2135E+00	0.2174E+00
100	0.1992E+00	0.2024E+00	0.2057E+00	0.2091E+00	0.2127E+00	0.2165E+00
105	0.1985E+00	0.2016E+00	0.2049E+00	0.2083E+00	0.2118E+00	0.2155E+00
110	0.1977E+00	0.2008E+00	0.2040E+00	0.2073E+00	0.2107E+00	0.2143E+00
115	0.1968E+00	0.1998E+00	0.2029E+00	0.2061E+00	0.2095E+00	0.2129E+00
120	0.1958E+00	0.1987E+00	0.2018E+00	0.2049E+00	0.2081E+00	0.2114E+00
125	0.1947E+00	0.1976E+00	0.2005E+00	0.2035E+00	0.2067E+00	0.2099E+00
120	0 10368+00	0 19638+00	0 10020+00	0 2021 =+00	0 2051 - 00	0 20925+00

0.1936E+00 0.1963E+00 0.1992E+00 0.2021E+00 0.2051E+00 0.2082E+00 130 135 0.1923E+00 0.1950E+00 0.1978E+00 0.2006E+00 0.2035E+00 0.2064E+00 140 0.1910E+00 0.1936E+00 0.1963E+00 0.1990E+00 0.2018E+00 0.2046E+00 145 0.1897E+00 0.1922E+00 0.1947E+00 0.1973E+00 0.2000E+00 0.2027E+00 150 0.1882E+00 0.1907E+00 0.1931E+00 0.1956E+00 0.1982E+00 0.2008E+00 155 0.1868E+00 0.1891E+00 0.1915E+00 0.1939E+00 0.1963E+00 0.1988E+00 160 0.1853E+00 0.1875E+00 0.1898E+00 0.1921E+00 0.1944E+00 0.1968E+00 165 0.1838E+00 0.1859E+00 0.1881E+00 0.1903E+00 0.1925E+00 0.1948E+00 170 0.1822E+00 0.1843E+00 0.1864E+00 0.1885E+00 0.1906E+00 0.1927E+00 175 0.1806E+00 0.1826E+00 0.1846E+00 0.1866E+00 0.1887E+00 0.1907E+00 180 0.1790E+00 0.1809E+00 0.1829E+00 0.1848E+00 0.1867E+00 0.1887E+00 0.1774E+00 0.1793E+00 0.1811E+00 0.1830E+00 0.1848E+00 0.1866E+00 185 190 0.1758E+00 0.1776E+00 0.1793E+00 0.1811E+00 0.1829E+00 0.1846E+00 195 0.1742E+00 0.1759E+00 0.1776E+00 0.1793E+00 0.1809E+00 0.1826E+00

Y

-50	0.1822E+00	0.1838E+00	0.1853E+00	0.1868E+00	0.1882E+00	0.1897E+00
-45	0.1843E+00	0.1859E+00	0.1875E+00	0.1891E+00	0.1907E+00	0.1922E+00
-40	0.1864E+00	0.1881E+00	0.1898E+00	0.1915E+00	0.1931E+00	0.1947E+00
-35	0.1885E+00	0.1903E+00	0.1921E+00	0.1939E+00	0.1956E+00	0.1973E+00
-30	0.1906E+00	0.1925E+00	0.1944E+00	0.1963E+00	0.1982E+00	0.2000E+00
-25	0.1927E+00	0.1948E+00	0.1968E+00	0.1988E+00	0.2008E+00	0.2027E+00
-20	0.1949E+00	0.1970E+00	0.1992E+00	0.2013E+00	0.2034E+00	0.2055E+00
-15	0.1970E+00	0.1993E+00	0.2016E+00	0.2038E+00	0.2061E+00	0.2083E+00
-10	0.1992E+00	0.2016E+00	0.2040E+00	0.2064E+00	0.2087E+00	0.2111E+00
-5	0.2013E+00	0.2038E+00	0.2064E+00	0.2089E+00	0.2114E+00	0.2140E+00
0	0.2034E+00	0.2061E+00	0.2087E+00	0.2114E+00	0.2142E+00	0.2169E+00
5	0.2055E+00	0.2083E+00	0.2111E+00	0.2140E+00	0.2169E+00	0.2198E+00
10	0.2075E+00	0.2104E+00	0.2134E+00	0.2165E+00	0.2196E+00	0.2227E+00
15	0.2095E+00	0.2126E+00	0.2157E+00	0.2190E+00	0.2223E+00	0.2256E+00
20	0.2114E+00	0.2146E+00	0.2180E+00	0.2214E+00	0.2249E+00	0.2285E+00
25	0.2132E+00	0.2166E+00	0.2201E+00	0.2237E+00	0.2274E+00	0.2313E+00
30	0.2149E+00	0.2185E+00	0.2222E+00	0.2260E+00	0.2299E+00	0.2340E+00
35	0.2165E+00	0.2202E+00	0.2241E+00	0 2281 - 00	0 23225+00	0.23668+00
40	0 21805+00	0.22025+00	0 22585+00	0.23005+00	0.23442+00	0.23005+00
45	0.2193E+00	0 2233E+00	0.2274 =+00	0.23185+00	0.23645+00	0.24125+00
50	0 2204 - 00	0.2235E+00	0.22995+00	0.23345+00	0.23822+00	0.24125+00
55	0.2214F+00	0.2256E+00	0.23005+00	0.2347E+00	0.23975+00	0.24322+00
60	0 22225+00	0.225555+00	0.23105+00	0.23585+00	0.24095+00	0.24435+00
65	0.22225+00	0.2203E+00	0.2317 ± 00	0.23565+00	0.24092+00	0.24035+00
70	0 22305+00	0.2274E+00	0.2321E+00	0.23705+00	0.24235+00	0.24795+00
75	0 22325+00	0 22765+00	0.232212+00	0.2372E+00	0.2425E+00	0.24752+00
80	0.22305+00	0.22745+00	0.23215+00	0.23705+00	0.24235+00	0.24012+00
85	0.222302+00	0.2274E+00	0.2321E+00	0.23665+00	0.24232+00	0.24735+00
90	0.222275+00	0.22658+00	0.2310E+00	0.23585+00	0.2410E+00	0.24/35+00
95	0.222225+00	0.22555+00	0.23105+00	0.23585+00	0.2409E+00	0.2403E+00
100	0.22145+00	0.22365+00	0.2300E+00	0.234/E+00	0.23975+00	0.24495+00
105	0.2204E+00	0.2245E+00	0.22895+00	0.23345+00	0.23642+00	0.2432E+00
110	0 21805+00	0.2233E+00	0.22585+00	0.2310E+00	0.2304E+00	0.23905+00
115	0 2165 - 00	0.22025+00	0.2230E+00	0.229005+00	0.23225+00	0.2356E+00
120	0.2109E+00	0.2202E+00	0.2231E+00	0.2260E+00	0.23222+00	0.2300E+00
125	0.2132E+00	0.2166E+00	0.2201E+00	0.2237E+00	0.2233E+00	0.2340E+00
130	0.2114E+00	0.2146E+00	0.2180E+00	0.2214E+00	0.2249E+00	0.2285E+00
135	0.2095E+00	0.2126E+00	0.2157E+00	0.2190E+00	0.2223E+00	0.2256E+00
140	0.2075E+00	0.2104E+00	0.2134E+00	0.2165E+00	0.2196E+00	0.2227E+00
145	0.2055E+00	0.2083E+00	0.2111E+00	0.2140E+00	0.2169E+00	0.2198E+00
150	0.2034E+00	0.2061E+00	0.2087E+00	0.2114E+00	0.2142E+00	0 21695+00
155	0 2013E+00	0 20385+00	0 2064 - 00	0 20895+00	0 211425+00	0.21095+00
160	0 19925+00	0.2016 - 00	0.2040 - 00	0.20642+00	0 20875+00	0.2140E+00
165	0.1970 - 100	0.19935+00	0.2016 - 00	0.20395+00	0.2007E+00	0.20935+00
170	0.1949=+00	0.1970=+00	0.1002000	0.2013=+00	0.20348+00	0 20555+00
175	0 1927 - 10	0 10495+00	0 10695100	0 10000100	0 20099100	0 20275+00
180	0 1006-+00	0 10250100	0 10448+00	0 1962	0 100200400	0 20005+00
195	0.1995-00	0.19255+00	0.10010+00	0 10200+00	0.10565+00	0 10735+00
100	0.18648+00	0.19035+00	0.19215700	0.19355+00	0 10218+00	
106	0.10045+00	0.10015+00	0.10768+00	0.19155+00	0.10070.00	0.10000.00
192	0.10435+00	0.1829E+00	0.18/5E+00	0.10316+00	0.130/E+00	0.1922E+00

X-> 10

15 20 25 30 35

4

v

	-50	0.1910E+00	0.1923E+00	0.1936E+00	0.1947E+00	0.1958E+00	0.1968E+00
	-45	0.1936E+00	0.1950E+00	0.1963E+00	0.1976E+00	0.1987E+00	0.1998E+00
	-40	0.1963E+00	0.1978E+00	0.1992E+00	0 2005 - 00	0.2018E+00	0.2029E+00
	_ 25	0 19905-00	0 2006 - 00	0.20218+00	0.2005E+00	0.20408+00	0.20618+00
	-35	0.19905+00	0.20005+00	0.20215+00	0.20355400	0.20495+00	0.20016+00
	-30	0.20165+00	0.20355+00	0.20516+00	0.2007E+00	0.20016+00	0.20955+00
	-25	0.20465+00	0.20045+00	0.20822+00	0.20992+00	0.21146+00	0.21296+00
	-20	0.20/5E+00	0.2095E+00	0.2114E+00	0.2132E+00	0.21496+00	0.21656+00
-	-15	0.2104E+00	0.2126E+00	0.2146E+00	0.2166E+00	0.2185E+00	0.2202E+00
	-10	0.2134E+00	0.2157E+00	0.2180E+00	0.2201E+00	0.2222E+00	0.2241E+00
	-5	0.2165E+00	0.2190E+00	0.2214E+00	0.2237E+00	0.2260E+00	0.2281E+00
	0	0.2196E+00	0.2223E+00	0.2249E+00	0.2274E+00	0.2299E+00	0.2322E+00
	5	0.2227E+00	0.2256E+00	0.2285E+00	0.2313E+00	0.2340E+00	0.2366E+00
	10	0.2258E+00	0.2290E+00	0.2321E+00	0.2352E+00	0.2382E+00	0.2410E+00
	15	0.2290E+00	0.2324E+00	0.2358E+00	0.2392E+00	0.2425E+00	0.2457E+00
	20	0.2321E+00	0.2358E+00	0.2395E+00	0.2432E+00	0.2469E+00	0.2505E+00
	25	0.2352E+00	0.2392E+00	0.2432E+00	0.2473E+00	0.2514E+00	0.2554E+00
	30	0.2382E+00	0.2425E+00	0.2469E+00	0.2514E+00	0.2559E+00	0.2604E+00
	35	0.2410E+00	0.2457E+00	0.2505E+00	0.2554E+00	0.2604E+00	0.2655E+00
(111)	40	0.2438E+00	0.2487E+00	0.2539E+00	0.2593E+00	0.2649E+00	0.2706E+00
	45	0 24635+00	0 2516 -00	0.25725+00	0.26305+00	0.26925+00	0.2756 - 00
	45	0.24035+00	0.25102+00	0.2572E+00	0.26505+00	0.20925+00	0.27502+00
	50	0.24852+00	0.25422+00	0.26012+00	0.26652+00	0.27322+00	0.2804E+00
	55	0.2505E+00	0.2564E+00	0.2628E+00	0.2696E+00	0.2769E+00	0.2848E+00
	60	0.2520E+00	0.2582E+00	0.2649E+00	0.2721E+00	0.2799E+00	0.2885E+00
	65	0.2532E+00	0.2596E+00	0.2665E+00	0.2740E+00	0.2823E+00	0.2914E+00
-	70	0.2539E+00	0.2604E+00	0.2675E+00	0.2752E+00	0.2837E+00	0.2932E+00
-	75	0.2542E+00	0.2607E+00	0.2678E+00	0.2756E+00	0.2842E+00	0.2939E+00
	80	0.2539E+00	0.2604E+00	0.2675E+00	0.2752E+00	0.2837E+00	0.2932E+00
	85	0.2532E+00	0.2596E+00	0.2665E+00	0.2740E+00	0.2823E+00	0.2914E+00
	90	0.2520E+00	0.2582E+00	0.2649E+00	0.2721E+00	0.2799E+00	0.2885E+00
	95	0.2505E+00	0.2564E+00	0.2628E+00	0.2696E+00	0.2769E+00	0.2848E+00
	100	0.2485E+00	0.2542E+00	0.2601E+00	0.2665E+00	0.2732E+00	0.2804E+00
	105	0.2463E+00	0.2516E+00	0.2572E+00	0.2630E+00	0.2692E+00	0.2756E+00
	110	0.2438E+00	0.2487E+00	0.2539E+00	0.2593E+00	0.2649E+00	0.2706E+00
	115	0.2410E+00	0.2457E+00	0.2505E+00	0.2554E+00	0.2604E+00	0.2655E+00
	120	0.2382E+00	0.2425E+00	0.2469E+00	0.2514E+00	0.2559E+00	0.2604E+00
	125	0 23525+00	0 23925+00	0 24325+00	0 24735+00	0.2514E+00	0 2554 - 00
	120	0.23315+00	0.23525+00	0.23955+00	0.24735+00	0.24695+00	0.25055+00
	130	0.23212+00	0.23582+00	0.23955400	0.24325+00	0.24092+00	0.25052+00
6-1	135	0.22902+00	0.2324E+00	0.23582+00	0.23928+00	0.24258+00	0.245/E+00
-	140	0.2258E+00	0.2290E+00	0.2321E+00	0.2352E+00	0.2382E+00	0.2410E+00
	145	0.2227E+00	0.2256E+00	0.2285E+00	0.2313E+00	0.2340E+00	0.2366E+00
	150	0.2196E+00	0.2223E+00	0.2249E+00	0.22/4E+00	0.2299E+00	0.2322E+00
	155	0.2165E+00	0.2190E+00	0.2214E+00	0.2237E+00	0.2260E+00	0.2281E+00
	160	0.2134E+00	0.2157E+00	0.2180E+00	0.2201E+00	0.2222E+00	0.2241E+00
	165	0.2104E+00	0.2126E+00	0.2146E+00	0.2166E+00	0.2185E+00	0.2202E+00
	170	0.2075E+00	0.2095E+00	0.2114E+00	0.2132E+00	0.2149E+00	0.2165E+00
	175	0.2046E+00	0.2064E+00	0.2082E+00	0.2099E+00	0.2114E+00	0.2129E+00
	180	0.2018E+00	0.2035E+00	0.2051E+00	0.2067E+00	0.2081E+00	0.2095E+00
	185	0.1990E+00	0.2006E+00	0.2021E+00	0.2035E+00	0.2049E+00	0.2061E+00
	190	0.1963E+00	0.1978E+00	0.1992E+00	0.2005E+00	0.2018E+00	0.2029E+00
	195	0.1936E+00	0.1950E+00	0.1963E+00	0.1976E+00	0.1987E+00	0.1998E+00
-							
		X-> 40	45	50	55	60	65
	Y						
	-						
	-50	0.1977E+00	0.1985E+00	0.1992E+00	0.1997 ± 00	0.20028-00	0.2005
	-45	0.2008E+00	0.2016 ± 00	0.2024E+00	0.2030 =+00	0 20355+00	0 20395+00

-40		٥.	2040E+00	0.2049E+00	0.2057E+00	0.2064E+00	0.2069E+00	0.2073E+00
-35		٥.	2073E+00	0.2083E+00	0.2091E+00	0.2099E+00	0.2104E+00	0.2109E+00
-30		٥.	2107E+00	0.2118E+00	0.2127E+00	0.2135E+00	0.2142E+00	0.2146E+00
-25		0.	2143E+00	0.2155E+00	0.2165E+00	0.2174E+00	0.2181E+00	0.2186E+00
-20		0.	2180E+00	0.2193E+00	0.2204E+00	0.2214E+00	0.2222E+00	0.2227E+00
-15		n .	2218E+00	0.2233E+00	0 2245 - 00	0 2256 - 00	0 2265 - 00	0.2271E+00
-10			22105+00	0.22335+00	0.2245E+00	0.22505+00	0.22055+00	0.22/16+00
-10			22505+00	0.22/45+00	0.2289E+00	0.2300E+00	0.23102+00	0.231/E+00
-5			23005+00	0.2318E+00	0.2334E+00	0.234/E+00	0.2358E+00	0.2366E+00
0		0.	2344E+00	0.2364E+00	0.2382E+00	0.2397E+00	0.2409E+00	0.2418E+00
5		0.	2390E+00	0.2412E+00	0.2432E+00	0.2449E+00	0.2463E+00	0.2473E+00
10		0.	2438E+00	0.2463E+00	0.2485E+00	0.2505E+00	0.2520E+00	0.2532E+00
15		0.	2487E+00	0.2516E+00	0.2542E+00	0.2564E+00	0.2582E+00	0.2596E+00
20		0.	2539E+00	0.2572E+00	0.2601E+00	0.2628E+00	0.2649E+00	0.2665E+00
25		Ο.	2593E+00	0.2630E+00	0.2665E+00	0.2696E+00	0.2721E+00	0.2740E+00
30		Ο.	2649E+00	0.2692E+00	0.2732E+00	0.2769E+00	0.2799E+00	0.2823E+00
35		٥.	2706E+00	0.2756E+00	0.2804E+00	0.2848E+00	0.2885E+00	0.2914E+00
40		0.	2765E+00	0.2823E+00	0.2879E+00	0.2932E+00	0.2979E+00	0.3016E+00
45		n .	2823E+00	0.2891E+00	0.2958E+00	0 30245+00	0 30835+00	0 3131E+00
50		n -	28795+00	0 20595+00	0 30408+00	0 31218+00	0.31095+00	0.32635+00
50		.	20792+00	0.29302+00	0.30405+00	0.31212+00	0.31985+00	0.3203E+00
55			29325+00	0.3024E+00	0.31212+00	0.32228+00	0.33232+00	0.3415E+00
60			29/92+00	0.30832+00	0.31985+00	0.33236+00	0.3458E+00	0.35916+00
65			3016E+00	0.3131E+00	0.3263E+00	0.3415E+00	0.3591E+00	0.3790E+00
70		0.	3040E+00	0.3163E+00	0.3307E+00	0.3481E+00	0.3698E+00	0.3982E+00
75		ο.	3048E+00	0.3174E+00	0.3323E+00	0.3506E+00	0.3741E+00	0.4073E+00
80		0.	3040E+00	0.3163E+00	0.3307E+00	0.3481E+00	0.3698E+00	0.3982E+00
85		٥.	3016E+00	0.3131E+00	0.3263E+00	0.3415E+00	0.3591E+00	0.3790E+00
90		٥.	2979E+00	0.3083E+00	0.3198E+00	0.3323E+00	0.3458E+00	0.3591E+00
95		٥.	2932E+00	0.3024E+00	0.3121E+00	0.3222E+00	0.3323E+00	0.3415E+00
100		٥.	2879E+00	0.2958E+00	0.3040E+00	0.3121E+00	0.3198E+00	0.3263E+00
105	(٥.	2823E+00	0.2891E+00	0.2958E+00	0.3024E+00	0.3083E+00	0.3131E+00
110		٥.	2765E+00	0.2823E+00	0.2879E+00	0.2932E+00	0.2979E+00	0.3016E+00
115		٥.	2706E+00	0.2756E+00	0.2804E+00	0.2848E+00	0.2885E+00	0.2914E+00
120		ς.	2649E+00	0.2692E+00	0.2732E+00	0.2769E+00	0.2799E+00	0.2823E+00
125		b .	25935+00	0.2630E+00	0.2665E+00	0.2696E+00	0.2721E+00	0.2740E+00
130		h	25398+00	0.2572E+00	0.2601E+00	0.26285+00	0.2649E+00	0.2665E+00
135		n	24875+00	0.2516F+00	0 25425+00	0 25645+00	0 25825+00	0 2596 - 00
140		.	24072+00	0.24635+00	0.23425+00	0.25055+00	0.25025+00	0.25305+00
145		.	23905+00	0.24035+00	0.24035+00	0.23035+00	0.2320E+00	0.23325+00
160		· ·	23905+00	0.24125+00	0.24325+00	0.24495700	0.2403E+00	0.24/35+00
150			23445+00	0.23045+00	0.23826+00	0.23975+00	0.24095+00	0.24105+00
155			23002+00	0.23182+00	0.2334E+00	0.234/E+00	0.2358E+00	0.23662+00
100			22585+00	0.22/4E+00	0.2289E+00	0.2300E+00	0.2310E+00	0.231/E+00
102			22185+00	0.2233E+00	0.2245E+00	0.2256E+00	0.2265E+00	0.22/1E+00
170			2180E+00	0.2193E+00	U.2204E+00	U.2214E+00	U.2222E+00	U.2227E+00
175			2143E+00	0.2155E+00	0.2165E+00	0.2174E+00	0.2181E+00	0.2186E+00
180	().	2107E+00	0.2118E+00	0.2127E+00	0.2135E+00	0.2142E+00	0.2146E+00
185).	2073E+00	0.2083E+00	0.2091E+00	0.2099E+00	0.2104E+00	0.2109E+00
190	().	2040E+00	0.2049E+00	0.2057E+00	0.2064E+00	0.2069E+00	0.2073E+00
195	().	2008E+00	0.2016E+00	0.2024E+00	0.2030E+00	0.2035E+00	0.2038E+00
	X-2	>	70	75	80	85	90	95
			_					
Y								
-								
-50		٥.	2007E+00	0.2008E+00	0.2007E+00	0.2005E+00	0.2002E+00	0.1997E+00
-45)	2040E+00	0.2041=+00	0.2040E+00	0.2038E+00	0.2035=+00	0.2030E+00
-40		5	2075	0.2076=+00	0.2075=+00	0.2073=+00	0.2069=+00	0 20649+00
-35			20,35+00 2111±±00	0 2112=+00	0 2111=+00	0.2100=+00	0.21048+00	0 20045+00
-30			21108.00	0.21125+00	0.2110-00	0.21055100	0.21045+00	
-30			21495+00 21495+00	U.2150E+00	U.2149E+00	0.2146E+00	U.2142E+00	U.2135E+00
-25		. ر	₹197E+00	0.2190E+00	0.2189E+00	0.2186E+00	0.2181E+00	0.21/4E+00

Ċ,

<u>ک</u>

1

Ċ,

×

٥Ì

۲Ì

	-20	Ο.	2230E+00	0.2232E+00	0.2230E+00	0.2227E+00	0.2222E+00	0.2214E+00
	-15	Ο.	2274E+00	0.2276E+00	0.2274E+00	0.2271E+00	0.2265E+00	0.2256E+00
	-10	0.	2321E+00	0.2322E+00	0.2321E+00	0.2317E+00	0.2310E+00	0.2300E+00
	-5	0.	2370E+00	0.2372E+00	0.2370E+00	0 2366E+00	0.2358E+00	0.2347E+00
	0	0.	2423E+00	0.2425E+00	0.2423E+00	0 24185+00	0 2409E+00	0.2397E+00
	5	ů.	24202+00	0 24915+00	0.24295+00	0.24105+00	0.24635+00	0.24495+00
	10	· · ·	24795+00	0.24016400	0.24792+00	0.24/32+00	0.24035400	0.244955+00
	10	0.	25395+00	0.25425+00	0.25392+00	0.25522+00	0.25205+00	0.2505E+00
_	15	0.	2604E+00	0.260/E+00	0.2604E+00	0.25968+00	0.25828+00	0.2564E+00
	20	0.	26/55+00	0.26/8E+00	0.26/5E+00	0.2665E+00	0.2649E+00	0.2628E+00
	25	0.	2752E+00	0.2756E+00	0.2752E+00	0.2740E+00	0.2721E+00	0.2696E+00
.	30	0.	2837E+00	0.2842E+00	0.2837E+00	0.2823E+00	0.2799E+00	0.2769E+00
	35	0.	2932E+00	0.2939E+00	0.2932E+00	0.2914E+00	0.2885E+00	0.2848E+00
	40	0.	3040E+00	0.3048E+00	0.3040E+00	0.3016E+00	0.2979E+00	0.2932E+00
-	45	0.	3163E+00	0.3174E+00	0.3163E+00	0.3131E+00	0.3083E+00	0.3024E+00
-	50	0.	3307E+00	0.3323E+00	0.3307E+00	0.3263E+00	0.3198E+00	0.3121E+00
	55	0.	3481E+00	0.3506E+00	0.3481E+00	0.3415E+00	0.3323E+00	0.3222E+00
	60	0.	3698E+00	0.3741E+00	0.3698E+00	0.3591E+00	0.3458E+00	0.3323E+00
	65	Ο.	3982E+00	0.4073E+00	0.3982E+00	0.3790E+00	0.3591E+00	0.3415E+00
	70	0.	4357E+00	0.4641E+00	0.4357E+00	0.3982E+00	0.3698E+00	0.3481E+00
	75	0.	4641E+00	0.6870E+00	0.4641E+00	0.4073E+00	0.3741E+00	0.3506E+00
-	80	0.	4357E+00	0.4641E+00	0.4357E+00	0.3982E+00	0.3698E+00	0.3481E+00
_	85	0.	3982E+00	0.4073E+00	0.3982E+00	0.3790E+00	0.3591E+00	0.3415E+00
	90	٥.	3698E+00	0.3741E+00	0.3698E+00	0.3591E+00	0.3458E+00	0.3323E+00
	95	Ο.	3481E+00	0.3506E+00	0.3481E+00	0.3415E+00	0.3323E+00	0.3222E+00
	100	Ο.	3307E+00	0.3323E+00	0.3307E+00	0.3263E+00	0.3198E+00	0.3121E+00
	105	Ο.	3163E+00	0.3174E+00	0.3163E+00	0.3131E+00	0.3083E+00	0.3024E+00
	110	Ο.	3040E+00	0.3048E+00	0.3040E+00	0.3016E+00	0.2979E+00	0.2932E+00
	115	0.	2932E+00	0.2939E+00	0.2932E+00	0.2914E+00	0.2885E+00	0.2848E+00
-	120	0.	2837E+00	0.2842E+00	0.2837E+00	0.2823E+00	0.2799E+00	0.2769E+00
	125	0.	2752E+00	0.2756E+00	0.2752E+00	0.2740E+00	0.2721E+00	0.2696E+00
	130	0.	2675E+00	0.2678E+00	0.2675E+00	0.2665E+00	0.2649E+00	0.2628E+00
	135	0.	2604E+00	0.2607E+00	0.2604E+00	0.2596E+00	0.2582E+00	0.2564E+00
	140	0.	2539E+00	0.2542E+00	0.2539E+00	0.2532E+00	0.2520E+00	0.2505E+00
	145	ů.	24795+00	0 24815+00	0.24795+00	0.24735+00	0.24635+00	0.24495+00
	150	0.	24735+00	0 2425 - 00	0.24735+00	0.24755+00	0.24095+00	0.23975+00
_	155	· · ·	23235+00	0.23725+00	0.23705+00	0.23665+00	0.23595+00	0.23975+00
	160	· · ·	23705+00	0.23725+00	0.2370E+00	0.2300E+00	0.23565+00	0.234/E+00
	160	0.	23216+00	0.23222+00	0.2321E+00	0.2317E+00	0.2310E+00	0.2300E+00
	105	0.	22745+00	0.22/6E+00	0.2274E+00	0.22/1E+00	0.22652+00	0.2256E+00
	170	0.	2230E+00	0.22328+00	0.2230E+00	0.222/E+00	0.22228+00	0.2214E+00
	1/5	0.	2189E+00	0.2190E+00	0.2189E+00	0.2186E+00	0.2181E+00	0.21/4E+00
200	180	0.	2149E+00	0.2150E+00	0.2149E+00	0.2146E+00	0.2142E+00	0.2135E+00
-	185	0.	2111E+00	0.2112E+00	0.2111E+00	0.2109E+00	0.2104E+00	0.2099E+00
	190	0.	20/5E+00	0.20/6E+00	0.20/5E+00	0.20/3E+00	0.2069E+00	0.2054E+00
	195	0.	2040E+00	0.2041E+00	0.2040E+00	0.2038E+00	0.2035E+00	0.2030E+00
			100	105	110		100	105
		X->	100	105	110	115	120	125
-								
	Ŷ							
		•						
-	-50	0.	1992E+00	0.1985E+00	0.1977E+00	0.1968E+00	0.1958E+00	0.1947E+00
-	-45	0.	2024E+00	0.2016E+00	0.2008E+00	0.1998E+00	0.1987E+00	0.1976E+00
	-40	0.	2057E+00	0.2049E+00	0.2040E+00	0.2029E+00	0.2018E+00	0.2005E+00
	-35	0.	2091E+00	0.2083E+00	0.2073E+00	0.2061E+00	0.2049E+00	0.2035E+00
	-30	0.	2127E+00	0.2118E+00	0.2107E+00	0.2095E+00	0.2081E+00	0.2067E+00
	-25	0.	2165E+00	0.2155E+00	0.21 43E +00	0.2129E+00	0.2114E+00	0.2099E+00
	-20	0.	2204E+00	0.2193E+00	0.2180E+00	0.2165E+00	0.2149E+00	0.2132E+00
محنز	-15	0.	2245E+00	0.2233E+00	0.2218E+00	0.2202E+00	0.2185E+00	0.2166E+00
	-10	0.	2289E+00	0.2274E+00	0.2258E+00	0.2241E+00	0.2222E+00	0.2201E+00
	-5	0.	2334E+00	0.2318E+00	0.2300E+00	0.2281E+00	0.2260E+00	0.2237E+00

						•
0	0.2382E+00	0.2364E+00	0.2344E+00	0.2322E+00	0.2299E+00	0.2274E+00
5	0.2432E+00	0.2412E+00	0.2390E+00	0.2366E+00	0.2340E+00	0.2313E+00
10	0.2485E+00	0.2463E+00	0.2438E+00	0.2410E+00	0.2382E+00	0.2352E+00
15	0.2542E+00	0.2516E+00	0.2487E+00	0.2457E+00	0.2425E+00	0.2392E+00
20	0.2601E+00	0.2572E+00	0.2539E+00	0.2505E+00	0.2469E+00	0.2432E+00
25	0 26658+00	0 26305+00	0.25925+00	0.25645+00	0.25145+00	0.24725+00
20	0.20055+00	0.20305+00	0.25955400	0.25546+00	0.25146+00	0.24/35+00
30	0.2/328+00	0.26926+00	0.2649E+00	0.2604E+00	0.25598+00	0.2514E+00
35	0.2804E+00	0.2756E+00	0.2706E+00	0.2655E+00	0.2604E+00	0.2554E+00
40	0.2879E+00	0.2823E+00	0.2765E+00	0.2706E+00	0.2649E+00	0.2593E+00
45	0.2958E+00	0.2891E+00	0.2823E+00	0.2756E+00	0.2692E+00	0.2630E+00
50	0.3040E+00	0.2958E+00	0.2879E+00	0.2804E+00	0.2732E+00	0.2665E+00
55	0.3121E+00	0.3024E+00	0.2932E+00	0.2848E+00	0.2769E+00	0.2696E+00
60	0.3198E+00	0.3083E+00	0.2979E+00	0.2885E+00	0.2799E+00	0.2721E+00
65	0.3263E+00	0.3131E+00	0.3016E+00	0.2914E+00	0.2823E+00	0.2740E+00
70	0.3307E+00	0.3163E+00	0.3040E+00	0.2932E+00	0.2837E+00	0.2752E+00
75	0 3323E+00	0 31745+00	0 30485+00	0.29395+00	0 28425+00	0.2756 - 00
<i>,</i> ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	0.33075+00	0.31632+00	0.30402+00	0.29395+00	0.28426+00	0.27502+00
80	0.330/E+00	0.3103E+00	0.30402+00	0.2932E+00	0.283/E+00	0.2/528+00
85	0.3263E+00	0.3131E+00	0.3016E+00	0.2914E+00	0.2823E+00	0.2740E+00
90	0.3198E+00	0.3083E+00	0.2979E+00	0.2885E+00	0.2799E+00	0.2721E+00
95	0.3121E+00	0.3024E+00	0.2932E+00	0.2848E+00	0.2769E+00	0.2696E+00
100	0.3040E+00	0.2958E+00	0.2879E+00	0.2804E+00	0.2732E+00	0.2665E+00
105	0.2958E+00	0.2891E+00	0.2823E+00	0.2756E+00	0.2692E+00	0.2630E+00
110	0.2879E+00	0.2823E+00	0.2765E+00	0.2706E+00	0.2649E+00	0.2593E+00
115	0.2804E+00	0.2756E+00	0.2706E+00	0.2655E+00	0.2604E+00	0.2554E+00
120	0.2732E+00	0.2692E+00	0.2649E+00	0.2604E+00	0.2559E+00	0.2514E+00
125	0.2665E+00	0.2630E+00	0.2593E+00	0.2554E+00	0.2514E+00	0.2473E+00
120	0.26012+00	0.25725+00	0.25305+00	0.25055+00	0.24695+00	0.24735+00
125	0.25012+00	0.25726+00	0.25552+00	0.25052+00	0.24092+00	0.24325+00
135	0.2542E+00	0.2516E+00	0.248/E+00	0.245/E+00	0.2425E+00	0.2392E+00
140	0.2485E+00	0.2463E+00	0.2438E+00	0.2410E+00	0.2382E+00	0.2352E+00
145	0.2432E+00	0.2412E+00	0.2390E+00	0.2366E+00	0.2340E+00	0.2313E+00
150	0.2382E+00	0.2364E+00	0.2344E+00	0.2322E+00	0.2299E+00	0.2274E+00
155	0.2334E+00	0.2318E+00	0.2300E+00	0.2281E+00	0.2260E+00	0.2237E+00
160	0.2289E+00	0.2274E+00	0.2258E+00	0.2241E+00	0.2222E+00	0.2201E+00
165	0.2245E+00	0.2233E+00	0.2218E+00	0.2202E+00	0.2185E+00	0.2166E+00
170	0.2204E+00	0.2193E+00	0.2180E+00	0.2165E+00	0.2149E+00	0.2132E+00
175	0.2165E+00	0.2155E+00	0.2143E+00	0.2129E+00	0.2114E+00	0.2099E+00
180	0.2127E+00	0.2118E+00	0.2107E+00	0.2095E+00	0.2081E+00	0.2067E+00
185	0.2091E+00	0.2083E+00	0.2073E+00	0.2061E+00	0.2049E+00	0.2035E+00
190	0.2057 ± 0.0	0 2049E+00	0.2040E+00	0 2029E+00	0.2018E+00	0 2005 - 00
105	0.203/5+00	0.20165+00	0.20085+00	0.19995+00	0.1997 ± 00	0.1976E+00
195	0.2024E+00	0.20162+00	0.20082+00	0.19985+00	0.198/6+00	0.19/62+00
	X- > 130	135	140	145	150	155
	x -> 150	135	140	145	150	155
v						
T						
E 0	0 10265-00	0 10035.00	0 10100.00	0 10075.00	0 10000.00	0 10000.00
-50	0.1936E+00	U.1923E+00	0.1910E+00	0.189/E+00	0.1882E+00	0.1808E+00
-45	0.1963E+00	0.1950E+00	0.1936E+00	0.1922E+00	0.1907E+00	0.1891E+00
-40	0.1992E+00	0.1978E+00	0.1963E+00	0.1947E+00	0.1931E+00	0.1915E+00
-35	0.2021E+00	0.2006E+00	0.1990E+00	0.1973E+00	0.1956E+00	0.1939E+00
-30	0.2051E+00	0.2035E+00	0.2018E+00	0.2000E+00	0.1982E+00	0.1963E+00
-25	0.2082E+00	0.2064E+00	0.2046E+00	0.2027E+00	0.2008E+00	0.1988E+00
20	0.2114E+00	0.2095E+00	0.2075E+00	0.2055E+00	0.2034E+00	0.2013E+00
-15	0.2146E+00	0.2126E+00	0.2104E+00	0.2083E+00	0.2061E+00	0.2038E+00
10	0.21805+00	0.2157E+00	0.2134E+00	0.2111E+00	0.2087E+00	0.2064E+00
5	0.2214 ± 0.00	0.2190	0.21658+00	0.2140 = -00	0.2114 ± 00	0.20895-00
0	0 22/05+00	0 222304+00	0 2106 - 10	0.21602+00	0 21/28+00	0 211/1 +00
5	0.22495+00	0.22235+00	0.21901700	0.21095100	0.21425+00	0.21145+00
5	U.2285E+00	U.2256E+U0	0.222/5+00	0.21985+00	0.2109E+00	0.21408+00
10	0.2321E+00	0.2290E+00	0.2258E+00	U.2227E+00	0.2196E+00	0.2165E+00
15	0.2358E+00	0.2324E+00	0.2290E+00	0.2256E+00	0.2223E+00	0.2190E+00

.

	20	0.2395E+	00 0.2358E+00	0.2321E+00	0.2285E+00	0.2249E+00	0.2214E+00
	25	0.2432E+	00 0.2392E+00	0.2352E+00	0.2313E+00	0.2274E+00	0.2237E+00
	30	0 24695+	00 0 24258+00	0 23825+00	0 22408+00	0 22998+00	0 22605+00
	30	0.24096+	00 0.24255+00	0.23026+00	0.23405+00	0.22995+00	0.2200E+00
	35	0.2505E+	00 0.245/E+00	0.2410E+00	0.2366E+00	0.2322E+00	0.2281E+00
	40	0.2539E+	00 0.2487E+00	0.2438E+00	0.2390E+00	0.2344E+00	0.2300E+00
	45	0.2572E+	00 0.2516E+00	0.2463E+00	0.2412E+00	0.2364E+00	0.2318E+00
	50	0.2601E+	00 0.2542E+00	0.2485E+00	0.2432E+00	0.2382E+00	0.2334E+00
	55	0.2628E+	00 0.2564E+00	0.2505E+00	0.2449E+00	0.2397E+00	0.2347E+00
	55	0.202051	00 0.25045+00	0.25052100	0.24495+00	0.24007.00	0.23475:00
	60	0.20496+	·00 0.2582E+00	0.2520E+00	0.2463E+00	0.24095+00	0.23562+00
	65	0.2665E+	00 0.2596E+00	0.2532E+00	0.2473E+00	0.2418E+00	0.2366E+00
	70	0.2675E+	00 0.2604E+00	0.2539E+00	0.2479E+00	0.2423E+00	0.2370E+00
	75	0.2678E+	00 0.2607E+00	0.2542E+00	0.2481E+00	0.2425E+00	0.2372E+00
	80	0.2675E+	00 0.2604E+00	0.2539E+00	0.2479E+00	0.2423E+00	0.2370E+00
	85	0.2665E+	00 0.2596E+00	0.2532E+00	0.2473E+00	0.2418E+00	0.2366E+00
	90	0 26495+	00 0 25828+00	0 25205+00	0 24635+00	0 24095+00	0 2358E+00
	50	0.20495+	00 0.2562E+00	0.25205+00	0.24035+00	0.24095+00	0.23305+00
	32	U.2628E+	00 0.2564E+00	0.2505E+00	0.24495+00	0.239/E+00	0.234/E+00
	100	0.2601E+	00 0.2542E+00	0.2485E+00	0.2432E+00	0.2382E+00	0.2334E+00
	105	0.2572E+	00 0.2516E+00	0.2463E+00	0.2412E+00	0.2364E+00	0.2318E+00
	110	0.2539E+	00 0.2487E+00	0.2438E+00	0.2390E+00	0.2344E+00	0.2300E+00
	115	0.2505E+	00 0.2457E+00	0.2410E+00	0.2366E+00	0.2322E+00	0.2281E+00
	120	0.2469E+	00 0.2425E+00	0.2382E+00	0.2340E+00	0.2299E+00	0.2260E+00
	105	0.24000	00 0 23035+00	0.23520.00	0.23125.00	0.22745+00	0.22275+00
	125	0.24326+	00 0.23925+00	0.2352E+00	0.2313E+00	0.22/46+00	0.223/E+00
	130	0.2395E+	00 0.2358E+00	0.2321E+00	0.2285E+00	0.2249E+00	0.2214E+00
_	135	0.2358E+	00 0.2324E+00	0.2290E+00	0.2256E+00	0.2223E+00	0.2190E+00
-	140	0.2321E+	00 0.2290E+00	0.2258E+00	0.2227E+00	0.2196E+00	0.2165E+00
	145	0.2285E+	00 0.2256E+00	0.2227E+00	0.2198E+00	0.2169E+00	0.2140E+00
	150	0.2249E+	00 0.2223E+00	0.2196E+00	0.2169E+00	0.2142E+00	0.2114E+00
	155	0 2214E+	00 0 21905+00	0 2165 - 00	0.2140 ± 00	0.2114E+00	0.2089E+00
-	100	0.22140+		0.21055+00	0.21100+00	0.20075.00	0.20030100
	100	0.21806+	00 0.215/E+00	0.2134E+00	0.21112+00	0.208/E+00	0.2064E+00
	165	0.2146E+	00 0.2126E+00	0.2104E+00	0.2083E+00	0.2061E+00	0.2038E+00
	170	0.2114E+	00 0.2095E+00	0.2075E+00	0.2055E+00	0.2034E+00	0.2013E+00
-	175	0.2082E+	00 0.2064E+00	0.2046E+00	0.2027E+00	0.2008E+00	0.1988E+00
	180	0.2051E+	00 0.2035E+00	0.2018E+00	0.2000E+00	0.1982E+00	0.1963E+00
	185	0.2021E+	00 0.2006E+00	0.1990E+00	0.1973E+00	0.1956E+00	0.1939E+00
	100	0 10025	00 0 1979 - 00	0 10635+00	0.1947 ± 00	0.1931 ± 00	0 1915 - 00
	190	0.19926+	00 0.19785+00	0.1903E+00	0.19478+00	0.19516+00	0.19132+00
	192	0.1963E+	00 0.1950E+00	0.19365+00	0.19225+00	0.190/E+00	0.18916+00
						_	
		X-> 160	165	170	175	180	185
-	Y						
-							
	-50	0.1853E+	00 0.1838E+00	0.1822E+00	0.1806E+00	0.1790E+00	0.1774E+00
	-45	0.1875E+	00 0.1859E+00	0.1843E+00	0.1826E+00	0.1809E+00	0.1793E+00
	-40	0 19995+	00 0 19918+00	0 19645+00	0 1946 - 00	0 19295+00	0 19115+00
_		0.10908+	00 0.1001E+00	0.10045+00	0.10402+00	0.10296+00	0.10116+00
	-35	0.19216+	00 0.1903E+00	0.1885E+00	0.18666+00	0.1848E+00	0.1830E+00
	-30	0.1944E+	00 0.1925E+00	0.1906E+00	0.1887E+00	0.1867E+00	0.1848E+00
	-25	0.1968E+	00 0.1948E+00	0.1927E+00	0.1907E+00	0.1887E+00	0.1866E+00
	-20	0.1992E+	00 0.1970E+00	0.1949E+00	0.1927E+00	0.1906E+00	0.1885E+00
	-15	0.2016E+	00 0.1993E+00	0.1970E+00	0.1948E+00	0.1925E+00	0.1903E+00
	-10	0.2040E+	00 0.2016E+00	0.1992E+00	0.1968E+00	0.1944E+00	0.1921E+00
	_5	0 20645	00 0 20295+00	0 20135+00	0 1000	0 19625+00	0 10205+00
		0.200467	00 0.2030ETUU	0.00045.00	0.19002+00	0.10000.00	0.105/
	0	0.208/E+	00 0.2061E+00	U.2034E+00	0.2008E+00	0.1982E+00	0.1956E+00
	5	0.2111E+	00 0.2083E+00	0.2055E+00	0.2027E+00	0.2000E+00	0.1973E+00
	10	0.2134E+	00 0.2104E+00	0.2075E+00	0.2046E+00	0.2018E+00	0.1990E+00
	15	0.2157E+	00 0.2126E+00	0.2095E+00	0.2064E+00	0.2035E+00	0.2006E+00
	20	0.2180E+	00 0.2146E+00	0.2114E+00	0.2082E+00	0.2051E+00	0.2021E+00
	25	0,2201 -	00 0.2166=+00	0.21325+00	0.20995+00	0.2067=+00	0 20255+00
	30	0 2220101	00 0 2100E+00	0.21405-00	0.21145:00	0.20076700	0.20356700
	50	0.22228+	00 0.2185E+00	0.21495+00	0.21148+00	0.2081E+00	U.2049E+00
	35	U.2241E+	UU 0.2202E+00	0.2165E+00	U.2129E+00	0.2095E+00	0.2061E+00

40	0.2258E+00	0.2218E+00	0.2180E+00	0.2143E+00	0.2107E+00	0.2073E+00
45	0.2274E+00	0.2233E+00	0.2193E+00	0.2155E+00	0.2118E+00	0.2083E+00
50	0.2289E+00	0.2245E+00	0.2204E+00	0.2165E+00	0.2127E+00	0.2091E+00
55	0.2300E+00	0.2256E+00	0.2214E+00	0.2174E+00	0.2135E+00	0.2099E+00
60	0.2310E+00	0.2265E+00	0.2222E+00	0.2181E+00	0.2142E+00	0.2104E+00
65	0.2317E+00	0.2271E+00	0.2227E+00	0.2186E+00	0.2146E+00	0.2109E+00
70	0.2321E+00	0.2274E+00	0.2230E+00	0.2189E+00	0.2149E+00	0.2111E+00
75	0.2322E+00	0.2276E+00	0.2232E+00	0.2190E+00	0.2150E+00	0.2112E+00
80	0.2321E+00	0.2274E+00	0.2230E+00	0.2189E+00	0.2149E+00	0.2111E+00
85	0.2317E+00	0.2271E+00	0.2227E+00	0.2186E+00	0.2146E+00	0.2109E+00
90	0.2310E+00	0.2265E+00	0.2222E+00	0.2181E+00	0.2142E+00	0.2104E+00
95	0.2300E+00	0.2256E+00	0.2214E+00	0.2174E+00	0.2135E+00	0.2099E+00
100	0.2289E+00	0.2245E+00	0.2204E+00	0.2165E+00	0.2127E+00	0.2091E+00
105	0.2274E+00	0.2233E+00	0.2193E+00	0.2155E+00	0.2118E+00	0.2083E+00
110	0.2258E+00	0.2218E+00	0.2180E+00	0.2143E+00	0.2107E+00	0.2073E+00
115	0.2241E+00	0.2202E+00	0.2165E+00	0.2129E+00	0.2095E+00	0.2061E+00
120	0.2222E+00	0.2185E+00	0.2149E+00	0.2114E+00	0.2081E+00	0.2049E+00
125	0.2201E+00	0.2166E+00	0.2132E+00	0.2099E+00	0.2067E+00	0.2035E+00
130	0.2180E+00	0.2146E+00	0.2114E+00	0.2082E+00	0.2051E+00	0.2021E+00
135	0.2157E+00	0.2126E+00	0.2095E+00	0.2064E+00	0.2035E+00	0.2006E+00
140	0.2134E+00	0.2104E+00	0.2075E+00	0.2046E+00	0.2018E+00	0.1990E+00
145	0.2111E+00	0.2083E+00	0.2055E+00	0.2027E+00	0.2000E+00	0.1973E+00
150	0.2087E+00	0.2061E+00	0.2034E+00	0.2008E+00	0.1982E+00	0.1956E+00
155	0.2064E+00	0.2038E+00	0.2013E+00	0.1988E+00	0.1963E+00	0.1939E+00
160	0.2040E+00	0.2016E+00	0.1992E+00	0.1968E+00	0.1944E+00	0.1921E+00
165	0.2016E+00	0.1993E+00	0.1970E+00	0.1948E+00	0.1925E+00	0.1903E+00
170	0.1992E+00	0.1970E+00	0.1949E+00	0.1927E+00	0.1906E+00	0.1885E+00
175	0.1968E+00	0.1948E+00	0.1927E+00	0.1907E+00	0.1887E+00	0.1866E+00
180	0.1944E+00	0.1925E+00	0.1906E+00	0.1887E+00	0.1867E+00	0.1848E+00
185	0.1921E+00	0.1903E+00	0.1885E+00	0.1866E+00	0.1848E+00	0.1830E+00
190	0.1898E+00	0.1881E+00	0.1864E+00	0.1846E+00	0.1829E+00	0.1811E+00
195	0.1875E+00	0.1859E+00	0.1843E+00	0.1826E+00	0.1809E+00	0.1793E+00

	x->	190	195
Y			
-50	ο.	1758E+00	0.1742E+00
-45	Ο.	1776E+00	0.1759E+00
-40	0.	1793E+00	0.1776E+00
-35	0.	1811E+00	0.1793E+00
-30	0.	1829E+00	0.1809E+00
-25	Ο.	1846E+00	0.1826E+00
-20	Ο.	1864E+00	0.1843E+00
-15	Ο.	1881E+00	0.1859E+00
-10	Ο.	1898E+00	0.1875E+00
-5	Ο.	1915E+00	0.1891E+00
0	Ο.	1931E+00	0.1907E+00
5	Ο.	1947E+00	0.1922E+00
10	ο.	1963E+00	0.1936E+00
15	0.	1978E+00	0.1950E+00
20	0.	1992E+00	0.1963E+00
25	0.3	2005E+00	0.1976E+00
30	0.3	2018E+00	0.1987E+00
35	0.	2029E+00	0.1998E+00
40	0.	2040E+00	0.2008E+00
45	0.1	2049E+00	0.2016E+00
50	0.	2057E+00	0.2024E+00
55	0.1	2064E+00	0.2030E+00
60	0.3	2069E+00	0.2035E+00

	65	0.2073E+00	0.2038E+00
	70	0.2075E+00	0.2040E+00
	75	0.2076E+00	0.2041E+00
-	80	0.2075E+00	0.2040E+00
-	85	0.2073E+00	0.2038E+00
	90	0.2069E+00	0.2035E+00
	95	0.2064E+00	0.2030E+00
	100	0.2057E+00	0.2024E+00
	105	0.2049E+00	0.2016E+00
	110	0.2040E+00	0.2008E+00
	115	0.2029E+00	0.1998E+00
	120	0.2018E+00	0.1987E+00
	125	0.2005E+00	0.1976E+00
	130	0.1992E+00	0.1963E+00
—	135	0.1978E+00	0.1950E+00
	140	0.1963E+00	0.1936E+00
	145	0.1947E+00	0.1922E+00
	150	0.1931E+00	0.1907E+00
	155	0.1915E+00	0.1891E+00
	160	0.1898E+00	0.1875E+00
-	165	0.1881E+00	0.1859E+00
	170	0.1864E+00	0.1843E+00
	175	0.1846E+00	0.1826E+00
(mm)	180	0.1829E+00	0.1809E+00
-	185	0.1811E+00	0.1793E+00
	190	0.1793E+00	0.1776E+00
	195	0.1776E+00	0.1759E+00

.

.

•

THIS PAGE INTENTIONALLY LEFT BLANK